As Filed with the Securities and Exchange Commission on May 12, 2000

> Registration No. 333-

SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM S-4

REGISTRATION STATEMENT

UNDER

THE SECURITIES ACT OF 1933

AVNET, INC.

2211 South 47th Street

Phoenix, Arizona 85034

(480) 643-2000

(Address, including zip code, and telephone number, including area code, of registrant's principal executive offices)

DAVID R. BIRK, Esq.

Senior Vice President,

General Counsel and Secretary

Avnet, Inc.

2211 South 47th Street

Phoenix, Arizona 85034

(480) 643-2000

(Name, address, including zip code, and telephone number, including area code, of agent for service)

With copies to:

New York (State or Other Jurisdiction of Incorporation or Organization) 5065 (Primary Standard Industrial Classification Code Number) 11-1890605 (I.R.S. Employer Identification Number)

Approximate date of commencement of proposed sale of the securities to the public:
As soon as practicable after the effective date of this Registration Statement and the effective time of the merger described herein.

If the securities being registered on this Form are being offered in connection with the formation of a holding company and there is compliance with General Instruction G, check the following box. $[\]$

If this Form is filed to register additional securities for an offering pursuant to Rule 462(b) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering.

If this Form is a post-effective amendment filed pursuant to Rule 462(d) under the Securities Act, check the following box and list the Securities Act registration statement number of the earlier effective registration statement for the same offering. [] _____

CALCULATION OF REGISTRATION FEE

JAMES E. ABBOTT, Esq. Carter, Ledyard & Milburn 2 Wall Street New York, New York 10005 (212) 732-3200

JORGE DEL CALVO, Esq. Pillsbury Madison & Sutro LLP 2550 Hanover Street Palo Alto, California 94304 (650) 233-4537

Title of Each Class of Securities to be Registered	Amount to be Registered	Proposed Maximum Offering Price Per Unit	Proposed Maximum Aggregate Offering Price	Amount of Registration Fee(5)
Common Stock, par value \$1.00 per	2,672,000 shares(1)	\$52.2379	\$139,579,811(3)	\$36,850
Common Stock, par value \$1.00 per Share	479,524 shares(2)	\$34.0662	\$ 16,335,606(4)	\$ 4,313
Totals	3,151,524 shares		\$155,915,417	\$41,163

The registrant hereby amends this Registration Statement on such date or dates as may be necessary to delay its effective date until the registrant shall file a further amendment which specifically states that this Registration Statement shall thereafter become effective in accordance with Section 8(a) of the Securities Act of 1933 or until the Registration Statement shall become effective on such date as the Commission, acting pursuant to said Section 8(a), may determine.

Table of Contents

SAVOIR TECHNOLOGY GROUP, INC.

PROXY STATEMENT/ PROSPECTUS

MERGER PROPOSED - YOUR VOTE IS VERY IMPORTANT.

Dear Fellow Stockholders:

The board of directors of Savoir Technology Group, Inc. has called a special meeting of stockholders for

2000 at which you will be asked to consider and to vote upon a proposal to merge our company with a wholly-owned subsidiary of Avnet, Inc. Avnet is one of the largest electronics components and computer products distribution businesses in the world.

The date, time and place of the special meeting are as follows:

10:00 a.m., 2000

In the merger, holders of Savoir common stock will receive, in exchange for each share they hold, between 0.15494 and 0.11452 of a share of Avnet common stock, depending upon the average closing price of Avnet common stock during the fifteen trading days ending five trading days before the date of the special meeting. Holders of Savoir series A preferred stock will receive, in exchange for each share they hold, a portion of a share of Avnet common stock equal to \$9.6581 divided by the average closing price of Avnet common stock during the five trading days before the date of the merger. Avnet will issue only full numbers of its shares to Savoir stockholders; fractional share interests will be settled in cash. The Avnet common stock is listed for trading on The New York Stock Exchange and the Pacific Exchange under the symbol "AVT." The merger will be tax-free to Savoir stockholders except to the extent that they receive cash instead of fractional Avnet shares.

Your vote is very important. We cannot proceed with the merger unless the stockholders of Savoir adopt the merger agreement. You are cordially invited to attend the special meeting. Whether or not you plan to attend, please complete and mail the enclosed proxy card to us or provide your voting instructions by telephone in accordance with the accompanying instructions. If you do not return your card or instruct your broker how to vote your shares held in your broker's name, the effect will be the same as a vote against the merger.

This proxy statement/prospectus gives you detailed information about the merger. It includes the merger agreement, which is attached as Appendix A. You can also obtain information about Savoir and Avnet from publicly available documents that they have filed with the Securities and Exchange Commission. We encourage you to read this entire document carefully.

YOUR BOARD OF DIRECTORS ENTHUSIASTICALLY AND UNANIMOUSLY SUPPORTS THIS MERGER AND RECOMMENDS THAT YOU VOTE "FOR" THE ADOPTION OF THE MERGER AGREEMENT.

- Represents the estimated maximum number of shares of the common stock of Avnet, Inc. ("Avnet") which may be issued to the holders of the common stock of Savoir Technology Group, Inc. ("Savoir") in connection with the merger and related transactions described herein, based on the highest possible exchange ratio in the merger (0.15494 of a share of Avnet common stock for each of 17,245,382 shares of Savoir common stock). Includes 1,561,575 shares of Avnet common stock which may be issued in connection with or after the merger upon exercise of currently outstanding warrants to purchase the common stock of Savoir, and 2,073,900 shares of Avnet common stock which may be issued upon the exercise of options outstanding under the Savoir 1994 Stock Option Plan.
 Represents the estimated maximum number of shares of Avnet common stock which may be issued to the holders of the series A preferred stock of Savoir in connection with the merger and related transactions described herein, based on an exchange ratio of 0.2592 shares of Avnet common stock for each of the 1,850,012 currently outstanding shares of Savoir series A preferred stock.
 Pursuant to Rule 457(f)(1) and 457(c) under the Securities Act of 1933, the proposed maximum aggregate offering price for the Savoir common stock is equal to 17,245,382 (the maximum number of shares of Savoir common stock to be converted in the merger and upon exercise of the warrants and options referred to in footnote (1), multiplied by the average (\$8.09375) of the high and low prices of a share of Savoir common stock as quoted on the Nasdaq National Stock Market on May 10, 2000.
 Pursuant to Rule 457(f)(2) under the Securities Act of 1933, the proposed maximum aggregate offering price for the Savoir series A preferred stock is equal to 1,850,012 (the number of shares of series A preferred stock currently outstanding) multiplied by \$8.83 (the book value of a share of Savoir series A preferred stock of March 31, 2000).
 Calcula

- (5) Calculated by multiplying .000264 by the proposed maximum aggregate offering price.

Please see page 11 for risk factors relating to the merger which you should

consider before you vote.

Neither the Securities and Exchange Commission nor any state securities commission has approved or disapproved of these securities or determined if this proxy statement/prospectus is truthful or complete. Any representation to the contrary is a criminal offense.

2000

and is first being distributed to stockholders of Savoir on or about

2000.

Table of Contents

[SAVOIR LOGO TO COME]

SAVOIR TECHNOLOGY GROUP, INC.

44931 Industrial Boulevard Fremont, California 94538

NOTICE OF SPECIAL MEETING OF STOCKHOLDERS

TO BE HELD ON 2000

To the Stockholders of Savoir Technology Group, Inc.:

Savoir Technology Group, Inc., a Delaware corporation, will hold a special meeting of stockholders on

2000 at 10:00 a.m., local time, at the company's offices at 44931 Industrial Boulevard, Fremont, California 94538 to vote on:

[Signature to come]
P. SCOTT MUNRO
Chairman of the Board

In the merger, Savoir stockholders generally will receive Avnet common stock (and cash instead of fractional shares). However, holders of Savoir series A preferred stock who do not vote in favor of adopting the merger agreement and who perfect their appraisal rights under Section 262 of the Delaware General Corporation Law will have the right to receive from Savoir a cash payment of the judicially determined fair value of their shares. See "Appraisal Rights" in the attached proxy statement/prospectus for a description of the procedures which must be followed to perfect appraisal rights under Section 262, a copy of which is included as Appendix E to this proxy statement/prospectus. Holders of Savoir common stock will not have appraisal rights.

person. Your vote is important. Whether or not you expect to attend the special meeting, please complete, date and sign the accompanying proxy and return it promptly in the enclosed, postage-paid envelope. You can find instructions for voting on the enclosed proxy card. Record holders may also submit their proxy with voting instructions by telephone in accordance with the instructions on the enclosed proxy card. Beneficial holders who hold shares in street name may be able to vote by telephone or through the Internet in accordance with the instructions they receive from the nominees holding their shares.

Please do not send in any of your Savoir stock certificates at this time. If the merger is completed, we will send you instructions regarding how to exchange your Savoir stock certificates for Avnet stock certificates.

Your Board of Directors unanimously recommends that you vote FOR adoption of the Amended and Restated Agreement and Plan of Merger.

- A proposal to adopt the Amended and Restated Agreement and Plan of Merger dated as of March 2, 2000, among Avnet, Inc., Tactful Acquisition Corp. and Savoir Technology Group, Inc., under which Avnet would acquire Savoir; and
- Any other business that may properly come before the special meeting or any adjournments, postponements, continuations or reschedulings of the special meeting.

2000

The Information Agent for Savoir stockholders is:

[Information Agent]

Call Toll Free [Number]

Table of Contents

Sources of Additional Information

This proxy statement/prospectus incorporates important business and financial information about Savoir and Avnet from documents that are not included in or delivered with this proxy statement/prospectus. This information is available to you without charge upon your written or oral request. You can obtain documents incorporated by reference in this proxy statement/prospectus, other than certain exhibits to those documents, by requesting them in writing or by telephone from the appropriate company at the following addresses:

By Order of the Board of Directors

[SIGNATURE TO COME]
P. SCOTT MUNRO
Secretary

If you would like to request documents, please do so by [meeting date $-\ 5$ business] in order to receive them before the special meeting.

Table of Contents

Table of Contents

Savoir Technology Group, Inc. 6550 North Loop 1604 East San Antonio, Texas 78247 Attention: Terry Johnson (210) 247-1125

Avnet, Inc. 2211 South 47th Street Phoenix, Arizona 85034 Attention: Raymond Sadowski (480) 643-2000

-iii-

Table of Contents

Questions and Answers About the Merger

	Page
Sources of Additional Information	ii
Questions and Answers About the Merger	iv
Summary	1
Selected Financial Information	7
Summary Historical and Pro Forma Per Share Data	8
Risk Factors	11
Cautionary Statement Regarding Forward-Looking Statements	12
The Special Meeting	13
General; Date, Time and Place	13
Record Date; Vote Required	13
Voting and Revocation of Proxies	14
Solicitation of Proxies	14
Letter of Transmittal	15
Security Ownership of Certain Beneficial Owners and Management	15
Market Price and Dividend Information	17
The Companies	18 18
Savoir	18 18
Avnet	20
The Merger General	20
Background to the Merger	20
Savoir's Reasons for the Merger; Recommendations of the Savoir Board of	20
Directors	22
Avnet's Reasons for the Merger	23
Opinion of Savoir's Financial Adviser	23
Interests of Certain Persons in the Merger and Possible Conflicts of	
Interest	27
Public Trading Markets	29
Accounting Treatment	29
The Merger Agreement	29
General	29
Terms of the Merger	29
Closing; Effective Time of the Merger	30
Certificate of Incorporation and Bylaws of the Surviving Corporation	30
Directors and Officers of the Surviving Corporation	30
Representations and Warranties	31
Conditions to the Merger Covenants	32 32
Termination	35
Termination Fees and Expenses	35
Amendment, Extension and Waiver	36
Other Agreements	36
Material Federal Income Tax Consequences of the Merger	38
Description of Avnet Common Stock	39
Description of Savoir Common Stock and Series A Preferred Stock	39
Comparison of Shareholder Rights	39
Voting on Business Combinations	40
State Takeover Legislation	40
Appraisal Rights	43

Amendments to Charters	44
Amendments to By-Laws	44
Duration of Proxies	45
Shareholder Action by Written Consent	45
Director Nominations and Shareholder Proposals	45
Special Shareholder Meetings	46
Removal of Directors	46
Vacancies	46
Indemnification of Directors and Officers	47
Limitation of Personal Liability of Directors	48
Dividends	49
Appraisal Rights	49
Experts	52
Validity of Shares	52
Additional Information for Savoir Stockholders	52
Where You Can Find More Information	53
Appendix A — Amended and Restated Agreement and Plan of Merger	A-1
Appendix B — Stock Option Agreement	B-1
Appendix C — Inducement Agreement	C-1
Appendix D — Opinion of Alliant Partners	D-1
Appendix E — Section 262 of the Delaware General Corporation Law	E-1
Appendix F — Savoir 1999 Annual Report on Form 10-K/A	F-1

Q: Who are the parties to this merger?

A: Tactful Acquisition Corp., a wholly-owned subsidiary of Avnet, Inc., will merge into Savoir Technology Group, Inc. and cease to exist. Savoir will remain after the merger as the surviving corporation and become a wholly-owned subsidiary of Avnet.

Q: What will Savoir stockholders receive?

A: Holders of Savoir common stock will receive, for each share of Savoir common stock, between 0.15494 and 0.11452 of a share of Avnet common stock, depending upon the average of the closing prices of Avnet common stock during the fifteen consecutive trading days ending five trading days before the date of the special meeting.

Holders of Savoir series A preferred stock (other than holders who perfect their appraisal rights under Delaware law) will receive, for each share of series A preferred stock, a portion of a share of Avnet common stock equal to \$9.6581 divided by the average of the closing prices of Avnet common stock during the five consecutive trading days ending on the trading day before the effective date of the merger.

Q: Why is Savoir merging?

A: The board of directors of Savoir believes that Savoir, by becoming a part of Avnet, should be able to:

- reduce its costs by combining its operations with Avnet's established infrastructure,
- expand its position with its customers by combining its marketing power and product offerings with those of Avnet, and
- benefit from the strength and experience of Avnet's senior management team.

- i v -

Table of Contents

Q: Why is the Savoir board of directors recommending that I vote for adoption of the merger agreement?

A: In reaching its decision to approve the merger agreement and the merger and to recommend that Savoir stockholders adopt the merger agreement, the Savoir board of directors consulted with Savoir management, as well as with Savoir's financial and legal advisors, and considered the terms of the proposed merger agreement and the transactions contemplated by the merger agreement. In addition, the Savoir board of directors considered each of the items set forth on pages to . Based on those consultations and considerations, the Savoir board of directors unanimously approved the merger agreement and the merger, and believes that the terms of the merger agreement and the merger are advisable and fair to, and in the best interests of, Savoir and its stockholders.

Q: What should I do?

A: After you have carefully read this proxy statement/prospectus, please mail your signed proxy card to us in the enclosed envelope, or submit your proxy with voting instructions by telephone in accordance with

the instructions on the accompanying proxy card, so that your shares will be voted at the special meeting.

- Q: If my shares are held in "street name" by my broker, will my broker vote my shares for me?
- A: Your broker will not be able to vote your shares without instructions from you. You should instruct your broker how to vote your shares following the directions provided by your broker. Failure to instruct your broker to vote your shares is equivalent to voting against the merger agreement.
- Q: Can I change my vote after I have submitted my proxy with voting instructions?
- A: Yes. There are three ways in which you may revoke your proxy and change your vote at any time before the special meeting:
 - You may send a written notice of revocation to the Corporate Secretary of Savoir at 44931 Industrial Boulevard, Fremont, California 94538, in time for it to be received before the meeting.
 - You may complete and submit by mail a new proxy card or submit your proxy with new voting instructions by telephone. Your latest dated proxy actually received by Savoir by mail or telephone before the special meeting will be recorded and any earlier votes will be revoked.
 - You may attend the special meeting and vote in person. However, simply attending the special meeting without voting will not revoke your proxy.

If you have instructed a broker to vote your shares which the broker holds of record, you must follow directions received from your broker to change or revoke your previous voting instructions.

- V -

Table of Contents

Summary

This summary, together with the preceding Questions and Answers section, highlights the important information contained in this proxy statement/ prospectus but may not contain all of the information that may be important to you. We urge you to read carefully this entire proxy statement/ prospectus and the other documents to which it refers to understand fully the merger. See "Where You Can Find More Information" on page . Each item in this summary includes a page reference directing you to a more complete description of that item.

The Companies

Savoir Technology Group, Inc. (page)

Savoir is a value-added wholesale distributor of commercial mid-range servers, peripheral equipment (including wireless networking equipment, storage products, printers and terminals) and software. Management of Savoir believes that it is one of the leading distributors of the commercial mid-range servers product lines of International Business Machines Corporation. The principal executive offices of Savoir are at 44931 Industrial Boulevard, Fremont, California 94538, and its telephone number at that address is (510) 413-0120.

```
Avnet, Inc. (page )
```

Avnet is one of the world's largest industrial distributors of electronic components and computer products, with net sales for its fiscal year ended July 2, 1999 of \$6.35 billion. It has distribution operations in the Americas, Europe, South Africa and the Asia/ Pacific region. The principal executive offices of Avnet are at 2211 South 47th Street, Phoenix, Arizona 85034, and its telephone number at that address is (480) 643-2000.

The Merger (page)

We are proposing a merger in which Tactful Acquisition Corp., a wholly-owned subsidiary of Avnet, will merge into Savoir and will cease to exist, and Savoir will remain as the surviving corporation. After the merger, Savoir will be a wholly-owned subsidiary of Avnet.

What You Will Receive (page

In the merger, each share of Savoir common stock will be converted into the right to receive between 0.15494 and 0.11452 of a share of Avnet common stock, depending upon the average of the closing prices of Avnet common stock during the fifteen trading days ending five trading days before the date of the Savoir special meeting. The following table illustrates how the average closing price of Avnet common stock will affect the exchange ratio:

Q: Should I send in my Savoir stock certificates now?

A: No. After the merger has been completed, you will receive instructions for exchanging your Savoir stock certificates for Avnet stock certificates. Please do not send in your stock certificates with your proxy card.

Q: How many Savoir shares will be converted into Avnet shares?

A: currently outstanding shares of Savoir common stock, and 1,850,012 currently outstanding shares of Savoir series A preferred stock, will be converted into Avnet common stock. Also, a maximum of additional shares of Savoir common stock could be issued before the merger upon exercise of employee stock options and warrants and would if so issued also be converted into Avnet common stock.

Q: Will the merger be consummated if stockholders of Savoir do not approve it?

A: No. The merger agreement must be adopted by a majority of the votes represented by all shares of Savoir series A preferred stock and common stock outstanding on the record date for the special meeting, voting together as a single class. If the stockholders of Savoir do not approve of the merger, it will not be consummated.

Q: Will the shares of Avnet common stock issued in the merger be listed on the New York Stock Exchange?

A: Yes. The Avnet common stock is currently listed on the New York Stock Exchange and the Pacific Exchange under the symbol "AVT." After the merger, the Avnet common stock (including the shares issued in the merger) will continue to be listed on both exchanges.

Q: Am I entitled to appraisal rights?

A: Holders of Savoir common stock are not entitled to appraisal rights in connection with the merger. Only holders of Savoir series A preferred stock will have appraisal rights. We describe the procedures for exercising appraisal rights in this proxy statement/ prospectus and we attach the provisions of Delaware law that govern appraisal rights as Appendix E to this proxy statement/prospectus.

Q: What are the tax consequences of the merger to Savoir stockholders?

A: The exchange of Savoir shares for Avnet shares in the merger will be a tax-free transaction for federal income tax purposes for Savoir stockholders, except to the extent that they receive any cash instead of fractional Avnet shares.

Q: Whom should Savoir stockholders call with questions?

A: Savoir stockholders should call Terry Johnson, Senior Vice President of Savoir, at (210) 247-1125.

To illustrate, assuming an average closing price for Avnet common stock of \$59, a Savoir stockholder who owns 1,000 shares of Savoir common stock would receive 133 shares of Avnet common stock and a cash payment in lieu of a 0.0508 fractional share of Avnet common stock.

In the merger, each share of Savoir series A preferred stock, other than shares held by stockholders who perfect their appraisal rights under Delaware law, will be converted into the right to receive a portion

1

Table of Contents

of a share of Avnet common stock equal to \$9.6581 divided by the average of the closing prices of Avnet common stock during the five trading days ending on the trading day before the effective date of the merger.

Recommendation to Savoir Stockholders (page

The board of directors of Savoir has determined that the merger is fair to and in the best interests of its stockholders, and unanimously recommends that you vote "FOR" the adoption of the merger agreement.

Opinion of Financial Adviser (pages

Alliant Partners, financial adviser to Savoir, delivered an opinion to the board of directors of Savoir as to the fairness of the consideration in the merger from a financial point of view. The full text of this opinion is attached as Appendix D to this proxy statement/prospectus and you should read it carefully in its entirety to understand the procedures followed, assumptions made, matters considered and limitations on the review undertaken by Alliant in providing its opinion.

Interests of Certain Persons in the Merger and Possible Conflicts of Interest (pages

In considering the recommendation of the Savoir board of directors that the merger agreement be adopted, Savoir stockholders should be aware that a number of Savoir executive officers and directors have interests in the merger that are, or may be, different from the interests of other Savoir stockholders. They include the following:

Closing Price of Avnet Common Stock

Number of Avnet Shares Received for Each Savoir Share

0.11452

Greater than \$68.5472

Greater than or equal to \$50.6654 and less than or equal to \$68.5472

\$7.85 Average Closing Price

Less than \$50.6654 0.15494 • P. Scott Munro, the Chairman of the Board and Chief Executive Officer of Savoir, has an employment agreement and an executive retention agreement with Savoir under which, upon a change of control (such as the merger), Savoir will forgive a debt which Mr. Munro owes to Savoir in the amount of \$3,600,000 plus accrued interest, and if within twelve months after the change of control Mr. Munro's employment is terminated without cause, or he resigns for good reason, Mr. Munro will receive a severance payment equal to 200% of the sum of his annual salary (which salary is currently \$495,000) plus his maximum incentive bonus (currently \$270,000). Under the agreements, if the benefits payable to Mr. Munro following a change of control would be subject to a "golden parachute" excise tax, Mr. Munro may require Savoir to reduce the amount paid to him to the extent necessary to maximize his after-tax income.

At the time of the merger agreement, Mr. Munro entered into a consulting and noncompetition agreement with Savoir and Tactful which provides that, as of and from the effective time of the merger, Mr. Munro's employment with Savoir will terminate and he will be entitled to all amounts and benefits payable to him under his employment agreement and executive retention agreement upon a termination after a change of control, that is, Savoir will forgive his debt in full and he will receive a severance payment equal to \$1,530,000, all subject to reduction at the request of Mr. Munro in order to maximize his after-tax income.

2

Table of Contents

- The merger will be a change in control of Savoir under the employment agreements of Carlton Joseph Mertens II, Savoir's President and Chief Operating Officer, Terry Johnson, Savoir's Chief Financial Officer and Robert O'Reilly, Savoir's Senior Vice President of Human Resources, triggering rights of those officers, including the right to receive cash payments if their employment is terminated without cause at any time or if there are material reductions in their responsibilities following the merger, and the immediate vesting of all unvested employee stock options they hold.
- Upon completion of the merger, Avnet will assume Savoir's obligations under Savoir's 1994 Stock Option Plan, so that the outstanding options to purchase Savoir common stock will become options to purchase Avnet shares in a number and at an exercise price adjusted to reflect the exchange ratio for the merger.

The Special Meeting

Time, Place and Matters to Be Voted Upon (page)

The special meeting of Savoir's stockholders will be held on

2000 at 10:00 a.m., local time, at Savoir's offices at 44931 Industrial Boulevard, Fremont, California 94538. At the special meeting, you will be asked:

• Avnet has agreed to preserve indemnification rights of Savoir's directors and officers for serving in their capacities as such and has agreed to maintain their directors' and officers' liability insurance coverage for up to four years following the merger.

Record Date and Vote Required (page)

You may cast one vote for each share of common stock of Savoir, and 1.1953125 votes for each share of Savoir series A preferred stock, that you owned at the close of business on [record date], the record date for the special meeting. Adoption of the merger agreement requires approval by a majority of the votes

represented by all shares of Savoir series A preferred stock and Savoir common stock outstanding on the record date voting together as a single class.

On [record date] there were shares of Savoir common stock, and shares of Savoir series A preferred stock outstanding and entitled to vote. Thus, a maximum of votes may be cast at the special meeting. As of the record date, all directors and officers of Savoir beneficially owned an aggregate outstanding shares of Savoir common stock (not including shares issuable upon exercise of stock options), which entitle them to cast % of all votes which approximately may be cast at the special meeting. Each executive officer and director of Savoir granted Avnet an irrevocable proxy to vote all of his shares in favor of the adoption of the merger agreement. See "Other Agreements" (page).

The Merger Agreement

The merger agreement is attached to this proxy statement/prospectus as Appendix A. Please read the merger agreement carefully and in its entirety. It is the legal document that governs the merger.

Effective Time of the Merger (page)

The merger will become effective as soon as practicable after all the conditions to the completion of the merger have been satisfied or waived. Although we can give you no assurances, we currently expect that the merger will become effective shortly after the conclusion of the special meeting if the Savoir stockholders adopt the merger agreement.

Conditions to the Merger (page)

The completion of the merger depends on a number of conditions being satisfied or, where applicable, waived, including:

- 1. to adopt the merger agreement; and
- 2. to act on any other business that may properly come before the special meeting or any adjournments, postponements, continuations or reschedulings of it.

3

Table of Contents

- approval of the merger agreement by the stockholders of Savoir;
- the accuracy in all material respects of the representations and warranties of Avnet and Savoir and compliance in all material respects by each of them with of all its covenants under the merger

agreement;

• receipt of legal opinions confirming that the United States federal income tax treatment will be as

Regulatory Matters (page)

described in this document; and

We furnished information and materials concerning Avnet, Savoir and the proposed merger to the Antitrust Division of the Department of Justice and the Federal Trade Commission under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, and the required waiting period has terminated. However, the Antitrust Division and the Federal Trade Commission still have the power to challenge the merger on antitrust grounds before or after the merger is completed. No other material federal or state regulatory requirements remain to be complied with, and no other federal or state regulatory approval must be obtained, in connection with the merger.

Termination (page)

Avnet and Savoir may agree in writing to terminate the merger agreement at any time before completing the merger, even after the stockholders of Savoir have approved it.

Also, either of us may decide, without consent of the other, to terminate the merger agreement if:

· receipt of all approvals required by law.

Savoir may terminate the merger agreement without the consent of Avnet for the purpose of entering into an agreement for an acquisition proposal superior to the merger with Avnet.

Avnet may terminate the merger agreement without the consent of Savoir if the board of directors of Savoir withdraws or adversely amends its recommendation of the merger agreement to the Savoir stockholders or recommends to them another acquisition proposal.

Fees and Expenses (page)

Whether or not the merger is completed, Avnet and Savoir each will pay their own fees and expenses.

Termination Fee (page)

Savoir will pay Avnet \$750,000 in cash if:

- any governmental entity has issued a final and non-appealable order prohibiting any of the transactions contemplated by the merger agreement;
- the merger has not been completed by September 15, 2000;
- Savoir fails to hold the special meeting or the Savoir stockholders do not adopt the merger agreement;
 or
- the other breaches any representation, warranty, covenant, agreement, condition or obligation in the merger agreement, and the breach or failure of the covenant, agreement, condition or obligation has or is likely to have a material adverse effect on the breaching party.

4

Table of Contents

Savoir will pay to Avnet an amount in cash equal to \$4,500,000, less any previous payment of \$750,000 as described above, if the merger agreement is terminated:

- Avnet terminates the merger agreement because the Savoir stockholders have not adopted the merger agreement, or because of a breach by Savoir of any of its representations, warranties and covenants in the merger agreement which has or would reasonably be likely to have a material adverse effect on Savoir, or because the board of directors of Savoir has withdrawn or adversely amended its recommendation of the merger agreement to the Savoir stockholders or recommended another acquisition proposal for Savoir; or
- Savoir terminates the merger agreement because the Savoir stockholders have not adopted the merger agreement, or Savoir terminates the merger agreement for the purpose of entering into a superior acquisition proposal;

and, in either case, within one year of such termination, Savoir enters into an agreement to effect another acquisition proposal.

Other Agreements (page)

At the same time as Avnet and Savoir first executed and delivered the merger agreement,

- · as described above; or
- by Avnet if the merger has not been consummated by September 15, 2000;

Material Federal Income Tax Consequences of the Merger (page $\,$)

In general, Savoir stockholders will recognize no gain or loss for federal income tax purposes as a result of the merger.

Avnet and Savoir have conditioned the merger on their receipt of legal opinions that the merger will be treated for federal income tax purposes as a "reorganization" within the meaning of Section 368(a) of the Internal Revenue Code of 1986.

Avnet and Savoir each could choose to waive receipt of its legal opinion. However, if the receipt of a legal opinion is waived and there is a material difference in the tax consequences to you from what we have described in this document, we will recirculate revised proxy materials and resolicit the vote of stockholders.

This tax treatment may not apply to certain Savoir stockholders and may depend on your specific situation and on variables not within our control. We urge you to consult your own tax adviser for a full understanding of the tax consequences of the merger.

Comparison of Shareholders' Rights (page)

The rights of Savoir stockholders currently are governed by Delaware law, the Savoir certificate of incorporation and the Savoir by-laws. Savoir stockholders will receive Avnet common stock in the merger, so that after the merger their rights as Avnet shareholders will be governed by New York law, the Avnet certificate of incorporation and the Avnet by-laws, which differ in material respects from Savoir's certificate of incorporation and by-laws.

Listing of Avnet Common Stock (page)

The Avnet common stock is listed on the New York Stock Exchange and the Pacific Exchange under the symbol "AVT." After consummation of the merger, the Avnet common stock will continue to be listed on these Exchanges.

Appraisal Rights (page)

A holder of series A preferred stock who delivers to Savoir a written demand for appraisal before the vote at the special meeting, who does not vote in favor of adopting the merger agreement, and who complies with all other applicable requirements of Delaware law will have the right to receive payment in cash of the "fair value" of his shares of series A preferred stock, as determined in a judicial proceeding in the Court of Chancery of the State of Delaware. The procedure for perfecting appraisal rights is

5

Table of Contents

summarized under the caption "Appraisal Rights" and the pertinent provision of Delaware law, Section 262 of the General Corporation Law, is included as Appendix E to this proxy statement/ prospectus. Holders of Savoir common stock will not have appraisal rights with respect to the merger.

Resales of Avnet Common Stock

The shares of Avnet common stock to be issued in the merger have been registered under the Securities Act of 1933 and therefore may be resold without restriction by all former stockholders of Savoir who are not deemed to be "affiliates" of either Avnet or Savoir. An affiliate of Savoir who receives shares of Avnet common stock in the merger would be unable to resell those shares in the absence of registration of such resales under the Securities Act or the availability of an exemption from such registration.

Selected Financial Information

The following tables present selected historical financial information of Savoir and Avnet. This information is based on the consolidated financial statements of Avnet and Savoir which are incorporated by reference in this proxy statement/prospectus, and should be read together with such historical financial statements and the notes thereto. See "Sources of Additional Information" (page) and "Where You Can Find More Information" (page).

Avnet, Inc.

- Avnet and Savoir entered into an option agreement under which Savoir granted Avnet an option to purchase up to 2,023,435 shares of Savoir common stock, representing about 15% of the outstanding shares of Savoir common stock, at an exercise price of \$6.83 per share, and
- Avnet entered into an inducement agreement with directors and executive officers holding 2,067,536 shares Savoir common stock, representing about 15% of the outstanding shares of Savoir common stock, in which those stockholders granted Avnet a proxy to vote their shares in favor of adoption of the merger agreement and granted Avnet an option to purchase those shares at an exercise price of \$7.85 per share.

	Half year	ended		Fi	scal year ende	d	
	December 31, 1999	January 1, 1999	July 2, 1999	June 26, 1998	June 27, 1997	June 28, 1996	June 30, 1995
			(In millions, ex	cept per share a	mounts)		
Income:			(,		
Sales	\$3,757.1	\$3,108.5	\$6,350.0	\$5,916.3	\$5,390.6	\$5,207.8	\$4,300.0
Gross profit	513.5(2)	460.7(3)	948.6(4)	980.4(5)	961.8	969.1	816.4
Income taxes	30.5(2)	32.4(3)	200.8(4)	115.9(5)	130.7	136.8	103.1
Net income	37.3(2)	42.2(3)	174.5(4)	151.4(5)	182.8	188.3	140.3
Financial position (at	` ,	, ,	` ,	. ,			
end of period):							
Working capital	\$1,586.8	\$1,482.0	\$1,517.5	\$1,461.3	\$1,319.0	\$1,293.9	\$1,057.1
Total assets	4,392.4	2,790.1	2,984.7	2,733.7	2,594.1	2,521.7	2,125.6
Total debt	1,686.3	884.0	791.5	810.9	514.6	497.5	419.5
Shareholders' equity	1,731.1	1,293.3	1,397.6	1,315.9	1,502.2	1,505.2	1,239.4
Per share:							
Basic earnings(1)	\$ 0.98(2)	\$ 1.17(3)	\$ 4.90(4)	3.85(5)	\$ 4.29	\$ 4.34	\$ 3.44
Diluted earnings(1)	0.97(2)	1.16(3)	4.86(4)	3.80(5)	4.25	4.31	3.32
Book value	40.60	36.80	39.70	36.09	36.55	34.67	30.38
Cash dividends	0.30	0.30	0.60	0.60	0.60	0.60	0.60

7

Table of Contents

- (1) Earnings per share have been restated to conform with the provisions of SFAS No. 128, "Earnings Per Share."
- (2) After incremental special charges associated with (a) the reorganization of Avnet's Electronics Marketing European operations amounting to \$6.1 pre-tax, (b) the integration of Marshall Industries amounting to \$18.4 pre-tax, (c) the reorganization of Avnet's Electronics Marketing Asian operations amounting to \$5.4 pre-tax and (d) the consolidation of Avnet's Electronics Marketing European warehousing operations and its lawsuit against Wyle Laboratories, Inc. amounting to \$4.2 pre-tax. The effect of these charges was to decrease income before taxes, net income and diluted earnings per share by \$34.1, \$21.5 and \$0.56, respectively. Of the \$34.1 pre-tax charge, \$23.8 is included in operating expenses and \$10.3 is included in cost of sales.
- (3) After \$26.5 pre-tax (\$7.9 cost of sales and \$18.6 operating expenses) and \$15.7 after-tax (\$0.43 per share on a diluted basis) of incremental special charges associated principally with the reorganization of Avnet's Electronics Marketing Group in Europe.
- (4) After (a) first half special charges discussed in footnote (3), (b) net gain on the sale of Avnet's former Allied Electronics subsidiary amounting to \$252.3 pre-tax and (c) charges recorded in connection with the disposition of the Avnet Setron catalog operation in Germany amounting to \$42.8 pre-tax. The effect of these items was to increase income before taxes, net income and diluted earnings per share by

approximately \$183.0, \$64.0 and \$1.78, respectively. Of the \$183.0 pre-tax gain related to special items, charges of \$56.1 are included in operating expenses and \$13.1 are included in cost of sales, and the \$252.3 net pre-tax gain on the sale of Allied Electronics is shown separately in Avnet's consolidated statement of income for fiscal 1999.

(5) After (a) gain on sale of Channel Master amounting to \$33.8 pre-tax, (b) operating expenses relating to the divestiture of Avnet Industrial, the closure of Avnet's corporate headquarters in Great Neck,

Savoir Technology Group, Inc.

New York and the anticipated loss on the sale of company-owned real estate amounting to \$13.3 in the aggregate and (c) \$35.4 pre-tax (\$9.7 cost of sales and \$25.7 operating expenses) of incremental special charges associated with the reorganization of the Electronics Marketing Group, primarily in the Americas. The effect of these items was to decrease income before taxes, net income and diluted earnings per share by approximately \$14.9, \$12.5 and \$0.32, respectively.

Year ended December 31,

	1999(3)	1998(4)	1997	1996	1995
		(In millions,	except per shar	e amounts)	
Income:	4-0- 0	* =00.0	****	****	****
Sales	\$767.2	\$593.3	\$237.9	\$131.7	\$106.5
Gross profit	84.0	67.2	32.8	17.3	13.0
Income taxes	6.9	8.3	0.3	0.3	_
Net income (loss) before extraordinary item	1.8	8.6	3.3	2.3	(5.1)
Financial position (at end of period):					, ,
Working capital	5.6	3.5	6.5	7.4	7.3
Total assets	324.7	308.9	186.9	63.3	35.9
Total debt	28.4	23.3	37.9	11.4	7.2
Convertible preferred stock subject to redemption	14.6	15.7	18.1	_	_
Stockholders' equity	98.2	77.8	28.9	15.7	11.0
Per share attributable to common stockholders:					
Basic earnings (loss) before extraordinary item(1)	\$ 0.03	\$ 0.37	\$ 0.57	\$ 0.55	\$(1.36)
Diluted earnings (loss) before extraordinary item(1)	0.03	0.34	0.55	0.52	(1.36)
Book value per common share on an as-if converted					
basis(2)	7.34	7.35	6.15	3.50	2.74
• •					

Summary Historical and Pro Forma Per Share Data

The following table presents selected pro forma per share data for Avnet and Savoir on an historical basis, for Avnet and Savoir on a pro forma basis giving effect to the merger, and for Savoir common stock on a pro forma equivalent share basis. The pro forma equivalent share data for Savoir common stock is based on an exchange of 0.13170 of a share of Avnet common stock for each outstanding share of Savoir common stock in the merger. This would be the exchange ratio if the average closing price of a share of Avnet common stock during the fifteen trading days ending five trading days before the date of the special meeting were \$59.6063, the midpoint between the range of \$50.6654 and \$68.5472 in the exchange ratio formula. Management of Avnet and Savoir are currently assessing non-recurring merger charges which could be material to the combined companies' results of operations and financial condition for the period in which the charges occur. No estimate of these charges is reflected in the Historical and Pro Forma Per Share Data provided below. In addition, the Historical and Pro Forma Per Share Data provided below.

8

Table of Contents

reflect assumptions that (a) the value of Avnet shares for the purpose of determining the purchase price is \$59.6063, (b) outstanding Savoir stock options and warrants are exercised immediately prior to the merger and the exercise proceeds are used to repurchase shares of Savoir stock and (c) convertible preferred stock subject to redemption is exchanged for common stock at the beginning of the period. See "What You Will Receive" on page .

The merger will be accounted for under the purchase accounting method and the pro forma data are derived in accordance with that method. The following information is based upon the historical financial statements of Avnet and Savoir and the related notes incorporated by reference into this proxy statement/prospectus, and should be read together with such historical financial statements and the notes thereto. This information is not necessarily indicative of the results of the future operations of Avnet after the merger or the actual results that would have occurred had the merger been consummated prior to the periods indicated.

Historical And Pro Forma

Per Share Data

- (1) Earnings per share have been restated to conform with the provisions of SFAS No. 128, "Earnings Per Share."
- (2) Includes shares of convertible preferred stock subject to redemption on an as-if converted basis into shares of common stock.
- (3) Income and earnings per share for 1999 include pre-tax charges totalling \$11.3 relating to the restructuring of corporate functions and the recognition of impaired assets. The effect of these charges was to decrease income before income taxes, net income using a normalized tax rate of 49% and diluted earnings per share by \$11.3, \$5.7 and \$0.44, respectively.
- (4) Income and earnings per share for 1998 are before an extraordinary item relating to prepayment of debt.

	Half year ended December 31, 1999	Fiscal year ended July 2, 1999
Avnet, Inc.		
Book value (at end of period):		
Historical	\$40.60	\$39.70
Pro forma combined	41.49	40.81
Pro forma per equivalent share of Savoir common stock	5.46	5.37
Cash dividends declared:		
Historical	0.30	0.60
Pro forma combined(1)	0.28	0.57
Pro forma per equivalent share of Savoir common stock	0.04	0.08
Earnings from continuing operations (diluted):		
Historical(2)	0.97	4.86
Pro forma combined	0.89	4.99
Pro forma per equivalent share of Savoir common stock(3)	0.12	0.66
avoir Technology Group, Inc. book value per common share on an as-if converted basis(5) (at end of period) luted earnings per share from continuing operations attributable to common stockho	olders	\$7.34 0.03
	Marc 200	
Closing price: Avnet common stock Savoir common stock Per share	\$65	7/16

Table of Contents

(1) Based on the total historical dividends paid by Avnet (Savoir has not paid dividends), divided by the pro forma weighted average number of shares outstanding as if the merger had occurred at the beginning

9

Table of Contents

Risk Factors

Savoir common stockholders cannot be sure what the market value of their Avnet common stock will be after the merger has been completed.

Avnet and Savoir agreed that the value of a share of Savoir common stock on March 2, 2000, the date on which they signed the merger agreement, was \$7.85. However, if the market value of Avnet common stock on the effective date of the merger is lower than the average closing price used to determine the merger exchange ratio, holders of Savoir common stock would receive Avnet common stock in the merger with an initial value less than \$7.85 per share of Savoir common stock. For example, if the average closing price of Avnet common stock used to calculate the exchange ratio is less than \$50.6654, the exchange ratio will be fixed at 0.15494. If this occurs, and the market price of Avnet common stock at the effective time of the merger is less than \$50.6654, the initial value of the Avnet common stock to be received by Savoir common stockholders will be less than \$7.85 for each share of Savoir common stock. Even if the average trading price used to determine the exchange ratio is greater than \$50.6654, Savoir common stockholders could still receive Avnet common stock with an initial value of less than \$7.85 for each share of Savoir common stock.

The market prices of Avnet common stock and Savoir common stock, and the value of Avnet common stock relative to Savoir common stock, may be substantially different on the date the merger agreement was signed, the date of this proxy statement/ prospectus, the date of the special meeting and the effective date of the merger. For example, during the first calendar quarter of 2000, the closing price of a share of Avnet common stock ranged from a low of \$50 to a high of \$73 1/2, as reported for New York Stock Exchange composite transactions, and the closing price of a share of Savoir common stock ranged from a low of \$6 to a high of \$8 5/8, as quoted on the Nasdaq National Stock Market. See "Market Price and Dividend Information" on page . These market prices may vary depending upon changes in the business, operations or prospects of Savoir or Avnet, market assessments of the likelihood that the merger will be consummated and the timing thereof, general market and economic conditions and other factors both within and beyond the control of Savoir and Avnet.

Avnet may not realize fully the cost savings and other benefits it expects to realize as a result of the merger. This may adversely affect Avnet's earnings and financial condition.

To achieve the benefits which Avnet expects from the merger, Avnet will need to integrate the operations of Savoir into its own operations quickly and efficiently. The integration of Savoir into Avnet will be complex and time consuming, and will require substantial attention from Avnet management. The diversion of management attention from other matters, and any difficulties encountered in the integration process, could have a material adverse effect upon the sales, level of expenses, operating results and financial condition of Avnet. See "The Merger — Savoir's Reasons for the Merger; Recommendations of the Savoir Board of Directors" on

page and "The Merger —
Avnet's Reasons for the Merger" on
page .

After the merger, customers of Avnet may seek alternative distributors.

After the merger, some customers of either Avnet and Savoir could transfer their business to another distributor. We cannot predict whether this will occur or to what extent.

After the merger, suppliers may terminate supply agreements with Avnet or add other distributors.

Some suppliers of Savoir or Avnet may terminate their supply agreements with Avnet after the merger. Alternatively, some other suppliers might consider adding additional distributors for their products. We cannot predict whether this will occur or to what extent.

11

Table of Contents

Cautionary Statement Regarding Forward-Looking Statements

This proxy statement/ prospectus contains forward-looking statements with respect to the financial condition, results of operations and business of each of Savoir and Avnet. These statements may be made directly in this proxy statement/ prospectus referring to Savoir or Avnet, or may be incorporated by reference in this proxy statement/ prospectus to other documents filed with the Securities and Exchange Commission by Savoir or Avnet, and may include statements for the period following the consummation of the merger. You can find many of these statements by looking for words like "believes," "expects," "anticipates," "estimates" or similar expressions in this document or in documents incorporated by reference.

These forward-looking statements are subject to numerous assumptions, risks and uncertainties. Factors that may cause actual results to differ materially from those contemplated by the forward-looking statements include the following:

- (2) See footnotes 2, 3 and 4 accompanying the Selected Financial Information of Avnet on page 7 of this proxy statement/ prospectus.
- (3) Before non-recurring charges and credits directly attributable to the merger.
- (4) The last trading day preceding the date on which the proposed general terms of the merger were publicly announced. The closing price of Savoir common stock per equivalent share of Avnet common stock was calculated by multiplying the closing price of Avnet common stock by the exchange ratio which would have applied if the merger had become effective on that date. The Avnet common stock is listed on the New York Stock Exchange and the Pacific Exchange, and the Savoir common stock is quoted on the Nasdaq National Stock Market. See "Market Price and Dividend Information" on page .
- (5) Includes shares of Savoir convertible preferred stock subject to redemption on as-if converted basis into shares of Savoir common stock.

Because forward-looking statements are subject to risks and uncertainties, Avnet's actual results may differ materially from those expressed or implied by them. We caution you not to place undue reliance on these statements, which speak only as of the date of this proxy statement/ prospectus or the date of any document incorporated by reference in this proxy statement/ prospectus.

All subsequent written and oral forward-looking statements attributable to Savoir and Avnet or any person acting on their behalf are expressly qualified in their entirety by the cautionary statements contained or referred to in this section. Neither Savoir nor Avnet undertakes any obligation to update publicly or revise any forward-looking statements, whether as a result of new information, future events or otherwise.

12

Table of Contents

The Special Meeting

General; Date, Time and Place

We are soliciting the accompanying proxy on behalf of the board of directors of Savoir Technology Group, Inc. for use at the special meeting of Savoir stockholders scheduled to be held on

2000 at 10:00 a.m., local time, at Savoir's offices at 44931 Industrial Boulevard, Fremont, California 94538. The purpose of the special meeting is to consider and to vote upon the adoption of the merger agreement and to transact any other business that may properly come before the meeting or any adjournments, postponements, continuations or reschedulings of the meeting. The expenses of the solicitation of proxies for the special meeting will be borne by Savoir.

The board of directors of Savoir Technology Group, Inc. has unanimously approved the merger agreement and recommends that Savoir stockholders vote FOR the adoption of the merger agreement.

Record Date; Vote Required

Record Date. The Savoir board of directors has fixed the close of business on [record date] as the record date for the determination of stockholders entitled to receive notice of and to vote at the special meeting. On the record date, there were shares

of Savoir common stock and

of series A preferred stock issued and outstanding. You may vote at the meeting only if you owned Savoir common stock or series A preferred stock at the close of business on the record date. You are entitled to one vote for each share of common stock and 1.1953125 votes for each share of series A preferred stock which you then owned.

Quorum. The presence either in person or by properly executed proxies of the holders of Savoir's common stock and

series A preferred stock representing a majority of the votes which may be cast at the special meeting will constitute a quorum at the special meeting.

Vote Required. The adoption of the merger agreement requires the affirmative vote of a majority of the votes represented by all shares of Savoir common stock and series A preferred stock outstanding at the close of business on the record date, voting as a single class. That is, at least

out of a total of votes will be required.

votes

Abstentions and Broker Non-Votes. Brokers who hold shares of Savoir common stock and series A preferred stock in "street" name for customers who are the beneficial owners of such shares may not give a proxy to vote such shares with respect to the proposal to adopt the merger agreement unless the brokers receive specific voting instructions from such customers. Shares of Savoir common stock and series A preferred stock represented by proxies returned by a broker holding such shares in nominee or "street" name will be counted for purposes of determining whether a quorum exists, even if such shares are not voted. Votes which are not cast by brokers because they received no instructions from their customers are known as "broker non-votes."

Because approval of the merger agreement requires approval by a majority of the votes represented by all outstanding shares of Savoir common stock and series A preferred stock, abstentions and broker non-votes will have the same effect as negative votes. Accordingly, the Savoir board of directors urges Savoir stockholders to respond to this proxy solicitation, whether by U.S. mail via the enclosed, postage-paid envelope or by telephone toll free, or by instructions given to your broker.

Directors' and Officers' Votes. As of the record date,

issued

and outstanding shares of Savoir common stock, representing in the aggregate approximately % of the votes which may be cast at the special meeting, were owned by directors and executive officers of Savoir. Each director and executive officer of Savoir has granted to Avnet an irrevocable proxy to vote his shares in favor of the adoption of the merger agreement. See "Other Agreements" on page .

13

Table of Contents

Voting and Revocation of Proxies

Savoir stockholders of record may submit proxies by completing and mailing the proxy card that accompanies this proxy statement/ prospectus or by submitting their voting instructions by telephone. Shares of Savoir common stock and series A preferred stock represented by a proxy properly signed or submitted by telephone as described below and received by Savoir at or before the special meeting, unless subsequently revoked, will be voted

in accordance with your instructions.

To submit a written proxy by mail, a record holder of Savoir common stock or series A preferred stock should complete, sign, date and mail the proxy card provided with this proxy statement/prospectus in accordance with the instructions set forth on the card. If a proxy card is signed, dated and returned without indicating any voting instructions, shares of Savoir common stock or series A preferred stock represented by the proxy will be voted "FOR" the adoption of the merger agreement.

Instead of submitting a signed proxy card, Savoir stockholders of record may also submit their voting instructions by telephone. To submit voting instructions by telephone, stockholders should follow the instructions that are set forth on their proxy card. Each stockholder of record has been assigned a unique control number which is printed on the holder's proxy card. Stockholders who submit proxies by telephone will be required to provide their control number before their proxy will be accepted. In addition to the instructions that appear on the proxy card, step-by-step instructions will be provided by recorded telephone message, and stockholders will receive confirmation that their proxies have been successfully submitted.

Beneficial owners whose shares are held of record by a broker, nominee, fiduciary or other custodian should follow the instructions they receive from the record holder of their shares with respect to voting.

A Savoir stockholder may revoke the stockholder's proxy and change the stockholder's vote, at any time before the meeting, in any one of three ways:

- Competitive pressures among distributors of electronic components and computer products may increase significantly through industry consolidation, entry of new competitors or otherwise.
- General economic or business conditions, domestic and foreign, may be less favorable than expected, resulting in lower sales than expected.
- Costs or difficulties related to the integration into Avnet of newly-acquired businesses, and other businesses Avnet expects to acquire, may be greater than expected.
- Avnet may lose customers or suppliers as a result of the merger and other recent acquisitions by Avnet.
- Legislative or regulatory changes may adversely affect the businesses in which Avnet is engaged.
- Adverse changes may occur in the securities markets.
- · Changes in interest rates and currency fluctuations may reduce profit margins.
- Avnet may be adversely affected by the allocation of products by suppliers.

The Savoir board of directors is not currently aware of any business to be acted upon at the special meeting other than the merger agreement. If, however, other matters are properly brought before the meeting, the persons appointed as proxies will have discretion to vote or to act according to their best judgment, unless otherwise indicated on any particular proxy. The persons appointed as proxies will have discretion to vote on adjournment of the special meeting. The adjournment may be for the purpose of soliciting additional proxies. However, shares represented by proxies voting against adoption of the merger agreement will be voted against a proposal to adjourn the Savoir special meeting for the purpose of soliciting additional proxies.

Savoir will solicit proxies for the special meeting primarily by mail. However, if necessary to ensure satisfactory representation at the special meeting, Savoir may also solicit proxies by telephone, telegraph and personal interview by employees of Savoir, none of whom will receive additional compensation for such services. Savoir has retained

to

assist in the solicitation of proxies on its behalf for a fee of approximately

plus out-of-pocket expenses. Also, we request brokerage houses, nominees, fiduciaries and other custodians holding Savoir common stock or series A preferred stock of record to forward soliciting materials to beneficial owners, and will reimburse them for their reasonable expenses incurred in sending materials to beneficial owners.

14

Table of Contents

Letter of Transmittal

As soon as practicable after the effective time of the merger, we will send a letter of transmittal to each holder of Savoir common stock and series A preferred stock. The letter of transmittal will contain instructions for the surrender of Savoir stock certificates in exchange for Avnet stock certificates. You should not forward your Savoir stock certificates to us until you receive the letter of transmittal following the merger.

Security Ownership of Certain Beneficial Owners and Management

Under its option agreement with Savoir and its inducement agreement with Savoir stockholders, Avnet has the right in some circumstances to purchase up to 4,090,971 shares of Savoir common stock, or about 30% of the Savoir common stock outstanding before exercise of those rights. See "Other Agreements." Avnet filed a Schedule 13D in connection with the option agreement and the inducement agreement in which Avnet disclaimed beneficial ownership of shares of Savoir stock subject to those agreements.

The following table sets forth information as to the beneficial ownership of Savoir's common stock and series A preferred stock as of the record date by: (i) each person known to Savoir to beneficially own more than five percent (5%) of Savoir's common stock or series A preferred stock; (ii) each of Savoir's directors and executive officers; and (iii) all executive officers and directors of Savoir as a group:

- The stockholder may send a written notice of revocation to the Corporate Secretary of Savoir at 44931 Industrial Boulevard, Fremont, California 94538, in time for it to be received before the meeting.
- The stockholder may complete and submit by mail a new proxy card or submit a proxy with new voting instructions by telephone. The latest dated proxy actually received by mail or telephone before the special meeting will be recorded and any earlier votes will be revoked.
- The stockholder may attend the special meeting and vote in person. However, simply attending the special meeting without voting will not revoke the stockholder's proxy.

Table of Contents

Common stock		Series A preferred stock		
Shares beneficially owned(1)	Percent(2)	Shares beneficially owned(1)	Percent	
1,877,430(3)	13.60%	207,000	11.19%	
1,487,112(4)	10.34%	533,000	28.81%	
1,259,697(5)	8.95%	387,012	20.92%	
1,200,000(6)	8.90%			
1,185,000(7)	8.78%			
960,826(8)	7.00%	160,000	8.65%	
841,573(9)	6.23%			
516,458(10)	3.82% 3.41%			
460,000(11) 24,888(12) 30,855(13)	*			
	Shares beneficially owned(1) 1,877,430(3) 1,487,112(4) 1,259,697(5) 1,200,000(6) 1,185,000(7) 960,826(8) 841,573(9) 516,458(10) 460,000(11)	Shares beneficially owned(1) Percent(2) 1,877,430(3) 13.60% 1,487,112(4) 10.34% 1,259,697(5) 8.95% 1,200,000(6) 8.90% 1,185,000(7) 8.78% 960,826(8) 7.00% 841,573(9) 6.23% 516,458(10) 3.82% 460,000(11) 24,888(12) *	Shares beneficially owned(1) Percent(2) Shares beneficially owned(1) 1,877,430(3) 13.60% 207,000 1,487,112(4) 10.34% 533,000 1,259,697(5) 8.95% 387,012 1,200,000(6) 8.90% 1,185,000(7) 8.78% 960,826(8) 7.00% 160,000 841,573(9) 6.23% 516,458(10) 3.82% 3,41% 460,000(11) 24,888(12) *	

	Common st	Series A preferred stock		
Name of Beneficial Owner	Shares beneficially owned(1)	Percent(2)	Shares beneficially owned(1)	Percent
Guy M. Lammle — Director	33,537	*		
J. Larry Smart — Director	26,000(14)	*		
K. William Sickler — Director	7,500(15)	*		
Angelo Guadagno — Director	2,500(16)	*		
All executive officers and directors as a group (9	, , ,	16.95%		
persons)	2,286,738(17)			

- * Less than one percent (1%).
- (1) Information with respect to beneficial ownership is based upon information obtained from the stockholders and their SEC filings and from Savoir's transfer agent. To the knowledge of Savoir, unless otherwise indicated in the footnotes below, each person and entity named in the table has sole voting and sole dispositive power with respect to all shares shown for such person or entity in this table, subject to community property laws where applicable. Beneficial ownership is determined in accordance with the rules of the Securities and Exchange Commission. Shares of common stock issuable upon conversion of shares of series A preferred stock, upon exercise of stock options currently exercisable or first becoming exercisable within 60 days of the record date, or upon exercise of warrants currently exercisable or first becoming exercisable within 60 days of the record date, are deemed to be outstanding and to be beneficially owned by the person entitled to convert or exercise for the purpose of computing that person's percentage ownership of Savoir common stock, but are not treated as outstanding for the purpose of computing the percentage ownership of any other person. Each share of series A preferred stock is currently convertible at any time into 1.1953125 shares of the common stock and is entitled to vote together with the common stock as a single class (1.1953125 votes per share of series A preferred stock).
- (2) Based on shares of common stock outstanding as of the record date plus the issuable shares as described in footnote (1).
- (3) Includes 247,430 shares of common stock issuable upon conversion of 207,000 shares of series A preferred stock, and 103,500 shares of common stock issuable upon exercise of warrants. Information is

based on a Schedule 13G/ A dated February 15, 2000, filed by Astoria Capital Partners, L.P. with the SEC.

- (4) Includes 637,101 shares of common stock issuable upon conversion of 533,000 shares of series A preferred stock, and 362,500 shares of common stock issuable upon exercise of warrants. Information is based on a Schedule 13G dated February 9, 2000, filed by Robert Fleming, Inc. with the SEC.
- (5) Includes 462,600 shares of common stock issuable upon conversion of 387,012 shares of series A preferred stock, and 193,506 shares of Common Stock issuable upon exercise of warrants. Information is based on a Schedule 13G/ A filed by Strome Susskind Investment Management, L.P. with the SEC.
- (6) Information is based on a Schedule 13D dated June 5, 1998 filed by Mr. Harkins with the SEC, as well as issuances of shares to Mr. Harkins thereafter.
- (7) Information is based on a Schedule 13D dated June 5, 1998 filed by Mr. Gunnells with the SEC, as well as issuances of shares to Mr. Gunnells thereafter.
- (8) Includes 191,250 shares of common stock issuable upon conversion of 160,000 shares of series A preferred stock, and 80,000 shares of common stock issuable upon exercise of warrants. Information is based on a Schedule 13G/ A dated February 15, 2000, filed by ROI Capital Management, Inc. with the SEC.
- (9) Information is based on a Schedule 13D/A filed on May 2, 2000 by Bear Stearns & Co. Inc.

16

Table of Contents

- (10) Includes 25,000 shares issuable upon exercise of stock options.
- (11) Information is based on a Schedule 13D dated June 5, 1998, filed by Mr. Mertens with the SEC.
- (12) Includes 18,599 shares issuable upon exercise of stock options.
- (13) Includes 28,750 shares issuable upon exercise of stock options.
- (14) Includes 12,500 shares issuable upon exercise of stock options.

Market Price and Dividend Information

The table below sets forth, for the calendar quarters indicated, the high and low sales prices per share for Savoir common stock as quoted in the Nasdaq National Stock Market, and for Avnet common stock as reported for New York Stock Exchange composite transactions:

- (15) Comprised of shares issuable upon exercise of stock options.
- (15) Comprised of shares issuable upon exercise of stock options.
- (16) Includes 97,500 shares issuable upon exercise of stock options.

The last sale price for a share of Avnet common stock as reported for New York Stock Exchange composite transactions on [record date] was

and the last sale price for a share of Savoir common stock on the Nasdaq National Stock Market on [record date] was

The closing sales prices per share of Savoir common stock and Avnet common stock on March 1, 2000, the last trading day before public announcement of the merger, were \$8 1/4 and \$65 7/16, respectively.

Avnet has paid a cash dividend on its common stock of 15 cents

per share during each calendar quarter in 2000, 1999 and 1998. In the same period Savoir paid no dividends on its common stock.

As of [record date], there were approximately stockholders of record of Savoir common stock and stockholders of record of Savoir series A preferred stock, as shown on the records of Savoir's transfer agent.

17

Table of Contents

The Companies

Savoir

Savoir is a value-added wholesale distributor of commercial mid-range servers, peripheral equipment (including wireless networking equipment, storage products, printers and terminals) and software. Savoir believes that it is one of the leading distributors of IBM's commercial mid-range servers product lines. The principal executive offices of Savoir are at 44931 Industrial Boulevard, Fremont, California 94538, and its telephone number at that address is (510) 413-0120.

For additional information about Savoir and its business, see its Annual Report on Form 10-K/A for the year ended December 31, 1999, a copy of which is attached as Appendix F to this proxy statement/prospectus.

Avnet

Avnet is one of the world's largest industrial distributors of electronic components and computer products, with sales for its fiscal year ended July 2, 1999 of \$6.35 billion. Including the fiscal 1999 sales of Marshall Industries, which Avnet acquired on October 20, 1999 (as described below), as well as the acquisitions of SEI Macro Group, Eurotronics B.V., PCD Italia S.r.l., Matica S.p.A. and Integrand Solutions Pty. Limited, Avnet's pro forma sales for fiscal 1999 were more than \$8.07 billion. Avnet is a vital link in the chain that connects suppliers of semiconductors, interconnect products, passives and electromechanical devices to original equipment manufacturers ("OEMs") and contract manufacturers that design and build the electronics equipment for end-market use, and to other industrial customers. In addition, Avnet distributes a variety of computer products and services to both the end user and the reseller channels. Through its electronic components distribution activities, Avnet acts as an extension of a supplier's sales force by marketing products to a larger base of customers than individual suppliers could do economically. While many suppliers can only serve a few hundred of the larger OEMs and contract manufacturers, Avnet is authorized to sell products of more than 100 of the world's leading component manufacturers to a global customer base of approximately 100,000 OEMs and contract manufacturers. Avnet ships electronic components as it receives them from Avnet's suppliers or with assembly or other value added. As part of its distribution activities, Avnet adds various processes that customize products to meet individual OEM customer specifications, and it provides material management and logistic

On October 20, 1999, Avnet acquired Marshall Industries and its wholly-owned subsidiary, Sterling Electronics. Marshall Industries was one of the world's largest distributors of electronic components and computer products. Marshall Industries had sales of \$1.72 billion in its last full fiscal year ended May 31, 1999. This acquisition was the largest in the history of the electronics distribution industry and makes Avnet's Electronics Marketing group the largest electronic components distribution business in the Americas, with fiscal 1999 sales of \$5.17 billion on a pro forma basis. The acquisition of Marshall Industries strengthens Avnet's line card and provides Avnet with a significant opportunity to enhance its operational efficiency with significant cost savings.

On January 3, 2000, Avnet completed its acquisition of the SEI Macro Group, an electronic components distributor headquartered in the United Kingdom, and Eurotronics B.V. (which does business under the name SEI), a pan-European electronic components distributor headquartered in the Netherlands. The combined annual sales of Eurotronics and the SEI Macro Group are approximately \$750 million.

One of Avnet's critical strengths is the breadth and quality of the suppliers whose products it carries. Listed below are the major product categories and the major suppliers in each category:

		Savoir Common Stock		
Calendar Quarter	High	Low	High	Low
1998				
First Quarter	\$12 3/4	\$9 5/8	\$66 1/4	\$57
Second Quarter	13 7/16	9 1/8	64 5/16	53 11/16
Third Quarter	12 3/4	4 5/16	58 1/2	35 1/4
Fourth Quarter	9 1/2	5 5/16	60 5/8	34 15/16
1999				
First Quarter	\$12 1/2	\$8 5/8	\$60 15/16	\$35 5/8
Second Quarter	10 3/4	7 1/2	51	34
Third Quarter	9 3/4	7 3/4	52 7/16	41 1/16
Fourth Quarter	8 3/8	4 5/16	60 1/2	37 5/16
2000				
First Quarter	8 5/8	6	73 1/2	50
Second Quarter (through May 10)	8 7/8	6 7/8	81 1/8	57 15/16

18

Table of Contents

• Semiconductors: Avnet's major suppliers of semiconductors are Advanced Micro Devices, Agilent Technologies, Analog Devices, Atmel, Conexant, Cypress, Dallas Semiconductor, Hitachi, Integrated Device Technology, Intel, Intersil, Lattice, Level One, Linear Technology, LSI Logic,

During fiscal year 1999 Avnet operated in two industry segments as described below.

Electronics Marketing ("EM") is Avnet's largest operating group, with fiscal year 1999 sales of \$4.80 billion, representing approximately 76% of Avnet's consolidated sales. EM is comprised of three regional operations: EM Americas, EM EMEA (Europe, Middle East and Africa) and EM Asia. EM distributes electronic components (semiconductors, connectors, passives and electromechanical devices), and EM offers an array of value-added services to its customers, such as inventory

Computer Marketing ("CM") is an international distributor of computer products to value-added resellers and end users focusing primarily on middle- to high-end, value-added computer products and services. CM's 1999 sales were \$1.55 billion, representing approximately 24% of Avnet's consolidated sales. CM is broadly split between two independent business units, Avnet Computer and Hall-Mark Global Solutions.

Avnet Computer sells industry leading high-end systems, mid-range servers, workstations, PCs, software, storage, networking, peripherals and services to end user customers. Avnet Computer is one of North America's leading technology solutions integrators, providing hardware, software, and services for corporate-wide applications.

Hall-Mark Global Solutions concentrates on sales of computer systems, peripherals and components to the reseller channel.

Management of Avnet believes that Hall-Mark Global Solutions is the industry's leading technical distributor of open systems in support of a limited line card of the foremost computer and peripherals manufacturers, which include Compaq, Hewlett-Packard, IBM and Intel. Hall-Mark Global Solutions provides those manufacturers' products to value-added resellers, along with complementary value-added solutions and in-house engineering support, complex systems integration and configuration services.

Hall-Mark Integrated Solutions was created in fiscal 1999 to focus on entry-level to mid-range servers, specifically for the Windows NT server market. The mission of this business unit is to recruit and serve Windows NT platform VARs by marketing bundled solutions from Avnet's key suppliers. These are new customers for CM who reach the fast growing small to medium business sector. Sales have been growing rapidly and Avnet believes they will surpass \$100 million in fiscal 2000.

CM has also created Avnet Direct, an Internet commerce company which sells computer systems to businesses and individuals on the World Wide Web. These computer systems are configured from thousands of name-brand computer and peripheral equipment products and software carried in CM's inventories.

Effective in fiscal year 2000, Avnet has formed a new group, Avnet Applied Computing ("AAC"). AAC focuses the resources of three existing business units, CM's OEM Business unit and EM's Personal Computer Components and OEM Systems units, with combined revenues of over \$1 billion. AAC provides technical solutions that include software, engineering services, leadership product, supply chain management, financing and physical distribution and integration of the end product.

Avnet has approximately 10,800 employees globally and maintains locations in the United States, Canada, Mexico, Europe, Asia, Australia, New Zealand, South Africa and South America. Avnet's

principal executive offices are located at 2211 South 47th Street, Phoenix, Arizona 85034, telephone number (480) 643-2000.

For additional information about Avnet and its business, see the documents identified in "Where You Can Find More Information" on page .

The Merger

This discussion of the merger and the principal terms of the merger agreement is subject to, and qualified in its entirety by reference to, the Amended and Restated Agreement and Plan of Merger dated as of March 2, 2000 by and among Avnet, Tactful Acquisition Corp. and Savoir, a copy of which is attached to this proxy statement/prospectus as Appendix A and incorporated herein by reference.

General

We are furnishing this proxy statement/prospectus to holders of Savoir common stock and series A preferred stock in connection with the solicitation of proxies by the board of directors of Savoir for use at its special meeting of stockholders, and at any adjournments, postponements, continuations or reschedulings of the meeting. At the special meeting, the Savoir stockholders will consider and vote upon a proposal to adopt the merger agreement.

The merger agreement provides for the merger of Tactful Acquisition Corp., a wholly-owned subsidiary of Avnet, into Savoir. At the effective time of the merger, the separate corporate existence of Tactful will cease and Savoir will remain as the surviving corporation in the merger and become a wholly-owned subsidiary of Avnet. Each share of Avnet common stock issued and outstanding before the merger will remain an issued and outstanding share of Avnet common stock after the merger. Savoir common stock and series A preferred stock outstanding at the effective time of the merger will be converted by the merger into Avnet common stock as described below. The merger will become effective when a certificate of merger is filed with the Secretary of State of Delaware, or at a subsequent date or time that Savoir and Avnet agree on and specify in the certificate of merger. The transaction is intended to qualify as a tax-free "reorganization" within the meaning of Section 368(a) of the Internal Revenue Code of 1986, as amended, for federal income tax purposes. See "Material Federal Income Tax Consequences of the Merger" on page

As of March 2, 2000, the date on which the merger agreement was first executed and delivered, Avnet and Savoir agreed that the value of a share of Savoir common stock was \$7.85. In the merger, each share of Savoir common stock will be converted into a right to receive between 0.15494 and 0.11452 of a share of common stock of Avnet. Within that range, the conversion ratio will be the quotient of \$7.85 divided by the average of the closing prices of a share of Avnet common stock during the fifteen trading days ending five trading days before the date of the special meeting. The closing price per share of Avnet common stock for New York Stock Exchange composite transactions on [record date] was

Because the merger would be a "change in control" as defined in Savoir's certificate of incorporation, the holders of series A preferred stock have a right to receive

\$9.6581 per share in cash or property having that value. Therefore, in the merger, each share of series A preferred stock will be converted into a right to receive a portion of a share of Avnet common stock equal to the quotient of \$9.6581 divided by the average of the closing prices of a share of Avnet common stock during the five trading days ending on the trading day immediately preceding the effective date of merger.

Background to the Merger

The board of directors and management of Savoir discussed the possibility of a merger or sale of Savoir at various times from January 1, 1998 through the signing of the merger agreement with Avnet. During that period, Savoir made and received initial overtures about the possibility of merging Savoir with a number of distribution companies. To facilitate those discussions, Savoir retained several investment banks, including Broadview Associates (to explore European opportunities), The Robinson-Humphrey

20

Table of Contents

Company, LLC and the Tucker Anthony Cleary Gull division of Tucker Anthony Inc. On several occasions, Savoir entered into confidentiality arrangements and exchanged financial information. However, in all cases, the discussions did not result in an offer for a merger or sale of Savoir.

A core element of Savoir's growth strategy had been to buy other businesses in exchange for shares of Savoir common stock. By October 1999, a decline in the market price of Savoir common stock had limited Savoir's ability to proceed with that strategy. That limit on Savoir's growth strategy was an important factor in the decision of the board of directors, in October 1999, to seek a buyer for Savoir.

Following the board's decision, Savoir's financial advisor, Alliant Partners, started to prepare a list of possible buyers and a confidential offering memorandum describing Savoir. The buyer list was divided into three categories:

Micron Semiconductors, Motorola, National Semiconductor, NEC, ON Semiconductor, Philips Semiconductor, ST Microelectronics, Texas Instruments, Toshiba and Xilinx.

- Computer Products: Avnet's major suppliers of computer products are Cabletron, Compaq Computer Corporation, Computer Associates, Hewlett-Packard, IBM, Intel and Oracle.
- Connectors: Avnet's major suppliers of connectors are AMP, Amphenol/Bendix, FCI, ITT Cannon, Molex, Thomas and Betts Components, and 3M.
- Passives, Electromechanical and Other: Avnet's major suppliers of these products are Aromat, AVX, Bourns, Kemet, Lambda, Leach, Murata, Nichicon, Panasonic, Pulse, Teledyne and Vishay.

At the same time, Alliant telephoned a number of approved prospects to determine their interest in a transaction with Savoir. Savoir and Alliant finished a confidential memorandum on November 19, 1999 and sent copies to 12 interested prospects.

During December 1999, representatives of Savoir and Alliant held a series of meetings with prospective buyers. The prospects indicating the greatest initial interest were either financial buyers or operating companies not directly in the computer distribution business. On December 20, 1999, a financial buyer made a written offer to buy Savoir at \$7 per share of common stock and \$ per share of series A preferred stock. Alliant described this offer, and gave a detailed report on the status of the sale process, to the board of directors of Savoir on December 21, 1999. The offer contained a number of contingencies that were not acceptable to Savoir, and its board directed Alliant to continue discussions with this prospect with the goal of bringing the board an acceptable offer.

Another financial buyer started a comprehensive due diligence review of Savoir, in anticipation of making a cash offer to acquire it. After completing the due diligence review, the second financial buyer concluded that it would not be able to finance a purchase of Savoir, and withdrew from the process without making an offer.

Savoir also held a series of meetings with a large company primarily engaged in the business of contract manufacturing. Those meetings, which started in early December 1999 and accelerated at the end of the year and first part of January 2000, focused on the value of Savoir and the structure of a possible transaction to buy Savoir. The prospect's key concern before moving into formal negotiations and making an offer to buy Savoir was to understand IBM's receptiveness to such a transaction, and the opportunities to expand Savoir's relationship with IBM in the future. On January 17, 2000, senior executives of IBM, the prospect and Savoir held a conference call to explore IBM's position regarding the contemplated transaction. The representatives of IBM participating in the call did not provide the prospect with the insight it deemed necessary to proceed with a transaction, and the prospect withdrew from the process without making an offer.

At the request of Savoir, representatives of Tucker Anthony made contact with Avnet on December 20, 1999. Avnet expressed interest in discussing a purchase of Savoir, and Andrew Bryant, Senior Vice President of Avnet, Richard Hamada, Vice President of Avnet, John Clark, Vice President of Avnet, Ed Kamins, Vice President of Avnet, and P. Scott Munro, Chairman of the Board and Chief Executive Officer of Savoir, Carlton Joseph Mertens II, President and Chief Operating Officer of Savoir, and Dennis Polk, then Chief Financial Officer of Savoir met in Phoenix on January 5, 2000. As a result of the meeting, Avnet expressed a preliminary interest in buying Savoir and Avnet started a due diligence review of Savoir. While Avnet continued with their internal processes, Alliant continued dialogue with several other prospective buyers.

21

Table of Contents

On January 17, 2000, Avnet provided a draft letter of intent, outlining a non-binding offer to acquire Savoir in a cash tender offer. A draft definitive agreement was prepared by counsel to Savoir based on the letter of intent; however, Avnet indicated that it was not prepared to proceed, until it had detailed discussions and negotiations with IBM. During the

subsequent two-week period, there were limited discussions between the parties.

On February 1, representatives of Avnet and Savoir held a conference call in an attempt to understand the status of the contemplated transaction. During the call Avnet explained that it was still in discussions with IBM, and that the management of Avnet was not yet committed to proceeding with a transaction. On February 11, 2000, Avnet management requested that a meeting be held in Phoenix as soon as possible to discuss a revised proposal to acquire Savoir. On February 14, 2000, a meeting was held at Avnet's corporate headquarters among David Birk, Avnet's Senior Vice President and General Counsel, Raymond Sadowski, its Senior Vice President and Chief Financial Officer, and Timothy Grant, its Corporate Acquisitions Director, and representatives of Alliant and Tucker Anthony. At the meeting, Avnet indicated that it had concluded its discussions with IBM and offered to buy Savoir at a price of \$7.00 per share of Savoir common stock (and a corresponding price per share of series A preferred stock), payable in Avnet common stock. The board of directors of Savoir considered the offer on February 15, 2000, and rejected it as not reflecting adequate value for Savoir's stockholders.

Representatives of Alliant and Avnet thereafter held a conference call on February 16, 2000 at which time Alliant advised that a minimum of \$7.80 to \$7.85 would be required to move forward with the transaction. Later that day, Avnet proposed improving their offer to \$7.85 per share, payable in Avnet common stock. Avnet's improved offer included a "collar" limiting the effect of an increase or decrease in the market price of Avnet common stock between the signing of an agreement and the closing of the merger. The Savoir board of directors considered the revised offer on February 17, 2000, and authorized management to proceed with a definitive agreement.

Counsel to Avnet presented a draft merger agreement to Savoir on February 22, 2000. From that time through the signing, representatives of Avnet and Savoir engaged in detailed negotiations of the merger agreement, an option agreement between Savoir and Avnet, an inducement agreement among Savoir executive officers and directors and Avnet, and a consulting and noncompetition agreement among Tactful, Savoir and P. Scott Munro. At the same time, Avnet and Savoir each continued their due diligence review of the other.

On February 29, 2000, the Avnet board of directors considered and approved the principal terms of the transaction and authorized Avnet management to finalize the negotiations and execute definitive documents. Final negotiations continued throughout the day on March 1, 2000, focusing primarily on the method of calculating the portion of a share of Avnet common stock to be issued for each share of Savoir common stock, and the terms of the price collar. Avnet and Savoir reached an agreement in principle on those issues that evening, and the Savoir board of directors then met to review the status of the transaction. At the meeting, Alliant orally presented their opinion that the proposed transaction with Avnet was fair to the stockholders of Savoir. The Savoir board then unanimously approved moving forward with the transaction and authorized the officers of Savoir to negotiate and sign the definitive agreements. Final negotiations continued that evening and the merger agreement, option agreement, inducement agreement and consulting agreement were signed on March 2, 2000.

After the merger agreement was signed by Avnet, Savoir and Tactful, they agreed to make various immaterial changes to the agreement and then signed an amended and restated merger agreement which is attached as Appendix A to this proxy statement/prospectus.

Savoir's Reasons for the Merger; Recommendations of the Savoir Board of Directors

The Savoir board of directors has unanimously determined that the merger is fair to, and in the best interest of, Savoir and its stockholders, and the board has unanimously adopted and approved the merger agreement. Accordingly, the Savoir board of directors recommends that Savoir stockholders vote in favor of the approval of the merger agreement.

22

Table of Contents

In reaching its determination, the Savoir board of directors concluded that the long-term value of stockholders' investment in Savoir, converted into Avnet common stock as a result of the merger, was likely to be greater than the value of their Savoir stock without the merger. Further, the Savoir board was of the view that the opportunities created by the merger to increase stockholder value more than offset the risks associated with the merger. In reaching these conclusions, the Savoir board gave significant consideration to a variety of factors, including those described below. In view of the wide variety of factors bearing on its decision, the Savoir board did not consider it practical to, nor did it attempt to, quantify or assign relative or specific weight to factors it considered in reaching its decision. Also, individual directors may have given differing weights to different factors. The Savoir board received the advice of its senior management, financial advisors and independent counsel throughout its consideration of the merger agreement.

In reaching the determination to approve the merger and the transactions contemplated by the merger, the Savoir board considered a number of factors, including:

- distribution companies;
- contract manufacturers; and
- firms (referred to below as "financial buyers") which are in the business of acquiring, developing and reselling businesses.

The Savoir board of directors unanimously recommends that Savoir stockholders vote "FOR" the adoption of the merger agreement.

Avnet's Reasons for the Merger

Avnet's goal is to provide the highest value relationships to its customers, suppliers, employees and shareholders, globally. Avnet's board of directors believes that the merger will serve Avnet's objectives to each of these constituencies.

Avnet believes that the merger will create a combined company that is the number one IBM mid-range value-added distributor in North America, and that Savoir's contract manufacturing business will give Avnet a complementary additional level of

technical sales and engineering expertise. Avnet also believes that Savoir and Avnet share a common culture which is focused on customer satisfaction.

Avnet believes that its shareholders will benefit from the merger because of anticipated cost savings and synergies as a result of combining the companies, for example, in consolidating the facilities and computer systems of the companies, and that shareholders will benefit from increased sales and earnings in the future.

The Avnet board of directors has unanimously approved the merger agreement.

Opinion of Savoir's Financial Adviser

Alliant Partners, an investment banking firm headquartered in Palo Alto, California, has provided Savoir with financial advisory services related to acquisitions since 1995. In November 1999, Savoir's board of directors decided to expand the scope of Alliant's activities to seek a strategic buyer for Savoir. In January 2000, Savoir and Alliant agreed in writing that Alliant would render an opinion regarding the fairness of a possible sale of Savoir, from a financial point of view, to the stockholders of Savoir.

On March 1, 2000, after the market close, the Savoir board of directors met and approved the merger. At this meeting, Alliant delivered to the Savoir board by teleconference its opinion that as of March 1, 2000, and based on the matters described therein, the total consideration received by Savoir was

23

Table of Contents

fair, from a financial point of view, to the stockholders of Savoir. Alliant noted that its presentation on the financial terms of the merger at the time of signing was based on an average closing price of Avnet common stock as of March 1, 2000, as calculated in the merger agreement. No limitations were imposed by Savoir on the scope of Alliant's investigations or the procedures to be followed by Alliant in rendering its opinion. The value per share of Savoir common stock at signing was determined through negotiations between the management of Savoir and Avnet. In furnishing its opinion, Alliant was not engaged as an agent or fiduciary of Savoir's stockholders or any other third party.

The full text of the Alliant opinion, as updated on April 19, 2000, which sets forth, among other things, assumptions made, matters considered and limitations on the review undertaken, is attached to this proxy statement/prospectus as Appendix D, and is incorporated to this proxy statement/prospectus by reference. Stockholders of Savoir are urged to read the Alliant opinion in its entirety. The Alliant opinion does not address the relative merits of the merger and any other transactions or business strategies discussed by the Savoir board as alternatives to the merger agreement or, except with respect to the fairness of the total consideration received by Savoir, from a financial point of view, to Savoir stockholders the underlying business decision of the Savoir board to proceed

with or effect the merger. This summary of the Alliant opinion is qualified in its entirety by reference to the full text of the Alliant opinion.

In connection with the preparation of the Alliant opinion, Alliant, among other things: (i) reviewed the terms of the merger agreement; (ii) reviewed the Savoir Form 10-K for the fiscal year ended December 31, 1998, including the audited financial statements included therein; the Savoir Form 10-Q for the nine months ended September 30, 1999, including the unaudited financial statements included therein; and the Savoir press release of its earnings for the fourth quarter and fiscal year ended December 31, 1999, including the unaudited financial statements included therein; (iii) reviewed certain internal financial and operating information relating to Savoir prepared by Savoir management; (iv) participated in discussions with Savoir management concerning the operations, business strategy, financial performance and prospects for Savoir; (v) reviewed the recent reported closing prices and trading activity for Savoir common stock; (vi) compared certain aspects of the financial performance of Savoir with comparable public companies; (vii) analyzed available information, both public and private, concerning other mergers and acquisitions comparable in whole or in part to the merger; (viii) reviewed the Avnet Form 10-K for the fiscal year ended July 2, 1999, and the Avnet Form 10-Q for the six months ended December 31, 1999; (ix) participated in discussions with Avnet management concerning the operations, business strategy, financial performance and prospects for Avnet and its strategic rationale for the merger; (x) reviewed the recent reported closing prices and trading activity for Avnet common stock; (xi) reviewed recent equity analyst reports covering Avnet; (xii) considered the effect of the merger on the future financial performance of the consolidated entity; (xiii) participated in discussions related to the merger among Avnet, Savoir and their financial and legal advisors; and (xiv) conducted other financial studies, analyses and investigations as Alliant deemed appropriate for the purposes of its opinion.

In conducting its review and arriving at its opinion, Alliant relied upon and assumed the accuracy and completeness of the financial statements and other information provided by Savoir and Avnet or otherwise made available to Alliant and did not assume responsibility independently to verify such information. Alliant further relied upon the assurances of Savoir's and Avnet's management that the information provided was prepared on a reasonable basis in accordance with industry practice and that such parties were not aware of any information or facts that would make the information provided to Alliant incomplete or misleading. Without limiting the generality of the foregoing, for the purpose of its opinion, Alliant assumed that neither Savoir nor Avnet was a party to any pending transaction, including external financing, recapitalization, acquisitions or merger discussions, other than the merger or in the ordinary course of business. Alliant also assumed that the merger would be free of federal tax to the holders of Savoir common stock and series A preferred stock.

In arriving at its opinion, Alliant did not perform any appraisals or valuations of specific assets or liabilities of Savoir or Avnet and was not furnished with any such appraisals or valuations.

Without limiting the generality of the foregoing, Alliant did not undertake any independent analysis of any pending or threatened litigation, possible unasserted claims or other contingent liabilities, to which Savoir, Avnet or any of their respective affiliates was a party or may be subject and, at Savoir's direction and with its consent, the Alliant opinion made no assumption concerning and therefore did not consider, the possible assertion of claims, outcomes or damages arising out of such matters. Although developments following the date of its opinion may affect the opinion, Alliant assumed no obligation to update, revise or reaffirm.

Following is a summary explanation of the various sources of information and valuation methodologies employed by Alliant in conjunction with rendering its opinion to the Savoir Board.

Comparable Company Analysis. Alliant compared certain financial information and valuation ratios relating to Savoir to corresponding publicly available data and ratios from a group of selected publicly traded companies deemed comparable to Savoir. The comparable companies selected included twelve publicly traded companies in the business of microcomputer and/or electrical components distribution. Financial information reviewed by Alliant included each company's: Enterprise Value, calculated as the market capitalization of the selected company, plus such company's long term debt, less such company's excess cash; LTM (latest twelve months) Revenue, EBITDA (Earnings Before Interest, Taxes, Depreciation and Amortization) and Earnings as reported as of the date of the Alliant opinion; prior year Revenue Growth Rate; and Projected Earnings Growth. Comparable companies included: Arrow Electronics, Inc.; Avnet, Inc.; Bell Microproducts, Inc.; Ingram Micro, Inc.; Jaco Electronics; Merisel, Inc.; MicroAge, Inc.; Nu Horizons Electronics; Pioneer Standard Electronics; SED International Holdings; All American Semiconductor; and Tech Data Corp.

The comparable companies had an Enterprise Value/ LTM Revenue ratio range of 0.01 to 0.56, with a weighted narrow average (narrow average excludes the highest and lowest estimates) of 0.25; Enterprise Value/ LTM EBITDA ratio range from 1.99 to 25.06, with a weighted narrow average of 9.63; and Equity Value/ Earnings ratio range of 8.22 to 46.77, with a weighted narrow average of 15.33. After making certain adjustments for differences in performance, liquidity and size and applying an acquisition control premium, this analysis yielded an implied Savoir Enterprise Value of \$132.3 million.

No company utilized as a comparison in the Comparable Company Analysis, with the exception of Avnet, is identical to Savoir or Avnet. In evaluating the comparable companies, Alliant made judgments and assumptions with regard to industry performance, general business, economic, market and financial conditions and other matters, many of which are beyond the control of Savoir or Avnet.

Comparable Transaction Analysis. Alliant reviewed eleven comparable merger and acquisition transactions from February, 1997, through the present, which involve sellers that share many characteristics with Savoir, including products offered and business model. These comparable transactions of companies in the microcomputer and/or electrical components distribution sectors are: (i) Arrow Electronics' acquisition of Consan; (ii) Arrow Electronics' acquisition of Support Net; (iii) Arrow Electronics' acquisition of Support Net; (iii) Arrow Electronics' acquisition of Scientific & Business Minicomputer; (iv) Bell Microproducts' acquisition of Tenex (a division of Axidata); (vi) Compucom Systems'

acquisition of Dataflex Corp; (vii) Miami Computer Supply Corp.'s acquisition of Dreher Business Products Corp; (viii) MicroAge's acquisition of Pride Technologies; (ix) Pioneer Standard Electronics' acquisition of Dickens Data Systems; (x) Savoir's acquisition of Target Solutions; and (xi) Savoir's acquisition of Varcity/ Unidirect Corp.

Estimated multiples paid in the comparable transactions were based on information obtained from public filings, public company disclosures, press releases, industry and popular press reports, databases and other sources. The Price/ Revenue multiples of the eleven transactions range from 0.05 to 0.35, with a weighted narrow average of 0.22. Price/ EBITDA and Price/ Earnings multiples were not used in the valuation analysis, as reliable EBITDA or Earnings numbers for these completed transactions were not available. This analysis yielded an implied Savoir Enterprise Value of \$129.8 million.

25

Table of Contents

No company, transaction or business in the Comparable Company Analysis or the Comparable Transaction Analysis is identical to Savoir or the Merger. Accordingly, an analysis of the results of the foregoing is not entirely mathematical; rather it involves complex considerations and judgments concerning differences in financial and operating characteristics and other factors that could affect the acquisition, public trading and other values of the comparable companies, comparable transactions or the business segment, company or transactions to which they are being compared.

Discounted Cash Flow Analysis. Alliant estimated the present value of the projected future cash flows of Savoir on a stand-alone basis using internal financial planning data prepared by Savoir management for the years ending December 31, 2000 through December 31, 2002, and a discount rate of 17.5%. Alliant obtained a terminal valuation based on the Gordon Growth Model, which assumed a long-term growth rate of 15%. This analysis yielded an estimated present value of Savoir's Enterprise Value of \$102.6 million.

Conclusion. While the foregoing summary describes certain analyses and factors that Alliant deemed material in its presentation to the Savoir board of directors, it is not a comprehensive description of all analyses and factors considered by Alliant. The preparation of a fairness opinion is a complex process that involves various determinations as to the most appropriate and relevant methods of financial analysis and the application of these methods to the particular circumstances and, therefore, such an opinion is not readily susceptible to summary description. Alliant believes that its analyses must be considered as a whole and that selecting portions of its analyses and of the factors considered by it, without considering all analyses and factors, would create an incomplete view of the evaluation process underlying the Alliant opinion. Several analytical methodologies were employed and no one method of analysis should be regarded as critical to the overall conclusion reached by Alliant. Each analytical technique has inherent strengths and weaknesses, and the nature of the available information may further affect the value of particular techniques. The conclusions reached by Alliant are based on all analyses and factors taken as a whole and also on application of Alliant's own experience and judgment. Such conclusions may

involve significant elements of subjective judgment and qualitative analysis. Alliant therefore gives no opinion as to the value or merit standing alone of any one or more parts of the analysis it performed. In performing its analyses, Alliant considered general economic, market and financial conditions and other matters, many of which are beyond the control of Savoir and Avnet. The analyses performed by Alliant are not necessarily indicative of actual values or future results, which may be significantly more or less favorable than those suggested by such analyses. Accordingly, analyses relating to the value of a business do not purport to be appraisals or to reflect the prices at which the business actually may be purchased. Furthermore, no opinion is being expressed as to the prices at which shares of Avnet common stock may trade at any future time.

Pursuant to the January 3, 2000 amendment to a letter agreement dated November 1, 1997, Alliant is to receive a fee of \$125,000 for the fairness opinion rendered to the Savoir board. Savoir has also agreed to reimburse Alliant for its out of pocket expenses and to indemnify and hold harmless Alliant and its affiliates and any person, director, employee or agent acting on behalf of Alliant or any of its affiliates, or any person controlling Alliant or its affiliates for certain losses, claims, damages, expenses and liabilities relating to or arising out of services provided by Alliant as financial advisor to Savoir. The terms of the fee arrangement with Alliant, which Savoir and Alliant believe are customary in transactions of this nature, were negotiated at arm's length between Savoir and Alliant, and the Savoir Board was aware of such fee arrangements.

Alliant was retained based on Alliant's experience as a financial advisor in connection with mergers and acquisitions and in securities valuations generally, as well as Alliant's investment banking relationship and familiarity with Savoir. Alliant has provided financial advisory and investment banking services to Savoir in the past.

As part of its investment banking business, Alliant is frequently engaged in the valuation of businesses and their securities in connection with mergers and acquisitions, sales and divestitures, joint ventures and strategic partnerships, private financings and other specialized studies.

26

Table of Contents

Interests of Certain Persons in the Merger and Possible Conflicts of Interest

In considering the recommendation of the Savoir board of directors, Savoir stockholders should be aware that certain members of Savoir's management and of the Savoir board of directors have interests in the merger that are different from, or in addition to, the interests of Savoir's stockholders generally. The members of the Savoir board of directors knew about these additional interests, and considered them, when they approved the merger agreement.

Stock Options

obligations under Savoir's 1994 Stock Option Plan, so that the outstanding Savoir options will become options to purchase Avnet shares in a number and at an exercise price adjusted to reflect the exchange ratio for the merger. See "The Merger Agreement — Covenants — Stock Options" on page . In addition, Avnet will, as promptly as practicable after the closing of the merger, register the Avnet shares issuable upon exercise of the options under the Securities Act of 1933.

As of the record date for the special meeting, the executive officers and directors of Savoir were the holders of options to purchase an aggregate of

shares

of Savoir common stock at prices ranging from to

per

share. The following table sets forth information as to these Savoir options and how they will be converted into Avnet options. The information on the Savoir options below assumes that 0.1317 of a share of Avnet common stock is issued for each share of Savoir common stock in the merger. That would be the exchange ratio if the average closing price of a share of Avnet common stock for purposes of the merger agreement was \$59.6063, the midpoint between the range of \$50.6654 and \$68.5472 in the exchange ratio formula.

- Economies of Scale. The market in which Savoir operates is increasingly competitive and price sensitive. By becoming a part of Avnet, Savoir should be able to combine its marketing, operational and other personnel with those of Avnet, and be able to more efficiently serve its customers.
- Market Presence. The opportunities for Savoir's business may be limited due to its size and capital resources, and its relatively narrow range of products. As a part of Avnet, Savoir should achieve more visibility in the market.
- Management Expertise. As a part of Avnet, Savoir will have the opportunity to benefit from the strength and experience of Avnet's senior management team.

Savoir options		Avnet options	
No. of Savoir shares	Exercise price	No. of Avnet shares	Exercise price

P. Scott Munro Carlton Joseph Mertens, II Terry Johnson Robert O'Reilly

Employment Agreements

Savoir has an employment agreement and an executive retention agreement with P. Scott Munro, the Chairman of the Board and Chief Executive Officer of Savoir. Under these agreements, Mr. Munro receives a base salary of \$495,000 per year and is eligible to receive a bonus of up to \$270,000 per year based on the achievement by Savoir of certain performance goals. If Mr. Munro's employment is terminated without cause, he will be entitled to receive his base salary for twelve months following his termination. If, within twelve months following a change in control, Savior terminates Mr. Munro's employment without cause, or Mr. Munro resigns with good reason, he will be entitled to receive a severance payment equal to 200% of his annual base salary plus his annualized target bonus, and any of his employee stock options which are then unvested will vest in full (all of his options are currently vested). The merger would be a change in control as defined in Mr. Munro's employment agreement.

Also under his executive retention agreement, on May 10, 1999 Mr. Munro obtained a recourse loan from Savoir for \$2,500,000 and on September 10, 1999, he obtained a recourse loan for an additional \$1,100,000. These loans bear interest at the rate of 4.9% per year. The principal of and accrued interest

Table of Contents

the loan currently amount to \$. At

the effective time of the merger Mr. Munro's obligation to pay the then outstanding balance of the loan and accrued interest thereon will be forgiven in full.

In connection with the merger agreement, Savoir, Tactful and Mr. Munro entered into a consulting and noncompetition agreement under which, as of and from the effective time of the merger:

- Mr. Munro's employment with Savoir will terminate;
- * Under the employment agreements and related agreements of Messrs. Mertens and O'Reilly, the merger will result in the acceleration of the vesting of this currently unvested option, so that when this option is converted into an Avnet option at the effective time of the merger, it will be fully vested and exercisable.

In his consulting and noncompetition agreement, Mr. Munro also made the following undertakings with respect to a legal proceeding by Lee Adams against Savoir which is pending in the Superior Court of Orange County, California:

- he will be entitled to all amounts and benefits payable to him under his employment agreement and
 retention agreement upon a termination after a change of control, that is, Savoir will forgive his debt
 in full, he will receive a severance payment equal to \$1,530,000, and his currently unvested employee
 stock options to purchase shares of Savoir common stock will vest;
- Mr. Munro will provide management consulting services to Savoir for one year; and
- for a period of five years, Mr. Munro will not be connected in any manner with any business or entity which is engaged in, or is in competition with, the businesses conducted by Avnet, Savoir or their affiliates on the date of the consulting and noncompetition agreement.

For a description of this legal proceeding, see Item 3 of Savoir's Annual Report on Form 10-K attached to this proxy statement/prospectus as Appendix F.

Savoir also has an employment agreement with Carlton Joseph Mertens II, President and Chief Operating Officer of Savoir, which will remain in effect after the merger. Under this agreement, Mr. Mertens receives a base salary of \$350,000 per year and is eligible to receive a bonus of up to \$150,000 per year based on the achievement by Savoir of certain performance goals. If Mr. Mertens' employment is terminated without cause at any time, or if Mr. Mertens terminates his employment with Savoir at any time for certain specified reasons, he will be entitled to receive his base salary for nine months following his termination. Such reasons include:

- Mr. Munro will indemnify Avnet and Savoir for losses incurred in the Adams legal proceeding, but he will not pay the first \$200,000 of such losses and his maximum payment obligation will be \$500,000; and
- Mr. Munro waives his right to be indemnified under the merger agreement and the certificate of incorporation and by-laws of Savoir both before and after the merger, and his right to coverage under directors' and officers' liability insurance (see below), in each case to the extent of any losses arising from the Adams legal proceeding for which he is liable, up to \$500,000.

Savoir also has an employment agreement with Terry Johnson, Savoir's Chief Financial Officer, which will remain in effect after the merger. Under this agreement, Mr. Johnson receives a base salary of \$195,000 per year, and is eligible to receive a bonus payment based on Savoir's operating income. If Savoir terminates the employment of Mr. Johnson without cause at any time, or if he terminates his employment

28

Table of Contents

for any reason after December 31, 2000, he will be entitled to receive his base salary for six months following his termination.

Savoir also has an employment agreement with Robert O'Reilly, Savoir's Senior Vice President of Human Resources, which will remain in effect after the merger. Under this agreement, Mr. O'Reilly receives a base salary of \$200,000 per year and is eligible to receive a bonus of up to \$50,000 per year. If the employment of Mr. O'Reilly is terminated without cause at any time, he will be entitled to receive his base salary for six months following his termination. If Mr. O'Reilly's responsibilities are reduced within twelve months following a change in control, such as the merger, and such reduction is not for cause, any resignation of employment as a consequence of such reduction in responsibilities will be treated as a termination of employment without cause.

Indemnification; Directors' and Officers' Insurance

Avnet has agreed to indemnify each present and former Savoir officer and director to the fullest extent permitted by Delaware law for at least four years after the effective time of the merger. Avnet has also agreed that it will maintain a policy of directors' and officers' liability insurance coverage for up to four years following the merger on terms no less advantageous than Savoir's existing insurance, subject to a cap on the amount of premiums Avnet will be required to pay for that coverage. See "The Merger Agreement — Covenants — Indemnification; Directors' and Officers' Insurance" on page

Public Trading Markets

The Savoir common stock is currently quoted on the Nasdaq National Stock Market under the symbol "SVTG." Upon consummation of the merger, the Savoir common stock will be delisted from the Nasdaq National Stock Market and deregistered under the Securities Exchange Act of 1934. The Savoir series A preferred stock is not publicly traded.

The Avnet common stock is currently listed on the New York Stock Exchange and the Pacific Exchange under the symbol "AVT" and will continue to be listed on both exchanges after the merger. See "Market Price and Dividend Information."

Accounting Treatment

Avnet will account for the acquisition of Savoir under the purchase method of accounting in accordance with the provisions of Accounting Principles Board Opinion No. 16, "Business Combinations." Accordingly, Avnet will record at its cost the acquired assets less liabilities assumed, with the excess of such cost over the estimated fair value of such net assets reflected as goodwill. Additionally, certain costs directly related to the acquisition will be reflected as additional purchase price in excess of the net assets acquired. Avnet's statement of income will include the operations of Savoir after the effective date of the merger.

The Merger Agreement

General

At the effective time of the merger, Tactful Acquisition Corp., a wholly-owned subsidiary of Avnet, will merge into Savoir, the separate corporate existence of Tactful will cease, and Savoir will remain as the surviving corporation in the merger and become a wholly-owned subsidiary of Avnet. The transaction is intended to qualify as a tax-free "reorganization" for federal income tax purposes within the meaning of Section 368(a) of the Internal Revenue Code.

Terms of the Merger

At the effective time of the merger, each share of Savoir common stock will be converted into shares of Avnet common stock at an exchange ratio based upon the average of the closing prices of a share

29

Table of Contents

Avnet common stock as reported on the New York Stock Exchange composite tape during the fifteen consecutive trading days ending on the fifth trading day before the date of the special meeting, as follows:

- assignment or alteration by Savoir of Mr. Mertens' duties, responsibilities or obligations materially inconsistent with his position with Savoir after notice of Mr. Mertens' objections thereto;
- failure of Savoir to provide to Mr. Mertens the salary or bonuses described above;
- relocation of Savoir's IBM operational headquarters outside of San Antonio, Texas;
- any requirement by Savoir for Mr. Mertens to relocate anywhere other than San Antonio, Texas; and
- instructions by Savoir to Mr. Mertens to violate any applicable law after notice of Mr. Mertens'

objections.

At the effective time of the merger, each share of Savoir series A preferred stock, other than shares held by persons who perfect their appraisal rights under Delaware law, will be converted into Avnet common stock at an exchange ratio equal to the quotient of \$9.6581 (the amount per share payable to the holders of series A preferred stock upon a "change in control" of Savoir, as provided in its certificate of incorporation) divided by the average of the closing prices of a share of Avnet common stock during the five consecutive trading days ending on the trading day immediately preceding the effective date of the merger.

Each share of Tactful issued and outstanding immediately prior to the effective time will be converted in the merger into one share of Savoir, the surviving corporation. Each share of Avnet common stock issued and outstanding immediately prior to the effective time will remain an issued and outstanding share of Avnet common stock.

Fractional shares of Avnet common stock will not be issued to Savoir stockholders pursuant to the merger agreement. Instead,

- if the fifteen-day average closing price is greater than \$68.5472, the exchange ratio will be 0.11452 of a share of Avnet common stock for each share of Savoir common stock;
- if the fifteen-day average closing price is greater than or equal to \$50.6654 and less than or equal to \$68.5472, the exchange ratio for each share of Savoir common stock will be the quotient of \$7.85 (the agreed upon value of a share of Savoir common stock in the merger agreement) divided by the fifteen-day average closing price; and
- if the fifteen-day average closing price is less than \$50.6654, the exchange ratio will be 0.15494 of a share of Avnet common stock for each share of Savoir common stock.

Closing; Effective Time of the Merger

The merger agreement provides that the closing of the merger will take place on the second business day after the day on which all the conditions set forth in the merger agreement are satisfied or waived, unless Avnet and Savoir agree to another time or date. On the closing date, Tactful and Savoir will file a certificate of merger with the Secretary of State of Delaware. The merger will become effective when the certificate of merger is duly filed or such other time as is set forth in the certificate of merger.

Certificate of Incorporation and Bylaws of the Surviving Corporation

The merger agreement provides that the certificate of incorporation and bylaws of Savoir as in effect immediately before the merger will be the certificate of incorporation and bylaws of Savoir as the surviving corporation at and after the effective time of the merger, except that the number of directors of Savoir, as specified in its bylaws, will be reduced to two.

Directors and Officers of the Surviving Corporation

The merger agreement provides that the persons serving as directors of Tactful immediately before the merger will be the initial directors of Savoir at and after the effective time of the merger, and that

Table of Contents

before the merger Avnet will name persons who will serve as the officers of Savoir at and after the effective time of the merger.

Representations and Warranties

The merger agreement contains representations and warranties of Savoir, Avnet and Tactful, most of which are customary for agreements of this nature and some of which are qualified as to materiality. Savoir represents and warrants to Avnet and Tactful, and Avnet and Tactful represent and warrant to Savoir, that, among other things:

- a holder of Savoir common stock will receive an amount in cash (without interest) equal to (1) the fraction of a share of Avnet common stock to which the holder would otherwise be entitled, multiplied by (2) the exchange ratio applicable to the Savoir common stock in the merger; and
- a holder of Savoir series A preferred stock will receive an amount in cash (without interest) equal to (1) the fraction of a share of Avnet common stock to which the holder would otherwise be entitled, multiplied by (2) the exchange ratio applicable to the series A preferred stock in the merger.

Also, Savoir represents and warrants to Avnet and Tactful as to the following matters, among others:

- they (and Savoir's subsidiaries) are corporations duly organized, validly existing and in good standing under the laws of the states where they are organized;
- Avnet, Savoir and Tactful have the corporate power and authority to execute, deliver and perform their
 obligations under the merger agreement, assuming, in the case of Savoir, that its stockholders adopt
 the merger agreement; and the merger agreement is a valid, binding and enforceable obligation of Avnet,
 Savoir and Tactful;
- the execution and delivery of the merger agreement and, in the case of Savoir, the option agreement, and the performance of the transactions contemplated by the merger agreement and option agreement, will not result in a breach, termination or violation of, or conflict with, their organizational documents or contracts or permits to which they are subject, or violate or conflict with any law, rule, regulation, judgment or decree to which they are subject;
- all consents and approvals required by any governmental entity or other person not a party to the merger agreement to permit the merger and other transactions contemplated by the merger agreement have been or will be obtained;
- since January 1, 1998, Avnet and Savoir have made all required filings with the Securities and Exchange Commission, and such filings were, and future filings will be, materially accurate when filed;
- except as disclosed by Savoir to Avnet or in Savoir's SEC filings, there have been no events since December 31, 1999, which resulted in, or are likely to result in, a material adverse effect on the businesses, financial condition, results of operations, business prospects or properties of Savoir and Avnet;
- Avnet's and Savoir's businesses are being conducted in compliance with all applicable laws; and
- there are no brokers and finders entitled to compensation in connection with the transactions contemplated in the merger agreement other than those brokers and finders specified in the merger agreement.

- Savoir's authorized, outstanding and reserved capital stock and its outstanding employee stock options and warrants;
- The absence of liens and encumbrances on Savoir's assets, or adverse changes to Savoir's material contracts, or adverse changes to Savoir's rights and obligations under any laws, rules or regulations to which it is subject, as a result of its execution and delivery of, and performance of its obligations under, the merger agreement;
- · tax matters;
- Savoir's material contracts, customers and suppliers;
- its employee benefits, employment agreements and labor matters;
- · legal proceedings to which Savoir or any of its subsidiaries is a party;
- insurance coverage;
- its title to real property, intellectual property and other property;

Conditions to the Merger

The obligations of each of Avnet and Savoir to effect the merger are subject to the fulfillment or waiver, at or before the effective date of the merger, of the following conditions, among others:

- environmental matters; and
- labor matters.

The obligations of each of Avnet and Savoir to effect the merger are also subject to the fulfillment or waiver by each of them of the following conditions:

- the adoption of the merger agreement by the stockholders of Savoir;
- the absence of any judgment, decree, injunction, ruling or order by any court or other governmental entity that prohibits, restricts or delays consummation of the merger;
- no issuance by the Securities and Exchange Commission of a stop order suspending the effectiveness of the Form S-4 registration statement of which this proxy statement/prospectus is a part; and
- the absence of any general disruption in the financial and securities markets, such as a general suspension of trading in securities on any national securities exchange or the over-the-counter market, the declaration of a banking moratorium, or any limitation by any governmental entity on the extension of credit by banks.

Avnet's obligations to effect the merger are also subject to the fulfillment or Avnet's waiver of the condition that Savoir must receive all consents necessary to allow the consummation of the merger and the continuation of Savoir's business after the merger from International Business Machines Corp., IBM Credit Corp. and their affiliates, and from certain other Savoir suppliers and other parties who otherwise may have a right to terminate their contracts with Savoir as a result of the merger.

Savoir's obligations to effect the merger are also subject to the requirement that Avnet common stock distributed in connection with the merger be accepted upon notice of issuance for listing on the New York Stock Exchange. Savoir has agreed to give Avnet's representatives full access during normal business hours before the effective time of the merger to all its properties, books, records, documents, personnel, auditors and legal counsel, and each of Avnet and Savoir will furnish promptly to the other all information concerning itself as the other may reasonably request. Avnet has agreed to provide Savoir's representatives such information as Savoir may reasonably request to determine the accuracy of Avnet's and Tactful's representation and warranties in the merger agreement.

32

Table of Contents

Conduct of Savoir's Business Prior to the Merger

Savoir has agreed that before the effective time of the merger (unless Avnet otherwise agrees), among other things:

- the representations and warranties of the other in the merger agreement must be true and correct in all material respects both as of the date of the merger agreement and on the effective date of the merger;
- the other must have complied in all material respects with all covenants requiring its compliance prior to the effective date;
- each of them must receive from its counsel an opinion stating that the merger will be treated for federal income tax purposes as a reorganization within the meaning of Section 368(a) of the Internal Revenue Code, that each of Savoir, Tactful and Avnet will be a party to that reorganization within the meaning of Section 368(b) of the Code, and in the case of the opinion to Avnet, that Avnet, Tactful and Savoir will not recognize any gain in the merger (see "Material Federal Income Tax Consequences of the Merger" (page)); and
- from the date of the merger agreement, the other shall not have suffered a material adverse effect on its business, financial condition, results of operation, business prospects or properties.
- it will operate its business only in the ordinary course of business consistent with past practices, and will use its best efforts to
 - (a) preserve its existing business organization, insurance coverage, material rights, material licenses or permits, advantageous business relationships, material agreements and credit facilities,
 - (b) retain and keep available the services of its present officers, employees and agents, and
 - (c) preserve the goodwill of its customers, suppliers and others having business dealings with it;
- Savoir and its subsidiaries will not
 - (a) enter into any material transaction or commitment, or dispose of or acquire material properties or assets, except purchases and sales of inventory in the ordinary course of business consistent with past practices,

- (b) implement any new employee benefit plan or employment, compensation, or severance agreement,
- (c) amend any existing employment plan or agreement,
- (d) take any action that would jeopardize their material supplier or customer relationships,
- (e) make any material change in the nature of their businesses and operations,
- (f) enter into any transaction or agreement with their officers, directors and affiliates,
- (g) incur or agree to incur any liability for payment of more than \$100,000 except transactions in the ordinary course of business, or
- (h) make any tax election or make any change in any method or period of accounting or any material change in any accounting policy, practice or procedure;

Agreement Not to Solicit Other Acquisition Proposals

Savoir has agreed that it will not, nor will it authorize or permit any of its subsidiaries, directors, officers or employees or any investment banker, financial adviser, attorney, accountant or other representative retained by it, directly or indirectly, to solicit, initiate or take any other action to facilitate knowingly any inquiries or the making of any proposal which is or may reasonably be expected to lead to an "acquisition proposal" or engage in any discussion or negotiations relating to an acquisition proposal. An acquisition proposal is any proposal or offer relating to any acquisition or purchase of all or a substantial part of the assets of Savoir or any of its subsidiaries or 15% or more of any class of equity securities of Savoir or any of its subsidiaries, or relating to a business combination, liquidation or similar transaction involving Savoir or any of its subsidiaries, other than the business combination with Avnet.

However, Savoir may comply with Securities and Exchange Commission rules with respect to a response to a tender offer. Savoir also may engage in discussions with a third party who, without any

33

Table of Contents

solicitation or encouragement by Savoir or its representatives, seeks to initiate such discussions, and may furnish the third party nonpublic information concerning Savoir if the third party has made a bona fide acquisition proposal and the board of directors of Savoir believes in good faith (after consultation with its financial advisor) that the proposal is a "superior proposal," meaning:

- Savoir and its subsidiaries will not amend their charter documents and by-laws;
- Savoir will not declare or pay any dividend or make any other distribution in respect of its capital stock; and
- Savoir will not split, combine or reclassify its capital stock or issue, redeem or acquire any of its

equity securities, options, warrants or convertible instruments except for existing commitments, and it will not grant any more options to purchase its common stock.

and Savoir's board of directors concludes in good faith that entering into discussions with, and furnishing nonpublic information to, the third party may be necessary for the Board of Directors to act in a manner consistent with its fiduciary duties under applicable law.

Stock Options

Currently, there are options to purchase an aggregate of shares

of Savoir common stock at prices ranging from

\$ to

\$ under Savoir's

1994 Stock Option Plan. At the effective time of the merger, Avnet will assume Savoir's obligations under Savoir's 1994 Stock Option Plan, so that the outstanding Savoir options will become options to purchase Avnet common stock. The exercise price per share of Avnet common stock issuable upon exercise of each such Avnet option will be equal to the per share exercise price of the related Savoir option divided by the exchange ratio for the conversion of Savoir common stock in the merger (see page

), and the number of shares of Avnet common stock issuable upon exercise of each such Avnet option will be equal to the number of shares of Savoir common stock that could have been acquired under the related Savoir option multiplied by the exchange ratio.

Warrants

Currently, there are warrants to purchase an aggregate of shares

of Savoir common stock at prices ranging from \$4.77 to \$9.69 per share. At the effective time of the merger, each such warrant will be converted into a right to acquire shares of Avnet common stock. The exercise price per share of Avnet common stock issuable upon exercise of each such warrant will be equal to the per share exercise price of the warrant before the effective time divided by the exchange ratio for the conversion of Savoir common stock in the merger, and the number of shares of Avnet common stock issuable upon exercise of each such warrant will be equal to the number of shares of Savoir common stock that could have been acquired under such warrant before the effective time multiplied by the exchange ratio.

Avnet Benefit Plans

When the merger is completed, the business of Savoir will be combined with that of Avnet and the Savoir employees will be offered employee benefits which, in the aggregate, are no less favorable to them than those Avnet provides to its own employees.

Indemnification; Directors' and Officers' Insurance

For at least four years after the effective time of the merger, Savoir and Avnet will indemnify each present and former director and officer of Savoir and their respective subsidiaries to the full extent permitted by Delaware law with respect to any alleged action or omission occurring before the effective time. Avnet and Savoir also have agreed that they will maintain directors' and officers' liability insurance coverage for such directors and officers up to four years following the merger on terms no less advantageous than Savoir's existing insurance; however, Avnet and Savoir will not be required to pay more

Table of Contents

than twice the annual premiums which Savoir currently pays for such insurance. Also, see "The Merger — Interests of Certain Persons in the Merger and Possible Conflicts of Interest" (p.).

Termination

The merger agreement may be terminated at any time before the effective time of the merger, whether before or after approval by Savoir's stockholders, by the mutual written consent of Savoir and Avnet. It may also be terminated by either Savoir or Avnet without the consent of the other if:

- that the proposal is reasonably capable of being completed, taking into account all relevant, legal, financial, regulatory and other aspects of the proposal and the source of its financing;
- that the proposal would, if consummated, result in a transaction more favorable to the stockholders of Savoir, from a financial point of view, than the transactions contemplated by the merger agreement with Avnet; and
- that the person making the proposal has, or is reasonably likely to have or obtain, any necessary funds or customary commitments to provide any funds necessary to consummate the proposal;

In addition, Savoir may terminate the merger agreement without the consent of Avnet for the purpose of entering into an agreement with another person that has made a "superior proposal". See "— Covenants — Agreement Not to Solicit Other Acquisition Proposals" on page .

Avnet may terminate the merger agreement without the consent of Savoir if the board of directors of Savoir withdraws, amends, modifies, conditions or qualifies in a manner adverse to Avnet its approval of the merger agreement or its recommendation of the merger agreement to the Savoir stockholders, or recommends to them another acquisition proposal for Savoir. See "— Covenants — Agreement Not to Solicit Other Acquisition Proposals" on page .

If the merger agreement is terminated as described above, the merger agreement will become void and have no effect, without any liability or obligation on the part of either Avnet or Savoir however, the termination will not relieve:

- any governmental entity has issued a final and non-appealable order prohibiting the consummation of any of the transactions contemplated by the merger agreement;
- the merger has not been consummated by September 15, 2000, unless due to the failure by the party seeking termination to perform any of its obligations under the merger agreement in any material respect;
- · Savoir fails to hold the special meeting or the Savoir stockholders do not adopt the merger agreement;
- Savoir or Avnet is not in breach of any provision of the merger agreement, the other breaches any representation, warranty, covenant, agreement, condition or obligation in the merger agreement, and the breach has or is reasonably likely to have a material adverse effect on the breaching party.

Termination Fees and Expenses

Whether or not the merger is consummated, each of Avnet and Savoir will pay its own expenses incurred in connection with the merger agreement and the transactions contemplated by the merger agreement.

Savoir will pay Avnet \$750,000 in cash if

- the liability of Avnet or Savoir for any willful breach of the merger agreement;
- the obligations of Savoir and Avnet to indemnify the other party with respect to the registration statement of which this proxy statement/prospectus is a part;
- the obligations of Savoir and Avnet to keep confidential all non-public information disclosed by the other in connection with the transactions contemplated by the merger agreement; or
- the obligation of Savoir to pay a termination fee (see below).
- Avnet terminates the merger agreement because:

35

Table of Contents

- 1. Savoir has failed to hold the special meeting or the stockholders of Savoir do not adopt the merger agreement; or
- 2. Avnet is not in breach of any provision of the merger agreement, Savoir has breached any of its representations and warranties in the merger agreement, and the breach has or would reasonably be likely to have a material adverse effect on Savoir; or
- 3. Avnet is not in breach of any provision of the merger agreement, Savoir has breached any of its covenants, agreements, conditions or obligations in the merger agreement, and the breach has or would reasonably be likely to have a material adverse effect on Savoir and is not or cannot be promptly cured; or
- 4. the board of directors of Savoir has withdrawn, amended, modified, conditioned or qualified in a manner adverse to Avnet its approval or recommendation of the merger, or has recommended another acquisition proposal for Savoir; or

and, in each case, Savoir or its stockholders are aware of another "acquisition proposal" as defined above under "— Covenants — Agreement Not to Solicit Other Acquisition Proposals."

Savoir will pay to Avnet \$4,500,000, less any previous payment of \$750,000 as described above, if the merger agreement is terminated:

• Savoir terminates the merger agreement because it has failed to hold the special meeting or the stockholders of Savoir do not adopt the merger agreement, or Savoir terminates the merger agreement for the purpose of entering into an agreement for a "superior proposal;"

and, in either case, within one year of such termination, Savoir enters into an agreement to effect another acquisition proposal.

Amendment, Extension and Waiver

Savoir and Avnet may amend the merger agreement in writing at any time before or after the Savoir stockholders adopt the merger agreement. However, after adoption of the merger agreement by the Savoir stockholders, the merger agreement cannot be amended in any way that would under Delaware law require further approval of the stockholders unless such further approval is obtained.

At any time before the effective time, Savoir or Avnet may extend the time for performance of any of the obligations or other acts of the other, waive any inaccuracies in the representations and warranties of the other contained in the merger agreement or in any document delivered pursuant to the merger agreement, or waive compliance by the other with any of the agreements or conditions contained in the merger agreement. Any agreement of extension or waiver must be in writing.

Other Agreements

At the time when Avnet and Savoir first executed and delivered the merger agreement, they also entered into an option agreement under which Savoir granted Avnet an option (the "Savoir Option") to purchase up to 2,023,435 shares of Savoir common stock, representing 15.0% of the issued and outstanding Savoir common stock before such purchase, at an exercise price of \$6.83 per share, subject to customary anti-dilution adjustments.

Avnet may not exercise the Savoir Option unless Avnet becomes entitled to receive from Savoir a termination fee under the merger agreement. See "The Merger Agreement — Termination Fees and Expenses" on page .

Also in connection with the merger agreement, Avnet entered into an inducement agreement with P. Scott Munro, Carlton Joseph Mertens II, Dennis Polk, Robert O'Reilly, J. Larry Smart, Angelo Guadagno, William Sickler, Michael N. Gunnels and Guy M. Lammle (collectively, the "Management Stockholders"), under which each of the Management Stockholders granted Avnet an option (the

36

Table of Contents

"Stockholders' Option") to purchase his shares of
Savoir common stock (collectively, 2,067,536 shares) at an
exercise price of \$7.85 per share, payable in cash or Avnet
common stock at Avnet's option. If Avnet exercises the
Stockholders' Option, the merger subsequently occurs and the
exchange ratio in the merger is based on an average closing
price which is less than \$59.6063 (the midpoint in the range
between \$50.6654 and \$68.5472 in the exchange ratio formula),
Avnet will issue to each Management Stockholder shares of Avnet
common stock equal in value to the difference between

(A) the value of the Avnet common stock which the Management Stockholder would have received in the merger for his Savoir common stock, and (B) the value of the Avnet common stock or the cash amount which the Management Stockholder received from the exercise of the Stockholders' Option. On the other hand, if Avnet exercises the Stockholders' Option, the merger subsequently occurs and the exchange ratio in the merger is based on an average closing price which is greater than \$59.6063, each Management Stockholder will deliver to Avnet, at his option, either shares of Avnet common stock having a value equal to, or cash in the amount of, the difference between (A) the value of the Avnet common stock or the cash amount which the Management Stockholder received from the exercise of the Stockholders' Option, and (B) the value of the Avnet common stock which the Management Stockholder would have received in the merger for his Savoir common stock. Thus, if Avnet exercises the Stockholders Option and the merger subsequently occurs, the Management Stockholders will receive consideration for their shares of Savoir common stock which is equal in value to the consideration which the other Savoir stockholders receive in the merger for their shares.

Each of the Management Stockholders:

- · as described above; or
- by Avnet if the merger has not been consummated by September 15, 2000;

Avnet may not exercise the Stockholders' Option until Savoir or its stockholders shall have received in writing, or there shall have been publicly proposed, an "acquisition proposal," as defined above under "— Covenants — Agreement Not to Solicit Other Acquisition Proposals."

The inducement agreement and the Savoir Option will terminate on the earliest to occur of (a) the effective time of the merger, (b) six months after Avnet becomes entitled to receive from Savoir a termination fee under the merger agreement (subject to extension if governmental authorities restrict exercise of the option), (c) termination of the merger agreement under circumstances which do not and cannot result in Avnet becoming entitled to receive a termination fee, and (d) twelve months after the termination of the merger agreement under circumstances which do or could result in Avnet becoming entitled to receive a termination fee, unless during such twelve month period Avnet does become entitled to a termination fee.

The option agreement is attached as Appendix B to this proxy statement/ prospectus and is incorporated herein by reference. The inducement agreement is attached as Appendix C to this proxy statement/ prospectus and is incorporated herein by reference. The foregoing summaries of the option agreement and the inducement agreement are not complete descriptions of all their terms, and are qualified in their entirety by reference to Appendix B and Appendix C.

37

The following summary discusses the material federal income tax consequences of the merger. The summary is based upon the Internal Revenue Code, applicable treasury regulations thereunder and administrative rulings and judicial authority as of the date of this proxy statement/ prospectus. All of the foregoing are subject to change, possibly with retroactive effect, and any change could affect the continuing validity of the discussion. The discussion and the opinion of Carter, Ledyard & Milburn assume that holders of shares of Savoir common stock and series A preferred stock hold such shares as capital assets. Further, the discussion does not address the tax consequences that may be relevant to a particular stockholder subject to special treatment under certain federal income tax laws, such as dealers in securities, traders in securities that elect to use a mark-to-market method of accounting, tax-exempt organizations, foreign persons, persons that hold Savoir common stock or series A preferred stock as part of a straddle or conversion transaction and persons who acquired shares of Savoir common stock or series A preferred stock through the exercise of employee stock options or rights or otherwise as compensation or through a tax-qualified retirement plan. This discussion does not address any consequences arising under the laws of any state, locality or foreign jurisdiction.

Tax Opinions. It is intended that the merger will be treated as a "reorganization" within the meaning of Section 368(a) of the Code. Avnet has received an opinion of Carter, Ledyard & Milburn that, based on the assumptions and conditions stated therein, the merger will be treated for federal income tax purposes as a reorganization within the meaning of Section 368(a) of the Code and that Avnet, Tactful and Savoir will each be a party to that reorganization within the meaning of Section 368(b) of the Code. It is a condition to the obligation of Avnet to complete the merger that its counsel confirm its opinion as of the merger date, and it is a condition to the obligation of Savoir to complete the merger that Savoir receive a similar opinion from its counsel as of the merger date. These opinions will not be binding on the Internal Revenue Service, and there can be no assurance that the IRS will agree with the conclusions expressed in the opinions. The opinions are and will be based in part upon certain assumptions and certain representations made by Savoir and Avnet customarily made and given in transactions of this type.

Tax Consequences of the Merger. In the opinion of Carter, Ledyard & Milburn, assuming that the assumptions and representations referred to in the preceding paragraph are true and complete as of the effective time, the merger will be treated as a "reorganization" within the meaning of Section 368(a) of the Code, and each of Avnet, Savoir and Tactful will be a party to that reorganization within the meaning of 368(b) of the Code. As a result, subject to the next paragraph below, in general, (1) no gain or loss will be recognized by a holder of Savoir common stock or series A preferred stock as a result of the conversion of shares of Savoir common stock or series A preferred stock into shares of Avnet common stock pursuant to the merger, (2) the aggregate $\ensuremath{\text{tax}}$ basis of the shares of Avnet common stock received in the merger will be the same as the aggregate tax basis of the shares of Savoir common stock or series A preferred stock converted and (3) the holding period of the shares of Avnet common stock received in the merger will include the holding period of shares of Savoir common stock or series A preferred stock converted.

Cash Received in Lieu of Fractional Shares. If a holder of shares of Savoir common stock or series A preferred stock receives cash in lieu of a fractional share of Avnet common stock, this fractional share interest will be treated as having been distributed to the holder, and the cash amount will be treated as received in redemption of the fractional share interest. In general, the holder will recognize capital gain or loss equal to the cash amount received for the fractional share reduced by the portion of the holder's tax basis in shares of Savoir common stock or series A preferred stock surrendered

that is allocable to the fractional share interest in Avnet common stock. The capital gain or loss will be long-term capital gain or loss if the holder's holding period in the fractional share interest for federal income tax purposes is more than one year.

The preceding summary does not purport to be a complete analysis or discussion of all potential tax effects relevant to the merger. Savoir stockholders are urged to consult their own tax advisers as to the

38

Table of Contents

specific tax consequences to them of the merger, including tax return reporting requirements, the applicability and effect of Federal, State, local, foreign and other applicable tax laws and the effect of any proposed changes in the tax laws.

Description of Avnet Common Stock

The holders of shares of Avnet's common stock have equal rights to dividends from funds legally available for dividends when, as and if declared by Avnet's board of directors, and are entitled, upon liquidation, to share ratably in any distribution in which holders of common stock participate. The Avnet common stock is not redeemable, has no preemptive or conversion rights and is not liable for assessments or further calls. The holders of shares of Avnet's common stock are entitled to one vote for each share at all meetings of shareholders.

The Transfer Agent and Registrar for Avnet's common stock is the Bank of New York. Avnet's common stock is listed on the New York Stock Exchange and the Pacific Exchange.

Description of Savoir Common Stock and Series A Preferred Stock

Holders of Savoir common stock are entitled to one vote for each share held of record on each matter submitted to stockholders. Holders of series A preferred stock are entitled to 1.1953125 votes for each share and vote, together with the holders of common stock, on all matters submitted to stockholders.

Subject to preferences that may be applicable to any shares of preferred stock outstanding at that time, holders of Savoir common stock are entitled to receive ratably such dividends as may be declared by the board of directors out of legally available funds and to share equally, on a per share basis, in all assets of Savoir remaining after satisfaction of all liabilities in the event of dissolution and liquidation. Common stockholders do not have preemptive rights to subscribe for common stock or other securities of Savoir and do not have any rights to convert their common stock into any other securities. All outstanding shares of common stock and series A preferred stock are fully paid and nonassessable.

The series A preferred stock is convertible at the option of the holders, at any time, into common stock of Savoir. Each share of series A preferred stock is currently convertible into 1.1953125 shares of common stock, subject to adjustment if Savoir issues any stock at less than the conversion price.

The series A preferred stock has an eight percent (8%) cumulative dividend (\$0.19125 per share per quarter), payable in cash or Savoir common stock at the election of Savoir.

Savoir may redeem the series A preferred stock:

- granted to Avnet an irrevocable proxy to vote his shares of Savoir common stock in favor of the adoption of the merger agreement and against any other proposal for any recapitalization, merger, sale of assets or other business combination between Savoir and any other person or entity other than Avnet or Tactful, or the taking of any action which would result in any of the conditions to Avnet's obligations under the merger agreement not being fulfilled;
- · agreed not to dispose of his shares of Savoir common stock;
- agreed that, except as permitted by the merger agreement, he will not directly or indirectly initiate, solicit or encourage any inquiries or the making or implementation of any alternative proposal to acquire Savoir common stock, or engage in any negotiations concerning, or provide any confidential information or data to, or have any discussions with, any person relating to such an alternative proposal; and
- agreed that any additional shares of Savoir common stock which he acquires, whether through a stock split, purchase, or other events, will be subject to the terms of the inducement agreement.

Upon the dissolution, liquidation or winding up of Savoir, the holders of series A preferred stock have the right to cash or other property with a value of \$9.5625 per share before the holders of common stock receive any assets. Upon a change of control of Savoir, such as the merger, the holders of series A preferred stock have the right to cash or other property with a value of \$9.6581 per share.

Comparison of Shareholder Rights

In connection with the merger, Savoir stockholders will receive common stock of Avnet. Savoir is a Delaware corporation, and Avnet is a New York corporation. The Savoir certificate of incorporation and by-laws differ from those of Avnet in significant ways. Because of differences between Delaware and New

39

Table of Contents

York law and the differences in the certificates of incorporation and by-laws of Savoir and Avnet, the rights of holders of Savoir common stock will change when they receive Avnet common stock in the merger.

Copies of the Savoir and Avnet certificates of incorporation and by-laws are available upon request. See "Where You Can Find More Information" on page .

Below is a summary of some of the material differences between Delaware and New York law and the certificates of incorporation and by-laws of Savoir and Avnet. It is not practical to summarize all differences in this proxy statement/ prospectus, but some of the principal differences which could materially affect the rights of Savoir stockholders include the following:

Voting on Business Combinations

Generally, under Delaware law, the approval by a majority of the votes represented by all of a corporation's outstanding shares entitled to vote is required for a merger or consolidation or sale, lease or exchange of all or substantially all the corporation's assets. Under the Savoir certificate of incorporation, holders of series A preferred stock are entitled to vote on any such transaction, together with the holders of Savoir common stock as a single class, and each share of series A preferred stock entitles the holder to 1.1953125 votes.

Under New York law, such transactions must be approved (a) in the case of corporations like Avnet that were in existence on February 22, 1998 and that do not expressly provide in their certificates of incorporation for majority approval of such transactions, by two-thirds of the votes of all outstanding shares entitled to vote on the transaction, and (b) in the case of all other corporations, by a majority of the votes of all outstanding shares entitled to vote thereon. The Avnet certificate of incorporation does not contain a provision expressly providing for majority approval of such transactions. New York law also provides that the holders of shares of a class, or series of a class, of capital stock of a corporation shall be entitled to vote together and to vote as a separate class on any merger or consolidation in which (a) such shares will remain outstanding after the merger or consolidation or will be converted into the right to receive shares of stock of the surviving or consolidated corporation or another corporation and (b) the charter of the surviving or consolidated corporation or such other corporation immediately after the effectiveness of the merger or consolidation would contain any provision that is not contained in the charter of the pre-merger corporation and that, if contained in an amendment thereto, would entitle the holders of shares of such class or series of a class to vote as a separate class pursuant to the procedures under New York law for class voting on charter amendments discussed under "- Amendments to Charters."

Though not applicable to the merger of Savoir with Tactful, under Delaware law, no vote of stockholders of the surviving corporation in a merger is required (and no dissenters' rights are available to such stockholders) if the agreement of merger does not amend in any respect the certificate of incorporation of the surviving corporation, there is no change in the outstanding shares of the surviving corporation as a result of the merger, and the number of common shares issued or issuable in the merger does not exceed 20% of the number outstanding immediately prior to the merger. New York law has no comparable provision.

State Takeover Legislation

Delaware law, in general, prohibits a business combination between a corporation and an "interested stockholder" for three years after the time stockholder became an interested stockholder, unless:

• Before September 19, 2001, if the market price for Savoir common stock is at least \$13.96875 for 30 consecutive trading days and the daily trading volume for Savoir common stock for at least 25 of those

trading days is at least 125,000 shares, at a price equal to the liquidation preference for the series A preferred stock (presently \$9.5625 per share); or

• any time on and after September 19, 2001, at a price equal to the liquidation preference for the series A preferred stock, plus a redemption premium of eight percent (8%), or \$0.7726 per share.

40

Table of Contents

- before that time, the board of directors of the corporation approved either the business combination or the transaction that resulted in the stockholder becoming an interested stockholder; or
- upon consummation of the transaction that resulted in the stockholder becoming an interested stockholder, the interested stockholder owned at least 85% of the voting stock of the corporation outstanding at the time the transaction commenced, exclusive of shares owned by directors who are also officers and by certain employee stock plans; or

The term "business combination" includes any merger or consolidation with or caused by the interested stockholder, and the sale, pledge, transfer or other disposition (including as part of a dissolution) of assets of the corporation having an aggregate market value equal to or greater than 10% of either the aggregate market value of all assets of the corporation on a consolidated basis or the aggregate market value of all the outstanding stock of the corporation; transactions that would increase the interested stockholder's proportionate share ownership of the stock of any class or series of the corporation or a subsidiary; and any receipt by the interested stockholder of the benefit of any loans, advances, guarantees, pledges or other financial benefits provided by or through the corporation or any subsidiary.

In general, an "interested stockholder" is any person who is the owner of 15% or more of the outstanding voting stock of the corporation, and the affiliates and associates of such person. The definition of interested stockholder does not include (a) persons who owned as of December 23, 1987 stock of the corporation representing 15% or more of the outstanding voting stock, (b) persons who received stock of the corporation representing 15% or more of the outstanding voting stock as a gift or bequest from a person who owned it before that date, or (c) persons whose ownership of the voting stock rises over the 15% threshold as a result of action taken by the corporation (such as stock repurchase) unless that person thereafter acquires additional shares. The term "owner" is broadly defined to include any person that individually or with or through such person's affiliates or associates, among other things, beneficially owns such stock, or has the right to acquire such stock (whether such right is exercisable immediately or only after the passage of time) pursuant to any agreement or understanding or upon the exercise of warrants or options or otherwise or has the right to vote such stock pursuant to any agreement or understanding, or has an agreement or understanding with the beneficial owner of such stock for the purpose of acquiring, holding, voting or disposing of such stock.

The restrictions of the Delaware business combination law do not apply to

 at or after the time that the stockholder became an interested stockholder, the business combination is approved by the board of directors and authorized by the affirmative vote at a stockholders' meeting of at least 66 2/3% of the outstanding voting stock that is not owned by the interested stockholder individually or with or through any of its affiliates and associates.

- a corporation that elects not to be subject to it, or
- with certain exceptions, a corporation whose voting stock is neither

Because the Savoir certificate of incorporation and the Savoir by-laws do not opt out of the Delaware business combination law, it would be applicable to the merger except that the Savoir board unanimously approved the merger and the transactions contemplated by the merger agreement before Avnet became an interested stockholder.

The business combination provisions of New York law are similar to the above provisions of Delaware law, except that (i) an "interested shareholder" under New York law is the direct or indirect beneficial owner of at least 20% (not 15%) of the corporation's voting stock, and (ii) the corporation may not engage in a business combination (the definition of which is similar to that under the Delaware law) with an interested shareholder for a period of five (rather than three) years, unless the business combination or the purchase of stock by means of which the interested shareholder became such is approved by the corporation's board of directors in advance of such stock purchase, or unless the interested shareholder was the beneficial owner of 5% or more of the corporation's outstanding voting stock on October 30, 1985, and remains so until becoming an interested shareholder.

After the five-year restricted period, an interested shareholder of a New York corporation may engage in a business combination with the corporation if the business combination is approved after the five-year period by the affirmative vote of the holders of a majority of the shares of the corporation's voting stock

41

Table of Contents

other than those beneficially owned by the interested shareholder and his affiliates and associates, or if the value of the aggregate consideration to be paid by the interested shareholder in connection with the business combination satisfies certain "fair price" formulas specified in the statute, and the interested shareholder, after becoming such, has not acquired any additional shares of voting of the corporation, except as provided in the statute.

The Delaware and New York business combination statutes provide only limited regulation of certain self-dealing transactions between a corporation and certain large stockholders. Moreover, wholly apart from the exceptions mentioned above, there are a number of additional methods whereby even an interested stockholder can benefit from a merger or sale of a corporation's assets during the first three years (Delaware) of five years (New York) after becoming an interested stockholder. For example, Delaware law provides that if, after an interested stockholder becomes such, an acquisition is proposed by a third party and is approved or not opposed by a majority of the board of directors who were directors before the interested stockholder became such, that interested stockholder is no longer subject to the three-year restriction and is given at least 20 days in which to develop a competing proposal. Also, under both Delaware law and New York law, a stockholder, prior to becoming an interested stockholder, may solicit proxies or stockholder consents to change the composition of the board of directors, and the new board which the stockholder has elected or helped to elect may approve a business combination with the stockholder without the statutory delay. Finally, an interested stockholder may simply acquire control of the corporation, remove the incumbent board and thereafter sell assets to third parties or even (in the case of Delaware) liquidate the corporation, provided that any dividends or liquidating distributions are made to all remaining stockholders pro rata, thus giving minority stockholders the opportunity to realize their fair share of the true value of the assets of the corporation.

It should be noted that the Delaware and New York business combination statutes

- listed on a national securities exchange or authorized for quotation on the Nasdaq Stock Market nor (a)
- held of record by more than 2,000 stockholders.

(b)

In addition to a business combination statute, New York law, but not Delaware law, includes a "greenmail" type of anti-takeover provision, designed to prevent a corporation from buying at a premium price a corporate raider's interest in the corporation. In general, no publicly held New York corporation may purchase or agree to purchase more than 10% of its stock from a shareholder for more than the market value thereof (as defined) unless the purchase or agreement to purchase is approved by the affirmative vote of the board of directors followed by the affirmative vote of the holders of a majority of the votes of all outstanding shares entitled to vote thereon, or such greater percentage as the corporation's certificate of incorporation may require. This does not apply when the corporation offers to purchase shares from all holders of its stock, or with respect to stock which the holder has owned beneficially for more than two years.

42

Table of Contents

Appraisal Rights

Under Delaware law, in general, stockholders of a constituent corporation in a merger or consolidation have the right to demand and receive payment of the fair value of their stock in a merger or consolidation. However, except as otherwise provided by Delaware law, stockholders do not have appraisal rights in a merger or consolidation if, among other things, their shares are:

- do not prohibit tender offers or in any way regulate when, how, at what price or by whom they may be made,
- do not in any way delay the purchase of shares in tender offers,
- do not interfere with the right of a stockholder, whether "interested" or not, to mount a proxy contest
 to remove incumbent management,
- do not prevent a stockholder from buying sufficient stock to replace existing management immediately, and

• unlike certain other anti-takeover statutes (e.g. Indiana's), do not eliminate or delay a stockholder's right to vote on the election of directors or on any other corporate matters except certain defined self-dealing transactions.

and, in each case, the consideration such stockholders receive for their shares in a merger or consolidation consists solely of:

- listed on a national securities exchange or, like the Savoir common stock, designated as a Nasdaq National Stock Market security; or
- held of record by more than 2,000 stockholders;

See "Appraisal Rights."

Shareholders of a New York corporation have the right to dissent and receive payment of the fair value of their shares, in the case of certain amendments or changes to the certificate of incorporation adversely affecting their shares, a merger or consolidation in which the corporation is a constituent corporation, any sale, lease, exchange or other disposition of all or substantially all the corporation's assets, and certain share exchanges. Shareholders do not have appraisal rights with respect to mergers, consolidations and share exchanges, if their shares are listed on a national securities exchange or designated as a Nasdaq National Stock Market security.

The procedures for perfecting appraisal rights are similar under New York law and Delaware law (see "Appraisal Rights") except that New York law provides a procedure for the corporation to make a written offer prior to the commencement of litigation (the "Offer") to each dissenting shareholder to pay cash for his or her shares at a specified, uniform price which the corporation considers to be their fair value. If the effective date of the corporate action dissented from has occurred, the Offer must be accompanied by an 80% advance payment of the Offer price to each dissenting shareholder who has submitted his or her stock certificates. If the effective date has not yet occurred, such advance payment shall be sent forthwith upon its occurrence. If the corporation and a dissenting shareholder agree upon a price to be paid for such dissenting shareholder's shares within 30 days after the making of the Offer, payment in full must be made by the corporation within 60 days of the date on which the Offer was made or within 60 days of the effective date, whichever is later. If any dissenting shareholder fails to agree with the corporation during the aforesaid 30-day period, or if an Offer is not made within a specified period of time, only then may a proceeding for judicial appraisal be commenced.

The concept of "fair value" in payment for shares upon exercise of dissenters' rights is different under New York law and Delaware law. Under Delaware law, "fair value" must be determined exclusive of any element of value arising from the accomplishment or expectation of the transaction in question. New York law does not exclude such element of value but mandates that the court should consider the nature of the transaction and its effect on the corporation and its shareholders, and the concepts and methods of valuation then customary in the relevant financial and securities markets.

Amendments to Charters

Under Delaware law, unless the certificate of incorporation requires a greater vote, a proposed amendment to the certificate of incorporation requires an affirmative vote of a majority of the votes of the outstanding stock entitled to vote thereon, and a majority of the outstanding stock of each class entitled by Delaware law or the certificate of incorporation to vote thereon as a class. Also, the approval of the holders of a majority of the outstanding shares of any class of capital stock of a corporation, voting separately as a class, is required, whether or not entitled to vote by the certificate of incorporation, if the amendment would

- shares of stock of the corporation surviving or resulting from such merger or consolidation;
- shares of stock of any other corporation that at the effective date of the merger or consolidation will be either listed on a national securities exchange, which is true in the case of the Avnet common stock to be issued pursuant to the merger, or designated as a national market system security on an interdealer quotation system by the NASD, or held of record by more than 2,000 stockholders;
- cash in lieu of fractional shares of the corporations described in the two immediately preceding bullet points; or
- any combination of shares of stock and cash in lieu of fractional shares described in the three immediately preceding bullet points.

The authorized number of shares of any class of stock may be increased or decreased (but not below the number of shares of such class outstanding) by the requisite vote described above if so provided in the original certificate of incorporation or in any amendment thereto that created such class of stock or that was adopted prior to the issuance of any shares of such class, or in an amendment authorized by a majority vote of the holders of shares of such class.

Under New York law, amendments to a certificate of incorporation generally must be approved by vote of a majority of the votes of all outstanding shares entitled to vote thereon at a meeting of shareholders. The approval of a majority of the votes of all outstanding shares of any class of capital stock of a corporation, voting separately as a class, is required to approve a proposed amendment to a corporation's certificate of incorporation, whether or not such holders are otherwise entitled to vote on such amendment by the certificate of incorporation, that:

- increase or decrease the aggregate number of authorized shares of such class (except as provided in the last sentence of this paragraph),
- increase or decrease the par value of the shares of such class, or
- alter or change the powers, preferences or special rights of the shares of such class so as to affect them adversely.

Amendments to By-laws

Under Delaware law, the power to adopt, alter and repeal by-laws is vested in the stockholders, except to the extent that a corporation's certificate of incorporation vests concurrent power in the board of directors. The Savoir certificate of incorporation authorizes the board of directors to make, amend, rescind or repeal its by-laws.

certificate of incorporation, by-laws may be amended, repealed or adopted by a majority of the votes cast by the shares at the time entitled to vote in the election of directors. When so provided in the certificate of incorporation or a by-law adopted by the shareholders, by-laws also may be amended, repealed or adopted by the board of directors by such vote as may be therein specified, which vote may be greater than the vote otherwise prescribed by New York law, but any by-law adopted by the board of directors may be amended or repealed by the shareholders entitled

44

Table of Contents

to vote thereon as provided by New York law. The Avnet certificate of incorporation authorizes the board of directors to adopt, amend or repeal its by-laws.

Duration of Proxies

Under Delaware law, no proxy is valid more than three years after its date unless otherwise provided in the proxy. A proxy shall be irrevocable if it states that it is irrevocable and if, and only as long as, it is coupled with an interest sufficient in law to support an irrevocable power. A proxy may be made irrevocable regardless of whether the interest with which it is coupled is an interest in the stock itself or an interest in the corporation generally.

Under New York law, no proxy is valid more than eleven months after its date unless otherwise provided in the proxy. Irrevocable proxies may be created for:

- would decrease the par value of the shares of such class, change any shares of such class into a different number of shares of the same class or into the same or a different number of shares of a different class, alter or change the designation, relative rights, preferences or limitations of the shares of such class, including the provision of new conversion rights or the alteration of any existing conversion rights, so as to affect them adversely;
- would exclude or limit the voting rights of such shares, except as such rights may be limited by voting rights given to new shares then being authorized of any existing or new class or series of shares; or
- would subordinate their rights by authorizing shares having preferences superior to the rights of such existing shares.

Shareholder Action by Written Consent

Under Delaware law, unless otherwise provided in the certificate of incorporation, any action required or permitted to be taken at a meeting of stockholders may be taken without a meeting, without prior notice and without a vote, if a written consent or consents setting forth the action taken is signed by the holders of outstanding stock having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote upon such action were present and voted. The Savoir certificate of incorporation does not include any provision limiting the power of stockholders to consent in writing without a meeting.

New York law provides that shareholder action may be taken without a meeting upon the written consent of the holders of all

outstanding shares entitled to vote, and also allows, if the certificate of incorporation so provides, shareholder action without a meeting upon the written consent of holders of outstanding shares having not less than the minimum number of votes that would be necessary to authorize such action at a meeting at which all shares entitled to vote thereon were present and voted. The Avnet certificate of incorporation does not authorize shareholders to act by less than unanimous written consent.

Director Nominations and Shareholder Proposals

The Savoir by-laws require that to properly bring a director nomination or stockholder proposal before the annual meeting, a stockholder must give notice thereof to the corporate secretary not less than 35 days prior to the meeting; however, if less than 50 days' notice of the meeting is given to stockholders, notice by the stockholder must be received by the earlier of the 15th day following such notice or two days prior to the date of the meeting.

The Avnet certificate of incorporation and by-laws do not impose any notice requirements for a director nomination or shareholder proposal. However, any such proposal is subject to the deadlines and conditions set out in SEC Rule 14a-8 if the proponent wants the proposal to be included in Avnet's proxy materials for a meeting of shareholders.

45

Table of Contents

Special Shareholder Meetings

Delaware law provides that a special meeting of stockholders may be called by the board of directors or by such person or persons as may be authorized by the certificate of incorporation or by the by-laws. The Savoir by-laws provide that a special meeting of stockholders will be held at any time, upon the call of the president or the board of directors at the request of the holders of a majority of the outstanding shares of capital stock entitled to vote.

New York law also provides that special meetings of shareholders may be called by the board of directors and by such persons as may be authorized in the certificate of incorporation or the by-laws. The Avnet by-laws provide that special meetings of the shareholders may be called by the board of directors or the chairman of the board, president or secretary at the written request of shareholders owning 75% of the shares entitled to vote at the meeting.

New York law provides that, if, for a period of one month after the date fixed by or under the by-laws for the annual meeting of shareholders or, if no date has been so fixed, for a period of 13 months after the last annual meeting, there is a failure to elect a sufficient number of directors to conduct the business of the corporation, the board of directors shall call a special meeting for the election of directors. If such special meeting is not called by the board of directors within two weeks after the expiration of such period or if it is called but there is a failure to elect such directors for a period of two months after

the expiration of such period, holders of 10% of the votes of the shares entitled to vote in an election of directors may, in writing, demand the call of a special meeting for the election of directors.

Under New York law, only such business may be transacted at a special meeting which is related to the purpose or purposes set forth in the notice of meeting.

Removal of Directors

Under Delaware law, a director or directors of Savoir may be removed with or without cause by the holders of a majority in voting power of the shares then entitled to vote in an election of directors.

New York law provides that any or all of Avnet's directors may be removed for cause by vote of the shareholders, and, if the certificate of incorporation or the specific provisions of a by-law adopted by the shareholders so provides, directors may be removed by action of the board of directors. If the certificate of incorporation or the by-laws so provide, any or all of the directors may be removed without cause by vote of the shareholders. The Avnet certificate of incorporation provides that directors may be removed without cause by the vote of shareholders holding a majority of the shares of Avnet common stock. An action to procure a judgment removing a director for cause may be brought by the attorney general of New York or by the holders of 10% of the outstanding shares, whether or not entitled to vote.

Vacancies

of incorporation or the by-laws, vacancies on a board of directors and newly created directorships resulting from an increase in the authorized number of directors may be filled by a majority of the directors then in office, although less than a quorum, or by the sole remaining director. In addition, if, at the time of the filling of any such vacancy or newly created directorship, the directors in office constitute less than a majority of the whole board of directors (as constituted immediately prior to any such increase), the Delaware Court of Chancery may, upon application of any stockholder or stockholders holding at least 10% of the total number of outstanding shares entitled to vote for such directors, summarily order an election to fill any such vacancy or newly created directorship, or replace the directors chosen by the directors then in office. The Savoir by-laws provide that any vacancies on the Savoir board caused by death, resignation or removal, or newly created directorships resulting from an increase in the number of directors, shall be filled by the affirmative vote of a majority of the remaining directors then in office, even though less than a quorum, or the sole remaining director.

46

Table of Contents

Under New York law, newly created directorships resulting from an increase in the number of directors and vacancies occurring on the board of directors for any reason, except the removal of

directors without cause, may be filled by vote of the board of directors. However, the certificate of incorporation or by-laws may provide that such newly created directorships or vacancies are to be filled by vote of the shareholders. Unless the certificate of incorporation or the specific provisions of a by-law adopted by the shareholders provide that the board of directors may fill vacancies occurring on the board of directors by reason of the removal of directors without cause, such vacancies may be filled only by vote of the shareholders. A director elected to fill a vacancy, unless elected by the shareholders, will hold office until the next meeting of shareholders at which the election of directors is in the regular order of business and until his or her successor has been elected and qualified. The Avnet by-laws provide that any vacancy on the Avnet board may be filled by a majority vote of the remaining directors, though less than a quorum. The by-laws also provide that vacancies resulting from the removal of directors by the shareholders with or without cause shall be filled by the shareholders.

Indemnification of Directors and Officers

Delaware law and New York law permit a corporation to indemnify its directors and officers for judgments, fines and other expenses incurred by them in connection with an action or proceeding other than an action by or in the right of the corporation (a "derivative action"). In connection with any pending, threatened or completed derivative action, Delaware law permits such indemnification only against expenses reasonably incurred, and New York law permits such indemnification only against amounts paid in settlement and reasonable expenses incurred. In New York and Delaware, indemnification is permitted only if the director or officer has acted in good faith and in a manner he or she reasonably believed to be in or (in Delaware) not opposed to the best interests of the corporation and, with respect to any criminal proceeding, had no reasonable cause to believe his or her conduct was unlawful. Furthermore, both state's laws provide that expenses incurred in defending any action or proceeding may be paid by the corporation in advance of the final disposition upon receipt of an undertaking by or on behalf of the director or officer to repay the amount if it is ultimately determined that he or she is not entitled to be indemnified by the corporation.

In both states, the statutory provisions for indemnification and advancement of expenses are not exclusive of any other rights to which those seeking indemnification or advancement of expenses may be entitled under (in Delaware) any by-law, agreement, vote of stockholders or disinterested directors or otherwise or (in New York) in the certificate of incorporation or by-laws or, when authorized thereby, in a resolution of directors or shareholders or an agreement providing for indemnification. Such other rights may, for example, provide for indemnification against (in New York) judgments and fines and (in Delaware) amounts paid in settlement which are incurred by the indemnified person in connection with derivative actions. New York law does not permit such other indemnification in any case where a judgment or final adjudication adverse to the director or officer establishes that his or her acts were committed in bad faith or were the result of active and deliberate dishonesty and were material to the cause of action so adjudicated, or that he personally gained in fact a financial profit or other advantage to which he was not legally entitled.

Avnet's by-laws and Savoir's certificate of incorporation and by-laws provide for mandatory indemnification of directors and officers and advancement of indemnified expenses to the full extent now or hereafter permitted by New York law and Delaware, respectively.

Insofar as indemnification for liabilities arising under the Securities Act may be permitted to directors, officers or persons controlling Avnet or Savoir pursuant to the foregoing provisions, Avnet and Savoir have been informed that, in the opinion of the SEC, such indemnification is against public policy as expressed in the Securities Act and is, therefore, unenforceable.

Both New York law and Delaware law permit a corporation to purchase and maintain insurance on behalf of any director or officer of the corporation against any liability asserted against him and incurred by him in such capacity, whether or not the corporation would have the power to indemnify him against

47

Table of Contents

such liability. The directors and officers of Avnet are currently covered as insureds under directors' and officers' liability insurance. Such insurance, subject to annual renewal and certain rights of the insurer to terminate, provides an aggregate maximum of \$50,000,000 of coverage for directors and officers of Avnet and its subsidiaries against claims made during the policy period.

Limitation of Personal Liability of Directors

Delaware law provides that a corporation's certificate of incorporation may include a provision limiting the personal liability of a director to the corporation or its stockholders for monetary damages for breach of fiduciary duty as a director. However, no such provision can eliminate or limit the liability of a director for:

- a pledgee;
- a person who has purchased or agreed to purchase the shares;
- a creditor of the corporation who extends credit in consideration of the proxy;
- a person who has contracted to perform services as an officer of the corporation if a proxy is required by the employment contract; and
- a person designated under a voting agreement.

The Savoir certificate of incorporation provides that a director's liability to Savoir for breach of duty to Savoir or its stockholders shall be limited to the fullest extent permitted by Delaware law.

New York law provides that a corporation's certificate of incorporation may contain a provision eliminating or limiting the personal liability of directors to the corporation or its shareholders for damages for any breach of duty in such capacity. However, no such provision can eliminate or limit the liability of any director:

- any breach of the director's duty of loyalty to the corporation or its stockholders;
- acts or omissions not in good faith or that involve intentional misconduct or a knowing violation of the law;

- violation of certain provisions of Delaware law;
- any transaction from which the director derived an improper personal benefit; or
- · any act or omission prior to the adoption of such a provision in the certificate of incorporation.

The Avnet certificate of incorporation provides that no director will be personally liable to Avnet or its shareholders for damages for any breach of duty as a director, except:

- if a judgment or other final adjudication adverse to such director establishes that such director's acts or omissions were in bad faith or involved intentional misconduct or a knowing violation of law, that the director personally gained in fact a financial profit or other advantage to which such director was not legally entitled, or that the director's acts violated certain provisions of New York law; or
- for any act or omission prior to the adoption of such a provision in the certificate of incorporation.

48

Table of Contents

Dividends

Delaware law generally permits dividends to be paid by a corporation (subject to any restrictions in its certificate of incorporation) out of (a) the corporations' surplus, defined as the excess of the net assets of the corporation over the amount determined to be the capital of the corporation by the board of directors (which amount cannot be less than the aggregate par value of all issued shares of capital stock), or (b) the corporation's net profits of the current or the preceding fiscal year, or both, unless net assets are less than the aggregate share capital of all outstanding preferred stock. Savoir's certificate of incorporation contains no restriction on the payment of dividends.

Under New York law, a corporation may declare and pay dividends, or make other distributions in cash or its bonds or its property, on its outstanding shares except when the corporation is insolvent or would thereby be made insolvent, or when the declaration, payment or distribution would be contrary to any restrictions contained in the certificate of incorporation. Avnet's certificate of incorporation contains no such restrictions. In general, dividends may be declared or paid and other distributions may be made out of surplus only, so that the net assets of the corporation remaining after such declaration, payment or distribution shall at least equal the amount of its stated capital.

Appraisal Rights

Section 262 of the Delaware General Corporation Law provides appraisal rights (sometimes referred to as "dissenters' rights") under certain circumstances to stockholders of a Delaware corporation that is involved in a merger. However, Section 262 appraisal rights are not available to stockholders whose securities are listed on a national securities exchange or the Nasdaq National Stock Market, or held of record by more than 2,000 holders, and who are not required to accept in exchange for their stock anything other than stock of another corporation listed on a national securities exchange or the Nasdaq National Stock Market, or stock of the

surviving corporation of the merger and, in either case, cash in lieu of fractional shares. Because the Savoir common stock is traded on the Nasdaq National Stock Market, and because holders of Savoir common stock will receive Avnet common stock in the merger, which is listed on the New York Stock Exchange, holders of Savoir common stock will not have appraisal rights with respect to the merger.

Record holders of Savoir series A preferred stock that follow the appropriate procedures are entitled to appraisal rights under Section 262 in connection with the merger.

The following discussion is not a complete statement of the law pertaining to appraisal rights under the Delaware General Corporation Law and is qualified in its entirety by the full text of Section 262, which appears as Appendix E to this proxy statement/prospectus. All references in Section 262 to a "stockholder" and in this discussion to a "record holder" are to the record holder of the shares of series A preferred stock immediately prior to the effective time of the merger. A person having a beneficial interest in shares of Savoir series A preferred stock held of record in the name of another person, such as a broker or nominee, must act promptly to cause the record holder to follow the steps summarized below properly and in a timely manner to perfect appraisal rights.

Under the Delaware General Corporation Law, record holders of series A preferred stock who follow the procedures set forth in Section 262 will be entitled to have their shares of series A preferred stock appraised by the Delaware Court of Chancery and to receive payment of the "fair value" of such shares, exclusive of any element of value arising from the accomplishment or expectation of the merger, together with a fair rate of interest, as determined by the Delaware Court of Chancery.

A holder of series A preferred stock wishing to exercise appraisal rights must deliver to Savoir, before the vote on the adoption of the merger agreement at the special meeting, a written demand for appraisal of such holder's series preferred stock. The demand must reasonably inform Savoir of the identity of the holder of record and that the holder intends to demand an appraisal of the fair value of the shares held. In addition, a holder of series A preferred stock wishing to exercise appraisal rights or wishing to preserve

49

Table of Contents

such holder's right to do so must hold of record such shares on the date the written demand for appraisal is made and must continue to hold such shares through the effective time of the merger.

Stockholders electing to exercise their appraisal rights under Section 262 must not vote for adoption of the merger agreement. A vote by a stockholder against adoption of the merger agreement is not required in order for that stockholder to exercise appraisal rights. However, if a stockholder returns a signed proxy but does not specify a vote against adoption of the merger agreement or a direction to abstain, the proxy, if not

revoked, will be voted for the adoption of the merger agreement, which will have the effect of waiving that stockholder's appraisal rights.

Only a holder of record of series A preferred stock is entitled to assert appraisal rights for series A preferred stock registered in such holder's name. A demand for appraisal should be executed by or on behalf of the holder of record, fully and correctly, as such holder's name appears on such holder's stock certificates, and must state that such holder intends thereby to demand appraisal of such holder's shares of series A preferred stock. If no number of shares of series A preferred stock is expressly mentioned, the demand will be presumed to cover all series A preferred stock held in the name of the record owner.

If the shares of series A preferred stock are owned of record in a fiduciary capacity, such as by a trustee, guardian or custodian, execution of the demand should be made in that capacity, and, if the shares of series A preferred stock are owned of record by more than one person, as in a joint tenancy or tenancy in common, the demand should be executed by or on behalf of all joint owners. An authorized agent, including one or more joint owners, may execute a demand for appraisal on behalf of a holder of record; however, the agent must identify the record owner or owners and expressly disclose the fact that, in executing the demand, the agent is agent for such owner or owners. A record holder, such as a broker who holds series A preferred stock as nominee for several beneficial owners, may exercise appraisal rights with respect to the shares of series A preferred stock held for one or more beneficial owners while not exercising such rights with respect to the series A preferred stock held for other beneficial owners. In such case, however, the written demand should set forth the number of shares of series A preferred stock as to which appraisal is sought.

All written demands for appraisal of series A preferred stock should be mailed or delivered to Savoir at 44931 Industrial Blvd., Fremont, California 94538, Attention: Corporate Secretary, so as to be received before the vote on the adoption of the merger agreement at the special meeting.

Within 10 days after the effective time of the merger, Savoir must send a notice as to the effectiveness of the merger to each person who delivered a demand for appraisal. Within 120 days after the effective time, Savoir, or any holder of series A preferred stock entitled to appraisal rights under Section 262 and who has complied with the foregoing procedures, may file a petition in the Delaware Court of Chancery demanding a determination of the fair value of such shares. Savoir is not under any obligation, and has no present intention, to file a petition with respect to the appraisal of the fair value of the series A preferred stock. Accordingly, it will be the obligation of the dissenting holders of the shares of series A preferred stock to initiate all necessary action to perfect their appraisal rights within the time prescribed in Section 262.

Within 120 days after the effective time, any record holder of Savoir series A preferred stock who has complied with the requirements for exercise of appraisal rights will be entitled to request in writing a statement from Savoir setting forth the aggregate number of shares of series A preferred stock with respect to which demands for appraisal have been received and the aggregate number of holders of such shares. Such statement must be mailed within 10 days after the written request has been received by Savoir or within 10 days after expiration of the period for delivery of demands for appraisal, whichever is later.

If a holder of series A preferred stock timely files a petition for appraisal and serves a copy of such petition upon Savoir, Savoir will then be obligated within 20 days after such service to file with the Delaware Register in Chancery in which the petition was filed a duly verified list containing the names and addresses of all stockholders who have demanded an appraisal of their shares and with whom agreements as to the value of their shares have not been reached by Savoir. After notice to such stockholders as

50

Table of Contents

required by the Delaware Court of Chancery, the Court is empowered to conduct a hearing on such petition to determine those stockholders who have complied with Section 262 and who have become entitled to appraisal rights. The Court of Chancery may require the holders of shares of Savoir series A preferred stock who demanded payment for their shares to submit their stock certificates to the Delaware Register in Chancery for notation of the pendency of the appraisal proceeding. If any stockholder fails to comply with such direction, the Delaware Court of Chancery may dismiss the proceedings as to such stockholder.

After determining the holders of series A preferred stock entitled to appraisal, the Delaware Court of Chancery will appraise the fair value of their shares of series A preferred stock, exclusive of any element of value arising from the accomplishment or expectation of the merger, together with a fair rate of interest, if any, to be paid upon the amount determined to be the fair value. Holders considering seeking appraisal should be aware that the fair value of their series A preferred stock as determined under Section 262 could be more than, the same as or less than the value of the Avnet common stock that they would otherwise receive in the merger if they did not seek appraisal of their series A preferred stock and that investment banking opinions as to fairness from a financial point of view are not necessarily opinions as to fair value under Section 262. The Delaware Supreme Court has stated that "proof of value by any techniques or methods which are generally considered acceptable in the financial community and otherwise admissible in court" should be considered in the appraisal proceedings. More specifically, the Delaware Supreme Court has stated that: "Fair value, in an appraisal context, measures" that which has been taken from the stockholder, viz., his proportionate interest in a going concern. In the appraisal process the corporation is valued 'as an entity,' not merely as a collection of assets or by the sum of the market price of each share of its stock. Moreover, the corporation must be viewed as an on-going enterprise, occupying a particular market position in the light of future prospects." In addition, Delaware courts have decided that the statutory appraisal remedy, depending on factual circumstances, may or may not be a stockholder's exclusive remedy. The Delaware Court of Chancery will also determine the amount of interest, if any, to be paid upon the amounts to be received by persons whose shares of series A preferred stock have been appraised. The costs of the action may be determined by the Delaware Court of Chancery and taxed upon the parties as the Delaware Court of Chancery deems equitable. The Delaware Court of Chancery may also order that all or a portion of the expenses incurred by any holder of series A preferred stock in connection with an appraisal, including, without limitation, reasonable attorneys' fees and the fees and expenses of experts $% \left(1\right) =\left(1\right) \left(1\right) \left($ utilized in the appraisal proceeding, be charged pro rata against the value of all of the shares of series A preferred stock

entitled to appraisal.

Any holder of series A preferred stock who has duly demanded an appraisal in compliance with Section 262 will not, after the effective time of the merger, be entitled to vote the shares of series A preferred stock subject to such demand for any purpose or be entitled to the payment of dividends or other distributions on those shares (except dividends or other distributions payable to holders of record of series A preferred stock as of a date prior to the effective time).

If any holder of series A preferred stock who demands appraisal of such holder's shares of series A preferred stock under Section 262 fails to perfect, or effectively withdraws or loses such holder's right to appraisal, as provided in the Section 262, the series A preferred stock of such holder will be converted into Avnet common stock in accordance with the merger agreement (without interest), as more fully described under "The Transaction Agreements — The Merger Agreement — Terms of the Merger." A holder of series A preferred stock will fail to perfect, or will effectively lose, the right to appraisal if no petition for appraisal is filed within 120 days after the effective time. A holder may withdraw a demand for appraisal by delivering to Savoir a written withdrawal of the demand for appraisal and an acceptance of the merger. Any such attempt to withdraw made more than 60 days after the effective time will, however, require the written approval of Avnet. Further, once a petition for appraisal is filed, the appraisal proceeding may not be dismissed as to any holder absent court approval.

Failure to follow the steps required by Section 262 for perfecting appraisal rights may result in the loss of such rights, in which event a holder of series A preferred stock will be entitled to receive only the

51

Table of Contents

consideration set forth in the merger agreement for each share of Savoir series A preferred stock owned by such holder immediately prior to the effective time of the merger.

Experts

The consolidated financial statements of Savoir at December 31, 1999 and 1998, and for each of the three years in the period ended December 31, 1999, included in Savoir's Annual Report on Form 10-K/A for the fiscal year ended December 31, 1999, which is incorporated by reference in and is attached as Appendix F to this proxy statement/prospectus, have been so included in reliance on the report of PricewaterhouseCoopers LLP, independent accountants, given on the authority of said firm as experts in auditing and accounting.

The consolidated financial statements and schedules of Avnet as of July 2, 1999 and June 26, 1998 and for each of the three years in the period ended July 2, 1999, incorporated by reference in this proxy statement/prospectus have been audited by Arthur Andersen LLP, independent public accountants, as

indicated in their reports with respect thereto, and are incorporated herein by reference in reliance upon the authority of that firm as experts in giving such reports.

The consolidated financial statements of Marshall Industries incorporated by reference in this proxy statement/prospectus from Avnet's Current Report on Form 8-K bearing cover date of October 20, 1999, for the fiscal years ended May 31, 1999, 1998 and 1997 have been audited by Arthur Andersen LLP, independent public accountants, as indicated in their report with respect thereto, and are incorporated herein by reference in reliance upon the authority of that firm as experts in giving such reports.

Validity of Shares

The validity of the Avnet common stock to be issued pursuant to the merger agreement will be passed upon by David R. Birk, Senior Vice President and General Counsel of Avnet. Mr. Birk beneficially owns 43,284 shares of Avnet common stock, which includes 40,625 shares issuable upon exercise of employee stock options.

Additional Information for Savoir Stockholders

Independent Auditors Will Be Present At Special Meeting

Representatives of PricewaterhouseCoopers LLP are expected to be present at the Savoir special meeting and will have an opportunity to make a statement if they desire to do so and will be available to respond to appropriate questions.

Stockholder Proposals for the Next Annual Meeting

Savoir has not scheduled an annual meeting for 2000 due to the pending merger. Assuming the merger is completed, no 2000 annual meeting will be held for Savoir. If the merger is not completed, an annual meeting of stockholders would be held before the end of calendar year 2000. In the event that such a meeting is held, any proposals of Savoir stockholders intended to be included in the Savoir proxy statement for such meeting would have to be received by Savoir at its corporate headquarters, 44931 Industrial Blvd., Fremont, California 94538 a reasonable time before Savoir begins to print and mail its proxy materials with respect to such meeting.

Stockholder proposals submitted outside the processes of Rule 14a-8 under the Securities Exchange Act of 1934 (i.e., a proposal to be presented at the next annual meeting of Savoir stockholders but not submitted for inclusion in the Savoir proxy statement for that meeting) must be timely submitted. Until further notice, to be timely with respect to the annual meeting (if one is held), a stockholder's notice of business to be brought before that annual meeting would have to be received in writing by Savoir at its principal executive office not less than thirty (30) days nor more than ninety (90) days prior to that

annual meeting. The notice must also contain some additional information required by Savoir's by-laws and otherwise comply with applicable legal requirements.

Where You Can Find More Information

Avnet has filed with the Securities and Exchange Commission a Registration Statement on Form S-4 under the Securities Act that registers the distribution to the Savoir stockholders of the Avnet common stock to be issued in the merger. The Registration Statement, including the attached exhibits and schedules, contains additional relevant information about Avnet common stock. The rules and regulations of the SEC allow us to omit certain information included in the Registration Statement from this proxy statement/prospectus.

In addition, Savoir and Avnet file reports, proxy statements and other information with the Commission under the Exchange Act. You may read and copy this information at the following locations of the Commission:

- if a judgment or other final adjudication adverse to him or her establishes that his or her acts or omissions were in bad faith or involved intentional misconduct or a knowing violation of law, that he or she personally gained in fact a financial profit or other advantage to which he or she was not legally entitled, or that his or her acts violated the provision of New York law that imposes liability on directors who vote for or concur in the following corporate actions: (1) the declaration of a dividend or other distribution to shareholders when the corporation is insolvent or would thereby be made insolvent or when the declaration or distribution would be contrary to any restrictions in the certificate of incorporation or when the stated capital of the corporation would be impaired thereby; (2) the purchase of shares of the corporation except out of surplus or, under certain limited circumstances, out of stated capital; (3) the distribution of assets to shareholders after dissolution of the corporation without providing for all known liabilities; or (4) the making of a loan to a director unless authorized by vote of the shareholders; or
- for any act or omission prior to the adoption of this provision by the shareholders of Avnet.

You may also obtain copies of this information by mail from the Public Reference Section of the Securities and Exchange Commission, 450 Fifth Street, N.W., Room 1024, Washington, D.C. 20549, at prescribed rates. Further information on the operation of the SEC's Public Reference Room in Washington, D.C. can be obtained by calling the SEC at 1-800-SEC-0330.

The SEC also maintains an Internet world wide web site that contains reports, proxy statements and other information about issuers, such as Savoir and Avnet, who file electronically with the SEC. The address of that site is http://www.sec.gov.

You can also inspect reports, proxy statements and other information about Savoir and Avnet at the offices of the New York Stock Exchange, 20 Broad Street, New York, New York 10005 and, in the case of Avnet, the offices of the Pacific Exchange, Inc., 301 Pine Street, San Francisco, California 94104.

The SEC allows us to "incorporate by reference" information into this proxy statement/prospectus, which means that we can disclose important information to you by referring you to another document filed separately with the SEC. The information incorporated by reference is deemed to be part of this proxy statement/prospectus, except for any information superseded by information in this proxy statement/prospectus. This proxy statement/prospectus incorporates by reference the documents set forth below that Avnet and Savoir previously filed with the SEC. These documents contain important information about Avnet, Savoir and their finances.

Public Reference Room 450 Fifth Street, N.W. Room 1024 Washington, D.C. 20549 New York Regional Office 7 World Trade Center Suite 1300 New York, New York 10048 Chicago Regional Office Citicorp Center 500 West Madison Street Suite 1400 Chicago, Illinois 60661-2511

53

Table of Contents

Savoir Filings (File No. 000-11560)

Period or Date

Annual Report on Form 10-K/A

Fiscal Year ended December 31, 1999

Avnet also incorporates by reference additional documents that it may file with the SEC between the date of this proxy statement/prospectus and the date of the Savoir special meeting. These documents include periodic reports, such as Annual Reports on Form 10-K, Quarterly Reports on Form 10-Q and Current Reports on Form 8-K, as well as proxy statements.

Savoir has supplied all information contained or incorporated by reference in this proxy statement/prospectus relating to Savoir, and Avnet has supplied all such information relating to Avnet.

54

Table of Contents

APPENDIX A

AMENDED AND RESTATED

AGREEMENT

AND

PLAN OF MERGER

dated as of

March 2, 2000

by and among

AVNET, INC.

and

TACTFUL ACQUISITION CORP.

and

SAVOIR TECHNOLOGY GROUP, INC.

A-1

Table of Contents

Table of Contents

A-2

Avnet Filings (File No. 1-4224)

Period or Date

Page

Annual Report on Form 10-K Quarterly Reports on Form 10-Q Current Reports on Form 8-K

Description of Avnet's common stock which appears in Avnet's Registration Statement for the registration of the common stock under Section 12(b) of the Securities Exchange Act of 1934.

Fiscal Year ended July 2, 1999 Quarters ended October 1, 1999 and December 31, 1999 Cover dates September 28, 1999, October 20, 1999, December 22, 1999, January 26, 2000, February 8, 2000 and April 25, 2000

Certain Agreements

Table of Contents

	ARTICLE I THE MERGER	
1.1	The Merger	A-6
1.2	Closing	A-0 A-7
1.3	Effective Time	A-7 A-7
1.4	Certificate of Incorporation	A-7 A-7
1.5	·	A-7 A-7
1.6	By-Laws Officers and Directors	A-7 A-7
1.7	Conversion of Shares	A-7 A-7
1.7	***************************************	A-7 A-8
	Surrender of Shares; Transfer Books	
1.9	Options and Warrants	A-10
1.10	Affiliates	A-10
1.11	Dissenting Shares	A-11
	ARTICLE II	
	REPRESENTATIONS AND WARRANTIES OF COMPANY	
2.1	Organization; Subsidiaries	A-11
2.2	Capitalization	A-12
2.3	Authority; Validity	A-12
2.4	No Conflict	A-13
2.5	Consents	A-13
2.6	Financial Statements; SEC Filings	A-13
2.7	Tax Matters	A-14
2.8	Absence of Certain Changes or Events	A-15
2.9	Material Contracts; Customers and Suppliers	A-15
2.10	Title and Related Matters	A-16
2.11	Employee Benefit Plans	A-16
2.12	Employment Agreements	A-18
2.13	Legal Proceedings	A-18
2.14	Compliance with Law; Accuracy of Certain Information	A-18
2.15	Accuracy of Proxy and Registration Statement and Other Information	A-19
2.16	Insurance	A-19
2.17	Environmental Matters	A-19
2.18	Certain Agreements	A-19

2.19	Intellectual Property	A-20
2.20	Product Warranties	A-21
2.21	Labor Matters	A-21
2.22	Related Party Transactions	A-21
2.23	State Takeover Statutes	A-21
2.24	Brokers; Advisors	A-22
2.25	Full Disclosure	A-22
	ARTICLE III	
	REPRESENTATIONS AND WARRANTIES OF PARENT AND BUYER	
3.1	Organization	A-22
3.2	Authority; Validity	A-22

A-3

Table of Contents

		Page
3.3	No Conflict	A-22
3.4	Consents	A-22
3.5	Legal Proceedings	A-22
3.6	Financial Statements, SEC Filings	A-23
3.7	No Material Adverse Effect	A-23
3.8	Compliance with Law	A-23
3.9	Accuracy of Proxy and Registration Statement and Other Information	A-23
3.10	Full Disclosure	A-24
3.11	No Brokers or Finders	A-24
	ARTICLE IV	
	COVENANTS	
4.1	Access and Information	A-24
4.2	Governmental Filings	A-25
4.3	Consents and Approvals	A-25
4.4	Meeting of Shareholders; Proxy and Registration Statement; Listing Application	A-25
	(a) Meeting of Shareholders	A-25 A-25
	(b) Proxy and Registration Statement	A-25 A-25
	(c) Indemnification	A-25 A-26
	(d) Listing Application	A-26 A-26
4.5	Conduct of Company Business	A-26
4.0	(a) Ordinary Course	A-26
	(b) Charter Documents	A-26
	(c) Dividends	A-26
	(d) Stock	A-26
4.6	Publicity	A-26
4.7	Notification of Defaults and Adverse Events	A-27
4.8	Satisfy Conditions to Closing	A-27
4.9	Termination Fee	A-27
4.10	Anti-takeover Statutes	A-27
4.11	Indemnification; Insurance	A-27
	(a) Indemnification	A-27
	(b) Insurance	A-28
4.12	Employee Benefits	A-28
4.13	No Solicitation	A-28
4.14	Consent of Holders of Options	A-29
4.15	Redemption of Series B Preferred Stock	A-29
4.16	Audited Financial Statements	A-29
	ARTICLE V CONDITIONS	
5.1	Conditions to Obligations of Company, Parent and Buyer	A-29
J.1	(a) Approval	A-29
	(b) Approval from Government Entities	A-30
	(c) Absence of Governmental Litigation	A-30
	(d) Effectiveness of Registration Statement	A-30
	(e) Market Conditions	A-30

A-4

Table of Contents

Annexes

		Page
5.2	Conditions to Obligations of Parent and Buyer (a) Representations and Compliance (b) Tax Opinion (c) No Material Adverse Effect	A-30 A-30 A-30 A-30
	(d) Material Contracts	A-30

5.3	(e) Consent of Option Holders(f) Audited Financial StatementsConditions to Obligations of Company(a) Representations and Compliance	A-30 A-30 A-31 A-31
	(b) Tax Opinion	A-31
	(c) Listing	A-31
	(d) No Material Adverse Effect	A-31
	ARTICLE VI	
C 1	TERMINATION, AMENDMENT AND WAIVER	A 24
6.1	Termination and Abandonment	A-31
6.2	Effect of Termination	A-32
6.3	Amendment	A-32
6.4	Extension; Waiver	A-32
	ARTICLE VII	
	MISCELLANEOUS	
7.1	Termination of Representations and Warranties	A-32
7.2	Expenses	A-32
7.3	Remedies	A-32
7.4	Notices	A-33
7.5	Further Assurances	A-33
7.6	Assignability	A-33
7.7	Governing Law	A-33
7.8	Interpretation	A-33
7.9	Counterparts	A-33
7.10	Integration	A-33
	ARTICLE VIII	
	DEFINITIONS	
8.1	Definitions	A-34

Schedules

Annex A Inducement Agreement
Annex B Option Agreement
Annex C Affiliate Letter

A-5

Table of Contents

Amended and Restated Agreement and Plan of Merger

THIS AMENDED AND RESTATED AGREEMENT AND PLAN OF MERGER (this "Agreement") is entered into as of March 2, 2000 by and among Avnet, Inc., a New York corporation ("Parent"), Tactful Acquisition Corp., a Delaware corporation and a wholly-owned subsidiary of Parent ("Buyer"), and Savoir Technology Group, Inc., a Delaware corporation ("Company").

Recitals

WHEREAS, Parent, Buyer and Company entered into an Agreement and Plan of Merger, dated as of March 2, 2000 (the "Original Merger Agreement");

WHEREAS, Parent, Buyer and Company entered into an Amendment No. 1 to the Original Merger Agreement, dated as of April 19, 2000 (the "Amendment No. 1");

WHEREAS, the Parties desire to set forth an amended and restated agreement and plan of merger incorporating the Original Merger Agreement, as amended by Amendment No. 1, and to further clarify certain of the provisions contained in the Original Merger Agreement, as amended;

WHEREAS, the Parties desire that the provisions of this Agreement shall be effective as of the date of the Original Merger Agreement;

WHEREAS, the respective Boards of Directors of Parent, Buyer and Company have each approved the acquisition of Company upon the terms and subject to the conditions set forth herein;

WHEREAS, to induce Parent and Buyer to enter into this Agreement, (i) certain beneficial and record holders of capital stock of Company are entering into an Inducement Agreement (the "Inducement Agreement") to vote their capital stock of Company in favor of the transactions contemplated by this Agreement, in the form of Annex A to this Agreement, and (ii) Company is entering into an Option Agreement (the "Option Agreement") granting Parent the right to acquire 2,023,435 shares of the common stock, par value \$.01 per share, of Company ("Company Common Stock") representing 15% of the currently outstanding shares of Company Common Stock upon the terms and conditions set forth therein, in the form of Annex B to this Agreement;

WHEREAS, Parent, Buyer and Company desire to make certain representations, warranties, covenants and agreements in connection with this Agreement and prescribe various conditions to the Merger (as such term is defined below); and

WHEREAS, Parent, Buyer and Company first entered into this Agreement as of March 2, 2000, and are amending and restating this Agreement as of the date hereof to conform it with the Certificate of Designations of Terms of the Series A Preferred Shares (as such term is defined in Section 1.7(b)), to provide for appraisal rights for the holders of Series A Preferred Shares, and to make certain additional amendments and corrections.

Agreement

NOW, THEREFORE, in consideration of the foregoing premises and of the representations, warranties, covenants and agreements contained herein, the parties hereto agree as follows:

ARTICLE I

THE MERGER

1.1 The Merger. Subject to the terms and conditions of this Agreement, at the Effective Time (as defined below), Buyer will be merged with and into Company and the separate corporate existence of Buyer will thereupon cease (the "Merger"). Company will be the surviving corporation in the Merger (sometimes hereinafter referred to as the "Surviving Corporation") and will continue to be governed by the laws of the State of Delaware. The separate corporate existence of Company with all its rights, privileges, immunities,

A-6

Table of Contents

Company will succeed to all of the rights and properties of Buyer and will be subject to all of the debts and liabilities of Buyer.

- 1.2 Closing. The closing of the transactions contemplated hereby (the "Closing") will take place (i) at the offices of Parent at 10:00 A.M., Pacific Standard time on the second business day after the day on which the last of the conditions set forth in Article V is fulfilled or waived in accordance with this Agreement or (ii) at such other place and time or on such other date as the parties hereto may agree (the date of the Closing, the "Closing Date").
- 1.3 Effective Time. Subject to the provisions of this Agreement and provided that this Agreement has not been terminated or abandoned pursuant to Article VI, a certificate of merger (the "Certificate of Merger") shall be duly prepared, executed and acknowledged by Company and thereafter filed with the Secretary of State of Delaware in accordance with Section 251 of the General Corporation Law of the State of Delaware (the "DGCL"), on or as soon as practicable after the Closing Date. The Merger will become effective immediately upon the filing of the Certificate of Merger (or, if the Certificate of Merger provide for a subsequent time for effectiveness, at the time thereafter so provided in the Certificate of Merger); the time of such effectiveness is hereinafter referred to as the "Effective Time"; and the date of such effectiveness is hereinafter referred to as the "Effective Date."
- 1.4 Certificate of Incorporation. The Certificate of Incorporation of Company in effect immediately prior to the Effective Time will be the Certificate of Incorporation of the Surviving Corporation at and after the Effective Time until duly amended in accordance with the terms thereof and the applicable provisions of the DGCL.
- 1.5 By-Laws. The By-Laws of Company in effect immediately prior to the Effective Time will be the By-Laws of the Surviving Corporation at and after the Effective Time until duly amended in accordance with the terms thereof and the applicable provisions of the DGCL; provided, however, that as of the Effective Time, the second sentence of Section 3.2 of the By-Laws of Company shall be amended to read in its entirety as follows: "The Board of Directors shall consist of two directors, until such time as the Board of Directors modifies such number by amendment to this Section 3.2."
- 1.6 Officers and Directors. The persons serving as directors of Buyer immediately prior to the Effective Time shall be the initial directors of the Surviving Corporation at and after the Effective Time, and the individuals specified by Parent in writing prior to the Effective Time shall be the initial officers of the Surviving Corporation at and after the Effective Time, in each case until their successors have been duly elected or appointed and qualified or until their earlier death, resignation or removal in accordance with the Surviving Corporation's Certificate of Incorporation and By-Laws.
- 1.7 Conversion of Shares. Except as otherwise provided herein, at the Effective Time:
- (a) Each share of Company Common Stock issued and outstanding immediately prior to the Effective Time (other than shares canceled pursuant to this Section 1.7) shall, by virtue of the Merger and without any action on the part of the holder thereof, be converted into the right to receive a number

of fully paid and nonassessable shares of Common Stock of Parent, par value \$1.00 per share ("Parent Stock") equal to the number derived by dividing \$7.85 by the Exchange Price (the "Stock Merger Consideration"); provided, however, that if between the date of this Agreement and the Effective Time the outstanding shares of Company Common Stock shall have been changed into a different number of shares or a different class, by reason of any stock dividend, subdivision, reclassification, recapitalization, split, combination, or exchange of shares, the Stock Merger Consideration to be received by the stockholders of the Company shall be appropriately and correspondingly adjusted to reflect such stock dividend, subdivision, reclassification, recapitalization, split, combination or exchange of shares.

```
Capitalization of Company and Company's Subsidiaries
Schedule 2.1
Schedule 2.2(a)
Schedule 2.4
                                                                   Rights to Acquire Company Common Stock
                                                                   Company Required Consents
Certain Events Company
Schedule 2.8
Schedule 2.9
Schedule 2.10
                                                                   Material Contracts; Customers and Suppliers
Certain Property or Assets
Company Employee Benefit Plans
Schedule 2.11(a)
                                                                  Certain Company Benefit Plan Operations
Certain Company Benefit Plan Contributions
Certain Company Benefit Plan Matters
Certain Company Benefit Plan Matters
Certain Company Benefits
Schedule 2.11(b)
Schedule 2.11(d)
Schedule
                 2.11(e)
Schedule 2.11(i)
Schedule 2.12
                                                                  Employment Agreements
Legal Proceedings
Schedule 2.13
                                                                  Compliance with Law
Accuracy of Certain Information
Company Insurance Policies
Certain Environmental Matters
Schedule 2.14
Schedule 2.14(ii)
Schedule 2.16
Schedule 2.17
                                                                   Intellectual Property
Schedule 2.19
Schedule 2.20
                                                                   Product Warranties
Schedule 2.22
                                                                   Related Party Transactions
```

(i) The Exchange Price shall be determined as follows:

A-7

Table of Contents

- (A) If the Closing Price is less than \$50.6654, the Exchange Price shall be \$50.6654.
- (B) If the Closing Price is equal to or greater than \$50.6654 and not greater than \$68.5472, the Exchange Price shall be the Closing Price.
 - (C) If the Closing Price is greater than \$68.5472, the Exchange Price shall be \$68.5472.
- (b) Each share of Series A Preferred Stock of Company (collectively, the "Series A Preferred Shares" and, collectively with the Company Common Shares, the "Company Shares") issued and outstanding immediately prior to the Effective Time shall, by virtue of the Merger and without any action on the part of the holder thereof, be converted into the right to receive the number of shares of Parent Stock derived by dividing \$9.6581, plus any dividends accrued and unpaid on such share to the Effective Time, by the average of the closing trade prices of Parent Stock for the five consecutive trading days ending on the trading day before the Effective Date, as reported on the New York Stock Exchange Composite Tape.
- (c) Each Company Common Share held immediately prior to the Effective Time by Company, Parent or Buyer or any of their

wholly-owned subsidiaries (other than shares held in trust or otherwise in a representative capacity) (the "Canceled Shares") shall be retired automatically, and no consideration shall be payable with respect thereto.

- (d) Each share of common stock of Buyer, par value \$.01 per share, issued and outstanding immediately prior to the Effective Time shall, by virtue of the Merger and without any action on the part of the holder thereof, be converted into one share of common stock, par value \$.01 per share, of the Surviving Corporation.
- 1.8 Surrender of Shares; Transfer Books.
- (a) Exchange Agent. Before the mailing of the Proxy and Registration Statement, Parent (with the consent of Company, which will not be unreasonably withheld) will appoint a bank or trust company to act as exchange agent (the "Exchange Agent") for the payment of the Merger Consideration. Parent will furnish the Exchange Agent forthwith upon the Effective Time with cash and certificates representing such number of shares of Parent Stock as the Exchange Agent shall require in order to transmit the Merger Consideration to shareholders surrendering certificates that immediately prior to the Effective Time represented Company Shares in accordance with paragraph (b) of this Section 1.8.
- (b) Exchange Procedures for Shares of Company Common Stock. As soon as practicable after the Effective Time, Parent shall cause the Exchange Agent to transmit to each holder of record of a certificate that immediately prior to the Effective Time represented Company Shares (i) a letter of transmittal (which shall specify that delivery shall be effected, and risk of loss and title to such certificates shall pass, only upon proper delivery of the certificates to the Exchange Agent and shall be in customary form) and (ii) instructions for use in effecting the surrender of such certificates in exchange for the Merger Consideration. Each holder of an outstanding certificate or certificates which immediately prior to the Effective Time represented Company Shares shall, upon surrender to the Exchange Agent of such certificate or certificates in accordance with such letter of transmittal, duly executed, and acceptance thereof by the Exchange Agent, be entitled to a certificate or certificates representing the number of full shares of Parent Stock, if any, to be received by the holder thereof pursuant to this Agreement and the cash, if any, payable in lieu of any fractional shares. The Exchange Agent will accept such certificates upon compliance with such reasonable terms and conditions as the Exchange Agent may impose to effect an orderly exchange thereof in accordance with normal exchange practices. After the Effective Time, there will be no further transfer on the records of the Surviving Corporation or its transfer agent of Company Shares which have been converted pursuant to this Agreement into the right to receive the Merger Consideration, and if certificates that immediately prior to the Effective Time represented Company Shares are presented to the Surviving Corporation for transfer, they will be canceled against delivery of certificates for Parent Stock (and cash to the extent required by **Section 1.8**(e)). If any certificate for such Parent Stock is to be issued in, or if cash is to be remitted to, a name other than that

in which the certificate that formerly represented Company Shares surrendered for exchange is registered, it will be a condition of such exchange that the certificate so surrendered will be properly endorsed, with signature guaranteed, or otherwise in proper form for transfer and that the person requesting such exchange will pay to the Surviving Corporation or its transfer agent any transfer or other taxes required by reason of the issuance of certificates for such Parent Stock in a name other than that of the registered holder of the certificate surrendered, or establish to the satisfaction of the Surviving Corporation or its transfer agent that such tax has been paid or is not applicable. Until surrendered as contemplated by this **Section 1.8**(b), each certificate that formerly represented Company Shares which have been converted into the right to receive the Merger Consideration will be deemed at any time after the Effective Time to represent only the right to receive upon such surrender the Merger Consideration as contemplated by **Section 1.7** and Section 1.8(e). No interest will be paid or will accrue on any cash payable in lieu of any fractional shares of Parent Stock.

- (c) Distributions with Respect to Unexchanged Shares. No dividends or other distributions with respect to Parent Stock with a record date after the Effective Time will be paid to the holder of any unsurrendered certificate that formerly represented Company Shares with respect to the shares of Parent Stock to be received in respect thereof and no cash payment in lieu of fractional shares will be paid to any such holder pursuant to Section 1.8(e) until the surrender of such certificate in accordance with this Article I Subject to the effect of applicable laws, following surrender of any such certificate, there will be paid to the holder of the certificate representing whole shares of Parent Stock issued in connection herewith, without interest, (i) at the time of such surrender the amount of any cash payable in lieu of a fractional share of Parent Stock to which such holder is entitled pursuant to **Section 1.8**(e) and the proportionate amount of dividends or other distributions with a record date after the Effective Time theretofore paid with respect to such whole shares of Parent Stock, and (ii) at the appropriate payment date, the proportionate amount of dividends or other distributions with a record date after the Effective Time but before such surrender and a payment date subsequent to such surrender payable with respect to such whole shares of Parent Stock.
- (d) No Further Ownership Rights in Company Common Stock.
 All Merger Consideration paid upon the surrender for exchange of certificates that formerly represented Company Shares in accordance with the terms of this Article I will be deemed to have been issued and paid in full satisfaction of all rights pertaining to the Company Shares represented by such certificates.
- (e) No Fractional Shares; Exchange Agent.
 - (ii) For purposes of calculating the Exchange Price in **Section 1.7**(a), the term **"Closing Price"** means the average of the closing trade prices of Parent Stock for the fifteen consecutive trading days ending on the fifth trading day before the date of the meeting of Company's shareholders to vote with respect to the Merger and this Agreement (the **"Company Shareholders Meeting"**), as reported on the New York Stock Exchange Composite Tape.

- (1) No Fractional Shares. No fractional shares of Parent Stock and no certificates or scrip representing fractional shares of Parent Stock will be issued in connection with the Merger, and such fractional share interests will not entitle the owner thereof to vote or to any rights of a shareholder of the Surviving Corporation after the Merger.
- (2) Cash Payment in Lieu of Fractional Shares. Each record holder of Company Shares converted pursuant to the Merger who would otherwise have been entitled to receive a fraction of a share of Parent Stock (after taking into account all Company Shares held by such holder) will be entitled to receive, in lieu thereof upon surrender of the certificates that immediately prior to the Effective Time represented Company Shares, a cash payment (without interest) in lieu of such fractional share in an amount equal to the product of such fraction multiplied by (A) the Exchange Price, in the case of the Company Common Stock, or (B) the average of the closing trade prices of Parent Stock for the five consecutive trading days ending on the trading day before the Effective Date, as reported on the New York Stock Exchange Composite Tape, in the case of the Series A Preferred Shares.
- (3) Termination of Exchange Agent's Duties. Any holders of certificates that immediately prior to the Effective Time represented Company Shares who have not complied with this Article I within six months after the Effective Time will thereafter look only to Parent for payment of the Merger Consideration.
- (4) No Liability. None of Parent, Buyer, Company or the Exchange Agent will be liable to any person in respect of any shares of Parent Stock (or dividends or distributions with respect thereto) or cash delivered to a public official pursuant to any applicable abandoned property, escheat or similar law. If any
- 1.9 Options and Warrants. (a) Prior to the Effective Time, the Board of Directors of Company and the Board of Directors of Parent (or, if appropriate, respective committees thereof) shall adopt appropriate resolutions and take such action as may be required, under the Company Option Plan (as defined in Section 2.2) and Parent's 1997 Stock Option Plan (the "Parent Option Plan"), or otherwise, such that, at the Effective Time,

certificates that immediately prior to the Effective Time represented Company Shares have not been surrendered before one year after the Effective Time or immediately before such earlier date on which any shares of Parent Stock, any cash, in lieu of fractional shares of Parent Stock, or any dividends or distributions with respect to shares of Parent Stock in respect of such certificate would otherwise escheat to or become the property of any governmental entity, any such shares, cash, dividends or distributions in respect of such certificate shall, to the extent permitted by applicable law, become the property of the Surviving Corporation, free and clear of all claims or interest of any person previously entitled thereto.

- (b) At the Effective Time, the Company Option Plan and any other employee stock option plans of Company shall be terminated automatically and no further stock awards, stock options or stock appreciation rights shall be granted thereunder subsequent to the Effective Time. Company will take all reasonable steps to insure that none of Parent, Buyer, Company or any of their respective subsidiaries is or will be bound by any Options, Warrants, or other options, warrants, rights or agreements that would entitle any person, other than Parent or its affiliates at or after the Effective Time to own or acquire any shares of any capital stock of the Surviving Corporation or any of its subsidiaries or to receive any payment in respect thereof other than as provided in this **Section 1.9**. Company will use its best efforts to obtain, prior to the Effective Time, written agreements of the holders of all Options and Warrants legally binding such holders to the foregoing.
- 1.10 Affiliates. Certificates representing Parent
 Stock issued to any person deemed by Parent to be an
 "affiliate," for purposes of Rule 145 under
 the Securities Act of 1933, as amended (the "Securities
 Act"), of Company ("Company Affiliates")
 will bear an appropriate restrictive legend. Within 90 days from
 the date of this Agreement, Company will deliver to Parent a list
 of all persons who are then Company Affiliates. Company will
 promptly amend or supplement this list as changes occur. Company
 will cause each person named in any such list, amendment or
 supplement and any other person that Parent considers to be a
 Company Affiliate to deliver promptly to Parent a letter

Table of Contents

- 1.11 Dissenting Shares.
- (a) If required under the DGCL, notwithstanding any other provision of this Agreement to the contrary, Series A Preferred Shares that are outstanding immediately prior to the Effective Time and which are held by shareholders who have voted against the Merger and who shall have demanded properly in writing appraisal for such shares in accordance with the DGCL and who shall not have withdrawn such demand or otherwise have for feited appraisal rights (collectively, the " ${f Dissenting}$ Shares") shall not be converted into or represent the right to receive the shares of Parent Stock. Such shareholders shall be entitled to receive payment of the appraised value of such Series A Preferred Shares in accordance with the provisions of the DGCL, except that all Dissenting Shares held by shareholders who shall have failed to perfect or who effectively shall have withdrawn or lost their rights to appraisal of such Series A Preferred Shares under the DGCL shall thereupon be deemed to have been converted into and to have become exchangeable, as of the Effective Time, for the right to receive, without any interest thereon, shares of Parent stock, upon surrender, in the manner provided in Section 1.7(b).
- (b) Company shall give Buyer (i) prompt notice of any demands for appraisal received by Company, withdrawals of such demands, and any other instruments served pursuant to the DGCL and received by Company and (ii) the opportunity to direct all negotiations and proceedings with respect to demands for appraisal under the DGCL. Company shall not, except with the prior written consent or Buyer, make any payment with respect to any demands for appraisal, or offer to settle, or settle, any such demands.

ARTICLE II

REPRESENTATIONS AND WARRANTIES OF COMPANY

- 2.1 Organization; Subsidiaries.
- (a) Company is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware, and, except as set forth in **Schedule 2.1**, each of its Subsidiaries is a corporation duly organized, validly existing and in good standing under the laws of its respective jurisdiction of incorporation and, except as set forth in **Schedule 2.1**, each of the foregoing is in good standing as a foreign corporation qualified to do business in each jurisdiction where the properties owned, leased or operated, or the business conducted by it require such qualification, except

for such failure to so qualify or to be in such good standing, which is not reasonably likely to have a Company Material Adverse Effect. Each of Company and its Subsidiaries has the requisite corporate power and authority to own, lease and operate its properties and to carry on its respective businesses as they are now being conducted.

- (b) Schedule 2.1 lists all Subsidiaries of Company and correctly sets forth the capitalization of each such Subsidiary, the jurisdiction in which Company and each such Subsidiary is organized or formed and each jurisdiction in which Company and each such Subsidiary is qualified or licensed to do business as a foreign corporation. As used in this Agreement, the term "Subsidiary" shall mean, with respect to any party, any corporation or other organization, whether incorporated or unincorporated or domestic or foreign to the United States of which (a) such party or any other Subsidiary of such party is a general partner (excluding such partnerships where such party or any Subsidiary of such party do not have a majority of the voting interest in such partnership) or (b) at least a majority of the securities or other interests having by their terms ordinary voting power to elect a majority of the board of directors or others performing similar functions with respect to such corporation or other organization is directly or indirectly owned or controlled by such party or by any one or more of its Subsidiaries, or by such party and one or more of its Subsidiaries. Schedule 2.1 correctly lists the current directors and executive officers of Company and of each of its subsidiaries. True, correct and complete copies of the respective charter documents and by-laws of Company and each of its Subsidiaries as in effect on the date hereof have been delivered to Buyer.
- (c) All outstanding securities or other ownership interests in each of the Subsidiaries listed on Schedule 2.1, except as set forth on Schedule 2.1, are owned of record and beneficially by Company or another

A-11

Table of Contents

of Company's wholly-owned subsidiaries, subject to no lien, claim, charge or encumbrance, and have been duly authorized and are validly issued, fully paid and nonassessable. Company does not directly or indirectly own or hold any interest in any corporation, association or business entity other than those listed on **Schedule 2.1**. There are no irrevocable proxies, voting agreements, or voting trusts with respect to any of the securities or other ownership interests of the Subsidiaries and no restrictions on Company to vote any of the stock or other securities of any of the Subsidiaries.

- 2.2 Capitalization.
- (a) The authorized capital stock of Company consists of
- (i) 25,000,000 shares of Company Common Stock; and
- (ii) 10,000,000 shares of Preferred Stock, par value \$0.01 per share (the "Preferred Stock"), of which there are designated 2,242,500 shares of Series A Preferred Stock, 10 shares of Series B Preferred Stock and the remaining shares of which have not been designated. As of February 24, 2000 (a) 13,489,569 shares of Company Common Stock were issued and outstanding, (b) 1,850,012 shares of Series A Preferred Stock and 10 shares of

Series B Preferred Stock were issued and outstanding, (c) 4,743,685 shares of Company Common Stock are reserved for issuance pursuant to Company's 1994 Stock Option Plan (the "Company Option Plan"), under which options to purchase 2,197,275 shares of Company Common Stock ("Options") have been granted, and (d) 458,741 shares of Company Common Stock are reserved for issuance under Company's 1995 Employee Stock Purchase Plan (the "ESPP"). All the outstanding shares of Company's capital stock are, and all of shares of Company Common Stock that may be issued pursuant to the exercise of outstanding Options will be, when issued in accordance with the terms thereof, duly authorized, validly issued, fully paid and non-assessable and not subject to preemptive or other similar rights. Schedule 2.2 sets forth a complete and correct list of the Options (including but not limited to Options granted under Company's Nonstatutory Stock Option Agreement and under the Option Agreement dated July 26, 1998 between Company and William H. Welling), including for each the name of the Option holder, the date of grant, the expiration date, and the number of shares subject to such Option. There are no bonds, debentures, notes or other instruments or evidences of indebtedness having the right to vote (or convertible into, or exercisable or exchangeable for, securities having the right to vote) on any matters on which the stockholders of Company or any of its Subsidiaries may vote ("Voting Debt") issued and outstanding. Schedule 2.2 sets forth a complete and correct list of all obligations of Company to issue shares pursuant to warrant agreements of Company ("Warrants"). Except as disclosed in Schedule 2.2, (i) there are no shares of capital stock of Company authorized, issued or outstanding, (ii) there are no existing options, warrants, calls, rights (including preemptive rights), subscriptions or other rights, agreements, arrangements or commitments of any character, obligating Company or any of its Subsidiaries to issue, transfer or sell or cause to be issued, transferred or sold any shares of capital stock or Voting Debt of, or other equity interests in, Company or any of its Subsidiaries or securities convertible into or exchangeable or exercisable for such shares, Voting Debt, or equity interests, or obligating Company or any of its Subsidiaries to grant, extend or enter into any such option, warrant, call, right, subscription or other right, agreement, arrangement or commitment, and (iii) there are no outstanding contractual or other obligations of Company or any of its Subsidiaries to repurchase, redeem or otherwise acquire any shares of Company Common Stock, or the capital stock of any Subsidiary or affiliate of Company or to provide funds to make any investment (in the form of a loan, capital contribution or otherwise) in any Subsidiary or any other entity. Except for the Inducement Agreement, there are not as of the date hereof and there will not be at the Effective Time any stockholder agreements, voting trusts or other agreements or understandings to which Company or any of the Subsidiaries is a party or by which any of them is bound relating to the voting of any shares of the capital stock of Company or any agreements, arrangements, or other understandings to which Company or any of its Subsidiaries is a party or by which it is bound that will limit in any way the solicitation of proxies by or on behalf of Company from or the casting of votes by, the stockholders of Company with respect to the Merger. There is no liability for any dividends or other distributions declared or accumulated but unpaid with respect to any capital stock of Company.

2.3 Authority; Validity. Company has the corporate power and authority to execute and deliver this Agreement and the Option Agreement and, subject to approval of this Agreement by the shareholders of Company in accordance with the applicable provisions of the DGCL (the "Company Shareholders' Approval"), to carry out its obligations hereunder and thereunder. The execution and delivery of this

Agreement and the consummation of the transactions contemplated hereby have been duly authorized by all necessary action on the part of Company's Board of Directors. Company's Board of Directors has directed that this Agreement and the transactions contemplated hereby be submitted to Company's shareholders for consideration at a meeting of such shareholders and, except for the Company Shareholders' Approval, no other corporate proceedings on the part of Company are necessary to authorize the execution and delivery of this Agreement, the Option Agreement and the Inducement Agreement and the consummation of the transactions contemplated hereby and thereby. Upon execution and delivery hereof (assuming that this Agreement is the legal, valid and binding obligation of Parent and Buyer), this Agreement will be the valid and binding obligation of Company enforceable against Company in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium and other similar laws and equitable principles relating to or limiting creditors rights generally.

2.4 No Conflict. Except as set forth on Schedule 2.4, neither Company nor any of its Subsidiaries nor any of its or their assets is subject to or obligated under any charter, bylaw, Contract, or other instrument or permit, or subject to any law, statute, rule, regulation, judgment, order, decree or award, which would be defaulted, breached, terminated, forfeited or violated by or in conflict with (or upon the failure to give notice or the lapse of time, or both, would result in a default, breach, termination, forfeiture or conflict with Company's execution, delivery and performance of this Agreement and the Option Agreement and the transactions contemplated hereby and thereby except where such event or occurrence is not reasonably likely to result in losses, liabilities, costs or expenses, damage or decline in value to the business, condition or properties of Company's Business (collectively, "Losses") that, individually or in the aggregate, would have a Company Material Adverse Effect. Except as set forth on Schedule 2.4 and except for compliance with the HSR Act, Company's execution, delivery and performance of this Agreement and the Option Agreement and the transactions contemplated hereby and thereby will not result in the creation of any lien, pledge, security interest or other encumbrance on the assets of Company or any of its Subsidiaries or result in any change in the rights or obligations of any party under or the acceleration of (with or without the giving of notice or the lapse of time), any provision of any Material Contract of Company or any of its Subsidiaries or any change in the rights or obligations of Company or any of its Subsidiaries under any law, statute, rule, regulation, judgment, order, decree or award or permit or license to which Company or any of its Subsidiaries is subject except where such encumbrance or change would not, individually or in the aggregate, reasonably be likely to have a Company Material Adverse Effect.

2.5 Consents. Except as set forth on Schedule 2.5 and other than (i) the filing of the Certificate of Merger as provided in Section 1.3, (ii) the filing with the SEC and the NYSE of the Proxy and Registration Statement, (iii) such consents (including the Company Shareholders' Approval), orders, approvals, authorizations, registrations, declarations and filings as may be required under the DGCL, applicable state securities laws and the securities laws of any foreign country, (iv) such filings as may be required under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "HSR Act") and (v) such local consents, orders, approvals, authorizations, registrations, declarations and filings which, if not obtained or made would not, individually or in the aggregate, reasonably be likely to have a Company Material Adverse Effect, and that would not impair, prohibit or prevent the consummation of the transactions contemplated hereby, no consent of any Person not a party to this Agreement, or consent of or filing with (including any waiting period) any domestic or

foreign court, commission, governmental body, regulatory agency, authority or tribunal (a "Governmental Entity") is required to be obtained or performed on the part of Company to permit the Merger and the other transactions contemplated hereby.

- 2.6 Financial Statements; SEC Filings.
- (a) Company has delivered to Buyer the unaudited consolidated balance sheet of Company as of December 31, 1999 and the consolidated statements of income, shareholders' equity and cash flows for the years ended December 31, 1999 and 1998. All financial statements delivered pursuant to this **Section 2.6**(a) are in accordance with the books and records of Company and have been prepared in accordance with generally accepted accounting principles consistently applied throughout the periods indicated. All consolidated balance sheets included in such financial statements present fairly in all material respects the consolidated financial position of Company as of the dates thereof. Except as and to the extent reflected or

A-13

Table of Contents

reserved against in such consolidated balance sheet (including the notes thereto), as of December 31, 1999, Company did not have any liabilities or obligations (absolute or contingent) of a nature required by generally accepted accounting principles to be reflected in a consolidated balance sheet as of such date. All consolidated statements of income present fairly in all material respects the consolidated results of operations of Company for the periods indicated. All accounts receivable appearing on such consolidated balance sheet arose from bona fide sales of products or services in the ordinary course of business consistent with past practice and are current and collectible in the amounts appearing thereon, net of any allowances for bad debts and customer returns. All inventories appearing on such consolidated balance sheet are owned by Company or its Subsidiaries free and clear of any liens or encumbrances and are of merchantable quality and in good condition and, to the extent such inventories are not of merchantable quality or in good condition, appropriate reserves have been provided therefor. Since December 31, 1999, Company has not made any changes in the accounting principles it has followed theretofore in preparing its consolidated financial statements, and all transactions have been recorded in Company's accounting records in the same manner as theretofore.

- (b) Since January 1, 1998, Company has filed with the SEC all forms, statements, reports and documents (including all exhibits, amendments and supplements thereto) required to be filed by it under the Securities Act, the Exchange Act and the respective rules and regulations thereunder (such forms, statements, reports and documents are collectively referred to as the "Company SEC Filings"). Company has delivered or made available to Buyer accurate and complete copies of all of the Company SEC Filings.
- (c) As of their respective dates, (i) each of Company's past Company SEC Filings was, and each of its future Company SEC Filings will be, prepared in compliance in all material respects with the applicable requirements of the Securities Act and the Exchange Act; and (ii) none of its

past Company SEC Filings did, and none of its future Company SEC Filings will, contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading.

- (d) Except as set forth on **Schedule 2.6**(d), neither Company nor any of its Subsidiaries is obligated to make earnout payments or other similar payments of cash or securities arising from completed acquisitions of businesses by Company and its Subsidiaries to the former owners of such businesses.
- 2.7 Tax Matters.
- (a) Company and each of its Subsidiaries has filed all Tax Returns required to be filed by it, except where failures to make such Tax Returns would not, individually or in the aggregate, have a Company Material Adverse Effect, and all such Tax Returns are complete and accurate in all material respects. Company and each of its Subsidiaries has paid (or Company has paid on its behalf) all Taxes shown as due on such Tax Returns. The most recent financial statements referred to in Section 2.6 reflect an adequate reserve for all Taxes payable by Company and its Subsidiaries for all taxable periods and portions thereof accrued through the date of such financial statements and no liabilities for Taxes have been incurred by Company or any of its Subsidiaries subsequent to such date other than in the ordinary course of its business.
- (b) No action, suit, investigation, audit, claim, assessment or adjustment is pending or proposed or threatened in writing with respect to a material amount of Taxes of Company or any of its Subsidiaries.
- (c) There is no agreement or other document extending, or having the effect of extending, the period of assessment or collection of any material amount of Taxes and no power of attorney with respect to any Taxes has been executed or filed with any taxing authority.
- (d) No material liens for Taxes exist with respect to any assets or properties of Company or any of its Subsidiaries, except for statutory liens for Taxes not yet due.
- (e) None of Company or any of its Subsidiaries is a party to or is bound by any Tax sharing agreement, Tax indemnity obligation or similar agreement, arrangement or practice with respect to a material amount of Taxes (including any advance pricing agreement, closing agreement or other agreement relating to Taxes with any taxing authority), and none of Company or any Subsidiary has been a member of any group of

A-14

on a combined or consolidated basis other than each such group of which it is currently a member.

- (f) None of Company or any of its Subsidiaries shall be required to include in a taxable period ending after the Effective Time a material amount of taxable income attributable to income that accrued in a prior taxable period but was not recognized in any prior taxable period as a result of the installment method of accounting.
- (g) All material amounts of Taxes which Company or any Subsidiary is required by law to withhold or to collect for payment have been duly withheld and collected, and have been paid or accrued.
- (h) Neither Company nor any of its Subsidiaries has filed a consent pursuant to Section 341(f) of the Internal Revenue Code of 1986, as amended, and the Treasury Regulations thereunder (the "Code") or agreed to have Section 341(f)(2) of the Code apply to any disposition of a subsection (f) asset (as such term is defined n Section 341(f)(2) of the Code) owned by Company or any of its Subsidiaries.
- (i) Neither Company nor any of its Subsidiaries is a party to any agreement, contract or arrangement that could result, separately or in the aggregate, in the payment of any "excess parachute payments" within the meaning of Section 280G of the Code or that would not be deductible pursuant to the terms of section 162(m) of the Code.
- (j) Neither Company nor any of its Subsidiaries is a "United States real property holding corporation" within the meaning of Section 897(c)(2) of the Code.
- (k) None of Company or any of its Subsidiaries has agreed, or is otherwise required, to make any adjustments pursuant to Code Section 481(a) or any similar provision of state, local or foreign law, or has any application pending with any taxing authority requesting permission for a change in accounting methods.
- 2.8 Absence of Certain Changes or Events. Except as set forth on Schedule 2.8 (or disclosed in the Company SEC Filings since December 31, 1999), since December 31, 1999, there has not been (a) any event which has resulted in, or is likely to result in, a Company Material Adverse Effect; (b) any declaration, setting aside or payment of any dividend (whether in cash, stock or property) in respect of the capital stock of Company, or any redemption or other acquisition of such stock by Company; (c) any increase in the compensation payable or to become payable by Company to its officers or key employees, except those occurring in the ordinary course of business, or any material increase in any bonus, insurance, pension or other employee benefit plan, payment or arrangement made to, for or with any such officers or key employees; or (d) any labor dispute, other than routine matters, none of which is material to Company's Business.
- 2.9 Material Contracts; Customers and Suppliers.

 Schedule 2.9 lists each of the Contracts to which
 Company or any of its Subsidiaries is a party or to which
 Company, any of its Subsidiaries or any of their respective
 properties is subject or by which any thereof is bound as of the
 date hereof (i) that would be required by
 Item 601(b)(10) of SEC Regulation S-K to be filed as an
 exhibit to an Annual Report on Form 10-K, (ii) that
 relates to MIS (data processing) equipment or other capital

equipment in the amount of \$50,000 or more, (iii) that is in the amount of \$50,000 or more and contains any provision prohibiting or limiting assignment thereof (including by merger or otherwise by operation of law) allowing another party to terminate or change any term or provision thereof in the event of any change in control or merger of Company or any of its subsidiaries, or (iv) pursuant to which Company or any of its Subsidiaries has the right to borrow money or obtain any financial accommodation (each a "Material Contract") except for those Contracts listed as exhibits to Company's Annual Report on Form 10-K for the fiscal year ending December 31, 1998 or disclosed in Company SEC Filings since December 31, 1998. Except as set forth on Schedule 2.9, no breach or default, alleged breach or default or event which would (with the passage of time, notice or both) constitute a breach or default thereunder by Company or any of its Subsidiaries, as the case may be, or, to Company's knowledge, any other party or obligor with respect thereto, has occurred, except where such breaches or defaults, individually or in the aggregate, would not be reasonably likely to have a Company Material Adverse Effect. To Company's knowledge, no party to any Material Contract intends to cancel, withdraw, modify or amend such Material Contract. **Schedule 2.9** also lists any customers to whom Company sold products in an amount in excess of \$2,300,000 during the fiscal year ended December 31, 1999 ("Principal Customers").

A-15

Table of Contents

Schedule 2.9 also lists any suppliers for whom Company sold products in an amount in excess of \$5,000,000 during the fiscal year ended December 31, 1999 ("Principal Suppliers"). Except as set forth on Schedule 2.9(ii), no Principal Customers or Principal Suppliers has advised Company that it intends to terminate its relationship with Company as a result of the consummation of the transaction contemplated by this Agreement. Except as set forth in Schedule 2.9, neither Company nor any of its affiliates is a party to, or in negotiations for the purpose of entering into, any agreement to acquire a majority of the voting securities of, or substantially all of the assets of, any Person or any business division of any Person.

2.10 Title and Related Matters.

Schedule 2.10 sets forth a list of parcels of real property owned or leased by Company or any of its Subsidiaries, including all sales and distribution facilities and all corporate support and distribution centers. Except as set forth in Schedule 2.10, Company has good and indefeasible title to all of its properties, interests in properties and assets, real and personal, reflected in Company's December 31, 1999 consolidated balance sheet or acquired after December 31, 1999 or necessary to conduct its business as now conducted (except properties, interests on properties and assets sold or otherwise disposed of since December 31, 1999 in the ordinary course of business), free and clear of all mortgages, liens, pledges, charges or encumbrances of any kind or character, except (a) the lien of current taxes not yet due and payable, (b) liens for mechanics', carriers', workmens', repairmens', landlord, statutory or common law liens either not delinquent or being contested in good faith, (c) such imperfections of title, liens, restrictions, variances and easements as do not materially detract from the value of or interfere with the value or the present or presently contemplated future use of the properties subject thereto or affected thereby, or otherwise materially impair the present or presently contemplated future business operations at such properties and (d) liens securing debt which is reflected on

Company's December 31, 1999 consolidated balance sheet. The plants and equipment of Company that are necessary to the operation of Company's Business are in good operating condition and repair (subject to normal wear and tear). All properties material to the operations of Company are reflected in Company's December 31, 1999 consolidated balance sheet to the extent generally accepted accounting principles require the same to be reflected.

2.11 Employee Benefit Plans.

- (a) Schedule 2.11(a) lists all employee benefit and compensation plans, programs, policies, agreements and commitments (other than oral employment agreements that are terminable at will by the employer without additional cost) covering any employee or former employee, or the beneficiary of either, of Company or any entity which would be aggregated at any relevant time with Company pursuant to Section 414(b), (c), (m), or (o) of the IRC (referred to herein as a "Company ERISA Affiliate"), providing benefits in the nature of pension, profit sharing, deferred compensation, retirement, severance, bonus, incentive compensation, stock option, stock bonus, stock purchase, health, medical, life, disability, sick leave, vacation, or other welfare or fringe benefits, including, without limitation, all employee benefit plans (as defined in Section 3(3) of ERISA (referred to herein as the "Company ERISA Plans"), and fringe benefit plans (as defined in IRC Section 6039D) (together with Company ERISA Plans referred to herein as the "Company Benefit Plans"). Except as set forth on Schedule 2.11(a), neither Company nor any Company ERISA Affiliate has ever contributed to, or been obligated to contribute to, (i) a multiemployer plan (as defined in Section 3(37) of ERISA) or (ii) a Company Benefit Plan subject to Title IV of ERISA.
- (b) Except as set forth on Schedule 2.11(b), each Company Benefit Plan is currently, and has been in the past, operated and administered in all material respects in compliance with its terms and with the requirements of all applicable Laws, including, without limitation, ERISA and the IRC. Each Company Benefit Plan which is, or was, intended to qualify under IRC $\,$ Section 401(a) (referred to herein as a "Qualified Plan") is, or was, so qualified and either (i) has been determined by the IRS to be so qualified by the issuance of a favorable determination letter a copy of which has been furnished to Parent, which remains in effect as of the date hereof and, to Company's knowledge, nothing has occurred since the date of such letter to cause such letter to be no longer valid, or (ii) is within the "remedial amendment period" as defined in IRC Section 401(b) and the regulations thereunder. Except as set forth on Schedule 2.11(b), all reports, notices and other documents required to be filed with any Governmental Entity or furnished to employees or participants with respect to the Company Benefit Plans have been timely filed or furnished. With respect to the most recent Form 5500 regarding each funded Company Benefit Plan, benefit liabilities do not exceed

A-16

- (c) No liability under Subtitle C or D of Title IV of ERISA has been or is expected to be incurred by Company or any of its subsidiaries with respect to any ongoing, frozen or terminated "single-employer plan", within the meaning of Section 4001(a)(15) of ERISA, currently or previously maintained by any of them, or the single-employer plan of any Company ERISA Affiliate. Company and its subsidiaries have not incurred and do not expect to incur any withdrawal liability with respect to a multiemployer plan under Subtitle E of Title IV of ERISA (regardless of whether based on contributions of an ERISA Affiliate). No notice of a "reportable event" within the meaning of Section 4043 of ERISA for which the 30-day reporting requirement has not been waived or extended, other than pursuant to PBGC Reg. Section 4043.66, has been required to be filed for any Company ERISA Plan or by any Company ERISA Affiliate within the 12-month period ending on the date hereof.
- (d) Except as set forth on Schedule 2.11(d), all contributions required to be paid on or prior to the date hereof to or with respect to any Company Benefit Plan by its terms or applicable law have been paid in full and proper form. Neither any Company ERISA Plan nor any single-employer plan of a Company ERISA Affiliate has an "accumulated funding deficiency" (whether or not waived) within the meaning of IRC Section 412 or Section 302 of ERISA and no Company ERISA Affiliate has an outstanding funding waiver. Neither Company nor any of its subsidiaries has provided, or is required to provide, security to any Company ERISA Plan or to any single-employer plan of a Company ERISA Affiliate pursuant to IRC Section 401(a)(29).
- (e) Except as set forth on **Schedule 2.11**(e), the transactions contemplated by this Agreement (i) do not constitute or result in a severance or termination of employment under any Company Benefit Plan for which severance or termination benefits may be payable with respect to any employee covered thereby, (ii) accelerate the time of payment or vesting or increase the amount of benefits due under any Company Benefit Plan or compensation to any employee of Company, or (iii) result in any payments (including parachute payments) or funding (through grants or otherwise) of compensation or benefits under any Company Benefit Plan or Law or result in any obligation or liability of Company to any employee of Company or any Company ERISA Affiliate or (iv) result in payments under any Company Benefit Plan which would not be deductible under IRC Section 162(m) or IRC Section 280G.
- (f) Neither Company nor any Company ERISA Affiliate, nor to Company's knowledge any other "disqualified person" or "party in interest" (as defined in IRC Section 4975 and Section 3(14) of ERISA, respectively) with respect to any Company ERISA Plan has engaged in any transaction in violation of Section 406 of ERISA for which no class, individual or statutory exemption exists or any "prohibited transaction" (as defined in IRC Section 4975(c)(1)) for which no class, individual or statutory exemption exists under IRC Section 4975(c)(2) or (d), nor has any such person who is a "fiduciary" (as defined in Section 3(21) of ERISA) of any Company ERISA Plan breached or participated in the breach of any fiduciary obligation imposed pursuant to Part 4 of Title I of ERISA.
- (g) There are no actions, suits, disputes or claims pending or, to Company's knowledge, threatened (other than routine claims for benefits) or legal, administrative or other proceedings or governmental investigations pending or, to Company's knowledge, threatened, against or with respect to any Company Benefit Plan or the assets of any Company Benefit Plan.

(h) Under each Company ERISA Plan which is a single-employer plan, as of the last day of the most recent plan year ended prior to the date hereof, the actuarially determined present value of all "benefit liabilities", within the meaning of Section 4001(a)(16) of ERISA (as determined on the basis of the actuarial assumptions contained in the Plan's most recent actuarial valuation), did not exceed the then current value of the assets of such Plan, and there has been no material change in the financial condition of such Plan since the last day of the most recent plan year. The withdrawal liability of Company and its Subsidiaries under each Company ERISA Plan which is a multiemployer plan to which Company, any of its subsidiaries or any ERISA Affiliate has contributed during the preceding 12 months, determined as if a "complete withdrawal," within the meaning of
Section 4203 of ERISA, had occurred as of the date hereof, does not exceed \$100,000.

A-17

Table of Contents

- (i) Except as set forth on **Schedule 2.11**(i), no Company Benefit Plan provides or provided health or medical benefits (whether or not insured) with respect to current or former employees of Company or its subsidiaries beyond their retirement or other termination of service (other than coverage mandated by statutory law). Company or its subsidiaries may amend or terminate any such Company Benefit Plan at any time without incurring any liability themselves.
- (j) Each Company ERISA Plan that is an "employee welfare benefit plan" as that term is defined in Section 3(1) of ERISA is either (as identified on Schedule 2.11(a)): (i) funded through an insurance company contract, (ii) funded throughout a tax-exempt "VEBA" trust or (iii) unfunded. There is no liability in the nature of a retroactive rate adjustment or loss-sharing or similar arrangement, with respect to any Company ERISA Plan which is an employee welfare benefit plan.
- (k) Company has provided to Parent true and complete copies of the following with respect to each of the Company Benefit Plans: (i) each plan document and summary plan description if any; (ii) each trust agreement, insurance policy or other instrument relating to the funding thereof; (iii) the most recent Annual Report (Form 5500 series) and associated schedules filed with the IRS or the United States Department of Labor for each Company Benefit Plan required to make such filing; (iv) the most recent audited financial statement report, if any; (v) the most recent actuarial report if any; and (vi) a description of each unwritten Company Benefit Plan and the individuals covered thereby.
- (1) All Company Benefit Plans maintained outside of the United States comply in all material respects with applicable local law. Company and its subsidiaries have no material unfunded liabilities with respect to any such Company Benefit Plan.
- (m) The Board of Directors of Company has adopted resolutions terminating the ESPP in accordance with Section 18 thereof, effective as of the date hereof, or as soon hereafter as practicable and legally permissible, and

providing that no shares of Company Common Stock shall be issued for the current participation period thereunder and that the amount in each participant's plan account thereunder shall be refunded to such participant in cash pursuant to Section 9(d) thereof.

- (n) The Board of Directors of Company (through its Options Committee or other body duly exercising the powers thereof) has adopted resolutions approving the conversion of the Options in the manner set forth in Section 1.9(a) and declaring that such conversion constitutes an assumption of the Options in accordance with Section 8(b) of the Company Option Plan.
- 2.12 Employment Agreements. Except as set forth on Schedule 2.12, none of Company or any of its Subsidiaries is a party to any employment, consulting, non-competition, severance, or indemnification agreement with any current or former executive officer or director or employee of Company or any of its Subsidiaries. True and complete copies of the agreements set forth on Schedule 2.12 have been furnished to Parent prior to the date hereof. Neither Company nor any of its Subsidiaries is a party to any collective bargaining agreement.
- 2.13 Legal Proceedings. Schedule 2.13
 sets forth a list of all legal proceedings pending to which
 Company or any of its Subsidiaries is a party, except
 (i) all workers' compensation claims which are fully
 insured, and (ii) receivables collections for less than
 \$50,000 where Company is the plaintiff and related countersuits.
 There is no pending or, to Company's knowledge, threatened
 judicial or administrative proceeding or investigation affecting
 Company or any of its Subsidiaries that if resolved adversely to
 it would reasonably be likely to result in a Company Material
 Adverse Effect or could reasonably be expected to impair
 Company's ability to consummate the Merger.
- 2.14 Compliance with Law; Accuracy of Certain
 Information. Except as set forth on Schedule 2.14
 , all licenses, franchises, permits and other governmental
 authorizations (collectively, "Licenses") held
 by Company and its Subsidiaries that are material in connection
 with Company's Business are valid and sufficient for all
 business presently carried on by Company. No suspension,
 cancellation or termination of any such material Licenses is
 threatened or imminent. Except as set forth on
 Schedule 2.14, Company's Business is being
 conducted in compliance with all applicable Laws and is not being
 conducted in violation of any Law, except for violations which
 either individually or in the aggregate are not reasonably likely
 to result in a

A-18

Table of Contents

Company Material Adverse Effect. Any and all past litigation concerning Licenses and all claims and causes of action raised therein have been finally adjudicated. No such License has been revoked, conditioned (except as may be customary) or restricted and no action (equitable, legal or administrative), arbitration or other process is pending, or to the Company's knowledge, threatened in writing which in any way challenges the validity of, or seeks to revoke, condition or restrict any such License.

2.15 Accuracy of Proxy and Registration Statement and Other Information. On the date on which Company mails to its shareholders the Proxy and Registration Statement, on the date the Company Shareholders Meeting is held, and on the Effective Date, the Proxy and Registration Statement will contain all material statements concerning Company which are required to be set forth therein in accordance with the Exchange Act; and at such respective times, the Proxy and Registration Statement will not include any untrue statement of a material fact or omit to state any material fact with respect to Company required to be stated therein or necessary to make the statements therein not misleading. Notwithstanding the foregoing, Company makes no representation or warranty with respect to any information concerning Buyer or any of Buyer's Subsidiaries or advisors included or incorporated by reference in the Proxy and Registration Statement.

2.16 Insurance. Schedule 2.16 lists all of Company's insurance policies. All properties of Company and its subsidiaries are insured for their respective benefits, in amounts customary and reasonable for the line of business of Company and against all risks usually insured against by Persons operating similar properties in the localities where such properties are located under valid and enforceable policies issued by insurers of recognized responsibility, except where failure to so insure would not be reasonably likely to have a Company Material Adverse Effect. Neither Company nor any of its subsidiaries is in default under any such policy or bond, and all such policies are in full force and effect, with all premiums due thereon paid. Company and its subsidiaries have timely filed claims with their respective insurers with respect to all matters and occurrences for which they believe they have coverage except for such failures as are not reasonably likely to result in a Company Material Adverse Effect.

2.17 Environmental Matters.

- (a) Each of the Company and each of its Subsidiaries is, and at all times has been, in compliance with all local, state and federal statutes, orders, rules, ordinances, regulations, codes and policies and all judicial or administrative interpretations thereof (collectively, "Environmental Laws") relating to pollution or protection of the environment, including, without limitation, laws relating to exposures, emissions, discharges, releases or threatened releases of Hazardous Substances (as hereinafter defined) into or on land, ambient air, surface water, groundwater, personal property or structures (including the protection, cleanup, removal, remediation or damage thereof), or otherwise related to the manufacture, processing, distribution, use, treatment, storage, disposal, transport, discharge or handling of Hazardous Substances. Neither the Company nor any Subsidiary has received any notice of any investigation, claim or proceeding against the Company relating to Hazardous Substances or any action pursuant to or violation or alleged violation under any Environmental Law, and the Company is not aware of any fact or circumstance that could involve the Company or any Subsidiary in any environmental litigation, proceeding, investigation or claim or impose any environmental liability upon the Company or any Subsidiary that would reasonably be expected to have a Company Material Adverse Fffect.
- (b) To the knowledge of the Company, there are no Hazardous Substances in, under or about the soil, sediment, surface water or groundwater on, under or around any properties at any time owned, leased or occupied by the Company or any Subsidiary. Neither the Company nor any Subsidiary has disposed of any Hazardous Substances on or about such properties. There is no present release or threatened release of any Hazardous Substances in, on, under or around such properties. Neither the Company nor any Subsidiary has disposed of any materials at any site being investigated or remediated for contamination or possible

contamination of the environment.

2.18 Certain Agreements. Except as set forth in Schedule 2.18, neither the Company nor any of its Subsidiaries is a party to any Benefit Plan, any of the benefits of which will be increased, or the vesting of the benefits of which will be accelerated, by the occurrence of any of the transactions contemplated by this Agreement or the value of any of the benefits of which will be calculated on the basis of any of the transactions

A-19

Table of Contents

contemplated by this Agreement, where the liability that would reasonably be expected to result would have a Company Material Adverse Effect.

2.19 Intellectual Property.

- (a) The Company and each of its Subsidiaries owns, or is licensed or otherwise possesses legally sufficient rights to use, all patents, trademarks, trade names, service marks, copyrights, and any applications therefor, technology, know-how, computer software programs or applications (in both source code and object code form) and tangible or intangible proprietary information or material that are used or proposed to be used in the business of the Company or such Subsidiary, as currently conducted in any material respect. Schedule 2.19 lists all current patents, registered and material unregistered copyrights, trade names and service marks, any domain name, and any applications therefor owned by the Company (the "Company Intellectual Property Rights"), and specifies the jurisdictions in which each such Company Intellectual Property Right has been issued or registered or in which an application for such issuance and registration has been filed, including the respective registration or application numbers and the names of all registered owners. Schedule 2.19 includes and specifically identifies all third-party patents, trademarks, and copyrights (including software) (the "Third Party Intellectual Property Rights") which are incorporated in, are, or form a part of, any product of the Company or any of its Subsidiaries. Schedule 2.19 lists (i) any requests the Company or any Subsidiary has received to make any registration of the type referred to in the penultimate sentence prior hereto, including the identity of the requester and the item requested to be so registered, and the jurisdiction for which such request has been made; (ii) all material licenses, sublicenses and other agreements as to which the Company or any of its Subsidiaries is a party and pursuant to which any person is authorized to use any Company Intellectual Property Right, or any trade secret material to the Company; and (iii) all material licenses, sublicenses and other agreements as to which the Company or any of its Subsidiaries is a party and pursuant to which the Company or any of its Subsidiaries is authorized to use any Third Party Intellectual Property Rights, or other trade secret of a third party in or as any product, and includes the identity of all parties thereto, a description of the nature and subject matter thereof, the applicable royalty and the term thereof.
- (b) Neither the Company nor any of its Subsidiaries is, nor will it be as a result of the execution and delivery of this Agreement or the performance of its obligations hereunder, in

violation of any license, sublicense or agreement described in Schedule 2.19. Except as set forth in Schedule 2.19, no claims with respect to the Company Intellectual Property Rights, any trade secret material to the Company or Third Party Intellectual Property Rights to the extent arising out of any use, reproduction or distribution of such Third Party Intellectual Property Rights by or through the Company or any of its Subsidiaries, are currently pending or, to the knowledge of the Company, are threatened by any person, (i) to the effect that the manufacture, sale, licensing or use of any product as now used, sold or licensed or proposed for use, sale or license by the Company or any of its Subsidiaries infringes on any patent, trademark, trade name, service mark, copyright, or trade secret; (ii) against the use by the Company or its Subsidiaries of any patent, trademarks, trade names, copyrights, trade secrets, technology, know-how or computer software programs and applications used in the business of the Company or any of its Subsidiaries as currently conducted or as proposed to be conducted; (iii) challenging the ownership, validity or effectiveness of any of the Company Intellectual Property Rights or other trade secret material to the Company and its Subsidiaries, or (iv) challenging the Company's or any of its Subsidiaries' license or legally enforceable right to use of the Third Party Intellectual Property Rights. To the Company's knowledge, after reasonable investigation, all patents, registered trademarks, trade names, service marks, and copyrights held by the Company or any of its Subsidiaries are valid and subsisting. To the Company's knowledge, there is no material unauthorized use, infringement or misappropriation of any of the Company Intellectual Property Rights by any third party, including any employee or former employee of the Company or any of its Subsidiaries. Neither the Company nor any of its Subsidiaries (A) has been sued or charged in writing as a defendant in any claim, suit, action or proceeding such involves a claim or infringement of trade secrets, any patents, trademarks, service marks, or copyrights and which has not been finally terminated prior the date hereof or been informed or notified by any third party that the Company may be engaged in such infringement or (B) has knowledge of any infringement liability with respect to, or

A-20

Table of Contents

infringement by, the Company or any of its Subsidiaries of any trade secret, patent, trademark, service mark, or copyright of another.

- (c) Neither the execution and delivery of this Agreement by the Company nor the consummation by the Company of the transactions contemplated by this Agreement will result in any breach of or constitute a default (or an event that with notice or lapse of time or both would become a default), or impair the Company's or any of its Subsidiaries' rights or alter the rights or obligations of any third party under, or give to others any rights of termination, amendment, acceleration or cancellation of, any license agreement, contract or other arrangement of any nature relating to Company Intellectual Property Rights or Third Party Intellectual Property Rights.
- (d) **Schedule 2.19** contains a complete and accurate list of any proceedings before any patent or trademark authority to which the Company or a Subsidiary is a party, a description of the subject matter of each proceeding, and the current status of each proceeding, including, without limitation, interferences, priority contests, opposition, and protests. Such list includes any pending applications for reissue or reexamination of a patent. The Company or a Subsidiary has the

- exclusive right to file, prosecute and maintain any such applications for patents, copyrights or trademarks and the patents and registrations that issue therefrom.
- (e) Each employee of the Company and its Subsidiaries has executed a confidentiality, invention and copyright agreement with the Company in the forms previously delivered to Parent.
- 2.20 Product Warranties. Except as set forth in Schedule 2.20, all products are sold or licensed by the Company and its Subsidiaries pursuant to terms that provide (a) the Company's disclaimer of all warranties, express or implied, including those of merchantability and fitness for a particular purpose; (b) the Company's disclaimer of all consequential damages arising from the use or possession of the product, regardless of whether such liability is based in tort, contract or otherwise; and (c) language stating that if the foregoing disclaimers are held to be unenforceable, the Company's maximum liability shall not exceed the amount of money(ies) paid for such product(s).
- 2.21 Labor Matters. There are no controversies pending or, to the knowledge of the Company, threatened, between the Company or any of its Subsidiaries and any group of their respective employees. Neither the Company nor any of its Subsidiaries is a party to any collective bargaining agreement or other labor union contract applicable to persons employed by the Company or its Subsidiaries nor does the Company know of any activities or proceedings of any labor union to organize any such employees; and neither the Company or any of its Subsidiaries has breached or otherwise failed to comply with any provision of any such agreement or contract and there are no grievances outstanding against any such parties under any such agreement or contract. There are no unfair labor practice complaints pending against the Company or any of its Subsidiaries before the National Labor Relations Board or any current union representation questions involving employees of the Company or any of its Subsidiaries. The Company has no knowledge of any strikes, slowdowns, work stoppages, lockouts, or threats thereof, by or with respect to any employees of the Company or any of its Subsidiaries. The Company has no knowledge of any strikes, slowdowns, work stoppages, lockouts, or threats thereof, by or with respect to any employees of the Company or any of its Subsidiaries. No consent of any union which is a party to any collective bargaining agreement with the Company or any of its Subsidiaries is required to consummate the transactions contemplated by this Agreement.
- 2.22 Related Party Transactions. Except as set forth in the Company SEC Filings or Schedule 2.22, no director, officer, or "affiliate" (as such term is defined in Rule 12b-2 under the Exchange Act) of the Company (a) has outstanding any indebtedness or other similar obligations of the Company or any of its Subsidiaries, other than ordinary course of business travel advances or (b) other than employment related benefits agreements contemplated by or disclosed in this Agreement, is a party to any legally binding contract, commitment or obligation to, from or with the Company or any of its Subsidiaries.
- 2.23 State Takeover Statutes. The action of the Board of Directors of the Company in approving the Merger, this Agreement, the Option Agreement and the Inducement Agreement and the transactions

contemplated hereby and thereby is sufficient to render inapplicable to the Merger, this Agreement, the Option Agreement and the Inducement Agreement the provisions of Section 203 of the DGCL.

- 2.24 Brokers; Advisors. No broker, investment banker, financial advisor or other person, other than Alliant Partners and Tucker, Anthony, Cleary, Gull, the fees and expenses of which will be paid by the Company (and are reflected in agreements between Alliant Partners and Tucker, Anthony, Cleary, Gull, respectively, and the Company, complete copies of which have been furnished to Parent), is entitled to any broker's, finder's, financial advisor's or other similar fee or commission in connection with the transactions contemplated by this Agreement based upon arrangements made by or on behalf of the Company. A true and complete estimate of the fees and costs of all financial and accounting advisors and legal counsel retained by Company in connection with the negotiation and consummation of the Merger is set forth in Schedule 2.24.
- 2.25 Full Disclosure. No representation or warranty by the Company herein (including the Schedules and Exhibits hereto) or in any certificate furnished by or on behalf of the Company to Parent in connection herewith, taken together with all the other information provided to Parent or its counsel in connection with the transactions contemplated hereby, contains or will contain any untrue statement of a material fact or omits or will omit to state a material fact necessary in order to make the statements herein or therein, in the light of the circumstances under which they were made, not misleading.

ARTICLE III

REPRESENTATIONS AND WARRANTIES OF PARENT AND BUYER

Parent and Buyer hereby represent and warrant to Company as follows:

- 3.1 Organization. Parent is a corporation duly organized, validly existing and in good standing under the laws of the State of New York, and has corporate power to carry on its business as it is now being conducted. Buyer is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware, and has corporate power to carry on its business as it is now being conducted.
- 3.2 Authority; Validity. Parent and Buyer have all corporate power and authority to execute and deliver this Agreement and to carry out their respective obligations hereunder. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly authorized by all necessary action on the part of the Board of Directors of Parent and Buyer, and no other corporate proceedings on the part of Parent or Buyer are necessary to authorize this Agreement and the transactions contemplated hereby. Upon execution and delivery hereof (assuming that this Agreement is the legal, valid and binding obligation of Company), this Agreement will be the valid and binding obligation of Parent and Buyer enforceable against Parent and Buyer in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium and other similar laws relating to or limiting creditors rights generally and equitable principles.

- 3.3 No Conflict. Neither Parent, Buyer nor any of their respective assets is subject to or obligated under any charter, bylaw, Contract, or other instrument or any permit, or subject to any statute, rule, order or decree, which would be defaulted, breached, terminated, forfeited or violated by or in conflict (or upon the failure to give notice or the lapse of time, or both, would result in a default, breach, termination, forfeiture or conflict) with its this Agreement and the transactions contemplated hereby except where such event or occurrence (i) as of the date hereof is not reasonably likely to result in Losses that, individually or in the aggregate, would reasonably be likely to have a Parent Material Adverse Effect; or (ii) between the date hereof and the Closing Date would not, individually or in the aggregate, reasonably be likely to have a Parent Material Adverse Effect.
- 3.4 Consents. Except as contemplated by this Agreement, no consent of any Person not a party to this Agreement, nor consent of or filing with (including any waiting period) any Governmental Entity, is required to be obtained or performed on the part of Parent and Buyer to permit the Merger.
- 3.5 Legal Proceedings. Except as otherwise disclosed by Parent to Company there is no pending or, to Parent's knowledge, threatened judicial or administrative proceeding or investigation affecting Parent that if

A-22

Table of Contents

resolved adversely to Parent would reasonably be likely to result in a Parent Material Adverse Effect or could reasonably be expected to impair its ability to consummate the Merger.

- 3.6 Financial Statements, SEC Filings.
- (a) Parent has delivered copies of the following financial statements to Company: (i) the consolidated balance sheet of Parent at July 2, 1999 and the consolidated statements of income, shareholders' equity and cash flows for the years ended July 2, 1999 and June 26, 1998, in each case including the notes thereto and the related report of Arthur Andersen LLP, independent certified public accountants, and (ii) the unaudited consolidated balance sheet of Parent at December 31, 1999 and the unaudited consolidated statements of income, shareholders' equity and cash flows for the six-month period ended December 31, 1999 in each case including any notes thereto.
- (b) All financial statements delivered pursuant to Section 3.6(a) hereof are in accordance with the books and records of Parent and have been prepared in accordance with generally accepted accounting principles consistently applied throughout the periods indicated (except as may be indicated in the notes thereto or, in the case of the unaudited statements, as permitted by Form 10-Q of the SEC). All consolidated balance sheets included on such financial statements present fairly in all material respects the consolidated financial position of Parent as of the dates thereof (subject, in the case of the unaudited statements, to customary reclassification year-end adjustments). Except as and to the

extent reflected or reserved against in such consolidated balance sheets (including the notes thereto) as of December 31, 1999, Parent did not have any liabilities or obligations (absolute or contingent) of a nature required by generally accepted accounting principles to be reflected in a consolidated balance sheet as of such date. All consolidated statements of income included on such financial statements present fairly in all material respects the consolidated results of operations of Parent for the periods indicated.

- (c) Since January 1, 1998, Parent has filed with the SEC all material forms, statements, reports and documents (including all exhibits, amendments and supplements thereto) required to be filed by it under the Securities Act, the Exchange Act and the respective rules and regulations thereunder (such forms, statements, reports and documents are collectively referred to as the "Parent SEC Filings"). Parent has delivered or made available to Company accurate and complete copies of all of the Parent SEC Filings.
- (d) As of their respective dates, (i) each of Parent's past Parent SEC Filings was, and each of its future Parent SEC Filings will be, prepared in compliance in all material respects with the applicable requirements of the Securities Act and the Exchange Act; and (ii) none of its past Parent SEC Filings did, and none of its future Parent SEC Filings will, contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading.
- 3.7 No Material Adverse Effect. Since December 31, 1999 there has not been any event which has had or is likely to have a Parent Material Adverse Effect.
- 3.8 Compliance with Law. All licenses, franchises, permits and other governmental authorizations held by Parent that are material in connection with Parent's Business are valid and sufficient for all business presently carried on by Parent except where the failure to maintain such a valid and sufficient permit would not reasonably be likely to result in a Parent Material Adverse Effect. No suspension, cancellation or termination of any such material, licenses, franchises, permits and other governmental authorizations is threatened or imminent. Parent's Business is not being conducted in violation of any Law, except for violations which either individually or in the aggregate are not reasonably likely to result in a Parent Material Adverse Effect.
- 3.9 Accuracy of Proxy and Registration Statement and Other Information. On the date on which Company mails to its shareholders the Proxy and Registration Statement, on the date the Company Shareholders Meeting is held, and on the Effective Date, the Proxy and Registration Statement will contain all material statements concerning Parent and Buyer which are required to be set forth therein in accordance with the Securities Act and the Exchange Act; and at such respective times, the Proxy and Registration Statement will not include any untrue statement of a material fact or omit to state any material fact required to be stated

therein or necessary to make the statements therein not misleading. Notwithstanding the foregoing, Parent and Buyer make no representation or warranty with respect to any information concerning Company or any of Company's Subsidiaries or advisors included or incorporated by reference in the Proxy and Registration Statement.

- 3.10 Full Disclosure. No representation or warranty by Parent or Buyer herein (including the Schedules and Exhibits hereto) or in any certificate furnished by or on behalf of Parent or Buyer to Company in connection herewith, taken together with all the other information provided to Company or its counsel in connection with the transactions contemplated hereby and all the information included in the Parent SEC Filings, contains or will contain any untrue statement of a material fact or omits or will omit to state a material fact necessary in order to make the statements herein or therein, in the light of the circumstances under which they were made, not misleading.
- 3.11 No Brokers or Finders. Neither Buyer nor any of its officers, directors or employees has employed any broker or finder or incurred any liability for any financial advisory, brokerage or finder's fees or commissions in connection with the transactions contemplated herein, except that Parent has retained Merrill Lynch & Co. as its financial advisor, whose fees and expenses will be paid by Parent.

ARTICLE IV

COVENANTS

4.1 Access and Information.

- (a) Subject to applicable laws and regulations, upon reasonable notice during the period from the date hereof through the Effective Time, Company will give to Parent and Buyer and Parent and Buyer's Representatives full access during normal business hours to all of its and its subsidiaries' properties, books, records, documents (including, without limitation, Tax Returns for all periods open under the applicable statute of limitations), personnel, auditors and counsel, and each party shall (and shall cause its subsidiaries to) furnish promptly to the other party all information concerning such party and its subsidiaries as such other party or such other party's Representatives may reasonably request. Subject to applicable laws and regulations, upon reasonable notice during the period from the date hereof through the Effective Time, Parent and Buyer will provide to Company and Company's Representatives such information as Company may reasonably request to determine the accuracy of Parent's and Buyer's representations and warranties in this Agreement and compliance by Parent and Buyer with their covenants in this Agreement.
- (b) All non-public information disclosed by any party (or its Representatives) whether before or after the date hereof, in connection with the transactions contemplated by, or the discussions and negotiations preceding, this Agreement to any other party (or its Representatives) shall be kept confidential by such other party and its Representatives and shall not be used by any such Persons other than as contemplated by this Agreement. Subject to the requirements of applicable Law, Parent, Buyer and Company will keep confidential, and each will cause their respective Representatives to keep confidential, all such non-public information and documents unless such information (i) was already known to Parent, Buyer or Company, as the

case may be, as long as such information was not obtained in violation of a confidentiality obligation (ii) becomes available to Parent, Buyer or Company, as the case may be, from other sources not known by Parent, Buyer or Company, respectively, to be bound by a confidentiality obligation, (iii) is independently acquired by Parent, Buyer or Company, as the case may be, as a result of work carried out by any Representative of Parent, Buyer or Company, respectively, to whom no disclosure of such information has been made, (iv) is disclosed with the prior written approval of Company or Parent, Buyer, as the case may be, or (v) is or becomes readily ascertainable from publicly available information. Upon any termination of this Agreement, each party hereto will collect and deliver to the other, or certify as to the destruction of, all documents obtained by it or any of its Representatives then in their possession and any copies thereof.

(c) Subject to applicable Law, if between the date hereof and the Effective Date any Governmental Entity shall commence any examination, review, investigation, action, suit or proceeding against any party hereto with respect to the Merger, such party shall (i) give the other parties prompt notice thereof, (ii) keep

A-24

Table of Contents

the other parties informed as to the status thereof and (iii) permit the other parties to observe and be present at each meeting, conference or other proceeding and have access to and be consulted in connection with any document filed or provided to such Governmental Entity in connection with such examination, review, investigation, action, suit or proceeding.

- 4.2 Governmental Filings. Subject to the terms and conditions herein provided, the parties hereto shall: (a) promptly make their respective filings and thereafter make any other required submissions under the HSR Act with respect to the Merger; (b) use all reasonable efforts to cooperate with one another in (i) determining which filings are required to be made prior to the Effective Time with, and which consents, approvals, permits or authorizations are required to be obtained prior to the Effective Time from, any Governmental Entity in connection with the execution and delivery of this Agreement and the consummation of the transactions contemplated hereby (ii) timely making all such filings and timely seeking all such consents, approvals, permits or authorizations; and (c) using all reasonable efforts to take, or cause to be taken, all other action and doing, or cause to be done, all other things necessary, proper or appropriate to consummate and make effective the transactions contemplated by this Agreement.
- 4.3 Consents and Approvals.
- (a) Company shall use its commercially reasonable efforts to obtain any and all consents from other parties to all Material Contracts, if necessary or appropriate to allow the consummation of the Merger. Each party hereto shall use its commercially reasonable efforts to obtain any and all permits or approvals of any Governmental Entity required by such party for the lawful consummation of the Merger.

- (b) Each party hereto shall, upon request, furnish each other with all information concerning themselves, their respective subsidiaries, directors, officers and shareholders and such other matters as may be reasonably necessary or advisable in connection with the Proxy and Registration Statement or any other statement, filing, notice or application made by or on behalf of Parent, Buyer or Company to any Governmental Entity in connection with the Merger and the other transactions contemplated hereby.
- (c) Each party hereto shall promptly furnish each other with copies of all written communications received by such party or any of their respective subsidiaries from, or delivered by any of the foregoing to, any Governmental Entity in respect of the transactions contemplated hereby.
- 4.4 Meeting of Shareholders; Proxy and Registration Statement; Listing Application.
- (a) Meeting of Shareholders. Company shall take all action necessary in accordance with applicable Law and its charter documents to duly call, give notice of, convene and hold a meeting of its shareholders as promptly as practicable to consider and vote upon this Agreement, the Merger and all matters related thereto. Company shall, through its Board of Directors, recommend to its shareholders approval of such matters (unless Company's directors after consultation with legal counsel, believe such action is inconsistent with the proper exercise by such directors of their fiduciary duties), and Company shall use its best efforts to obtain such approval by its shareholders. Company agrees to adjourn, postpone or delay its meeting, or to convene a second meeting, as appropriate, in the event insufficient voting shares are present to conduct the meeting.
- (b) Proxy and Registration Statement. As promptly as reasonably practicable, Company, Parent and Buyer shall prepare and file with the SEC a proxy statement and registration statement on Form S-4 under the Securities Act and the rules and regulations promulgated thereunder with respect to the Parent Stock to be issued in the Merger (the "Proxy and Registration Statement" or "Registration **Statement"**) for use in connection with the ${\tt Company's\ shareholder\ meeting.\ The\ Proxy\ and\ Registration}$ Statement shall not be filed, and no amendment or supplement to the Proxy and Registration Statement shall be made by either Company or Parent, without prior consultation with the other party and its counsel. Company and Parent shall cooperate and use all reasonable efforts to have the Registration Statement declared effective by the SEC. Parent will, as promptly as practicable, provide any written comments received from the SEC with respect to the Registration Statement and advise Company of any verbal comments received from the SEC with respect thereto. Parent shall also take any action (other than qualifying to do business in any jurisdiction in which it is now not so qualified) required to be taken under the securities or "blue" sky" laws of the various States in connection with the issuance of the Parent Stock pursuant to the Merger.

A-25

(each an "Indemnifying Party") agrees to indemnify and hold harmless the other, each person who controls the other within the meaning of the Securities Act, and each director and officer of the other, against any losses, claims, damages, liabilities or expenses (including reasonable counsel fees and costs of investigation and defense) that are based on the ground or alleged ground that the Registration Statement includes an untrue statement of a material fact or omits to state a material fact required to be stated therein or necessary in order to make the statements therein not misleading. This indemnification obligation extends, however, only insofar as any such statement or omission was made in reliance upon, and in conformity with, any written information furnished by the Indemnifying Party for use in the preparation of such materials. These indemnity obligations will remain in force after any termination of this Agreement.

- (d) Listing Application. Parent shall use its best efforts to cause the Parent Stock distributed in connection with the Merger to be authorized for listing on the NYSE and shall make all necessary blue sky law filings in connection therewith.
- 4.5 Conduct of Company Business. Company covenants and agrees that after the date hereof and prior to the Effective Time (unless Parent and Buyer shall have agreed in writing):
 - (i) each Option (as defined in **Section 2.2**) issued under the Company Option Plan (a **"Plan Option"**), whether or not then exercisable, will be converted into an option to acquire shares of Parent Stock, under, and subject to the terms and conditions of, the Parent Option Plan (or, at the election of Parent, subject to the terms and conditions of such Plan Option); provided, that (A) the number of shares of Parent Stock subject to such option at acquire Parent Stock shall be determined by multiplying the number of shares of Company Common Stock subject to such Plan Option by a fraction, of which the numerator shall be \$7.85 and the denominator shall be the Exchange Price (with such share number rounded down to the nearest whole number); and (B) the exercise price of such option to acquire Parent Stock shall be determined by multiplying the exercise price of such Plan Option by a fraction, of which the numerator shall be the Exchange Price and the denominator shall be \$7.85 (with such exercise price rounded up to the nearest penny), and
 - (ii) each Warrant (as defined in **Section 2.2**), and each Option other than a Plan Option, whether or not then exercisable, will be converted into a right to acquire shares of Parent Stock, subject to terms and conditions materially equivalent, in the reasonable judgment of the Board of Directors of Parent, to the terms and conditions set out in such Warrant or Option; provided, that (A) the number of shares of Parent Stock subject to such right shall be determined by multiplying the number of Company Common Shares subject to such Warrant or Option by a fraction, of which the numerator shall be \$7.85 and the denominator shall be the Exchange Price; and (B) the exercise price of such right shall be determined by multiplying the exercise price of such Warrant by a fraction, of which the numerator shall be the Exchange Price and the denominator shall be \$7.85.
- 4.6 Publicity. Company and Parent must mutually agree upon the initial press releases. Thereafter, Company and Parent shall coordinate all publicity relating to the transactions contemplated by this Agreement and no party shall issue any press release, publicity statement or other public notice relating to this Agreement, or the transactions contemplated by this Agreement, without prior consultation with both Company and Parent, except to the extent that the disclosing party is advised by its counsel that such action is required by applicable Law, and then, if practicable, only after consultation with the other party.

- 4.7 Notification of Defaults and Adverse Events.

 Company and Parent will promptly notify each other if, subsequent to the date of this Agreement and prior to the Effective Date:
 (i) an event occurs that may be reasonably likely to result in a Company Material Adverse Effect or a Parent Material Adverse Effect, respectively, or (ii) any suit, action or proceeding is instituted or, to the knowledge of Company or Parent, threatened against or affecting Company or Parent or any of their respective subsidiaries which, if adversely determined, would be reasonably likely to result in a Company Material Adverse Effect or a Parent Material Adverse Effect. Each of Company and Parent will promptly notify the other if it determines it is or will be unable to comply with any of its obligations under this Agreement or fulfill any conditions under its control.
- 4.8 Satisfy Conditions to Closing. Parent and Company shall each use its reasonable best efforts to cause all conditions to Closing to be satisfied.
- 4.9 Termination Fee. If this Agreement is terminated (a) by Parent pursuant to **Section 6.1**(d), Section 6.1(e), Section 6.1(f) or Section 6.1(h), or (b) by Company pursuant to Section 6.1(d) or Section 6.1(g) (in each case only if the Company or its stockholders have received in writing, or there shall have been publicly disclosed, an Acquisition Proposal on or before the date of such termination), then Company shall pay to Parent, upon demand, \$750,000 in cash. If within one year of (A) any such termination, or (B) any termination by Parent pursuant to Section 6.1(c)(if prior to such termination pursuant to **Section 6.1**(c), (i) the SEC shall have declared effective the Proxy and Registration Statement, (ii) the waiting period under the HSR Act shall have expired, in each case no later than August 1, 2000, and (iii) the Company or its stockholders shall have received in writing, or there shall have been publicly disclosed, an Acquisition Proposal), Company enters into an agreement to effect an Acquisition Proposal, then Company shall pay to Parent, upon demand, an amount in cash equal to \$4,500,000, less any amount paid pursuant to the immediately preceding sentence (the total of all amounts payable under this Section 4.9 being the ("Termination Fee").
- 4.10 Anti-takeover Statutes. If any anti-takeover or similar statute is applicable to the transactions contemplated hereby, Company will grant such approvals and take such actions as are necessary so that the transactions contemplated hereby may be consummated as promptly as practicable on the terms contemplated hereby and otherwise act to eliminate the effects of such anti-takeover or similar statute on the transactions contemplated hereby.
- 4.11 Indemnification; Insurance.
- (a) Indemnification. For not less four
 (4) years after the Effective Time, Parent and the Surviving Corporation shall indemnify, defend and hold harmless, each present and former director and officer of Company and each such person's personal representative, estate, testator or intestate successors (the "Indemnified Parties") against any and all losses, claims, damages, liabilities, costs, expenses, judgments and amounts paid in settlement with the approval of Parent and the Surviving Corporation (which approval shall not be unreasonably withheld) in connection with any actual or threatened claim, action, suit, proceeding or investigation arising out of or pertaining to any action or omission occurring prior to the Effective Time (including without limitation, any which arise out of or relate to the transactions contemplated by this Agreement), whether asserted or claimed prior to, or on or

after, the Effective Time, to the full extent Company would be permitted under the DGCL or Company's Certificate of Incorporation or By-Laws in effect as of the date of this Agreement (to the extent consistent with applicable law), including, without limitation, provisions relating to advances of expenses incurred in the defense of any action or suit, provided that any determination required to be made with respect to whether an Indemnified Party's conduct complies with the standards set forth under Delaware law and the Company's Certificate of Incorporation and By-laws shall be made by independent counsel mutually acceptable to Parent and the Indemnified Party. In addition, Parent and the Surviving Corporation shall pay expenses incurred by an Indemnified Party in advance of the final disposition of any such action or proceeding upon receipt of an undertaking by or on behalf of such Indemnified Party to repay such amount if it shall ultimately be determined that he is not entitled to be indemnified. Without limiting the foregoing, in the event any claim, action, suit, proceeding or investigation is brought against any Indemnified Party, Parent and the Surviving Corporation shall be entitled to assume the defense of any such action or proceeding. Upon assumption by Parent and the Surviving Corporation of the

A-27

Table of Contents

defense of any such action or proceeding, the Indemnified Party shall have the right to participate in such action or proceeding and to retain its own counsel, but neither Parent nor the Surviving Corporation shall be liable for any legal fees or expenses subsequently incurred by the Indemnified Party in connection with the defense thereof unless (i) Parent and the Surviving Corporation have agreed to pay such fees and expenses, (ii) the Indemnified Party shall have been advised by counsel that representation of the Indemnified Party by counsel provided by Parent and the Surviving Corporation is not possible due to conflicts of interest among Parent and the Surviving Corporation and the Indemnified Party, or (iii) Parent and the Surviving Corporation shall have failed in a timely manner to assume the defense of the matter. Neither Parent nor the Surviving Corporation shall be liable for any settlement of any claim effected without its written consent. Neither Parent nor the Surviving Corporation shall, except with the written consent of the Indemnified Party, consent to entry of any judgment or enter into any settlement which does not include as an unconditional term the release by the claimant or plaintiff of such Indemnified Party from all further liability in respect of such claim. Any Indemnified Party wishing to claim indemnification under this Section 4.11(a), upon learning of any such claim, action, suit, proceeding or investigation, shall notify Parent and the Surviving Corporation (but the failure so to notify Parent and the Surviving Corporation shall not relieve it from any liability which it may have under this **Section 4.11**(a) except to the extent such failure materially prejudices Parent and the Surviving Corporation). In addition to the foregoing, and without limiting in any manner the foregoing, after the Effective Time Parent and the Surviving Corporation shall assume the obligations of Company under the indemnification agreements set forth in Section 4.4(c), but only to the extent Company would be permitted under the DGCL to perform its obligations under such indemnification agreements.

(b) Insurance. For a period of not less than four (4) years after the Effective Date, Parent and the Surviving Corporation shall cause to be maintained officers' and directors' liability insurance covering Company's existing officers and directors who are currently covered in such capacities by Company's existing officers' and directors' liability insurance policies on terms substantially no less advantageous to such officers and directors than such existing insurance provided, further, that in no event shall Parent and the Surviving Corporation be required to expend in excess of 200% of the annual premium currently paid by the Company for such coverage (or obtain coverage in excess of the coverage that is available for such 200% of such annual premium).

- 4.12 Employee Benefits. (a) As soon after the date hereof as practicable and legally permissible, Company shall make a cash refund of all participant plan accounts under the ESPP pursuant to the resolutions described in Section 2.11(m).
- (b) At the Effective Time, the Surviving Corporation shall offer all persons who were theretofore employees of Company and its subsidiaries benefits under Parent Benefit Plans which, in the aggregate, are no less favorable to such employees than those that Company currently provides to its own employees. Each Parent Benefit Plan (i) shall give credit for purposes of eligibility to participate and vesting to employees of Company and its subsidiaries for service prior to the Effective Time with Company and its subsidiaries (and their predecessors, to the extent credit for service with such predecessors was given by Company) to the same extent that such service was recognized under a comparable Company Benefit Plan and (ii) shall, if applicable, waive any pre-existing condition or limitation applicable to the addition of such employees to any Parent Benefit Plan to the same extent that such condition or limitation would be waived under a comparable Company Benefit Plan.
- 4.13 No Solicitation. From and after the date hereof, Company shall not, and shall not authorize or permit any of its subsidiaries or Representatives to, directly or indirectly, solicit or initiate (including by way of furnishing information) or take any other action to facilitate knowingly any inquiries or the making of any proposal which constitutes or may reasonably be expected to lead to an Acquisition Proposal from any person or entity, or engage in any discussion or negotiations relating thereto or accept any Acquisition Proposal; provided, however, that notwithstanding any other provision hereof, Company may (a) comply with Rule 14e-2 promulgated under the Exchange Act with regard to a tender or exchange offer; and (b) at any time prior to the Closing, (i) engage in discussions or negotiations with a third party who (without any solicitation, initiation, encouragement, discussion or negotiation, directly or indirectly, by or with Company or any of its subsidiaries or Representatives after the date hereof) seeks to initiate such discussions or

A-28

Table of Contents

negotiations, and may furnish such third party nonpublic information concerning Company and its business, properties and assets if, and only to the extent that, (A) (1) the third party has first made a bona fide Acquisition Proposal that the Board of Directors of Company believes in good faith (after consultation with its financial advisor) is reasonably capable of being completed, taking into account all relevant, legal, financial, regulatory and other aspects of the Acquisition Proposal and the source of its financing, and believes in good faith (after consultation with its financial advisor and after considering all of the terms, conditions, representations, warranties and covenants which are included in such Acquisition Proposal) that such Acquisition Proposal would, if consummated,

Company, from a financial point of view, than the transactions contemplated by this Agreement and believes in good faith (after consultation with its financial advisor) that the person making such Acquisition Proposal has, or is reasonably likely to have or obtain, any necessary funds or customary commitments to provide any funds necessary to consummate such Acquisition Proposal (any such more favorable Acquisition Proposal being referred in this Agreement as a "Superior Proposal") and (2) Company's Board of Directors shall conclude in good faith, after considering applicable provisions of state law, that such action may be necessary for the Board of Directors to act in a manner consistent with its fiduciary duties under applicable law, and (B) forty-eight hours prior to furnishing such information to or entering into discussions or negotiations with such person or entity, Company
(1) provides prompt notice to Parent to the effect that it is furnishing information to or entering into discussions or negotiations with such person or entity and (2) receives from such person or entity an executed confidentiality agreement in reasonably customary form on terms not materially more favorable to such person or entity than the terms contained in the Confidentiality Agreement, and/or (ii) accept a Superior Proposal from a third party, provided that the conditions set forth in clauses (i)(A) and (i)(B) above have been satisfied and Company complies with and terminates this Agreement pursuant to Section 6.1(g). Company shall immediately cease and terminate any existing solicitation, initiation, encouragement, activity, discussion or negotiation with any persons or entities conducted heretofore by Company or any of its subsidiaries or Representatives with respect to the foregoing. The Company shall notify Parent orally and in writing of any such inquiries, offers or proposals (including the terms and conditions of any such proposal and the identity of the person making it) within twenty-four hours of the receipt thereof, and shall keep Parent informed of the status and details of any such inquiry, offer, or proposal.

result in a transaction more favorable to the shareholders of

- 4.14 Consent of Holders of Options. Company shall use its best reasonable efforts to obtain and deliver to Parent, prior to the Effective Time, binding agreements from the holders of all of the Options, agreeing to the conversion of such Options on the terms described in **Section 1.9**.
- 4.15 Redemption of Series B Preferred Shares.
 Prior to the Effective Time, Company will redeem all outstanding shares of its Series B Preferred Stock, and will obtain the prior written consent of the holders of Series A Preferred Shares to such redemption.
- 4.16 Audited Financial Statements. As soon as practicable, Company shall deliver the consolidated balance sheet of Company as of December 31, 1999 and the consolidated statements of income, shareholders' equity and cash flows for the years ended December 31, 1999 and 1998, in each case including the notes thereto and the related report of PriceWaterhouse Coopers LLP, independent certified public accountants.

ARTICLE V

CONDITIONS

5.1 Conditions to Obligations of Company, Parent and Buyer. The respective obligations of the parties to effect the Merger are subject to the fulfillment at or prior to the Effective Date of the following conditions unless waived in writing by all parties:

- (a) Ordinary Course. Company will, and will cause its subsidiaries to, operate Company's Business only in the ordinary course of business consistent with past practices and use its best efforts to (i) preserve its existing business organization, insurance coverage, material rights, material licenses or permits, advantageous business relationships, material agreements and credit facilities; (ii) retain and keep available the services of its present officers, employees and agents; and (iii) preserve the goodwill of its customers, suppliers and others having business relations with it. Company and its subsidiaries will not: (A) enter into any material transaction or commitment, or dispose of or acquire any material properties or assets, except purchases and sales of inventory in the ordinary course of business consistent with past practices; (B) implement any new employee benefit plan, or employment, compensatory or severance agreement; (C) amend any existing employee benefit plan or employment agreement except as required by Law or by this Agreement; or (D) take any action that would jeopardize the continuance of its material supplier or customer relationships; (E) make any material change in the nature of their businesses and operations; (F) enter into any transaction or agreement with any officer, director or affiliate of Company or any of its subsidiaries; (G) incur or agree to incur any obligation or liability (absolute or contingent) that individually calls for payment by Company or any of its subsidiaries of more than \$100,000 in any specific case in the aggregate, excluding transactions in the ordinary course of business; or (H) make any Tax election or make any change in any method or period of accounting or any material change in any accounting policy, practice or procedure.
- (b) Charter Documents. Company will not amend its Certificate of Incorporation or By-Laws and will not permit any of its Subsidiaries to amend their charter documents or by-laws.
- (c) $\it Dividends$. Company will not declare or pay any dividend or make any other distribution in respect of its capital stock.
- (d) Stock. Company will not split, combine or reclassify any shares of its capital stock, or issue, redeem or acquire (or agree to do so) any of its equity securities, options, warrants, or convertible instruments, except (i) pursuant to existing obligations under Company Benefit Plans, (ii) pursuant to the existing commitments or conversion rights listed on Schedule 2.2(a) or (iii) pursuant to Section 4.15. Company will not grant any Options.

A-29

Table of Contents

- (a) Approval. All corporate actions necessary to authorize the execution, delivery and performance of this Agreement and the Merger shall have been duly and validly taken by the other parties. The shareholders of Company shall have approved this Agreement and the Merger in accordance with applicable Law.
- 5.2 Conditions to Obligations of Parent and Buyer.
 The obligations of Parent and Buyer to effect the Merger are subject to the fulfillment at or prior to the Effective Date of the following conditions except to the extent waived in writing by Parent and Buyer:
 - (b) Approval from Government Entities. All approvals required by any Governmental Entity and all other actions required to effect the Merger and related transactions shall have been obtained. The waiting period under the HSR Act shall have expired, or early termination of the waiting period under the HSR Act shall have been granted.
 - (c) Absence of Governmental Litigation. No Governmental Entity shall have instituted a proceeding seeking injunctive or other relief in connection with the Merger and related transactions. There shall not be any judgment, decree, injunction, ruling or order of any Governmental Entity that prohibits, restricts, or delays consummation of the Merger.
 - (d) Effectiveness of Registration Statement. The Registration Statement covering the Parent Stock shall have been declared effective under the Securities Act and no stop order suspending the effectiveness of the Registration Statement shall have been issued with respect thereto.
 - (e) Market Conditions. There shall not have occurred and be continuing (i) any general suspension of, or limitation on prices for, trading in securities on any national securities exchange or the overthe-counter market, (ii) a declaration of a banking moratorium or any suspension of payments in respect of banks in the United States, (iii) any limitation (whether or not mandatory) by any government or Governmental Entity of the United States on the extension of credit by banks or other lending institutions, or (iv) in the case of any of the foregoing existing at the time of the execution of this Agreement, a material acceleration or worsening thereof.

Table of Contents

- 5.3 Conditions to Obligations of Company. The obligations of Company to effect the Merger are subject to the fulfillment at or prior to the Effective Date of the following conditions unless waived in writing by Company:
 - (a) Representations and Compliance. The representations and warranties of Company in this Agreement shall be true and correct as of the date of this Agreement and on the Effective Date with the same effect as though made on and as of such date, except where failure to be so true and correct would not (in the aggregate for all representations and warranties of Company) have Company Material Adverse Effect (other than representations and warranties that are already so qualified, which in each such case shall be true and correct as written), and except for any changes contemplated by this Agreement; Company shall have complied in all material respects with all covenants requiring compliance by it prior to the Effective Date; and Buyer shall have received an officer's certificate signed by the Chief Executive Officer of Company certifying as to each of the foregoing.
 - (b) Tax Opinion. Parent shall have received an opinion from Carter, Ledyard & Milburn, counsel to Parent, based upon reasonably requested representation letters and dated the Effective Date, that the Merger will constitute a reorganization for United States federal income tax purposes within the meaning of IRC Section 368(a), that each of Parent, Buyer and Company will be a party to that reorganization within the meaning of IRC Section 368(b) and that neither Parent, Buyer nor Company will recognize any gain on the Merger.
 - (c) No Material Adverse Effect. From the date hereof, there shall not have occurred any event which has resulted or is likely to result in a Company Material Adverse Effect.
 - (d) Material Contracts. All consents from International Business Machines Corp., IBM Credit Corp. and their affiliates, and from other parties to the Material Contracts listed on **Schedule 2.4** if necessary to allow the consummation of the Merger and the continuation of Company's Business in the ordinary course after consummation of the Merger, shall have been received.
 - (e) Consent of Option Holders. Company shall have delivered to Parent binding agreements from the holders of at least 90% of the Options (by number of underlying shares of Company Common Stock), agreeing to the conversion of such Options on the terms described in **Section 1.9**.
 - (f) Audited Financial Statements. The corresponding portions of the audited financial statements delivered pursuant to **Section 4.16** shall not differ, in any manner adverse to Parent and Buyer, from the unaudited financial statements described in **Section 2.6**(a).

ARTICLE VI

TERMINATION, AMENDMENT AND WAIVER

- 6.1 Termination and Abandonment. This Agreement may be terminated and the Merger may be abandoned at any time prior to the Effective Time, whether before or after approval by the shareholders of Company:
 - (a) Representations and Compliance. The representations and warranties of Parent and Buyer in this Agreement shall be true and correct in all material respects as of the date of this Agreement and on the Effective Date with the same effect as though made on and as of such date, except where failure to be so true and correct would not (in the aggregate for all representations and warranties of Parent and Buyer) have a Parent Material Adverse Effect (other than representations and warranties that are already so qualified, which in each such case shall be true and correct as written), except for any changes contemplated by this Agreement; Parent and Buyer shall have complied in all material respects with all covenants requiring compliance by it prior to the Effective Date; and Company shall have received an officer's certificate signed by the Chief Executive Officer of Parent certifying as to each of the foregoing.
 - (b) Tax Opinion. Company shall have received an opinion from Pillsbury Madison & Sutro LLP,

counsel to Company, based upon reasonably requested representation letters and dated the Effective Date, that the Merger will constitute a reorganization for United States federal income tax purposes within the meaning of IRC Section 368(a) that each of Parent, Buyer and Company will be a party to that reorganization within the meaning of IRC Section 368(b).

- (c) Listing. The Parent Stock distributed in connection with the Merger shall have been accepted upon notice of issuance for listing on the NYSE.
- (d) No Material Adverse Effect. From the date hereof, there shall not have occurred any event which has resulted or is likely to result in a Parent Material Adverse Effect.

A-31

Table of Contents

- (a) by mutual consent of Parent and Company;
- (b) by either Parent or Company upon written notice to the other party if any Governmental Entity of competent jurisdiction shall have issued a final nonappealable order denying, enjoining or otherwise prohibiting the consummation of any of the transactions contemplated by this Agreement;
- (c) by either Parent or Company if the Merger shall not have been consummated on or before September 15, 2000 unless the failure of the Merger to occur by such date shall be due to the failure of the party seeking to terminate this Agreement to perform or observe in any material respect the covenants and agreements of such party set forth herein;
- (d) by either Company or Parent if any approval of the shareholders of Company required for the consummation of the Merger shall not have been obtained by reason of the failure to hold the Company Shareholders Meeting or to obtain the required vote of shareholders of Company at the Company Shareholders' Meeting or at any adjournment or postponement thereof;
- (e) by either Parent or Company (so long as the terminating party is not then in breach of any representation, warranty, covenant or other agreement contained herein) if there shall have been a breach of any of the representations or warranties set forth in this Agreement on the part of the other party which has or would reasonably likely have a Company Material Adverse Effect (if the terminating party is Parent) or a Parent Material Adverse Effect (if the terminating party is Company);
- (f) by either Parent or Company (so long as the terminating party is not then in breach of any representation, warranty, covenant or other agreement contained herein) if there shall have been a breach
- 6.2 Effect of Termination. Except as provided in Section 4.9 and Section 7.2 hereof with respect to expenses and fees (including the Termination Fee), and except as provided in Section 4.1(b) hereof with respect to information obtained in connection with the transactions contemplated hereby, and except as provided in Section 4.4(c), in the event of the termination of this Agreement and the abandonment of the Merger, this Agreement shall thereafter become null and void and have no effect, and no party hereto shall have any liability to any other party hereto or its shareholders or directors or officers in respect thereof, and each party shall be responsible for its own expenses, except that nothing herein shall relieve any party for liability for any willful breach hereof.
- 6.3 Amendment. This Agreement may be amended by the parties hereto at any time before or after approval hereof by the shareholders of Company and Buyer, but, after any such approval, no amendment shall be made which would under the DGCL require the approval of the shareholders of Company or Buyer, respectively, without such further approval of such shareholders. This Agreement may not be amended except by an instrument in writing signed on behalf of each of the parties hereto.
- 6.4 Extension; Waiver. At any time prior to the Effective Date, the parties hereto may (a) extend the time for the performance of any of the obligations or other acts of the other parties hereto, (b) waive any inaccuracies in the

representations and warranties contained herein or in any document delivered pursuant hereto and (c) waive compliance with any of the agreements or conditions contained herein. Any agreement on the part of a party hereto to any such extension or waiver shall be valid if set forth in an instrument in writing signed on behalf of such party.

ARTICLE VII

MISCELLANEOUS

- 7.1 Termination of Representations and Warranties.
 The representations and warranties of each party will terminate on the Effective Date.
- 7.2 Expenses. Subject to **Section 4.9**, each party will pay its own expenses relating to this Agreement and the transactions contemplated hereby.
- 7.3 Remedies. If, in accordance with the terms of the parenthetical contained in the second sentence of Section 4.4(a), Company's Board of Directors fails to recommend this Agreement (and the transactions contemplated hereby, including the Merger) to Company's shareholders, or amends, modifies, withdraws, conditions or qualifies, in a manner adverse to Parent, its approval and recommendation thereof to Company's shareholders, Parent's sole remedy in connection therewith under this Agreement (without prejudice to the remedies of Parent under the Option Agreement and the Inducement Agreement) shall be Company's payment of the Termination Fee to Parent pursuant to Section 4.9.

A-32

Table of Contents

7.4 Notices. All notices and other communications hereunder shall be in writing and shall be deemed given if delivered in person, sent by registered or certified mail (return receipt requested), or telecopied to the parties at the following addresses (or at such other address for a party as shall be specified by like notice):

of any of the covenants or agreements or conditions or obligations set forth in this Agreement on the part of the other party which has or would reasonably likely have a Company Material Adverse Effect (if the terminating party is Parent) or a Parent Material Adverse Effect (if the terminating party is Company), and which breach shall not have been cured within ten days following receipt by the breaching party of written notice of such breach from the other party hereto or which breach, by its nature, cannot be cured prior to the Effective Time;

- (g) by Company, prior to the consummation of the transactions contemplated hereby, for the purpose of entering into an agreement with a Person that has made a Superior Proposal; and
- (h) by Parent prior to the consummation of the transactions contemplated hereby if the Board of Directors of Company shall have withdrawn, amended, modified, conditioned or qualified in a manner adverse to Parent its approval or recommendation of this Agreement or shall have recommended another Acquisition Proposal or offer for the purchase of Company Common Stock.
- 7.5 Further Assurances. Buyer and Company each agree to execute and deliver such other documents, certificates,

agreements and other writings and to take such other actions as may be reasonably necessary or desirable in order to expeditiously consummate or implement the transactions contemplated by this Agreement.

- 7.6 Assignability. Neither this Agreement nor any rights or obligations under it are assignable.
- 7.7 Governing Law. This Agreement will be governed by the laws of the State of Delaware without regard to conflict of law principles.
- 7.8 Interpretation. Headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement.
- 7.9 Counterparts. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same instrument.
- 7.10 Integration. This Agreement and the Schedules hereto constitute the entire agreement and supersede all prior agreements and understandings (including the Confidentiality Agreement), both written and oral, among the parties with respect to the subject matter hereof.

A-33

Table of Contents

ARTICLE VIII

DEFINITIONS

8.1 Definitions. For all purposes of this Agreement, except as otherwise expressly provided or unless the context otherwise requires: (a) the terms defined in this Article VIII have the meaning assigned to them in this Article VIII and include the plural as well as the singular; (b) all accounting terms not otherwise defined herein have the meanings assigned under generally accepted accounting principles; (c) all references in this Agreement to designated "Articles," "Sections" and other subdivisions are to the designated Articles, Sections and other subdivisions of the body of this Agreement; (d) pronouns of either gender or neuter shall include, as appropriate, the other pronoun forms; and (e) the words "herein," "hereof" and "hereunder" and other words of similar import refer to this Agreement as whole and not to any particular Article, Section nor other subdivision.

As used in this Agreement and the Schedules delivered pursuant to this Agreement, the following definitions shall apply.

"Acquisition Proposal" means any proposal or offer from any Person relating to any direct or indirect acquisition or purchase of all or a substantial part of the assets of the Company or any of its subsidiaries or of over 15% of any class or series of equity securities of the Company or any of its subsidiaries, any tender offer or exchange offer that if consummated would result in any Person beneficially owning 15% or more of any class or series of equity securities of Company or any of its subsidiaries, any merger, consolidation, business combination, sale of all or substantially all of the assets, recapitalization, liquidation, dissolution or similar transaction involving Company or any of its subsidiaries, other than the transactions contemplated by this Agreement.

"Buyer" has the meaning set forth in the first paragraph hereof.

"Canceled Shares" has the meaning set forth in Section 1.7(d).

"Certificate of Merger" has the meaning set forth in Section 1.3.

"Closing" has the meaning set forth in Section 1.2.

"Closing Date" has the meaning set forth in Section 1.2.

"Code" means the Internal Revenue Code of 1986, as amended.

"Company" has the meaning set forth in the first
paragraph hereof.

"Company Affiliates" has the meaning set forth in Section 1.10.

"Company Benefit Plans" has the meaning set forth in Section 2.11(a).

"Company Common Stock" has the meaning set forth
in the second Recital hereto.

"Company ERISA Affiliate" has the meaning set forth in Section 2.11(a).

"Company ERISA Plans" has the meaning set forth in Section 2.11(a).

"Company Material Adverse Effect" means a material adverse effect on the business, financial condition, results of operation, business prospects or properties of Company and its subsidiaries or the Surviving Corporation and its subsidiaries, in each case taken as a whole. For purposes of this Agreement, a Company Material Adverse Effect does not include a material adverse effect on the business, financial condition, results of operation or properties of Company as a result of (i) the transactions contemplated hereby or the public announcement thereof, or (ii) changes in the conditions or prospects of Company and its subsidiaries, taken as a whole, which are consistent with general economic conditions or general changes affecting the electronic components, computer products or production supplies distribution industries or electronics manufacturing industry, or (iii) any matter disclosed in Company SEC Filings (as defined in Section 2.6) made before the execution of this Agreement or in the Schedules to this Agreement.

"Company SEC Filings" has the meaning set forth in Section 2.6(b).

A-34

Table of Contents

"Company Shareholders' Approval" has the meaning set forth in Section 2.3.

"Company Common Shares" means shares of Company
Common Stock.

"Company's Business" means the business of Company and its subsidiaries, taken as a whole.

"Company Option Plan" has the meaning set forth in Section 2.2(a).

"Company Shares" has the meaning set forth in Section 1.7(c).

"Confidentiality Agreement" means the confidentiality agreement of recent date by and between Company and Parent.

"Contract" means any agreement, arrangement, bond, commitment, franchise, indemnity, indenture, instrument, lease, license or understanding, whether or not in writing.

"Effective Date" has the meaning set forth in Section 1.3.

"Effective Time" has the meaning set forth in Section 1.3.

"Environmental Regulations" means, collectively, all Laws, regulations, orders and other requirements of any Governmental Entity relating to the protection of human health or the environment or to Hazardous Substances and the use, storage,

treatment, disposal, transport, generation, release of, and exposure of others to, Hazardous Substances.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"ESPP" has the meaning set forth in Section 2.2(a).

"Exchange Act" means the Securities Exchange Act
 of 1934, as amended.

"Exchange Agent" has the meaning set forth in Section 1.8(a).

"Exchange Price" is that price calculated pursuant to the terms set forth in Section 1.7(a).

"Governmental Entity" means any governmental agency, district, bureau, board, commission, court, department, official political subdivision, tribunal or other instrumentality of any government, whether federal, state or local, domestic or foreign.

"Hazardous Substances" means (but shall not be limited to) substances that are defined or listed in, or otherwise classified pursuant to, any applicable Laws as "hazardous substances," "hazardous materials," "hazardous wastes" or "toxic substances," or any other formulation intended to define, list or classify substances by reason of deleterious properties such as ignitability, corrosivity, reactivity, radioactivity, carcinogenicity, reproductive toxicity or "EP toxicity," and petroleum and drilling fluids, produced waters and other wastes associated with the exploration, development, or production of crude oil, natural gas or geothermal energy, and lead, asbestos, PCBs or other substances regulated under Environmental Regulations.

"HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

"Indemnified Parties" has the meaning set forth in Section 4.11(a).

"Indemnifying Party" has the meaning set forth in Section 4.4(c).

"IRS" means the Internal Revenue Service.

"Laws" means any constitutional provision, statute, ordinance, or other law, code, common law, rule, regulation or interpretation of any Governmental Entity and any decree, injunction, judgment, award, order, ruling, assessment or writ.

"Losses" has the meaning set forth in

"Merger" has the meaning set forth in Section 1.1.

"Merger Consideration" means the Stock Merger Consideration and cash payable in lieu of fractional shares, if any, pursuant to **Section 1.8**(e).

A-35

Table of Contents

"NYSE" means the New York Stock Exchange.

"Options" has the meaning set forth in Section 2.2(a).

"Parent Benefit Plans" means collectively, all employee benefit plans, programs and commitments that Parent makes generally available to its employees and their beneficiaries, providing benefits in the nature of pension, retirement, severance, stock purchase, health, medical, life, disability, sick leave, vacation, or other welfare or fringe benefits, including, without limitation, all employee benefit plans (as defined in Section 3(3) of ERISA) and fringe benefit plans (as defined in IRC Section 6039D).

"Parent's Business" means the business of Parent and its subsidiaries, taken as a whole.

"Parent Material Adverse Effect" means a material adverse effect on the business, financial condition, results of operation, business prospects or properties of Parent and its subsidiaries, taken as a whole. For purposes of this Agreement, a Parent Material Adverse Effect does not include a material adverse effect on the business, financial condition, results of operation or properties of Parent as a result of (i) the transactions contemplated hereby or the public announcement hereof, or (ii) changes in the conditions or prospects of Parent and its subsidiaries, taken as a whole, which are consistent with general economic conditions or general changes affecting the electronic components or computer products, distribution industries or electronics manufacturing industry, or (iii) any matter disclosed in the Parent SEC Filings (as defined in Section 3.6) or in the Schedules to this Agreement.

"Parent Stock" has the meaning set forth in Section 1.7(a).

"Person" means any individual, partnership, joint venture, corporation, bank, trust, unincorporated organization or other entity.

"Plan Option" has the meaning set forth in Section 2.2(a).

"Qualified Plan" has the meaning set forth in Section 2.11(b).

"Registration Statement" has the meaning set forth in Section 4.4(b).

"Representatives" means a Person's or any
of its Subsidiaries' officers, directors, employees,
consultants, investment bankers, accountants, attorneys and other
advisors, representatives and agents.

"Securities Act" means the Securities Act of 1933, as amended.

"SEC" means the Securities and Exchange Commission.

"Superior Proposal" has the meaning set forth in Section 4.13.

"Surviving Corporation" has the meaning set forth in Section 1.1.

"Tax" or "Taxes" means any foreign, federal, state, county or local income, sales, use, excise, franchise, ad valorem, real and personal property, transfer, gross receipt, stamp, premium, profits, customs, duties, windfall profits, capital stock, production, business and occupation, disability, employment, payroll, severance or withholding taxes, fees, assessments or charges of any kind whatever imposed by any Governmental Entity, and interest and penalties (civil or criminal), additions to tax, payments in lieu of taxes or additional amounts related thereto or to the nonpayment thereof, and any loss in connection with the determination, settlement or litigation of any Tax liability.

"Tax Return" means a declaration, statement report, return or other document or information required to be filed or supplied with respect to Taxes, including, where permitted or required, combined or consolidated returns for any group of entities that includes Company or any of its subsidiaries.

"Termination Fee" has the meaning set forth in Section 4.9.

IN WITNESS WHEREOF, the parties have duly executed this Agreement as of the date first written above.

if to Parent or Buyer:

Avnet, Inc. 2211 South 4th Street Phoenix, Arizona 85034

Attention: David Birk, General Counsel

Telecopy: (480) 643-7929

with a copy to:

Carter, Ledyard & Milburn 2 Wall Street New York, New York 10005 Attention: Jim Abbott, Esq. Telecopy: (212) 732-3200

if to Company:

Savoir Technology Group, Inc. 254 East Hacienda Avenue Campbell, California 95008 Attention: P. Scott Munro, CEO Telecopy: (408) 370-4597

with a copy to:

Pillsbury Madison & Sutro LLP 2550 Hanover Street Palo Alto, California 94304 Attention: Jorge del Calvo Telecopy: (650) 233-4545

AVNET, INC.

By: /s/ RAYMOND SADOWSKI

Name: Raymond Sadowski

Title Senior Vice President and Chief Financial Officer

TACTFUL ACQUISITION CORP.

By: /s/ DAVID R. BIRK

Name: David R. Birk Title: President

SAVOIR TECHNOLOGY GROUP, INC.

By: /s/ P. SCOTT MUNRO

Name: P. Scott Munro

A-37

STOCK OPTION AGREEMENT

STOCK OPTION AGREEMENT, dated as of March 2, 2000 (the "Agreement"), by and between Savoir Technology Group, Inc., a Delaware corporation ("Issuer"), and Avnet, Inc., a New York corporation ("Grantee").

WHEREAS, Issuer, Grantee and Tactful Acquisition Corp., a
Delaware corporation ("Sub"), which is a direct wholly
owned subsidiary of Grantee, propose to enter into an Agreement
and Plan of Merger, dated as of the date hereof (the "Merger
Agreement"; capitalized terms used but not defined herein
shall have the meanings set forth in the Merger Agreement),
providing for, among other things, a merger (the
"Merger") of Sub with and into Issuer;

WHEREAS, as a condition and inducement to Grantee's willingness to enter into the Merger Agreement, Grantee has requested that Issuer agree, and Issuer has agreed, to grant Grantee the Option (as defined below); and

NOW, THEREFORE, in consideration of the foregoing and the respective representations, warranties, covenants and agreements set forth herein and in the Merger Agreement, Issuer and Grantee agree as follows:

Title: Chairman and Chief Executive Officer

B-1

Table of Contents

- 1. Grant of Options. Subject to the terms and conditions set forth herein, Issuer hereby grants to Grantee an irrevocable option (the "Option") to purchase up to 2,023,435 shares (the "Option Shares") of common stock, par value \$0.01 per share, of Issuer (the "Issuer Common Stock") (being 15% of the number of shares of Issuer Common Stock outstanding on February 24, 2000 before such issuance), at a purchase price of \$6.83 per Option Share (such price, as adjusted if applicable, the "Purchase Price"), payable in cash or in shares of common stock, par value \$1.00 per share, of Grantee (the "Grantee Common Stock"), at the election of Grantee. The number and nature of Option Shares that may be received upon the exercise of the Option and the Purchase Price are subject to adjustment as set forth herein.
- 2. Exercise of Option. (a) Grantee may exercise the Option, in whole or in part, at any time or from time to time following the occurrence of a Purchase Event (as defined below); provided that, except as otherwise provided herein, the Option shall terminate and be of no further force and effect upon the earliest to occur of (i) the Effective Time, (ii) 6 months after the first occurrence of a Purchase Event (or if, at the expiration of such 6-months after the first occurrence of a Purchase Event, the Option cannot be exercised by reason of any applicable judgment, decree, order, law or regulation, 10 business days after such impediment to exercise shall have been removed, but in no event under this clause (ii) later than the first anniversary of the Purchase Event), (iii) termination of the Merger Agreement under circumstances which do not and cannot result in Grantee's becoming entitled to receive the Termination Fee from Issuer pursuant to Section 4.9 of the Merger Agreement; and (iv) 12 months after the termination of the Merger Agreement under circumstances which do or could result in Grantee's becoming entitled to receive the Termination Fee from Issuer pursuant to Section 4.9 of the Merger Agreement, unless during such 12-month period, a Purchase Event shall occur. The Grantee also may terminate the Option, in whole or in part, upon notice to Issuer. The termination of the Option shall not affect any rights hereunder which by their terms extend beyond the date of such termination.
- (b) As used herein, a "Purchase Event" means an event the result of which is that the Grantee becomes entitled to receive the Termination Fee from Issuer pursuant to Section 4.9 of the Merger Agreement.

(c) In the event Grantee wishes to exercise the Option, it shall send to Issuer a written notice (the "Exercise Notice"; the date of which being herein referred to as the "Notice Date") specifying (i) the total number of Option Shares it intends to purchase pursuant to such exercise and (ii) a place and date not earlier than three business days nor later than 10 business days from such Notice Date for the closing of such purchase (a "Closing"; and the date of such Closing, a "Closing Date"); and (iii) whether it

elects to pay the Purchase Price in cash or in shares of Grantee Common Stock; provided that such closing shall be held only if (A) such purchase would not otherwise violate or cause the violation of applicable law (including the HSR Act), (B) no law, rule or regulation shall have been adopted or promulgated, and no temporary restraining order, preliminary or permanent injunction or other order, decree or ruling issued by a court or other governmental authority of competent jurisdiction shall be in effect, which prohibits delivery of such Option Shares (and the parties hereto shall use their reasonable best efforts to have any such order, injunction, decree or ruling vacated or reversed) and (C) any prior notification to or approval of any other regulatory authority in the United States or elsewhere required in connection with such purchase shall have been made or obtained, other than those which if not made or obtained would not reasonably be expected to result in a significant detriment to the Grantee and its Subsidiaries taken as a whole or the Issuer and its Subsidiaries taken as a whole. If the Closing cannot be consummated by reason of a restriction set forth in clause (A), (B) or (C) above, notwithstanding the provisions of Section 2(a), the Closing shall be held within 5 business days following the elimination of such restriction.

- 3. Payment and Delivery of Certificates. (a) On each Closing Date, if Grantee has elected to pay the Purchase Price therefor in cash, Grantee shall pay to Issuer in immediately available funds by wire transfer to a bank account designated by Issuer an amount equal to the Purchase Price multiplied by the Option Shares to be purchased on such Closing Date.
- (b) On each Closing Date, if Grantee has elected to pay the Purchase Price therefor in shares of Grantee Common Stock ("Purchase Shares"), Grantee shall deliver to Issuer a certificate or certificates representing the Purchase Shares to be delivered at such Closing, which Purchase Shares shall be free and clear of all liens, charges or encumbrances ("Liens"), and Issuer shall deliver to Grantee a letter agreeing that Issuer shall not offer to sell or otherwise dispose of such Purchase Shares in violation of applicable law or the provisions of this Agreement. The number of Purchase Shares issuable at such Closing shall be obtained by multiplying the number of Option Shares specified in the Exercise Notice therefor by a fraction, of which the numerator shall be the Purchase Price, and the denominator shall be price per share of Grantee Common Stock on the five trading days immediately preceding the Notice Date therefor.
- (c) At each Closing, simultaneously with the delivery of immediately available funds as provided in Section 3(a) or Purchase Shares as provided in Section 3(b), Issuer shall deliver to Grantee a certificate or certificates representing the Option Shares to be purchased at such closing, which Option Shares shall be free and clear of all Liens, and Grantee shall deliver to Issuer a letter agreeing that Grantee shall not offer to sell or otherwise dispose of such Option Shares in violation of applicable law or the provisions of this Agreement.
- (d) Certificates for the Option Shares and Purchase Shares (collectively, "Shares") delivered at each Closing shall be endorsed with a restrictive legend which shall read substantially as follows:

THE TRANSFER OF THE STOCK REPRESENTED BY THIS CERTIFICATE IS SUBJECT TO RESTRICTIONS ARISING UNDER THE SECURITIES ACT OF 1933, AS AMENDED, AND PURSUANT TO THE TERMS OF A STOCK OPTION AGREEMENT DATED AS OF MARCH , 2000. A COPY OF SUCH AGREEMENT WILL BE PROVIDED TO THE HOLDER HEREOF WITHOUT CHARGE UPON RECEIPT BY THE ISSUER OF A WRITTEN REQUEST THEREFOR.

B-2

Table of Contents

It is understood and agreed that (i) the reference to restrictions arising under the Securities Act in the above legend shall be removed by delivery of substitute certificate(s) without such reference if the issuer of such Shares shall have received a copy of a letter from the staff of the SEC, or an opinion of counsel in form and substance reasonably satisfactory to such issuer and its counsel, to the effect that such legend is not required for purposes of the Securities Act and (ii) the reference to restrictions pursuant to this Agreement in the above legend shall be removed by delivery of substitute certificate(s) without such reference if the Shares evidenced by certificate(s) containing such reference have been sold or

Table of Contents

- 4. Authorized Stock. (a) Issuer hereby represents and warrants to, and covenants with, Grantee that Issuer has taken all necessary corporate and other action to authorize and reserve and to permit it to issue, at all times from the date hereof until the obligation to deliver Shares upon the exercise of the Option terminates, and will have reserved for issuance, upon exercise of the Option, all of the Option Shares issuable to Grantee upon exercise of the Option, and Issuer will take all necessary corporate and other action to authorize and reserve for issuance and to permit it to issue all additional shares of Issuer Common Stock or other securities which may be issued pursuant to Section 6 upon exercise of the Option. The Option Shares to be issued upon due exercise of the Option, including all additional shares of Issuer Common Stock or other securities which may be issuable upon exercise of the Option pursuant to Section 6, upon issuance pursuant hereto, shall be duly and validly issued, fully paid and nonassessable, and shall be delivered free and clear of all Liens, including any preemptive rights of any stockholder of Issuer.
- (b) Grantee hereby represents and warrants to, and covenants with, Issuer that prior to each Closing at which Grantee will issue Purchase Shares, Grantee will take all necessary corporate and other action to authorize and reserve and to permit it to issue, and will have reserved for issuance, all of the Purchase Shares issuable to Issuer at such Closing, and Grantee will take all necessary corporate and other action to authorize and reserve for issuance and to permit it to issue all additional shares of Grantee Common Stock or other securities which may be issued pursuant to Section 6 upon exercise of the Option. The Purchase Shares to be issued upon due exercise of the Option, including all additional shares of Grantee Common Stock or other securities which may be issuable upon exercise of the Option pursuant to Section 6, upon issuance pursuant hereto, shall be duly and validly issued, fully paid and nonassessable, and shall be delivered free and clear of all Liens, including any preemptive rights of any stockholder of Grantee.
- 5. Purchase Not For Distribution. Grantee hereby represents and warrants to Issuer that any Option Shares or other securities acquired by Grantee upon exercise of the Option will not be taken with a view to the public distribution thereof and will not be transferred or otherwise disposed of except in a transaction registered or exempt from registration under the Securities Act. Issuer hereby represents and warrants to Grantee that any Purchase Shares or other securities acquired by Issuer upon exercise of the Option will not be taken with a view to the public distribution thereof and will not be transferred or otherwise disposed of except in a transaction registered or exempt from registration under the Securities Act.
- 6. Adjustment Upon Changes in Capitalization, Etc. (a) In the event of any change in Shares by reason of reclassification, recapitalization, stock split, split-up, combination, exchange of shares, stock dividend, dividend, dividend payable in any other securities, or any similar event, the type and number of Shares or securities subject to the Option, and the Purchase Price therefor (including for purposes of repurchase thereof pursuant to Section 7), shall be adjusted appropriately, and proper provisions shall be made in the agreements governing such transaction, so that Grantee and Issuer each shall receive upon exercise of the Option the number and class of shares or other securities or property that Grantee would have received in respect of Shares if the Option had been exercised immediately prior to such event or the record date therefor, as applicable. If any additional shares of Issuer Common Stock are issued after the date of this Agreement (other than pursuant to an event described in the immediately preceding sentence), the number of shares of Issuer Common Stock subject to the Option shall be adjusted so that immediately prior to such issuance, it equals 15% of the number of Shares then issued and outstanding. In no event shall the number of shares of Issuer Common Stock subject to the Option exceed 15% of the number of shares of Issuer Common Stock subject to the exercise (without giving effect to any shares subject or issued pursuant to the Option).
- (b) Without limiting the foregoing, whenever the number of Option Shares purchasable upon exercise of the Option is adjusted as provided in this Section 6, the Purchase Price per Option Share shall

be adjusted by multiplying the Purchase Price by a fraction, the numerator of which is equal to the number of Option Shares purchasable prior to the adjustment and the denominator of which is equal to the number of Option Shares purchasable after the adjustment.

(c) Without limiting the parties' relative rights and obligations under the Merger Agreement, in the event that Issuer enters into an agreement (i) to consolidate with or merge or convert into any Person, other than Grantee or one of its Subsidiaries, and Issuer will not be the continuing or

surviving corporation in such consolidation, conversion, or merger, (ii) to permit any Person, other than Grantee or one of its Subsidiaries, to merge into Issuer and Issuer will be the continuing or surviving corporation, but in connection with such merger, the shares of Company Common Stock outstanding immediately prior to the consummation of such merger will be changed into or exchanged for stock or other securities of Issuer or any other Person or cash or any other property, or (iii) to sell or otherwise transfer all or substantially all of its assets to any Person, other than Grantee or one of its Subsidiaries, then, and in each such case, the agreement governing such transaction will make proper provision so that the Option will, upon the consummation of any such transaction and upon the terms and conditions set forth herein, be converted into, or exchanged for, an option with identical terms appropriately adjusted to acquire the number and class of shares or other securities or property that Grantee would have received in respect of Option Shares had the Option been exercised immediately prior to such consolidation, conversion, merger, sale or transfer or the record date therefor, as applicable. Issuer shall take such steps in connection with such consolidation, merger, conversion, sale, transfer, or other such transaction as may be reasonably necessary to assure that the provisions hereof shall thereafter apply as nearly as possible to any securities or property thereafter deliverable upon exercise of the Option.

- 7. Repurchase of Option. (a) Notwithstanding the provisions of Section 2(a), at any time upon or after the first occurrence of a Purchase Event and prior to termination of the Option in accordance with Section 2, Issuer shall at the request of Grantee (any such request, a "Cash Exercise Notice"), repurchase from Grantee the Option or a portion thereof (if and to the extent not previously exercised or terminated) at a price which, subject to Section 10 below, is equal to the excess, if any, of (x) the Applicable Price (as defined below) as of the Section 7 Request Date (as defined below) for an Option Share over (y) the Purchase Price (subject to adjustment pursuant to Section 6), multiplied by all or such portion of the Option Shares subject to the Option as the Grantee shall specify in the Cash Exercise Notice (the "Option Repurchase Price").
- (b) Notwithstanding the provisions of Section 2(a), at any time following the occurrence of a Purchase Event, Issuer (or any successor entity thereof) may, at its election (notice of which shall be given to Grantee), repurchase the Option (if and to the extent not previously exercised or terminated) at the Option Repurchase Price; provided that the aggregate number of Option Shares as to which the Option may be repurchased shall not exceed 1,348,957. For purposes of this Agreement, an exercise of the Option shall be deemed to occur on the Closing Date and not on the Notice Date relating thereto.
- (c) In connection with any exercise of rights under this Section 7, Issuer shall, within 5 business days after the Section 7 Request Date, pay the Option Repurchase Price in immediately available funds, and Grantee shall surrender to Issuer the Option or the applicable portion thereof. Upon receipt by the Grantee of the Option Repurchase price, the obligations of the Issuer to deliver Option Shares pursuant to Section 3 of this Agreement shall be terminated with respect to the number of Option Shares specified in the Cash Exercise Notice or the number of Option Shares as to which the Option is repurchased under Section 7(b).
 - (d) For purposes of this Agreement, the following terms have the following meanings:

B-4

Table of Contents

(i) "Applicable Price", as of any date, means the highest of (A) the highest price per Share paid or proposed to be paid by any third party for Shares or the consideration per Share received or to be received by holders of Shares, in each case pursuant to any Acquisition Proposal for or with Issuer made on or prior to such date or (B) the average closing price per Share as reported by Nasdaq National Market ("NNM") or if the Shares are not listed on the NNM, the highest bid price per Share as quoted on the National Association of Securities Dealers Automated Quotation

System or, if the Shares are not quoted thereon, on the principal trading market on which such Shares are traded as reported by a recognized source, during the 10 trading days preceding such date. If the consideration to be offered, paid or received pursuant to the foregoing clause (A) shall be other than in cash, the value of such consideration shall be determined in good faith by an independent nationally recognized investment banking firm selected by Grantee and reasonably acceptable to Issuer.

(ii) "Section 7 Request Date" means the date on which Issuer gives notice of its election to repurchase the Option pursuant to Section 7(b) or Grantee provides a Cash Exercise Notice, as the case may be.

- Registration Rights. Issuer shall, if requested by Grantee or any Subsidiary of Grantee which is the owner of Option Shares (collectively with Grantee, the "Owners") at any time and from time to time within two years of the first exercise of the Option, as expeditiously as possible prepare and file up to two registration statements under the Securities Act if such registration is necessary in order to permit the sale or other disposition of any or all shares or other securities that have been acquired by or are issuable to such Owners upon exercise of the Option ("Registrable Securities") in accordance with the intended method of sale or other disposition stated by such Owners, including a "shelf" registration statement under Rule 415 under the Securities Act or any successor provision, and Issuer shall use all reasonable efforts to qualify such Registrable Securities under any applicable state securities laws. Issuer shall use all reasonable efforts to cause each such registration statement to become effective, to obtain all consents or waivers of other parties which are required therefor and to keep such registration statement effective for such period at least 90 days from the day such registration statement first becomes effective as may be reasonably necessary to effect such sale or other disposition. The obligations of Issuer hereunder to file a registration statement and to maintain its effectiveness may be suspended for a period of time not exceeding 90 days in the aggregate if the Board of Directors of Issuer shall have determined in good faith that the filing of such registration statement or the maintenance of its effectiveness would require disclosure of nonpublic information that would materially and adversely affect Issuer (but in no event shall Issuer exercise such postponement right more than once in any 12-month period). Any registration statement prepared and filed under this Section 8, and any sale covered thereby, shall be at Issuer's expense except for underwriting discounts or commissions, brokers' fees and the reasonable fees and disbursements of Owners' counsel related thereto. The Owners shall provide all information reasonably requested by Issuer for inclusion in any registration statement to be filed hereunder. If during the time period referred to in the first sentence of this Section 8 Issuer effects a registration under the Securities Act capital stock of the same class as the Registrable Securities for its own account or for any other stockholders of Issuer (other than on Form S-4 or Form S-8, or any successor form), it shall allow the Owners the right to participate in such registration, and such participation shall not affect the obligation of Issuer to effect two registration statements for the Owners under this Section 8; provided that, if the managing underwriters of such offering advise Issuer in writing that in their opinion the number of Registrable Securities requested to be included in such registration exceeds the number which can be sold in such offering without adversely affecting the offering price, Issuer and the Owners shall each reduce on a pro rata basis the Registrable Securities to be included therein on their respective behalf. In connection with any registration pursuant to this Section 8, Issuer and the Owners shall provide each other and any underwriter of the offering with customary representations, warranties, covenants, indemnification and contribution in connection with such registration.
- 9. Additional Covenants of Issuer. (a) If Shares or any other securities to be acquired upon exercise of the Option are then listed on the NNM or any other securities exchange or market, Issuer, upon the request of any Owner, will promptly file an application to list the Shares or other securities to be acquired upon exercise of the Options on the NNM or such other securities exchange or market and will use its reasonable best efforts to obtain approval of such listing as soon as practicable.
- (b) Issuer will use its reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable under applicable laws and regulations to permit the exercise of the Option in accordance with the terms and conditions hereof, as soon as practicable after

B-6

Table of Contents

the date hereof, including making any appropriate filing pursuant to the HSR Act and any other applicable law, supplying as promptly as practicable any additional information and documentary material that may be requested pursuant to the HSR Act and any other applicable law, and taking all other actions necessary to cause the expiration or termination of the applicable waiting periods under the HSR Act as soon as practicable.

- (c) Issuer agrees not to avoid or seek to avoid (whether by charter amendment or through reorganization, consolidation, conversion, merger, issuance of rights, dissolution or sale of assets, or by any other voluntary act) the observance or performance of any of the covenants, agreements or conditions to be observed or performed hereunder by it.
- (d) Issuer shall take all such steps as may be required to cause any acquisitions or dispositions by Grantee (or any affiliate who may become subject to the reporting requirements of Section 16(a) of the Exchange Act) of any Shares acquired in connection with this Agreement (through conversion or exercise of the Option or otherwise) to be exempt under Rule 16b-3 promulgated under the Exchange Act.

- 10. Limitation of Grantee Profit. (a) Notwithstanding any other provision in this Agreement, in no event shall Grantee's Total Profit (as defined below) exceed \$5,100,000 (the "Maximum Profit") and, if it otherwise would exceed such amount, Grantee, at its sole discretion, shall either (i) reduce the number of Shares subject to the Option, (ii) deliver to Issuer for cancellation Shares (or other securities into which such Option Shares are converted or exchanged) previously purchased by Grantee, (iii) pay cash to Issuer, or (iv) any combination of the foregoing, so that Grantee's actually realized Total Profit shall not exceed the Maximum Profit after taking into account the foregoing actions.
- (b) Notwithstanding any other provision of this Agreement, the Option may not be exercised for a number of Option Shares as would, as of any Notice Date, result in a Notional Total Profit (as defined below) of more than the Maximum Profit and, if exercise of the Option otherwise would result in the Notional Total Profit exceeding such amount, Grantee, at its discretion, may (in addition to any of the actions specified in Section 10(a) above) (i) reduce the number of Shares subject to the Option or (ii) increase the Purchase Price for that number of Option Shares set forth in the Exercise Notice so that the Notional Total Profit shall not exceed the Maximum Profit; PROVIDED that nothing in this sentence shall restrict any exercise of the Option permitted hereby on any subsequent date at the Purchase Price set forth in Section 1 hereof.
- (c) For purposes of this Agreement, "Total Profit" shall mean: (i) the aggregate amount (before taxes) of (A) any excess of (x) the net cash amounts or fair market value of any property received by Grantee pursuant to a sale of Option Shares (or securities into which such shares are converted or exchanged) over (y) the Grantee's aggregate Purchase Price for such Option Shares (or other securities), plus (B) any amounts received by Grantee on the repurchase of the Option by Issuer pursuant to Section 7, plus (C) any Termination Fee paid by Issuer and received by Grantee pursuant to Section 4.9 of the Merger Agreement, minus (ii) the amounts of any cash previously paid by Grantee to Issuer pursuant to this Section 10 plus the value of the Option Shares (or other securities) previously delivered by Grantee to Issuer for cancellation pursuant to this Section 10.
- (d) For purposes of this Agreement, "Notional Total Profit" with respect to any number of Option Shares as to which Grantee may propose to exercise the Option shall mean the Total Profit determined as of the Notice Date assuming that the Option was exercised on such date for such number of Option Shares specified in the Exercise Notice and assuming that such Option Shares, together with all other Option Shares previously acquired upon exercise of the Option and held by Grantee as of such date, were sold for cash at the closing price per Share on the NNM as of the close of business on the preceding trading day (less customary brokerage commissions).
- (e) Notwithstanding any other provision of this Agreement, nothing in this Agreement shall affect the ability of Grantee to receive, nor relieve Issuer's obligation to pay, the Termination Fee provided for in Section 4.9 of the Merger Agreement; provided that if and to the extent the Total Profit received by

B-7

Table of Contents

Grantee would exceed the Maximum Profit following receipt of such payment, Grantee shall be obligated to promptly comply with the terms of Section 10(a).

- (f) For purposes of Section 10(a) and clause (ii) of Section 10(c), the value of any Option Shares delivered by Grantee to Issuer shall be the Applicable Price of such Option Shares.
- 11. Loss, Theft, Etc. of Agreement. This Agreement (and the Option granted and evidenced hereby) is exchangeable, without expense, at the option of Grantee, upon presentation and surrender of this Agreement at the principal office of Issuer for other Agreements providing for Options of different denominations entitling the holder thereof to purchase in the aggregate the same number of Shares purchasable hereunder. The terms "Agreement" and "Option" as used herein include any other Agreements and related Options for which this Agreement (and the Option granted hereby) may be exchanged. Upon receipt by Issuer of evidence reasonably satisfactory to it of the loss, theft, destruction or mutilation of this Agreement, and (in the case of loss, theft or destruction) of reasonably satisfactory indemnification, and upon surrender and cancellation of this Agreement, if mutilated, Issuer will execute and deliver a new Agreement of like tenor and date. Any such new Agreement executed and delivered shall constitute an additional contractual obligation on the part of Issuer, whether or not the Agreement so lost, stolen, destroyed or mutilated shall at any time be enforceable by anyone.

12. Miscellaneous.

- (a) Expenses. Except as otherwise provided herein or in the Merger Agreement, each of the parties hereto shall bear and pay all expenses incurred by it or on its behalf in connection with the transactions contemplated hereunder, including fees and expenses of its own financial consultants, investment bankers, accountants and counsel.
 - (b) Waiver and Amendment. Any provision of this Agreement may be waived at any time by the party

that is entitled to the benefits of such provision. This Agreement may not be modified, amended, altered or supplemented except upon the execution and delivery of a written agreement executed by the parties hereto.

- (c) Entire Agreement; No Third-Party Beneficiary; Severability. Except as otherwise set forth in the Merger Agreement, this Agreement, together with the Merger Agreement (a) constitutes the entire agreement and supersedes all prior agreements and understandings, both written and oral, between the parties with respect to the subject matter hereof and (b) is not intended to confer upon any person other than the parties hereto any rights or remedies hereunder. If any term, provision, covenant or restriction of this Agreement is held by a court of competent jurisdiction or a federal or state regulatory agency to be invalid, void or unenforceable, the remainder of the terms, provisions, covenants and restrictions of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated. If for any reason such court or regulatory agency determines that the Option does not permit Grantee to acquire, or does not require Issuer to issue or repurchase, the full number of Shares, or all or the relevant portion of the Option, as the case may be and as provided in Sections 2 and 7, as adjusted pursuant to Section 6, it is the express intention of Issuer to allow Grantee to acquire or to require Issuer to issue or repurchase such lesser number of Shares, or such lesser portion of the Option, as the case may be, as may be permissible without any amendment or modification hereof.
- (d) Governing Law. THIS AGREEMENT SHALL BE GOVERNED AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF DELAWARE (WITHOUT GIVING EFFECT TO CHOICE OF LAW PRINCIPLES).
- (e) Descriptive Headings. The descriptive headings contained herein are for convenience of reference only and shall not affect in any way the meaning or interpretation of this Agreement.
- (f) Notices. All notices and other communications hereunder shall be in writing and shall be deemed given as set forth in Section 7.4 of the Merger Agreement.
- (g) Counterparts. This Agreement and any amendments hereto may be executed in two counterparts, each of which shall be considered one and the same agreement and shall become effective when

IN WITNESS WHEREOF, Issuer and Grantee have caused this Stock Option Agreement to be signed by their respective officers thereunto duly authorized, all as of the date first above written.

both counterparts have been signed by each of the parties and delivered to the other party, it being understood that both parties need not sign the same counterpart.

- (h) Assignment. Grantee may not, without the prior written consent of Issuer (which shall not be unreasonably withheld), assign this Agreement or the Option to any other person. This Agreement shall not be assignable by Issuer except by operation of law. Subject to the preceding sentence, this Agreement shall be binding upon, inure to the benefit of and be enforceable by the parties and their respective successors and assigns.
- (i) Representations and Warranties. The representations and warranties contained in Sections 2.1 through 2.5 of the Merger Agreement are incorporated herein by reference, mutatis mutandis.
- (j) Further Assurances. In the event of any exercise of the Option by Grantee, Issuer and Grantee shall execute and deliver all other documents and instruments and take all other action that may be reasonably necessary in order to consummate the transactions provided for by such exercise.
- (k) Enforcement. The parties agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms. It is accordingly agreed that the parties shall be entitled to specific performance of the terms hereof, this being in addition to any other remedy to which they are entitled at law or in equity. Both parties further agree to waive any requirement for the securing or posting of any bond in connection with the obtaining of any such equitable relief and that this provision is without prejudice to any other rights that the parties hereto may have for any failure to perform this Agreement.
- (1) Captions. The Article, Section and paragraph captions herein are for convenience only, do not constitute part of this Agreement and shall not be deemed to limit or otherwise affect any of the provisions hereof.
- (m) Confidentiality Agreement. Issuer hereby waives the restrictions on Grantee's acquisition of Shares contained in the Confidentiality Agreement to the extent necessary to permit Grantee to exercise the Option and purchase the Option Shares as herein provided.

SAVOIR TECHNOLOGY GROUP, INC. By: /s/ P. SCOTT MUNRO

P. Scott Munro Chairman and Chief Executive Officer

AVNET, INC. By: /s/ RAYMOND SADOWSKI

B-8

Table of Contents

APPENDIX C

INDUCEMENT AGREEMENT

This Inducement Agreement (the "Agreement"), dated as of March 2, 2000, by and among Avnet, Inc., a New York corporation ("Parent"), and the stockholders listed on the signature page hereof (each such stockholder being referred to herein as a "Stockholder" and, collectively with each other Stockholder, the "Stockholders").

WITNESSETH

WHEREAS, each Stockholder is the sole record and beneficial owner of, and has the sole right to vote with respect to, certain shares of common stock, par value \$.01 per share (the "Company Common Stock") of Savoir Technology Group, Inc., a Delaware corporation ("Company") (together with any shares of Company Common Stock acquired by a Stockholder after the date hereof, the "Company Shares");

WHEREAS, Parent, Tactful Acquisition Corp., a Delaware corporation ("Buyer"), and Company propose on the date hereof to enter into an Agreement and Plan of Merger (the "Merger Agreement"), pursuant to which Buyer will be merged with and into Company (the "Merger"), on the terms and subject to the conditions contained in the Merger Agreement; and

WHEREAS, in order to induce Buyer and Parent to enter into the Merger Agreement and to incur the obligations set forth therein, the Stockholders are entering into this Agreement pursuant to which each Stockholder is granting: (i) an irrevocable proxy to Parent to vote in favor of the Merger, and to make certain agreements with respect to such Stockholders' Company Shares, upon the terms and conditions set forth herein, and (ii) an option to Parent to purchase all of the Company Shares owned by such Stockholders.

NOW THEREFORE, for and in consideration of the foregoing and the mutual promises contained herein, and upon and subject to the terms and conditions set forth below, the parties hereto agree as follows:

Table of Contents

Section 1. Grant of Irrevocable Proxy. Each Stockholder hereby irrevocably appoints and constitutes Parent or any designee of Parent, with full power of substitution, the lawful agent, attorney and proxy of the Stockholder (each an "Irrevocable Proxy") during the term of this Agreement to vote in its sole discretion all of the shares of Company Common Stock of which such Stockholder is or becomes the owner of record or has the power to vote in the following manner for the following purposes: (i) to call one or more meetings of the stockholders of Company in accordance with the By-Laws of Company and applicable law for the purpose of considering the transactions contemplated by the Merger Agreement such that the stockholders shall have the full opportunity to approve the Merger Agreement and any and all amendments, modifications and waivers thereof and the transactions contemplated thereby; (ii) in favor of the Merger Agreement or any of the transactions contemplated by the Merger Agreement at any stockholders meetings of Company held to consider the Merger Agreement (whether annual or special and whether or not an adjourned meeting); (iii) against any other proposal for any recapitalization, merger, sale of assets or other business combination between Company and any other person or entity other than Buyer or Parent or the taking of any action which would result in any of the conditions to Parent's obligations under the Merger Agreement not being fulfilled; and (iv) as otherwise necessary or appropriate to enable Buyer and Parent to consummate the transactions contemplated by the Merger Agreement and, in connection with such purposes, to otherwise act with respect to the Shares which the Stockholder is entitled to vote. THIS IRREVOCABLE PROXY HAS BEEN GIVEN IN CONSIDERATION OF THE UNDERTAKINGS OF BUYER AND PARENT IN THE MERGER AGREEMENT AND SHALL BE IRREVOCABLE AND COUPLED WITH AN INTEREST UNTIL THE EXPIRATION DATE AS DEFINED IN SECTION 2 HEREOF. This Agreement shall revoke all other proxies granted by the Stockholders with respect to their Shares.

Section 2. Expiration Date. This Irrevocable Proxy shall expire on the date (the "Expiration Date") of the earlier to occur of (i) the effective time of the Merger, (ii) 6 months after the first occurrence of a Purchase Event (or if, at the expiration of such 6-months after the first occurrence of a

C-2

Table of Contents

Purchase Event, the Option (as defined below) cannot be exercised by reason of any applicable judgment, decree, order, law or regulation, 10 business days after such impediment to exercise shall have been removed, but in no event under this clause (ii) later than the first anniversary of the Purchase Event), (iii) termination of the Merger Agreement under circumstances which do not and cannot result in Parent becoming entitled to receive the Termination Fee from Company pursuant to Section 4.9 of the Merger Agreement; and (iv) 12 months after the termination of the Merger Agreement under circumstances which do or could result in Parent's becoming entitled to receive the Termination Fee from Issuer pursuant to Section 4.9 of the Merger Agreement, unless during such 12-month period, a Purchase Event shall occur. As used herein, a "Purchase Event" means an event the result of which is that Parent becomes entitled to receive the Termination Fee from Company pursuant to Section 4.9 of the Merger Agreement.

Section 3. Grant of Option. Subject to the conditions herein set forth, each Stockholder hereby grants to Parent an irrevocable option (the "Option") to acquire such Stockholder's Shares (the "Option Shares"), at the Exercise Price, payable in cash or in shares of common stock, par value \$1.00 per share, of Parent (the "Parent Stock"), at the election of Parent. The number and nature of Option Shares that may be received upon the exercise of the Option, and the Exercise Price, are subject to adjustment as set forth herein. The Option may be exercised from and after the date Company or its stockholders shall have received in writing, or there shall have been publicly disclosed, an "Acquisition Proposal" (as such term is defined in the Merger Agreement). The Option shall terminate and be of no further force and effect upon the Expiration Date.

Section 4. Exercise of Option. Parent may exercise the Option by delivery of written notice of exercise (an "Exercise Notice") to the Stockholder with respect to which the Option is being exercised (the "Called Stockholder"), binding Parent to acquire the Called Stockholder's Option Shares on the terms set forth herein, signed by an officer of Parent, to the Called Stockholder at the executive offices of Company in Campbell, California, prior to the Expiration Date.

Section 5. Exercise Price. (a) The "Exercise Price" shall be \$7.85, subject to adjustment pursuant to this Section 5. In the event of any change in Option Shares by reason of reclassification,

recapitalization, stock split, split-up, combination, exchange of shares, stock dividend, dividend, dividend payable in any other securities, or any similar event, the type and number of Option Shares or securities subject to the Option, and the Exercise Price therefor, shall be adjusted appropriately, and proper provisions shall be made in the agreements governing such transaction, so that Parent shall receive upon exercise of each Option the number and class of shares or other securities or property that Parent would have received in respect of Option Shares if such Option had been exercised immediately prior to such event or the record date therefor.

- (b) If the Merger shall occur, under the Merger Agreement as entered into on the date hereof, at an "Exchange Price" (as such term is defined in the Merger Agreement) less than \$59.6063, and if any Option shall have been exercised prior to the Merger, then Parent shall issue to each Stockholder with respect to which the Option was exercised, promptly after the Merger, an additional number of shares of Parent Stock (plus cash in lieu of fractional shares) equal to the difference (the "Collar Difference") between (i) the number of shares of Parent Stock (A) issued upon such exercise of the Option (if Parent shall have elected to pay the Exercise Price in shares of Parent Stock) or (B) derived by dividing the amount of cash received upon such exercise of the Option (if Parent shall have elected to pay the Exercise Price in cash) by the Exchange Price and (ii) the number of shares of Parent Stock to which such Stockholder would have been entitled in respect of such Option Shares upon the Merger.
- (c) If the Merger shall occur, under the Merger Agreement as entered into on the date hereof, at an Exchange Price greater than \$59.6063 and if any Option shall have been exercised prior to the Merger, then each Stockholder with respect to which the Option was exercised, promptly after the Merger, shall deliver to Parent a number of shares of Parent Stock (plus cash in lieu of fractional shares) equal to the Collar Difference, or, in the election of the Stockholder, cash equal to the product of the Collar Difference times the Exchange Price.

Section 6. Closing of Option. (a) The closing of each purchase and sale of Option Shares (the "Closing") shall occur at the offices of Carter, Ledyard & Milburn in New York, New York, at 10 a.m. on the second business day following the delivery of the Exercise Notice therefor.

- (b) At each Closing, if Parent shall have elected to pay the Exercise Price therefor in shares of Parent Stock ("Purchase Shares"), Parent shall deliver to the Called Stockholder a certificate or certificates representing the Purchase Shares to be delivered at such Closing, which Purchase Shares shall be free and clear of all liens, charges or encumbrances ("Liens"), plus cash in lieu of fractional shares, and the Called Stockholder shall deliver to Grantee a letter agreeing that Issuer shall not offer to sell or otherwise dispose of such Purchase Shares in violation of applicable law or the provisions of this Agreement. The number of Purchase Shares issuable at such Closing shall be obtained by multiplying the number of Option Shares specified in the Exercise Notice therefor by a fraction, of which the numerator shall be the Exercise Price, and the denominator shall be price per share of Parent Common Stock on the five trading days immediately preceding the date of the Exercise Notice therefor.
- (c) At each Closing, if Parent shall have elected to pay the Exercise Price therefor in cash, Grantee shall pay to the Called Stockholder in immediately available funds by wire transfer to a bank account designated by Issuer an amount equal to the Exercise Price multiplied by the number of Option Shares to be purchased at such Closing.
- (d) At each Closing, simultaneously with the delivery of immediately available funds as provided in Section 6(b) or Purchase Shares as provided in Section 6(c), the Called Stockholder shall deliver to Parent a certificate or certificates representing the Option Shares to be purchased at such closing, which Option Shares shall be free and clear of all Liens, and Parent shall deliver to the Called Stockholder a letter agreeing that Parent shall not offer to sell or otherwise dispose of such Option Shares in violation of applicable law or the provisions of this Agreement.
- (e) Certificates for the Option Shares and Purchase Shares delivered at each Closing shall be endorsed with a restrictive legend which shall read substantially as follows:

THE TRANSFER OF THE STOCK REPRESENTED BY THIS CERTIFICATE IS SUBJECT TO RESTRICTIONS ARISING UNDER THE SECURITIES ACT OF 1933, AS AMENDED, AND PURSUANT TO THE TERMS OF AN INDUCEMENT AGREEMENT DATED AS OF MARCH , 2000. A COPY OF SUCH AGREEMENT WILL BE PROVIDED TO THE HOLDER HEREOF WITHOUT CHARGE UPON RECEIPT BY THE ISSUER OF A WRITTEN REQUEST THEREFOR.

It is understood and agreed that (i) the reference to restrictions arising under the Securities Act in the above legend shall be removed by delivery of substitute certificate(s) without such reference if the issuer of such Shares shall have received a copy of a letter from the staff of the Securities and Exchange Commission, or an opinion of counsel in form and substance reasonably satisfactory to such issuer and its counsel, to the effect that such legend is not required for purposes of the Securities Act and (ii) the reference to restrictions pursuant to this Agreement in the above legend shall be removed by delivery of substitute certificate(s) without such reference if the Shares evidenced by certificate(s) containing such reference have been sold or transferred in compliance with the provisions of this Agreement under circumstances that do not require the retention of such reference.

Table of Contents

- (a) sell, transfer, pledge, hypothecate, encumber, assign, tender or otherwise dispose of, or enter into any contract, option or other arrangement or understanding with respect to the sale, transfer, pledge, hypothecation, encumbrance, assignment, tender or other disposition of, any of his Company Shares or any interest therein;
- (b) other than as expressly contemplated by this Agreement, grant any powers of attorney or proxies or consents in respect of any of such Stockholder's Company Shares, deposit any of such

Company Shares into a voting trust, enter into a voting agreement with respect to any of such Company Shares or otherwise restrict or take any action adversely affecting the ability of such Stockholder freely to exercise all voting rights with respect thereto; or

(c) except as permitted by the Merger Agreement, directly or indirectly through his or her agents and representatives, initiate, solicit or encourage, any inquiries or the making or implementation of any alternative proposal (an "Alternative Proposal") to acquire the Company Shares or engage in any negotiations concerning, or provide any confidential information or data to, or have any discussions with, any person relating to an Alternative Proposal, or otherwise facilitate any effort or attempt to make or implement an Alternative Proposal; and such Stockholder shall (i) immediately cease and cause to be terminated any existing activities, including discussions or negotiations with any parties, conducted heretofore with respect to any of the foregoing and will take the necessary steps to inform his or her agents and representatives of the obligations undertaken in this Section 7(c), and (ii) notify Parent immediately if any such inquiries or proposals are received by, any such information is requested from, or any such negotiations or discussions are sought to be initiated or continued with, him or her.

Section 8. Covenants of Parent. Parent covenants and agrees for the benefit of the Stockholders that (a) immediately upon execution of this Agreement, Parent shall enter into the Merger Agreement, and (b) until the Expiration Date, it shall use all reasonable efforts to take, or cause to be taken, all action, and do, or cause to be done, all things necessary or advisable in order to consummate and make effective the transactions contemplated by this Agreement and the Merger Agreement, consistent with the terms and conditions of each such agreement; provided, however, that nothing in this Section 8 or any other provision of this Agreement is intended, nor shall it be construed, to limit or in any way restrict Parent's right or ability to exercise any of its rights under the Merger Agreement.

Section 9. Representations and Warranties of the Stockholders. Each Stockholder represents and warrants to Parent that:

C-4

Table of Contents

- (a) the execution, delivery and performance by such Stockholder of this Agreement will not conflict with, require a consent, waiver or approval under, or result in a breach or default under, any of the terms of any contract, commitment or other obligation (written or oral) to which such Stockholder is bound;
- (b) such Stockholder has full right, power and authority to enter into and execute this Agreement and to perform his obligations hereunder;
- (c) this Agreement has been duly executed and delivered by such Stockholder and constitutes a legal, valid and binding obligation of such Stockholder enforceable against him in accordance with its terms;

- (d) such Stockholder is the sole record and beneficial owner of, and has the sole right to vote with respect to, the number of Company Shares set forth opposite such Stockholder's name on Schedule A hereto, and such Company Shares represent all shares of Company Common Stock of or with respect to which such Stockholder is the sole owner or has the right to vote at the date hereof;
- (e) except for the Company Shares listed on Schedule A hereto, such Stockholder does not have any right to acquire, nor is he or she the "beneficial owner" (as such term is defined in Rule 13d-3 under the Securities Exchange Act of 1934, as amended) of, any other shares of any class of capital stock of Company or any securities convertible into or exchangeable or exercisable for any shares of any class of capital stock of Company (other than shares subject to options or other rights granted by Company as set forth on Schedule B hereto);
- (f) such Stockholder's Company Shares are duly authorized, validly issued, fully paid and non-assessable, and such Stockholder owns its Company Shares free and clear of all Liens, other than as provided by this Agreement, and good and valid title to its Company Shares, free and clear of any

Lien, will pass to Parent upon Closing or exercise of the Option granted pursuant to Section 4 hereof; and

(g) The Board of Directors of Company has approved the granting of the Option to Parent.

The representations and warranties contained herein shall be made as of the date hereof and as of the Closing.

Section 10. Representations and Warranties of Parent. Parent represents and warrants to the Stockholders that:

- (a) It has all requisite corporate power and authority to enter into and perform all of its obligations under this Agreement;
- (b) The execution, delivery and performance of this Agreement by it and all transactions contemplated hereby have been duly authorized by all necessary corporate action on its part, and this Agreement constitutes the legal, valid and binding contract of Parent enforceable against it in accordance with its terms;
- (c) Parent will not acquire the Option Shares with a view to the distribution thereof as that term is used in the Securities Act of 1933; and
- (d) The Purchase Shares issuable hereunder, when issued to a Called Stockholder in accordance with Section 6(b) hereof, will be duly authorized, validly issued, fully paid and non-assessable shares of common stock of Parent, and good and valid title to such shares of Parent Stock, free and clear of any Encumbrance, will pass to the Stockholders upon exercise of the Option.

C-5

Table of Contents

The representations and warranties contained herein shall be made as of the date hereof and as of the Closing.

Section 10. Adjustments; Additional Shares. In the event of any stock dividend, stock split, merger (other than the Merger), recapitalization, reclassification, combination, exchange of shares or the like of the capital stock of Company on, of or affecting the Company Common Stock then the terms of this Agreement shall apply to the shares of capital stock or other instruments or documents that the Stockholders own or have the right to vote immediately following the effectiveness of such event as though they were Shares hereunder.

Section 11. Registration Rights. If Parent shall have elected to pay the Exercise Price for the purchase of any Option Shares in Purchase Shares, Parent shall, as expeditiously as reasonably possible after the Closing therefor, prepare and file a registration statement under the Securities Act if such registration is necessary in order to permit the sale or other disposition of any or all such Purchase Shares in accordance with the intended method of sale or other disposition stated by the holder of such Purchase Shares, including a "shelf" registration statement under Rule 415 under the Securities Act or any successor provision, and Parent shall use all reasonable efforts to list such Purchase Shares on the New York Stock Exchange and to qualify such Purchase Shares under any applicable state securities laws.

Parent shall use all reasonable efforts to cause such registration statement to become effective, to obtain all consents or waivers of other parties which are required therefor and to keep such registration statement effective for such period at least 90 days from the day such registration statement first becomes effective as may be reasonably necessary to effect such sale or other disposition. The obligations of Parent hereunder to file a registration statement and to maintain its effectiveness may be suspended for a period of time not exceeding 90 days in the aggregate if the Board of Directors of Parent shall have determined in good faith that the filing of such registration statement or the maintenance of its effectiveness would require disclosure of nonpublic information that would materially and adversely affect Parent (but in no event shall Parent exercise such postponement right more than once in any 12-month period). Any registration statement prepared and filed under this Section 11, and any sale covered thereby, shall be at Parent's expense except for underwriting discounts or commissions, brokers' fees and the reasonable fees and disbursements of counsel to the Purchase Share holders. The Purchase Share holders shall provide all information reasonably requested by Parent for inclusion in any registration statement to be filed

hereunder. In connection with any registration pursuant to this Section 11, Parent and the Purchase Share holders shall provide each other and any underwriter of the offering with customary representations, warranties, covenants, indemnification and contribution in connection with such registration.

Section 12. Specific Performance. The parties hereto agree that the Shares are unique and that money damages are an inadequate remedy for breach of this Agreement because of the difficulty of ascertaining the amount of damage that will be suffered by Parent in the event that this Agreement is breached. Therefore, each of the Stockholders agrees that in addition to and not in lieu of any other remedies available in Parent at law or in equity, Parent may obtain specific performance of this Agreement.

Section 13. Assignment. Parent's rights and obligations under this Agreement may not be assigned without the consent of each affected Stockholder, except that Parent may assign the same to any of its direct or indirect wholly-owned subsidiaries upon delivery of written notice of such assignment to such affected Stockholder(s).

Section 14. Amendments. Amendment or waiver of any provision of this Agreement or consent to departure therefrom shall be effective unless in writing and signed by Parent and all affected Stockholders, in the case of an amendment, or by the party which is the beneficiary of any such provision, in the case of a waiver or a consent to depart therefrom.

Section 15. Notices. Any notices or other communications hereunder shall be in writing and shall be deemed to have bee duly given (and shall be deemed to have been duly received if so given) if personally delivered or sent by telecopier or by registered or certified mail, postage paid, addressed to the respective parties as follows:

If to Parent:

Avnet, Inc. 2211 South 4th Street Phoenix, Arizona 85034

Attention: David Birk, General Counsel

Telecopy: (480) 643-7929

with a copy to:

Carter, Ledyard & Milburn 2 Wall Street New York, New York 10005 Attention: James E. Abbott Telecopier No.: 212-732-3232

If to a Stockholder:

with a copy to:

Pillsbury Madison & Sutro LLP 2550 Hanover Street Palo Alto, California 94304 Attention: Jorge del Calvo Telecopy: (650) 233-4545

or to such other address as either party may have furnished to the other in writing in accordance herewith, except that notices of change of address shall only be effective upon receipt.

C-6

Table of Contents

Section 16. *Miscellaneous.* All references herein to time shall mean New York, New York time. All amounts payable hereunder are in United States Dollars.

C - 7

Table of Contents

IN WITNESS WHEREOF, Parent and each of the Stockholders have duly executed this Agreement as of the date and year first above written.

Section 17. Governing Law. This Agreement shall be governed by and construed in accordance with the internal substantive laws of the State of Delaware, without regard to the conflict of laws principles thereof.

Section 18. Binding Effect. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors, personal representatives, executors, heirs and permitted assigns.

Section 19. *Headings*. The Section headings herein are for convenience of reference only and shall not affect the construction hereof.

Section 20. Counterparts. This Agreement may be executed in several counterparts, each of which shall be an original, but all of which together shall constitute one and the same Agreement.

AVNET, INC.

By: /s/ RAYMOND SADOWSKI

Name: Raymond Sadowski

Title: Senior Vice President and

APPENDIX D

[LOGO TO COME]

April 19, 2000

Board of Directors

Savoir Technology Group, Inc.

254 E. Hacienda Avenue

Campbell, CA 95008

Ladies and Gentlemen:

You have requested our opinion as to the fairness, from a financial point of view, to the shareholders of Savoir Technology Group, Inc. ("Savoir" or the "Company") of the consideration received in the acquisition (the "Acquisition") of Savoir by Avnet, Inc. ("Avnet"). As contemplated by the Amended and Restated Agreement and Plan of Merger dated as of March 2, 2000 (the "Merger Agreement"), holders of Savoir common stock will receive, for each share of Savoir common stock, between 0.15494 and 0.11452 of a share of Avnet common stock, depending upon the average of the closing prices of Avnet common stock during the fifteen trading days ending five days before the date of the special meeting. Holders of Savoir Series A Preferred stock will receive, for each share of Series A Preferred stock, a number of shares of Avnet common stock equal to \$9.6581 divided by the average of the closing prices of Avnet common stock during the five trading days ending on the day before the effective date of the merger. Based on an Exchange Price at signing of \$59.61 in Avnet common stock and 16.294 million Savoir shares outstanding (on a fully diluted basis), consideration paid to Savoir shareholders is valued at \$127.9 million.

For purposes of the opinion set forth herein, we have:

Chief Financial Officer

/s/ P. SCOTT MUNRO

P. Scott Munro

/s/ CARLTON JOSEPH MERTENS II

Carlton Joseph Mertens II

/s/ DENNIS POLK

Dennis Polk

/s/ BOB O'REILLY

Bob O'Reilly

/s/ LARRY SMART

Larry Smart

/s/ ANGELO GUADAGNO
Angelo Guadagno
/s/ BILL SICKLER

Bill Sickler

/s/ MIKE GUNNELS

Mike Gunnels

/s/ GUY LAMMLE

Guy Lammle

We have assumed and relied upon, without independent verification, the accuracy and completeness of the information reviewed by us for the purposes of this opinion. With respect to the financial projections of

D-1

Table of Contents

Board of Directors
Savoir Corporation

April 19, 2000

Page 2

Savoir, we have assumed that they have been reasonably prepared on bases reflecting the best currently available estimates and judgments of the future financial performance of the Company. The financial and other information regarding Savoir reviewed by Alliant Partners in connection with the rendering of this opinion was limited to information provided by Savoir's management and certain discussions with Savoir's senior management regarding the Company's financial condition and prospects as well as the strategic objectives of the Acquisition and strategic alternatives available to Savoir. In addition, we have assumed that the Acquisition will be consummated in accordance with the terms set forth in the Agreement. We have not made any independent valuation or appraisal of the assets or liabilities of Savoir, nor have we been furnished with any such appraisals. Our opinion is necessarily based on economic, market and other conditions as in effect on, and the information made available to us as of, the date hereof.

Our opinion addresses only the fairness of the proposed Acquisition, from a financial point of view, to the stockholders of Savoir, and we do not express any views on any other terms of the proposed Acquisition or the business and strategic bases underlying the Merger Agreement.

Alliant Partners has received fees for this transaction as well as for other Savoir transactions over the past several years.

Based upon and subject to the foregoing, it is our opinion that, as of the date hereof, the total consideration received by the Savoir stockholders pursuant to the Agreement and Plan of Merger

is fair, from a financial point of view, to the Savoir stockholders.

- (a) Reviewed public financial statements and other information concerning Savoir and Avnet as well as selected analyst reports discussing historical and projected future performance of Avnet;
- (b) Reviewed certain internal financial statements and other financial and operating data concerning Savoir that was prepared by Savoir's management;
 - (c) Analyzed certain financial projections prepared by the management of Savoir;
- (d) Discussed the past and current operations, financial condition, and the prospects of Savoir with senior executives of Savoir;
- (e) Discussed with the senior management of Savoir the strategic objectives of the Acquisition and the strategic alternatives available to Savoir;
 - (f) Discussed with the senior management of Avnet the strategic objectives of the Acquisition;
- (g) Compared the financial performance of Savoir with that of certain comparable publicly-traded companies and the prices paid for securities in those publicly-traded companies;
- (h) Reviewed the financial terms, to the extent publicly available, of certain comparable acquisition transactions of comparable companies;
- (i) Assessed Savoir's value based upon a forecast of future cash flows using a discounted cash flow analysis;
- (j) Reviewed the Merger Agreement and discussed the proposed terms of the transaction with managements of both Savoir and Avnet; and
- (k) Performed such other analyses and considered such other factors as we have deemed appropriate.

D-2

Table of Contents

APPENDIX E

Section 262 of the Delaware General Corporation Law — Appraisal Rights

(a) Any stockholder of a corporation of this State who holds shares of stock on the date of the making of a demand pursuant to subsection (d) of this section with respect to such shares, who continuously holds such shares through the effective date of the merger or consolidation, who has otherwise complied with subsection (d) of this section and who has neither voted in favor of the merger or consolidation nor consented thereto in writing pursuant to § 228 of this title shall be entitled to an appraisal by the Court of Chancery of the fair value of the stockholder's shares of stock under the circumstances described in subsections (b) and (c) of this section. As used in this section, the word "stockholder" means a holder of record of stock in a stock corporation and also a member of record of a nonstock corporation; the words "stock" and "share" mean and include what is ordinarily meant by those words and also membership or membership interest of a member of a nonstock corporation; and the words "depository receipt" mean a receipt or other instrument issued by a depository representing an interest in one or more shares, or fractions thereof, solely of stock of a corporation, which stock is deposited with the depository.

(b) Appraisal rights shall be available for the shares of any class or series of stock of a constituent corporation in a merger or consolidation to be effected pursuant to § 251 (other than a merger effected pursuant to § 251(g) of this title), § 252, § 254, § 257, § 258, § 263 or § 264 of this title:

Very truly yours,
/s/ Alliant Partners

- (1) Provided, however, that no appraisal rights under this section shall be available for the shares of any class or series of stock, which stock, or depository receipts in respect thereof, at the record date fixed to determine the stockholders entitled to receive notice of and to vote at the meeting of stockholders to act upon the agreement of merger or consolidation, were either (i) listed on a national securities exchange or designated as a national market system security on an interdealer quotation system by the National Association of Securities Dealers, Inc. or (ii) held of record by more than 2,000 holders; and further provided that no appraisal rights shall be available for any shares of stock of the constituent corporation surviving a merger if the merger did not require for its approval the vote of the stockholders of the surviving corporation as provided in subsection (f) of § 251 of this title.
- (2) Notwithstanding paragraph (1) of this subsection, appraisal rights under this section shall be available for the shares of any class or series of stock of a constituent corporation if the holders thereof are required by the terms of an agreement of merger or consolidation pursuant to §§ 251, 252, 254, 257, 258, 263 and 264 of this title to accept for such stock anything except:
 - a. Shares of stock of the corporation surviving or resulting from such merger or consolidation, or depository receipts in respect thereof;
 - b. Shares of stock of any other corporation, or depository receipts in respect thereof, which shares of stock (or depository receipts in respect thereof) or depository receipts at the effective date of the merger or consolidation will be either listed on a national securities exchange or designated as a national market system security on an interdealer quotation system by the National Association of Securities Dealers, Inc. or held of record by more than 2,000 holders;
 - c. Cash in lieu of fractional shares or fractional depository receipts described in the foregoing subparagraphs a. and b. of this paragraph; or
 - d. Any combination of the shares of stock, depository receipts and cash in lieu of fractional shares or fractional depository receipts described in the foregoing subparagraphs a., b. and c. of this paragraph.

E-1

Table of Contents

(c) Any corporation may provide in its certificate of incorporation that appraisal rights under this section shall be available for the shares of any class or series of its stock as a result of an amendment to its certificate of incorporation, any merger or consolidation in which the corporation is a constituent corporation or the sale of all or substantially all of the assets of the corporation. If the certificate of incorporation contains such a provision, the procedures of this section, including those set forth in subsections (d) and (e) of this section, shall apply as nearly as is practicable.

(d) Appraisal rights shall be perfected as follows:

Table of Contents

- (e) Within 120 days after the effective date of the merger or consolidation, the surviving or resulting corporation or any stockholder who has complied with subsections (a) and (d) hereof and who is otherwise entitled to appraisal rights, may file a petition in the Court of Chancery demanding a determination of the value of the stock of all such stockholders. Notwithstanding the foregoing, at any time within 60 days after the effective date of the merger or consolidation, any stockholder shall have the right to withdraw such stockholder's demand for appraisal and to accept the terms offered upon the merger or consolidation. Within 120 days after the effective date of the merger or consolidation, any stockholder who has complied with the requirements of subsections (a) and (d) hereof, upon written request, shall be entitled to receive from the corporation surviving the merger or resulting from the consolidation a statement setting forth the aggregate number of shares not voted in favor of the merger or consolidation and with respect to which demands for appraisal have been received and the aggregate number of holders of such shares. Such written statement shall be mailed to the stockholder within 10 days after such stockholder's written request for such a statement is received by the surviving or resulting corporation or within 10 days after expiration of the period for delivery of demands for appraisal under subsection (d) hereof, whichever is later.
- (f) Upon the filing of any such petition by a stockholder, service of a copy thereof shall be made upon the surviving or resulting corporation, which shall within 20 days after such service file in the office of the Register in Chancery in which the petition was filed a duly verified list containing the names and addresses of all stockholders who have demanded payment for their shares and with whom agreements as to the value of their shares have not been reached by the surviving or resulting corporation. If the petition shall be filed by the surviving or resulting corporation, the petition shall be accompanied by such a duly verified list. The Register in Chancery, if so ordered by the Court, shall give notice of the time and place fixed for the hearing of such petition by registered or certified mail to the surviving or resulting corporation and to the stockholders shown on the list at the addresses therein stated. Such notice shall also be given by 1 or more publications at least 1 week before the day of the hearing, in a newspaper of general circulation published in the City of Wilmington, Delaware or such publication as the Court deems advisable. The forms of the notices by mail and by publication shall be approved by the Court, and the costs thereof shall be borne by the surviving or resulting corporation.
- (g) At the hearing on such petition, the Court shall determine the stockholders who have complied with this section and who have become entitled to appraisal rights. The Court may require the stockholders who have demanded an appraisal for their shares and who hold stock represented by certificates to submit their certificates of stock to the Register in Chancery for notation thereon of the pendency of the appraisal proceedings; and if any stockholder fails to comply with such direction, the

- (h) After determining the stockholders entitled to an appraisal, the Court shall appraise the shares, determining their fair value exclusive of any element of value arising from the accomplishment or expectation of the merger or consolidation, together with a fair rate of interest, if any, to be paid upon the amount determined to be the fair value. In determining such fair value, the Court shall take into account all relevant factors. In determining the fair rate of interest, the Court may consider all relevant factors, including the rate of interest which the surviving or resulting corporation would have had to pay to borrow money during the pendency of the proceeding. Upon application by the surviving or resulting corporation or by any stockholder entitled to participate in the appraisal proceeding, the Court may, in its discretion, permit discovery or other pretrial proceedings and may proceed to trial upon the appraisal prior to the final determination of the stockholder entitled to an appraisal. Any stockholder whose name appears on the list filed by the surviving or resulting corporation pursuant to subsection (f) of this section and who has submitted such stockholder's certificates of stock to the Register in Chancery, if such is required, may participate fully in all proceedings until it is finally determined that such stockholder is not entitled to appraisal rights under this section.
- (i) The Court shall direct the payment of the fair value of the shares, together with interest, if any, by the surviving or resulting corporation to the stockholders entitled thereto. Interest may be simple or compound, as the Court may direct. Payment shall be so made to each such stockholder, in the case of holders of uncertificated stock forthwith, and the case of holders of shares represented by certificates upon the surrender to the corporation of the certificates representing such stock. The Court's decree may be enforced as

F-3

Table of Contents

other decrees in the Court of Chancery may be enforced, whether such surviving or resulting corporation be a corporation of this State or of any state.

- (j) The costs of the proceeding may be determined by the Court and taxed upon the parties as the Court deems equitable in the circumstances. Upon application of a stockholder, the Court may order all or a portion of the expenses incurred by any stockholder in connection with the appraisal proceeding, including, without limitation, reasonable attorney's fees and the fees and expenses of experts, to be charged pro rata against the value of all the shares entitled to an appraisal.
- (k) From and after the effective date of the merger or consolidation, no stockholder who has demanded appraisal rights as provided in subsection (d) of this section shall be entitled to vote such stock for any purpose or to receive payment of dividends or other distributions on the stock (except dividends or other distributions payable to stockholders of record at a date which is prior to the effective date of the merger or consolidation); provided, however, that if no petition for an appraisal shall be filed within the time provided in subsection (e) of this section, or if such stockholder shall deliver to the surviving or resulting corporation a written withdrawal of such stockholder's demand for an appraisal and

an acceptance of the merger or consolidation, either within 60 days after the effective date of the merger or consolidation as provided in subsection (e) of this section or thereafter with the written approval of the corporation, then the right of such stockholder to an appraisal shall cease. Notwithstanding the foregoing, no appraisal proceeding in the Court of Chancery shall be dismissed as to any stockholder without the approval of the Court, and such approval may be conditioned upon such terms as the Court deems just.

(1) The shares of the surviving or resulting corporation to which the shares of such objecting stockholders would have been converted had they assented to the merger or consolidation shall have the status of authorized and unissued shares of the surviving or resulting corporation.

E-4

Table of Contents

APPENDIX F

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-K/A

(Amendment No. 1)

(Mark One)

- (1) If a proposed merger or consolidation for which appraisal rights are provided under this section is to be submitted for approval at a meeting of stockholders, the corporation, not less than 20 days prior to the meeting, shall notify each of its stockholders who was such on the record date for such meeting with respect to shares for which appraisal rights are available pursuant to subsection (b) or (c) hereof that appraisal rights are available for any or all of the shares of the constituent corporations, and shall include in such notice a copy of this section. Each stockholder electing to demand the appraisal of such stockholder's shares shall deliver to the corporation, before the taking of the vote on the merger or consolidation, a written demand for appraisal of such stockholder's shares. Such demand will be sufficient if it reasonably informs the corporation of the identity of the stockholder and that the stockholder intends thereby to demand the appraisal of such stockholder's shares. A proxy or vote against the merger or consolidation shall not constitute such a demand. A stockholder electing to take such action must do so by a separate written demand as herein provided. Within 10 days after the effective date of such merger or consolidation, the surviving or resulting corporation shall notify each stockholder of each constituent corporation who has complied with this subsection and has not voted in favor of or consented to the merger or consolidation of the date that the merger or consolidation has become effective; or
- (2) If the merger or consolidation was approved pursuant to § 228 or § 253 of this title, each constituent corporation, either before the effective date of the merger or consolidation or within ten days thereafter, shall notify each of the holders of any class or series of stock of such constituent

corporation who are entitled to appraisal rights of the approval of the merger or consolidation and that appraisal rights are available for any or all shares of such class or series of stock of such constituent corporation, and shall include in such notice a copy of this section; provided that, if the notice is given on or after the effective date of the merger or consolidation, such notice shall be given by the surviving or resulting corporation to all such holders of any class or series of stock of a constituent corporation that are entitled to appraisal rights. Such notice may, and, if given on or after the effective date of the merger or consolidation, shall, also notify such stockholders of the effective date of the merger or consolidation. Any stockholder entitled to appraisal rights may, within 20 days after the date of mailing of such notice, demand in writing from the surviving or resulting corporation the appraisal of such holder's shares. Such demand will be sufficient if it reasonably informs the corporation of the identity of the stockholder and that the stockholder intends thereby to demand the appraisal of such holder's shares. If such notice did not notify stockholders of the effective date of the merger or consolidation, either (i) each such constituent corporation shall send a second notice before the effective date of the merger or consolidation notifying each of the holders of any class or series of stock of such constituent corporation that are entitled to appraisal rights of the effective date of the merger or consolidation or (ii) the surviving or resulting corporation shall send such a second notice to all such holders on or within 10 days after such effective date; provided, however, that if such second notice is sent more than 20 days following the sending of the first notice, such second notice need only be sent to each stockholder who is entitled to appraisal rights and who has demanded appraisal of such holder's shares in accordance with this subsection. An affidavit of the secretary or assistant secretary or of the transfer agent of the corporation that is required to give either notice that such notice has been given shall, in the absence of fraud, be prima facie evidence of the facts stated therein. For purposes of determining the stockholders entitled to receive either notice, each constituent corporation may fix, in advance, a record date that shall be not more than 10 days prior to the date the notice is given, provided, that if the notice is given on or after the effective date of the merger or consolidation, the record date shall be such effective date. If no record date is fixed and the notice is given prior to the effective date, the record date shall be the close of business on the day next preceding the day on which the notice is given.

For the Fiscal Year Ended December 31, 1999

[X] ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from to

Commission File Number 000-11560

Savoir Technology Group, Inc.

(Exact name of registrant as specified in its charter)

]TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Registrant's telephone number, including area code: (408) 379-0177

Securities registered pursuant to Section 12(b) of the Act: None

Securities registered pursuant to Section 12(g) of the

Delaware (State or other jurisdiction of incorporation or organization)

94-2414428 (I.R.S. Employer Identification No.)

44931 Industrial Boulevard, Fremont, CA (Address of principal executive offices)

94538 (Zip Code) all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes [X] No [].

Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K (§229.405 of this chapter) is not contained herein, and will not be contained, to the best of registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10-K. []

The aggregate market value of the registrant's voting stock held by non-affiliates on February 24, 2000, based on the last reported sales price for the registrant's Common Stock on the Nasdaq National Market on such date, was approximately \$34,265,881. For purposes of the foregoing calculation only, the registrant has included in the shares owned by affiliates the beneficial ownership of Common Stock of officers and directors of the registrant and members of their families. Such inclusion shall not be construed as an admission that any such person is an affiliate for any other purpose.

As of February 24, 2000, there were 13,489,569 outstanding shares of Common Stock, \$0.01 par value.

F-1

SAVOIR TECHNOLOGY GROUP, INC.

TABLE OF CONTENTS

TABLE OF CONTENTS

Title of each class

Name of each exchange on which registered

Common Stock, par value \$0.01 per share

Nasdaq National Market

1999 FORM 10-K

PART I

<u>Item 1. Bu</u>siness

Item 2. Properties

<u>Item 3. Legal Proceedings</u>

Item 4. Submission of Matters to a Vote of Security Holders

PART II

Item 5. Market for Registrant's Common Stock and Related Stockholder Matters

Item 6. Selected Consolidated Financial Data

Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations

<u>Item 7A. Quantitative and Qualitative Disclosures About Market Risk</u>

Item 8. Financial Statements and Supplementary Data
Item 9. Changes in and Disagreements With Accountants on Accounting and Financial Disclosure
PART III

<u>Item</u> 10. <u>Directors and Executive Officers of the Registrant</u>

Item 11. Executive Compensation

Item 12. Security Ownership of Certain Beneficial Owners and Management

Item 13. Certain Relationships and Related Transactions

Item 14. Exhibits, Financial Statement Schedules, and Reports on Form 8-K

This Report contains forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended (the "Securities Act"), and Section 21E of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), regarding future events and the Company's plans and expectations that involve risks and uncertainties. When used in this Report, the words "estimate," "project," "intend," "expect" and "anticipate" and similar expressions are intended to identify such forward-looking statements. Such statements are subject to certain risks and uncertainties, including those discussed below, that could cause actual results to differ materially from those projected. Factors that may cause or contribute to such differences include, but are not limited to, those discussed below under "Risk Factors," as well as those discussed elsewhere in this Report and in the documents incorporated herein by reference. In light of the important factors that can materially affect results, including those set forth in this paragraph and below, the inclusion of forward-looking information herein should not be regarded as a representation by the Company or any other person that the objectives or plans for the Company will be achieved. The reader is therefore cautioned not to place undue reliance on the forward-looking statements contained herein, which speak only as of the date hereof. Savoir Technology Group, Inc. (the "Company" or "Savoir") undertakes no obligation to publicly release updates or revisions to these statements.

F-2

Table of Contents

PART I

Item 1. Business

Except for historical information contained herein, the following discussion contains forward-looking statements that involve risks and uncertainties. Such forward-looking statements include, but are not limited to, statements regarding future events and our plans and expectations. Our actual results could differ materially from those discussed herein. Factors that could cause or contribute to such differences include, but are not limited to, those discussed below in "RISK FACTORS" as well as those discussed elsewhere in this Report.

Company Overview

We were incorporated in California in 1975 as Silicon Valley Services, Inc. and were renamed Western Micro Technology, Inc. in April 1977. In August 1997, we reincorporated in Delaware. We changed our name to Savoir Technology Group, Inc. in We are a value-added wholesale distributor of commercial mid-range servers, peripheral equipment (including wireless networking equipment, storage products, printers and terminals) and software. We believe that we are one of the leading distributors of IBM's commercial mid-range servers product lines (AS/400 and RS/6000). We also distribute commercial mid-range servers, peripheral equipment and software manufactured by NCR, Telxon, Aironet and the Santa Cruz Operation ("SCO"). We primarily distribute commercial mid-range servers and related products to value-added resellers ("VARs") who generally combine our products with commercial applications software and sell integrated computer systems to end user customers. We also integrate and configure personal computers, workstations and departmental servers for original equipment manufacturers ("OEMs") and VARs and provide and remarket installation and technical support services. Through twelve acquisitions since December 1994 and internal growth, we have expanded our products and services, increased our geographic market coverage, strengthened our management and technical personnel and increased our operating leverage. As a result, our net sales increased from \$131.7 million in 1996 to \$767.2 million in 1999.

On March 2, 2000 we agreed to merge with a subsidiary of Avnet, Inc., so that we would become a wholly owned subsidiary of Avnet. The acquisition has been approved by the boards of directors of both entities and is subject to approval by the stockholders of Savoir and regulatory review. If the merger is completed, depending upon the average of the closing trading price of Avnet's common stock during the 15 days ending on the fifth day prior to the vote of our stockholders, our common stockholders will receive between 0.15494 and 0.11452 of a share of Avnet common stock for each share of Savoir stock they own. The transaction is expected to close in June. If the merger is not completed, we may be required to pay Avnet a termination fee of \$750,000, and we may be required to pay Avnet an additional fee of \$3.75 million if we enter into an acquisition agreement with another company within a year of termination of the Avnet merger agreement. Avnet, Inc. is a Phoenix, Arizona based distributor of semiconductors, interconnect, passive and electromechanical components and computer products. At its last fiscal year end, Avnet reported sales of approximately \$6.35 billion.

Products and Vendors

Two of our business divisions, the Mid-Range Systems Division and the Computer and Peripherals Group, generate most of our net sales.

Mid-Range Systems Division. The distribution of commercial mid-range servers in 1999 accounted for approximately 86% of our total net sales. The principal goal of our commercial mid-range systems distribution business is to provide customers with both rapid, accurate delivery of products and quality configuration and technical support. Products distributed by this Division include mid-range servers which run on Unix, OS/400 and NT operating systems, peripheral equipment (including wireless networking equipment, storage products, printers and terminals) and software. In addition to selling new equipment, we also distribute refurbished IBM AS/400 and RS/6000 equipment. Our Mid-Range Systems Division represents four major manufacturers: IBM, NCR, Aironet and Telxon. During the years ended December 31, 1997, 1998, and 1999, approximately

65%, 80% and 82%, respectively, of our net sales was generated from the sale of IBM products (including mid-range and peripheral equipment). Our mid-range IBM product line includes the AS/400 and RS/6000 families of mid-range servers. We have received the IBM business partner leadership award for the past five six-month periods. This award is given by IBM to its top distributor in North America once every six months. Additionally, IBM has named Savoir as one of only five certified e-business partners (and the only distributor) in North America. We are also one of several distributors to qualify for IBM's Authorized Assembly Program ("AAP"). The AAP certification allows us to utilize our integration facilities to assemble custom RS/6000 configurations, allowing for shorter delivery times to customers and a reduction in required inventory levels of pre-configured systems.

Computer and Peripherals Group. We offer OEMs and departmental server customers a single source for their hardware, software and service needs through our Computer and Peripherals Group ("CPG"). CPG accounted for approximately 14% of our total net sales in 1999. Through CPG, we offer our customers a wide variety of value-added systems integration services up to, and including, the actual installation at the end user site (e.g., "turnkey" systems assembly of departmental servers, workstations, hardware and software "bundling" and light manufacturing). CPG's more advanced products include fault tolerant software, serial port expansion devices and disk striping and mirroring solutions for the SCO/ UNIX operating environments. Products assembled and manufactured by CPG include special purpose PC-based subcomponents of larger systems, private-label departmental and small enterprise servers and related peripherals. CPG purchases components manufactured by Intel Corporation, IBM System Storage Division, Sony Electronics, Inc., SCO, Hitachi America, Ltd. and Microsoft Corporation, among others. Through CPG, we also specialize in building systems under long-term contracts for customers seeking fully compatible configurations that remain consistent over time.

Value-Added Services

In addition to the products we offer, we also provide a variety of value-added services, including the following:

Item No.		Page
	PART I	
1.	Business	F-3
2.	Properties	F-16
3.	Legal Proceedings	F-17
4.	Submission of Matters to a Vote of Security Holders	F-17
	PART II	
5.	Market for the Registrant's Common Stock and Related Stockholder Matters	F-18
6.	Selected Consolidated Financial Data	F-19
7.	Management's Discussion and Analysis of Financial Condition and Results of	
	Operations	F-20
7A.	Quantitative and Qualitative Disclosures About Market Risk	F-27
8.	Financial Statements and Supplementary Data	F-28
9.	Changes In and Disagreements With Accountants on Accounting and Financial	
	Disclosure	F-50
	PART III	
10.	Directors and Executive Officers of the Registrant	F-50
11.	Executive Compensation	F-52
12.	Security Ownership of Certain Beneficial Owners and Management	F-56
13.	Certain Relationships and Related Transactions	F-58
	PART IV	
14.	Exhibits, Financial Statement Schedules and Reports on Form 8-K	F-59

Integration Services. We perform light manufacturing or technical integration services, ranging from simple hardware and software integration, burn-in and testing to building customized systems to our customer's specifications.

Technical Support Services. We currently offer our customers pre-sale technical assistance, configuration review and verification, consulting services, network design, implementation and installation services and site planning, telephone support and help desk, patch/bug isolation and identification, certification requirements and preparation and system administration assistance. We also remarket certain vendor maintenance and consulting services and reseller training programs.

Logistical and Inventory Management Services. We offer ordering and purchasing services, including order acknowledgment, order management, contract purchasing and end-of-life buy programs. We also offer inventory services such as expedited delivery, kitting and bill-of-material services, warehousing and storage services, bonded inventory programs, consignment programs and customer on-site operations. In addition, we offer various delivery options and services, including drop shipments, blind shipments, custom packaging, consolidated shipping services, special handling services, personnel services and exporting assistance.

Marketing Services. We make current and updated information on our products and services available to our customers through fax broadcast services and our web site. We also customize and provide Internet web sites for certain of our customers. In addition, we offer ready to execute demand generation campaigns, assistance with such campaigns, assistance with organizing advertising campaigns and joint marketing funds.

Financing, Credit and Leasing Services. We offer our customers various financing and credit options, including open account terms, electronic funds transfer, standby letters of credit, security interest/ UCC filings, personal guarantees and end user lock box services. We also offer end user financing

Customers

We currently have approximately 2,600 active customer accounts. One of our customers, Sirius Computer Solutions, Ltd. ("Sirius"), accounted for approximately 11%, 18% and 16% of our net sales in 1997, 1998 and 1999, respectively. No other single customer accounted for more than 10% of our net sales. Our sales to Sirius are made under the Industry Remarketer Affiliate Agreement between the Company and Sirius dated as of September 30, 1997, as amended by Amendments Nos. 1 and 2 thereto, dated December 31, 1998 and November 30, 1999, respectively (the "Sirius Agreement"), pursuant to which we appointed Sirius as one of our industry remarketer affiliates of IBM products. The Sirius Agreement provides that Sirius may not enter into any similar arrangement with any third party for the purpose of selling IBM products to its end user customers and also provides a favorable pricing structure to Sirius. As a result, Sirius is expected to remain our largest customer for the duration of the Sirius Agreement and to account for approximately the same percentage of our net sales in 2000 as it represented in 1999. The Sirius Agreement expires on December 31, 2000, but may be terminated earlier under certain conditions, not including termination at will.

We divide our significant customers into the following three broad categories:

programs through third parties, including leasing programs, joint purchase orders, payment agreements and inventory financing programs.

Sales and Marketing

We focus on selling and marketing high-quality commercial mid-range servers and integrated computer system products from a relatively small number of vendors. In general, sales, sales support and product management organizations are organized by vendor into business units that sell and support only products

offered by that particular vendor. We believe that our customers require ongoing support from technically trained sales professionals who (1) are dedicated to certain vendors, and, in certain instances, to a particular product line, and (2) can provide technical support on the increasingly complex mid-range servers and systems we offer our customers.

Our sales professionals require the technical expertise to work with customers and our mid-range product purchasing specialists to provide the computer system solutions required by our customers and, ultimately the end user. Our sales professionals regularly participate in vendor-sponsored training and certification programs. Within the mid-range distribution business units, we maintain a salesperson to technical support person ratio of approximately 4 to 1. We utilize directed telemarketing programs, maintain a database of current and

F-5

Table of Contents

potential customers, participate in cooperative advertising with vendors, participate in trade shows and advisory councils and utilize print media as part of our sales and marketing efforts.

Our CPG sells primarily to OEMs. Technical expertise within CPG's sales force is critical during the relatively long sales cycles required to develop new commercial products. Once the products are developed, the sales force and technical support personnel must carefully manage and review the ongoing forecasting, manufacture, delivery and installation of these systems. Within CPG, we maintain a salesperson to technical support person ratio of approximately 2 to 1. Due to the complex nature of the products offered by CPG, new customers are primarily solicited using targeted print advertising and customer referrals.

As of December 31, 1999, we had approximately 140 direct sales personnel. We have a North American presence served by sales offices in Campbell and Lake Forest, California, Chicago, Illinois, Boston, Massachusetts, Huntsville, Alabama, Atlanta, Georgia, Monterrey and Mexico City, Mexico, San Antonio, Texas and Ontario, Canada. Centralized marketing departments located in Campbell and San Antonio support our sales offices. We generally compensate our sales and marketing personnel based on attainment of specified gross profit targets.

Operations and Infrastructure

Information Systems

Our corporate information system is a scalable, centralized processing system capable of supporting numerous operational functions, including purchasing, receiving, order processing, shipping, inventory management, sales analysis and accounting. Our customers and sales representatives rely on the information system for on-line, real-time information on product pricing, inventory availability and order status. The fully integrated modular system provides customers and sales representatives on-line access to the status of the backlog of shipments we expect to receive, thereby significantly reducing back office telephone investigation time. After product pricing and availability have been determined, the integrated order entry

system automatically places an order for shipment or, if necessary, allocates the inventory to the assembly operations. The system then instructs warehouse personnel to pull products for shipment and directs them to the location of the inventory. In order to optimize the use of warehouse space, we use a random access system whereby inventory is stored in the first available location within the warehouse.

Inventory Control

For both the Mid-Range Systems Division and CPG, our computer system automatically determines price and availability of inventory and can allocate inventory to bills of material. This computer system manages all of our inventories at all of our locations throughout the United States. Inventories are overseen by a dedicated group of product specialists, assigned by product line, whose responsibility it is to appropriately manage inventory levels and turnover.

Warehouse and Integration Facilities; Shipping

We maintain inventory stocking locations in Lake Forest and Fremont, California, Atlanta, Georgia, Chicago, Illinois, Boston, Massachusetts, and San Antonio, Texas. In addition, we have a major integration facility in Fremont, California, adjacent to our warehouse, with other integration facilities in Atlanta, Lake Forest, Chicago and San Antonio. Our Fremont integration facility is ISO 9002 certified. We currently ship products from our warehouses via FedEx, UPS and other common carriers. In addition, we distribute a significant portion of our products by having our vendors drop-ship the products to our customers.

Financial Services

We offer a number of flexible leasing and financing alternatives to our customers, including a variety of leasing options, inventory flooring options and end user lock-box arrangements. We also maintain credit insurance in order to more effectively manage the risk of extending credit to our customers.

F-6

Table of Contents

Competition

The markets in which we operate are highly competitive. Competition is based primarily on product availability, price, credit availability, speed of delivery, ability to tailor specific solutions to customer needs, breadth and depth of product lines and services, technical expertise and pre- and post-sale service and support. Increased competition may result in further price reductions, reduced gross profit margins and loss of market share, any of which could materially and adversely affect our business, financial condition and results of operations.

Through the Mid-Range Systems Division, we compete with national, regional and local distributors, including, but not limited to, Gates/ Arrow Commercial Systems, a division of Arrow Electronics, Inc., Hall-Mark Global Solutions, a subsidiary of Avnet, Inc.,

and Pioneer Standard Electronics, Inc. In some limited circumstances, we compete with our own vendors. In the distribution of storage products, we compete with national, regional and local distributors. Through CPG, we compete with contract manufacturers, systems integrators and certain assemblers of computer products. We have experienced, and expect to continue to experience, increased competition from current and potential competitors, many of which have substantially greater financial, technical, sales, marketing and other resources, as well as greater name recognition and a larger customer base than we do. Accordingly, such competitors or future competitors may be able to respond more quickly to new or emerging technologies and changes in customer requirements or to devote greater resources to the development, promotion and sale of their products than we can. Competitors which are larger than us may be able to obtain more favorable pricing and terms from vendors than we can. As a result, we may be at a disadvantage when competing with these larger companies. If we fail to compete effectively, our business, financial condition and results of operations would be materially and adversely affected.

Employees

As of February 24, 2000, we had approximately 585 full-time employees. We are not a party to any collective bargaining agreement and consider our employee relations to be good.

Risk Factors

Risks Related to the Proposed Acquisition of Us by Avnet

The expected benefits of combining Avnet and Savoir may not be realized

We entered into the merger agreement with Avnet with the expectation that the merger will result in benefits to the combined companies, including reduced costs as a percentage of sales due to increased economies of scale, greater access to capital, stronger buying power with our vendors and broader offerings of products to our customers. If we are not able to integrate effectively our technology, operations and personnel in a timely and efficient manner, then the benefits of the merger will not be realized, and, as a result, the combined companies' operating results and the market price of Avnet's common stock may be adversely affected. In particular, if the integration is not successful:

Value-Added Resellers. VARs typically install their own or other vendors' software, configure completed systems and integrate their service offerings with hardware. For example, Sirius purchases IBM mid-range servers from us and typically bundles these servers with software provided by J.D. Edwards. As a result, Sirius is usually able to provide its end user customers with a complete turnkey computer systems package. Similarly, NxTrend, Inc. purchases IBM mid-range servers from us and sells the servers to its end user customers combined with its own proprietary distribution applications software.

OEMs. These manufacturers similarly integrate or have us integrate our products with their own products prior to distribution to their end user customers. One example of our OEM customers is eShare Technologies ("eShare"), a provider of customer contact and telephone call management systems. eShare uses our capabilities to configure its proprietary software on a preconfigured system which can be shipped directly to eShare's customer. In addition, customers such as Divicom, Inc. and Wang Laboratories, Inc. have found it more efficient to outsource certain specialized products to us as opposed to creating their own internal infrastructure.

Systems Integrators. Systems integrators focus on delivering non-industry specific solutions to the end user customer. The solutions may include electronic commerce, networking, Intranet/ Internet configurations and application-specific solutions. For example, Innovative Business Systems, Inc. designs and installs network systems solutions in a variety of application environments that incorporate commercial mid-range servers purchased from Savoir.

In addition, the attention and effort devoted to the integration of the two companies will significantly divert management's

attention from other important issues, and could significantly harm the combined companies' business and operating results.

Failure to complete the proposed merger could harm our stock price and future business and operations

If the merger is not completed for any reason, we may be subject to a number of material risks, including the following:

- we may lose key personnel; and
- we may not be able to retain our customers.

F-7

Table of Contents

• we may be required to pay Avnet a termination fee of \$750,000, and we may be required to pay Avnet an additional fee of \$3.75 million if we enter into an acquisition agreement with another company within a year of termination of the Avnet merger agreement;

In addition, our customers may, in response to the announcement of the merger, delay or defer purchasing decisions. Any delay or deferral in purchasing decisions by our customers could materially and adversely affect our business, regardless of whether the merger is ultimately completed. Similarly, our current and prospective employees may experience uncertainty about their future role with Avnet until Avnet's strategies with regard to Savoir are announced or executed. This may adversely affect our ability to attract and retain key management, marketing and technical personnel.

Further, if the merger is terminated and our board of directors determines to seek another merger or business combination, we may be unable to find a partner willing to pay an equivalent or more attractive price than the price proposed to be paid in the merger. In addition, while the merger agreement is in effect and subject to limited exceptions, we are prohibited from soliciting, initiating or inducing or entering into certain extraordinary transactions, such as a merger, sale of assets or other business combination, with any party other than Avnet. We granted Avnet an option to purchase shares of our common stock, and if the merger agreement is terminated, because of that option, we may not be able to account for future transactions as a "pooling of interests."

Risks Related to Our Business

The following risk factors include risks related to our business without taking into account the proposed merger. Risks such as those relating to strategic acquisitions may not apply if the proposed merger is completed.

We are dependent upon IBM as our principal vendor

Our business, financial condition and results of operations are

highly dependent upon our relationship with International Business Machines Corporation ("IBM") and upon the continued market acceptance of IBM commercial mid-range servers, storage products and other peripheral equipment. During the years ended December 31, 1997, 1998 and 1999, approximately 65%, 80% and 82%, respectively, of our net sales were generated from the sale of IBM products and we expect this percentage to remain constant in 2000. Our agreement with IBM is non-exclusive and may be unilaterally modified by IBM upon 30 days' written notice, renews automatically but may be terminated by IBM upon written notice given not less than 90 days prior to the renewal date (January 1, 2001), provides no franchise rights and may not be assigned by us. The continued consolidation of wholesale distributors of commercial mid-range servers may also result in IBM raising the sales volume threshold required to maintain most favorable volume discount status. As part of our business strategy, and in order to maintain most favorable volume discount status with IBM, we have completed several acquisitions and we are actively engaged in an ongoing search for additional acquisitions. We are also seeking to make minority equity investments in potential large customers for similar purposes. However, we cannot assure you that we will be successful in completing any future acquisitions or in making any equity investments. If we are unable to complete other acquisitions or make equity investments, or are otherwise unable to increase our sales volume through internal growth, we could lose our most favorable volume discount status with IBM, which would, in turn, have a material adverse effect on our relationship with IBM and on our business, financial condition and results of operations. The occurrence of any of the following events could have a material adverse effect upon our business, financial condition and results of operations:

- the price of our common stock may decline to the extent that the current market price of our common stock reflects a market assumption that the merger will be completed; and
- costs related to the merger, such as legal, accounting and financial advisor fees, must be paid even if the merger is not completed.

F-8

Table of Contents

We receive market development funds and other incentives from IBM. These market development and incentive funds directly affect our gross profit and our sales and marketing expenses. Any change in the availability of these market development and incentive funds would have a material adverse effect on our business, financial condition and results of operations.

Our operating results may fluctuate from quarter to quarter

Our quarterly net sales and operating results may vary significantly as a result of a variety of factors, including, but not limited to:

- any disruption, change or termination in our relationship with IBM or in the manner in which IBM distributes its products;
- the failure of IBM to develop new products which are accepted by our customers;
- our failure to continue to achieve sufficient sales volumes of certain IBM products or to maintain the required infrastructure, in each case as required to maintain most favorable volume discount status;

• the addition of other wholesale distributors by IBM.

Our operating results could also be adversely affected by:

- changes in the supply and demand for commercial mid-range servers, peripheral equipment, software and related services;
- the cost, timing and integration of acquisitions;
- the addition or loss of a key vendor or customer;
- · the introduction of new technologies;
- · changes in manufacturers' prices, price protection policies or stock rotation (return) privileges;
- changes in market development or other promotional funds;
- · product supply shortages;
- · disruption of warehousing or shipping channels;
- inventory adjustments;
- increases in the amount of accounts receivable written off;
- price competition; and
- changes in the mix of products sold through distribution channels and in the mix of products purchased by OEMs.

In addition, historically a substantial portion of our net sales has been made in the last few days of a quarter. Our quarterly operating results are therefore difficult to predict and delays in the closing of sales near the end of a quarter could cause quarterly net sales to fall substantially short of anticipated levels and, to a greater degree, adversely affect profitability. Thus, we believe that period-to-period comparisons of our operating results are not necessarily meaningful and should not be relied upon as an indication of our future performance. Our future operating results are expected to fluctuate as a result of these and other factors, which could have a material adverse effect on our business, financial condition and results of operations and on the price of our common stock. It is possible that in future periods our operating results may be below the expectations of securities analysts and investors. If this happens, it is likely that the market price of our common stock would be materially and adversely affected.

We face substantial competition

The markets in which we operate are highly competitive. Competition is based primarily on:

- general economic and other conditions affecting the timing of customer orders and capital spending;
- a downturn in the market for commercial mid-range servers; and
- order cancellations or rescheduling.

- product availability;
- price;
- credit availability;

Increased competition may result in further price reductions, reduced gross profit margins and loss of market share, any of which could materially and adversely affect our business, financial condition and results of operations.

Through our Mid-Range Systems Division, we compete with national, regional and local distributors, including Gates/ Arrow Commercial Systems, a division of Arrow Electronics, Inc., Hall-Mark Global Solutions, a subsidiary of Avnet, Inc., and Pioneer Standard Electronics, Inc. In some limited circumstances, we also compete with our own vendors. In the distribution of storage products, we compete with national, regional and local distributors. Through our Computers and Peripherals Group, we compete with contract manufacturers, systems integrators and assemblers of computer products.

We have experienced, and expect to continue to experience, increased competition from current and potential competitors, many of which have substantially greater financial, technical, sales, marketing and other resources, as well as greater name recognition and a larger customer base than we do. Accordingly, present or future competitors may be able to respond more quickly to new or emerging technologies and changes in customer requirements or to devote greater resources to the development, promotion and sale of their products than we can. Competitors that are larger than Savoir may be able to obtain more favorable pricing and terms from vendors than we can. As a result, we may be at a disadvantage when competing with these larger companies. If we fail to compete effectively, our business, financial condition and results of operations would be materially and adversely affected.

Sirius Computer Solutions, Ltd. accounts for 16% of our net sales

During the years ended December 31, 1997, 1998 and 1999, sales to Sirius Computer Solutions, Ltd. accounted for approximately 11%, 18% and 16%, respectively, of our net sales. Our sales to Sirius are made under the Industry Remarketer Affiliate Agreement between Savoir and Sirius dated as of September 30, 1997, under which we appointed Sirius as one of our industry remarketer affiliates for IBM products. This agreement provides that Sirius may not enter into any similar arrangement with any third party for the purpose of selling IBM products to its end user customers and also provides a favorable pricing structure to Sirius. As a result, Sirius is expected to remain our largest customer for the duration of this agreement and to account for approximately the same percentage of our net sales in 2000 as it represented in 1999. The agreement with Sirius expires on December 31, 2000, but may be terminated earlier upon the happening of specified events. This agreement may not be unilaterally terminated by either Savoir or Sirius. Any disruption, change or termination of our relationship with Sirius or a reduction in Sirius's purchases from us could have a material adverse effect upon our business, financial condition and results of operations.

Integration of acquired companies and our business may not be successful

Since December 1994, we have completed twelve acquisitions.

The combination of our business and acquired businesses requires, among other things:

- speed of delivery;
- ability to tailor specific solutions to customer needs; and
- breadth and depth of product lines and services, technical expertise and pre-sale and post-sale service and support.

The difficulties of such integration may be increased by the necessity of coordinating geographically separate organizations. The integration of operations will require the dedication of management resources which may temporarily divert attention away from the day-to-day business of the combined company. We cannot assure

F-10

Table of Contents

you that the required coordination and integration will be accomplished smoothly or successfully. Our inability to integrate successfully the operations of acquired businesses could have a material adverse effect on our business, financial condition and results of operations. In addition, during the integration phase, aggressive competitors may attempt to attract our customers and recruit our key employees. We cannot assure you that acquisitions will not materially and adversely affect the selling patterns of vendors and the buying patterns of our present and potential customers, and that any change in these patterns will not materially and adversely affect our business, financial condition and results of operations.

Our ability to achieve the anticipated benefits of our acquisitions depends in part upon whether the integration of our business and any acquired business is accomplished in an efficient and effective manner, and we cannot assure you that this will occur. Our previous acquisitions and investments have placed and will, together with future acquisitions, continue to place, substantial demands on our management team and financial resources. The integration of the operations of acquired companies has on occasion been slower, more complex and more costly than we originally anticipated. We will encounter similar uncertainties and risks in any future acquisitions and investments. Although we expect to realize cost savings and sales enhancements as a result of the recent and proposed acquisitions, we cannot assure you that these savings or enhancements will be realized in full or when anticipated, or that any cost savings will not be offset by increases in other expenses.

- integration of the respective management teams and sales and other personnel;
- coordination of sales and marketing efforts;
- conversion of computer systems (including inventory control, order entry and financial reporting); and
- integration of the businesses' products and physical facilities.

Acquisitions have played an important role in the implementation of our business strategy, and we believe that additional acquisitions are important to our growth, development and continued ability to compete effectively in the marketplace. We

evaluate potential acquisitions and strategic investments on an ongoing basis. We cannot assure you as to our ability to compete successfully for available acquisition or investment candidates or to complete future acquisitions and investments or as to the financial effect on us of any acquired businesses or equity investments. Any future acquisitions and investments we might make may involve significant cash expenditures and may result in increased indebtedness, interest and amortization expense or decreased operating income, any of which could have a material adverse effect on our business, financial condition and results of operations. In addition, future growth will require additional financing to fund the working capital requirements of our business and to finance future acquisitions and strategic equity investments, if any. We cannot assure you that we will be able to raise financing on satisfactory terms and conditions, if at all. Should we be unable to implement successfully our acquisition and investment strategy, our business, financial condition and results of operations could be materially and adversely affected.

We may have difficulty in managing our growth

Since 1997, we have experienced significant growth in the number of our employees and in the scope of our operating and financial systems, resulting in increased responsibilities for our management. To manage future growth effectively, we will need to continue to improve our operational, financial and management information systems, procedures and controls and expand, train, motivate, retain and manage our employee base. We cannot assure you that we will be successful in managing any future expansion or identifying, attracting and retaining key personnel, and failure to do so could have a material adverse effect on our business, financial condition and results of operations.

We are dependent on key personnel

Our future success depends in part on the continued service of our key management, technical, sales and marketing personnel and our ability to identify and hire additional personnel.

Competition for qualified management, technical, sales and marketing personnel is intense and we cannot assure you that we can retain and recruit adequate personnel to operate our business. Our success is largely dependent on the skills, experience and efforts of our key personnel, particularly P. Scott Munro, Chairman of the Board, Chief Executive Officer and Secretary, and Carlton Joseph Mertens, II, President and Chief Operating Officer, each of whom has entered into an employment agreement with us. The loss of either of these individuals or other key personnel could have a material adverse effect on our business, financial condition and results of

F-11

Table of Contents

operations. We maintain life insurance on Mr. Munro and Mr. Mertens in the amounts of \$7.9 million and \$10.0 million, respectively.

We may not be able to complete the future acquisitions and expansion that we believe are important to the growth of our business

Our operations to date have required substantial amounts of working capital to finance accounts receivable and product

inventories. Although we believe that we have sufficient funds, or alternate sources of funds, to carry on our business as presently conducted through 2000, we will need to raise additional amounts through public or private debt or equity financings in order to achieve the growth contemplated by our business plan. We cannot assure you that additional financing of any type will be available on acceptable terms, or at all, and failure to obtain such financing could have a material adverse effect upon our business, financial condition and results of operations.

We are dependent upon the availability of credit and our present credit facility

In order to obtain necessary working capital, we rely primarily on a line of credit that is collateralized by substantially all of our assets. The amount of credit available to us may be adversely affected by numerous factors beyond our control, such as:

We have significant future capital needs, and the availability of additional financing is uncertain

Any decrease or material limitation on the amount of capital available to us under our line of credit or other financing arrangements will limit our ability to fill existing sales orders or expand our sales levels and, therefore, would have a material adverse effect on our business, financial condition and results of operations. In addition, any significant increase in interest rates will increase our cost of financing and could have a material adverse effect on our business, financial condition and results of operations. We are dependent on the availability of accounts receivable financing on reasonable terms and at levels that are high relative to our equity base in order to maintain and increase our sales. We cannot assure you that such financing will continue to be available to us or available under terms acceptable to us. Our inability to have continuous access to such financing at reasonable costs would materially and adversely impact our business, financial condition, results of operations and cash flows.

We have primarily funded our working capital requirements through a \$125.0 million Inventory and Working Capital Agreement with IBM Credit Corporation. Borrowings under this credit facility are collateralized by substantially all of our assets, including accounts receivable, inventories and equipment. This credit facility provides that the outstanding interest-bearing cash advance balance is subject to interest at the annual rate of prime plus 1.875% (10.375% at December 31, 1999) and expires on August 31, 2000. IBM Credit Corporation may terminate this credit facility at any time upon the occurrence of, and subsequent failure to cure, an "Event of Default" (as that term is defined in the documentation for the credit facility). In the event of termination, the outstanding borrowings under the credit facility become immediately due and payable. The termination of this credit facility and our subsequent inability to secure a replacement credit facility on terms and conditions no less favorable than those contained in our present credit facility would have a material adverse effect on our business, financial condition and results of operations.

- delays in collection or deterioration in the quality of our accounts receivable;
- economic trends in the technology industry;
- the obsolescence of our inventory;
- interest rate fluctuations; and
- the lending policies of our creditors.

The terms of our credit facility with IBM Credit Corporation require that we obtain the consent of IBM Credit Corporation prior to incurring some types of additional indebtedness, including any additional senior or subordinated debt. We may incur additional indebtedness without IBM's consent through capital leases and general business commitments if the terms are commercially reasonable and consistent with our prior business

F-12

Table of Contents

practices. Our present credit facility and our anticipated cash flows may not provide funding sufficient to achieve the growth contemplated by our business plan. We may therefore need to obtain the consent of IBM Credit Corporation to incur additional indebtedness. While we have no reason to believe that IBM will not so consent, we cannot assure you that IBM Credit Corporation will give its consent. Failure to obtain IBM's consent or to obtain an alternate credit facility could have a material adverse effect on our business, financial condition and results of operations.

Our present credit facility limits our ability to incur additional indebtedness

Since we acquire inventory in advance of product orders and shipments, there is a risk that we will forecast incorrectly and stock excessive or insufficient inventory of particular products. The markets for products that we sell are extremely competitive and are characterized by declining selling prices over the life of a particular product and rapid technological change. Therefore, our business, like that of other wholesale distributors, is subject to the risk that the value of our inventory will decline as a result of price reductions by manufacturers or due to technological changes affecting the usefulness or desirability of our product inventory. It is the policy of many manufacturers of technology products to protect wholesale distributors such as Savoir from the loss in value of inventory due to technological change or reductions in the manufacturers' prices. Under the terms of most of our distribution agreements, vendors will generally credit us for inventory losses resulting from the vendor's price reductions if we comply with the conditions set forth in those agreements. In addition, generally under such agreements, we have the right to return for credit or exchange for other products a portion of our slow moving or obsolete inventory items within designated periods of time. We cannot assure you that, in every instance, we will be able to comply with all necessary conditions or manage successfully our price protection or stock rotation opportunities, if available. Also, a manufacturer that elects to terminate a distribution agreement generally will repurchase its products carried in a wholesale distributor's inventory. These industry practices are sometimes not included in written agreements and do not protect us in all cases from declines in inventory value, excess inventory or product obsolescence. We cannot assure you that manufacturers will continue these protective practices or that we will be able to manage successfully our existing and future inventories. Historically, we have not experienced losses due to obsolete inventory in excess of established inventory reserves. Significant declines in inventory value in excess of established inventory reserves or dramatic changes in prevailing technology could have a material adverse effect on our business, financial condition and results of operations.

As a result of price competition, we have low gross profit and operating income margins. These low margins magnify the impact on operating results of variations in net sales and operating costs. We have partially offset the effects of our low gross profit margins by increasing net sales, availing ourselves of large volume purchase discount opportunities and reducing selling, general and administrative expenses as a percentage of net sales. However, we cannot assure you that we will maintain or increase net sales, continue to avail ourselves of large volume purchase discount opportunities or further reduce selling, general and administrative expenses as a percentage of net sales. Future gross profit margins may be materially and adversely affected by changes in product mix, vendor pricing actions and competitive and economic pressures.

We may experience product supply shortages

We are dependent upon the supply of products available from our vendors. From time to time, the industry has experienced shortages of some of the products that we distribute due to vendors' difficulty in projecting demand. When product shortages occur, we typically receive an allocation of product from the vendor. We cannot assure you that our vendors will be able to maintain an adequate supply of products to fulfill all of our orders on a timely basis. If we fail to obtain adequate product supplies, or if product supplies are available to competitors but not to us, it would have a material adverse effect on our business, financial condition and results of operations.

F-13

Table of Contents

We extend credit to customers without requiring collateral

We sell products to a broad geographic and demographic base of customers and offer unsecured credit terms to our customers. Sirius accounted for approximately 16% of our outstanding accounts receivable at December 31, 1999. No other single customer accounted for more than 10% of our outstanding accounts receivable at December 31, 1999. To reduce our credit risk, we perform ongoing credit evaluations of our customers, maintain an allowance for doubtful accounts and have credit insurance. Historically, we have not experienced losses from write offs in excess of established reserves. Should our customers increase the rate at which they default on payments due to us, and should we be unable to collect our accounts receivable at a rate consistent with our present experience, it could have a material adverse effect on our business, financial condition and results of operations.

Our business is seasonal

The computer distribution industry experiences seasonal trends and, within each quarter, a substantial amount of product is generally sold in the last few days of the quarter. Our largest vendor, IBM, historically has sold up to 35% of its products in the last calendar quarter, and the continuation of this pattern could have an effect on our quarterly net sales. Historically, a substantial portion of our net sales has been made in the last

few days of a quarter. Due to our recent significant growth through acquisitions and our increased dependence on the sale of IBM products, sales variations may be magnified in the future and could have a material adverse effect on our business, financial condition and results of operations.

Our ability to expand our service capabilities is uncertain

We are expanding the nature and scope of our value-added services. We cannot assure you that new value-added services will be integrated successfully with our commercial mid-range server and related products distribution business. If we are unable to provide value-added services effectively, we may be unable to compete for the business of customers that demand services as a condition to purchasing products from us. In addition, we will be subject to risks commonly associated with a value-added services business, including dependence on reputation, fluctuations in workload and dependence on the ability to identify, recruit and retain qualified technical personnel. The expansion of our value-added services is expected to require a significant capital investment, including an increase in the number of technical employees. We cannot assure you that difficulties encountered in connection with the expansion of our value-added services will not have a material adverse effect on our business, financial condition and results of operations.

We are dependent on third-party shippers

We presently ship a majority of our products from our warehouses via Federal Express Corporation, but we also ship via United Parcel Service of America, Inc. and other common carriers. In addition, we drop-ship products from our vendors directly to our customers via these carriers. Changes in shipping terms or the inability of Federal Express, United Parcel Service or any other third-party shipper to perform effectively (whether as a result of mechanical failure, casualty loss, labor stoppage, other disruption or any other reason) could have a material adverse effect on our business, financial condition and results of operations. We cannot assure you that we can maintain favorable shipping terms or replace our present shipping services on a timely or cost-effective basis.

Our planned international expansion may not be successful

One of the elements of our business strategy is to expand internationally. We are currently distributing IBM products in Canada and Mexico. We cannot assure you that we will be able to continue to expand our international business successfully. Risks inherent in doing business on an international level include:

Our business is subject to rapid technological change, price reductions and inventory risk

F-14

Table of Contents

- unexpected changes in regulatory requirements;
- · export restrictions;

Any of these risks could adversely impact the success of our international operations. We cannot assure you that difficulties encountered with one or more of these factors will not have a material adverse effect on our future international operations and, consequently, on our business, financial condition and results of operations.

- tariffs and other trade barriers;
- · difficulties in staffing and managing foreign operations;
- · longer payment cycles;
- problems in collecting accounts receivable;
- political instability;
- fluctuations in currency exchange rates; and
- potentially adverse tax consequences.

We have never declared or paid a cash dividend on our common stock. We currently anticipate that we will retain all available funds for use in the operation of our business, including possible acquisitions, and we do not intend to pay any cash dividends in the foreseeable future. The payment of any future dividends will be at the discretion of our Board of Directors and will depend upon, among other factors:

We have not paid and do not presently intend to pay cash dividends on Savoir common stock

Further, our ability to pay cash dividends is currently restricted by the terms of our credit facility with IBM Credit Corporation. The terms of future credit facilities or other agreements may also contain similar restrictions. In addition, our Certificate of Designation with respect to the Series A Preferred Stock prohibits the payment of dividends on our common stock unless and until dividends are paid on the Series A Preferred Stock in accordance with its terms.

- · future earnings and cash flow;
- · operations;
- capital requirements;
- acquisitions and strategic investment opportunities;
- our general financial condition; and
- general business conditions.

We may issue additional shares of our common stock or other equity or convertible debt securities to effect future acquisitions or for other corporate purposes. Upon the issuance of additional capital stock, the percentage ownership of our stockholders will be reduced and stockholders may experience additional dilution.

Our stock price has been volatile historically

The market price of our common stock has been and is likely to continue to be highly volatile and may be significantly affected by factors such as:

Table of Contents

It is possible that in some future quarter, our operating results may be below the expectations of securities analysts and investors. If this occurs, the price of our common stock would likely decline, perhaps substantially. In addition, the stock market has from time to time experienced significant price and volume fluctuations that have particularly affected the market prices of the stocks of technology companies. These broad market fluctuations may adversely affect the market price of our common stock. In the past, following periods of volatility in the market price of a particular company's securities, securities class action litigation has often been brought. We cannot assure you that similar litigation will not occur in the future with respect to us and our securities. Any litigation relating to our securities could result in substantial costs and a diversion of management's attention and resources, which could have a material adverse effect upon our business, financial condition and results of operations.

- actual or anticipated fluctuations in our quarterly operating results;
- announcements of technological innovations;
- industry conditions and trends;
- · changes in or our failure to meet the expectations of securities analysts and investors; and
- general market conditions and other factors.

Provisions of our Certificate of Incorporation and of our Bylaws may make it more difficult for a third party to acquire, or may discourage a third party from attempting to acquire, control of Savoir. These provisions could limit the price that investors may be willing to pay for shares of our common stock. We presently have 1,850,012 shares of Series A Preferred Stock outstanding and 10 shares of Series B Preferred Stock outstanding and, without any further vote or action by the stockholders, have the authority to issue up to an additional 8,149,978 shares of preferred stock and to determine the price, rights, preferences, qualifications, limitations and restrictions, including voting rights, of this additional preferred stock. The issuance of additional preferred stock, while providing desirable flexibility in connection with possible acquisitions and other corporate purposes, could delay or prevent a third party from acquiring a majority of our outstanding voting stock. Further, Section 203 of the General Corporation Law of Delaware prohibits us from engaging in various types of business combinations with interested stockholders. These provisions may delay or prevent a change in control of Savoir without action by the stockholders, and therefore could adversely affect the market price of our common stock.

Item 2. Properties

summarizes the principal properties occupied by us.

Provisions of our charter and bylaws and Delaware law may make Savoir a less attractive acquisition candidate

F-16

Table of Contents

Location	Principal Use	Approximate Square Footage	Lease Expiration Date
Campbell, California	Corporate Headquarters and Sales	04.000	2000
San Antonio, Texas	Office Savoir Operational Headquarters and Sales, Warehouse, Distribution, Integration Center, Marketing and Technical Support	24,000	2000
	Office	120,000	2003
Fremont, California	Warehouse, Distribution and Integration Center	66,500	2003
Atlanta, Georgia	Warehouse, Distribution, Integration Center and Sales	40.500	
Lake Forest, California (a suburb of Los Angeles)	Office Warehouse, Distribution, Integration Center, Sales Office	16,500	2002
	and Technical Support Office	13,300	2004

We believe our facilities are suitable for our uses and are generally adequate to support our current level of operations. We believe that lease extensions or replacement space may be obtained for all of our leased facilities upon the expiration of the current lease terms, in most cases at rates not materially higher than those currently in effect.

Item 3. Legal Proceedings

Other than as set forth below, we are not a party to any material pending legal proceeding, nor is our property the subject of any material pending legal proceeding, except ordinary routine legal proceedings arising in the ordinary course of our business and incidental to our business, none of which are expected to have a material adverse impact upon our business, financial position or results of operations.

On June 18, 1999 a complaint was filed in the Superior Court of Orange County, California by Lee Adams against Western Micro Technology, Inc., WMT Acquisition Corp. and us. Mr. Adams' complaint purports to allege causes of action for breach of contract, specific performance, accounting, common count and false promise, arising out of an Agreement and Plan of Reorganization, dated March 17, 1997, by and among Western Micro Technology, Inc., WMT Acquisition Corp., Target Solutions, Inc. and Lee Adams. In addition to equitable relief, Mr. Adams' complaint seeks to recover compensatory damages in the amount of \$10.0 million and punitive damages according to proof. Mr. Adams has subsequently dismissed without prejudice his cause of action for accounting. We believe the complaint is without merit and intend to defend ourselves vigorously against this complaint. In addition, we have filed a cross-complaint for breach of contract in which we are seeking compensatory damages in the amount of \$1.5 million. The case has been set for trial on July 17, 2000.

Not applicable.

F-17

Table of Contents

PART II

Item 5. Market for Registrant's Common Stock and Related Stockholder Matters

(a) Common Stock Price Range

Our Common Stock is quoted on The Nasdaq National Market under the symbol "SVTG." Prior to November 24, 1997, our Common Stock was quoted on The Nasdaq National Market under the symbol "WSTM." The following table sets forth, for the periods indicated, high and low sales prices for our Common Stock as reported by The Nasdaq National Market.

Location	Principal Use	Approximate Square Footage	Lease Expiration Date
Burr Ridge, Illinois (a suburb of Chicago)	Warehouse, Distribution, Integration Center and Technical		
	Support and Sales Office	16,900	2003
Framingham, Massachusetts (a suburb of Boston)	Sales Office	11,200	2000
Huntsville, Alabama	Sales Office	6,000	2002
Monterrey, Mexico	Sales Office	2,700	2000
Mississauga, Ontario, Canada	Sales Office	3,300	2001

(b) Holders

As of February 24, 2000, there were approximately 260 stockholders of record and approximately 1,900 beneficial stockholders of our Common Stock.

(c) Dividends

We have never declared or paid any cash dividends on our Common Stock. We currently anticipate that we will retain all available funds for use in the operation of our business, including possible acquisitions, and do not intend to pay any cash dividends in the foreseeable future.

(d) Recent Sales of Unregistered Securities, Use of Proceeds from Registered Securities

Not applicable.

Table of Contents

Item 6. Selected Consolidated Financial Data

	High	Low
Year Ended December 31, 1998		
First Quarter	\$12.75	\$9.63
Second Quarter	13.44	9.13
Third Quarter	12.75	4.31
Fourth Quarter	9.50	5.31
Year Ended December 31, 1999		
First Quarter	\$12.50	\$8.63
Second Quarter	10.75	7.50
Third Quarter	9.75	7.75
Fourth Quarter	8.38	4.31

Year Ended December 31,

				/	
	1995	1996	1997	1998	1999
		(In thous	ands, except per s	hare data)	
Statements of Operations Data:	0.100 100	#404 CO7	****	#500.044	# 707 040
Net sales Cost of goods sold	\$106,462 93,416	\$131,697 114,389	\$237,884 205,089	\$593,341 526,113	\$767,243 683,292
Gross profit Selling, general and administrative	13,046	17,308	32,795	67,228	83,951
expenses Restructuring charges	13,694 3,600	13,716 —	25,969 —	46,002 —	61,083 11,287
Operating income (loss) Interest expense	(4,248) 850	3,592 978	6,826 3,181	21,226 4,318	11,581 2,842
Income (loss) before income taxes	(5,098)	2,614	3,645	16,908	8,739
Income tax expense		276	335	8,268	6,948
Income (loss) before extraordinary item Extraordinary item, net of tax	(5,098)	2,338	3,310	8,640 (2,338)	1,791
Net income (loss) Dividends on convertible preferred	(5,098)	2,338	3,310	6,302	1,791
stock subject to redemption			(484)	(5,439)	(1,432)
Net income (loss) attributable to common stockholders	\$ (5,098)	\$ 2,338	\$ 2,826	\$ 863	\$ 359
Net income (loss) per share attributable to common stockholders: (1)					
- Basic - Diluted Number of shares used in per share calculations:(1)	\$ (1.36) \$ (1.36)	\$ 0.55 \$ 0.52	\$ 0.57 \$ 0.55	\$ 0.10 \$ 0.09	\$ 0.03 \$ 0.03
- Basic - Diluted	3,756 3,756	4,255 4,513	4,902 5,976	8,714 9,343	12,387 12,951

As of December 31,

1995	1996	1997	1998	1999
		(In thousands)		
\$ 7,312	\$ 7,448	\$ 6,454	\$ 3,455	\$ 5,569
35,899	63,276	186,888	308, 902	324,693
7,126	11,335	15,579	22, 260	23, 987
117	53	22, 330	1,087	4,376
_	_	18,132	15,729	14,580
11,004	15,714	28,948	77,838	98,210
	\$ 7,312 35,899 7,126 117	\$ 7,312 \$ 7,448 35,899 63,276 7,126 11,335 117 53	\$ 7,312 \$ 7,448 \$ 6,454 35,899 63,276 186,888 7,126 11,335 15,579 117 53 22,330 - 18,132	(In thousands) \$ 7,312 \$ 7,448 \$ 6,454 \$ 3,455 35,899 63,276 186,888 308,902 7,126 11,335 15,579 22,260 117 53 22,330 1,087 - 18,132 15,729

Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations

The following discussion and analysis should be read in conjunction with "Selected Consolidated Financial Data" and the Consolidated Financial Statements and Notes thereto appearing elsewhere in this Report.

Except for historical information contained herein, the following discussion contains forward-looking statements that involve risks and uncertainties. Such forward-looking statements include, but are not limited to, statements regarding future events and our plans and expectations. Our actual results could differ materially from those discussed herein. Factors that could cause or contribute to such differences include, but are not limited to, those discussed previously in "Risk Factors," as well as those discussed elsewhere in this Report or incorporated herein by reference.

Overview

Recognizing the consolidation trend in the commercial mid-range systems distribution industry, we commenced our acquisition strategy in December 1994 with the acquisition of First Computer Corporation. Since then, we have focused upon expanding our commercial mid-range server distribution business through internal growth and strategic acquisitions. Since December 1994, we have completed twelve acquisitions, which have expanded our products and services, increased our geographic market coverage, strengthened our management and technical personnel and increased our operating leverage. The following table summarizes our acquisition history:

(1) See Note 8 of Notes to Consolidated Financial Statements for information concerning the computation of net income (loss) per share attributable to common stockholders.

[Additional columns below]

[Continued from above table, first column(s) repeated]

		Initial Cons	sideration	Actual Earn	out Paid(1)
Acquisition Target	Transaction Date	Cash	Shares of Common Stock	Cash	Shares of Common Stock
Enlaces ("Enlaces")(2) Infinite Solutions, Inc.	4/27/99	\$ 5,200,000	235,638		
("Infinite")(3) REAL Applications, Ltd.	1/4/99	2,750,000	88,560	_	_
("REAL")(4) MCBA Systems, Inc. ("MCBA")	9/8/98	12,875,000	-	_	-
(5) UniDirect Corporation ("UDC")	6/5/98	_	852,854	_	1,500,000
(6) Star Management Services,	5/15/98	2,900,000	_	_	_
<pre>Inc. ("SMS")(7) Target Solutions, Inc.</pre>	9/30/97	42,150,000	460,000	\$5,000,000	_
("TSI")(8) International Data Products,	3/17/97	_	220, 273	_	_
LLC ("IDP")(9) Star Technologies, Inc.	11/29/96	265,000	_	_	140,000
("STI")(10) R&D Hardware Systems Company	11/7/96	_	113, 263	_	166,800
of Colorado ("R&D")(11) International Parts, Inc.	1/2/96	1,000,000	125,000	_	78,587
("IPI")(12) First Computer Corporation	11/18/95	_	300,000	_	42,516
("FCC")(13)	12/1/94	_	328,943	_	=

Potential Consideration(1)

Acquisition Target	Cash	Shares of Common Stock
Enlaces ("Enlaces")(2) Infinite Solutions, Inc.	\$5,000,000	
("Infinite")(3) REAL Applications, Ltd. ("REAL")	_	_
(4)	_	_
MCBA Systems, Inc. ("MCBA")(5)	_	_
UniDirect Corporation ("UDC")(6) Star Management Services, Inc.	_	_
("SMS")(7)	_	_
Target Solutions, Inc. ("TSI")(8) International Data Products, LLC	- -	_
("IDP")(9) Star Technologies, Inc. ("STI")	_	_
(10) R&D Hardware Systems Company of	_	_
Colorado ("R&D")(11) International Parts, Inc. ("IPI")	_	_
(12) First Computer Corporation	_	_
("FCC")(13)	_	_

F-20

Table of Contents

- (1) As of February 24, 2000.
- (2) For the year ended December 31, 1998, Enlaces had unaudited revenues of approximately \$17 million.
- (3) For the year ended December 31, 1998, Infinite had unaudited revenues of approximately \$19 million.
- (4) For the year ended April 30, 1998, REAL had audited revenues of approximately \$80 million.
- (5) We also canceled indebtedness of \$487,000 owed to the Company by MCBA at the time of acquisition. For the year ended December 31, 1997, MCBA had audited revenues of approximately \$27 million.
- (6) For the year ended December 31, 1997, UDC had unaudited revenues of approximately \$18 million. We also issued a \$1.7 million promissory note to the former parent company of UDC. At December 31, 1999, the balance on the note was \$800,000.
- (7) For the eleven months ended September 30, 1997, SMS had audited revenues of approximately \$87 million.
- (8) For the year ended December 31, 1996, TSI had unaudited revenues of approximately \$15 million. The TSI agreement provides for a total earnout potential consideration of \$10,000,000 in cash and stock.
- (9) Excludes assumed liabilities of \$424,000. For the year ended December 31, 1995, IDP had unaudited revenues of \$5 million.

Income Taxes

Due to the use of net operating loss carryforwards generated in 1995 and years prior, our effective tax rate was 9.2% in 1997, as compared to a normal combined effective tax rate of approximately 40% for federal and state income taxes. As of December 31, 1997, we had used substantially all of our available federal net operating loss carryforward amounts. Due to the significant amount of non-tax deductible amortization expense related to goodwill incurred in certain acquisitions, our effective tax rates were 48.9% in 1998 and 79.5% in 1999. The 1999 tax rate was also affected by the write off of nondeductible intangible assets as part of the restructuring charge.

Results of Operations

The following table sets forth for the periods indicated certain income and expense items as a percentage of net sales:

- (10) For the year ended June 30, 1996, STI had unaudited revenues of approximately \$8 million.
- (11) For the year ended December 31, 1995, R&D had unaudited revenues of approximately \$10 million.
- (12) For the year ended December 31, 1994, IPI had unaudited revenues of approximately \$15 million.
- (13) For the year ended December 31, 1993, FCC had unaudited revenues of approximately \$6 million.

Comparisons of Years Ended December 31, 1997, 1998 and 1999

Net Sales. Net sales consists of sales of commercial mid-range servers, integrated personal computers, workstations, peripheral equipment, storage products, software and remarketed installation and technical support services, net of sales discounts and returns. Net sales increased by 29.3% to \$767.2 million in 1999 from \$593.3 million in 1998, and 149.4% in 1998 from \$237.9 million in 1997. Sales increased due to the continued expansion of our computer systems distribution business. The acquisitions of Infinite and Enlaces also contributed to the rise in our revenue. Infinite and Enlaces accounted for approximately \$37.4 million of net sales in 1999. Sales also increased due to the recruitment of new customers, hiring of additional sales representatives and higher storage product sales. Despite the increase in revenue during the third and fourth quarters of 1999 compared to the third and fourth quarters of 1998, we did incur a slowdown in sales in the third and fourth quarters of 1999 in our largest product line, the IBM AS/400™. We attribute the slowdown to the Year 2000 issue and its effect on purchasing patterns in end user accounts. We have determined that larger end user accounts reduced their purchases from our value-added reseller customers as they focused on ensuring that their systems are Year 2000 compliant. We expect historical buying patterns to return after the first quarter of 2000. The acquisitions of MCBA and REAL, the opening of a sales office in Canada and increased software sales, primarily as a result of the acquisition of UDC, contributed to the rise in revenue in 1998. MCBA, REAL and UDC accounted for approximately \$64.2 million of net sales in 1998.

For the years ended December 31, 1997, 1998 and 1999, sales to Sirius accounted for approximately 11%, 18% and 16%, respectively, of our net sales. Our sales to Sirius are made under the Sirius Agreement, pursuant to which we appointed Sirius as one of our industry remarketer affiliates of IBM products. The Sirius Agreement provides that Sirius may not enter into any similar arrangement with any third party for the purpose of selling IBM products to its end user customers and also provides a favorable pricing structure to Sirius. As a result, Sirius is expected to remain our largest customer for the duration of the Sirius Agreement and to account for approximately the same percentage of our net sales in 2000 as it represented in 1999. The

Sirius Agreement expires on December 31, 2000, but may be terminated earlier under certain conditions, not including termination at will.

Gross Profit. Cost of sales is comprised of purchase costs, net of early payment and volume discounts and product freight and does not include any depreciation or amortization expense. Gross profit as a percentage of net sales is affected by several factors including the mix of high margin and low margin products and services and the proportion of large orders on which we extend volume discounts to our customers. Gross profit increased by 24.9% to \$84.0 million in 1999 from \$67.2 million in 1998, and by 105.0% in 1998 from \$32.8 million in 1997. Gross profit as a percentage of net sales was 10.9% in 1999, 11.3% in 1998 and 13.8% in 1997. The decreases are a result of a higher proportion of large orders on which we extended volume discounts to our customers.

Operating Expenses. Operating expenses include: salaries and commissions paid to sales representatives; compensation paid to marketing, product management, technical and administrative personnel; depreciation of infrastructure costs, including our information system and leasehold improvements; amortization of intangibles resulting from goodwill recorded from acquisitions; facility lease expenses; telephone and data line expenses and provision for bad debt losses. Fluctuations in operating expenses as a percentage of net sales can result from planned expenditures by us for additional sales, marketing, technical support and administrative personnel, efficiencies gained through higher sales volumes and resulting economies of scale and the timing of acquisitions.

Selling, general and administrative expenses (excluding depreciation and amortization expense and restructuring charge) increased by 31.5% to \$52.2 million in 1999 from \$39.7 million in 1998, and by 70.3% in 1998 from \$23.3 million in 1997. Selling, general and administrative expenses as a percentage of net sales were 6.7% in 1999, 6.7% in 1998 and 9.8% in 1997. In 1999, selling, general and administrative expenses increased due to the acquisitions of Infinite and Enlaces, a full year of expenses related to the 1998 acquisitions of MCBA, REAL and UDC, necessary personnel increases as a result of higher systems sales and the costs associated with reviewing potential international acquisitions that were not completed. As a percentage of net sales, selling, general and administrative expenses were essentially the same in 1999 as in 1998 as a result of the revenue shortfall experienced in the third and fourth quarter of 1999. In 1998, selling, general and administrative expenses as a percentage of net sales decreased due to the economies of scale generated from acquisitions and rapid net sales growth.

Depreciation and amortization expense increased by 41.3% to \$8.9 million in 1999 from \$6.3 million in 1998, and by 137.1% in 1998 from \$2.7 million in 1997. Depreciation and amortization expense as a percentage of net sales was 1.2% for 1999 and 1.1% for 1998 and 1997. Depreciation and amortization increased during 1999 and 1998 as a result of higher amortization expense from increased goodwill related to larger acquisitions and increased depreciation costs due to leasehold improvements and computer equipment additions.

The consolidated statements of operations for 1999 include pre-tax charges totaling \$11.3 million related to the restructuring of corporate functions and the recognition of impaired assets. Of this amount, \$2.8 million represents costs associated with the consolidation of facilities and elimination of redundant back-office functions. The costs associated with the restructuring principally include lease termination costs (\$500,000), leasehold and other asset disposals (\$1.4 million)

and personnel (a total of 38 employees, primarily in administration, finance and information technology) severance costs (\$900,000). The remaining \$8.5 million represents asset impairment charges related to recorded goodwill and fixed assets. These impairment costs principally include the write off of goodwill (\$7.0 million) related to unprofitable business activities acquired in previous purchases (Target Solutions, Inc., International Data Products, LLC and UniDirect, Inc.), the operations of which we have terminated as part of our restructuring. The balance of the impairment charge relates to the write off of unproductive electronic commerce assets (\$1.5 million).

We decided to restructure our operations as a result of internal reviews of all significant corporate functions. Upon completion of the review, we determined, as a result of multiple acquisitions over the past three years, that there were functions, facilities and personnel that were duplicative and unnecessary if properly consolidated.

F-23

Table of Contents

The impairment charges were a result of our ongoing evaluation of our assets and the related useful lives and contributions to our company of these assets. The goodwill portion of the charge relates to the acquisitions of Target Solutions, Inc., International Data Products, L.L.C. and UniDirect, Inc. The business lines associated with each of these acquisitions have had deteriorating results over the past year, with a significant drop off in the second and third quarters of 1999. The decline is attributable to the fact that these business lines were not within the core distribution competency of our company. As a result, we terminated these operations as part of our restructuring. The fixed asset portion of the impairment charge relates to unsuccessful investments made by our company in an electronic commerce attachment to our business process system.

Operating Income. Operating income decreased by 45.3% to \$11.6 million in 1999 from \$21.2 million in 1998, and increased in 1998 by 211.0% from \$6.8 million in 1997. Operating income as a percentage of net sales was 1.5% in 1999, 3.5% in 1998 and 2.9% in 1997. The decrease in operating income in 1999 was primarily due to the restructuring charge in the third quarter of 1999 and revenue shortfall experienced in the third and fourth quarters of 1999. Operating income increased in 1998 due to higher net sales and economies of scale.

Interest Expense. Interest expense decreased by 34.9.% to \$2.8 million in 1999 from \$4.3 million in 1998 and increased by 35.7% from \$3.2 million in 1997. Interest expense as a percentage of net sales was 0.4% in 1999, 0.6% in 1998 and 1.4% in 1997. Interest expense decreased in 1999 due to a more effective use of working capital to fund operations and limiting the use of our line-of-credit. In 1998 interest expense increased due to additional borrowings necessary to fund acquisitions, infrastructure additions and overall growth. In addition, interest expense in 1998 also increased due to the amortization of a discount on warrants issued in September 1997. The warrants were originally determined to have a fair market value of \$1,330,000, which was recorded as discount on notes payable. During the first quarter of 1998, the warrants were revalued at \$2,721,000. We recorded approximately \$330,000 in interest expense for the discount on the warrants in 1998. We will not incur an additional expense resulting from the warrants as the

unamortized value was expensed as a result of our public offering and subsequent payoff of the debt underlying the warrants.

Income Taxes. Income tax expense was \$6.9 million, \$8.3 million and \$335,000 in 1999, 1998 and 1997, respectively, reflecting effective tax rates of 79.5%, 48.9% and 9.2%, respectively. In 1999 and 1998, our effective tax rates were higher than the federal statutory rate (35.0%) due to non-deductible goodwill and other expenses, state income taxes and foreign taxes in excess of the federal statutory rate. Our effective tax rates were lower than the statutory rates in 1997 due to the utilization of net operating loss carryforwards from losses generated in 1995 and prior and the release of the valuation allowance against the deferred tax assets. These net operating loss carryforwards were fully applied in 1997.

In connection with the repayment of our outstanding debt obligations, we recorded an extraordinary charge in 1998 of \$2,338,000, net of tax, resulting from a prepayment penalty, the write off of unamortized discounts relating to certain warrants issued to debt holders and other related expenses.

Ouarterly Results of Operations

The following tables set forth certain unaudited quarterly financial data and such data expressed as a percentage of net sales for the quarters of 1998 and 1999. In our opinion, this information has been presented on the same basis as the audited consolidated financial statements appearing elsewhere in this Report, and all necessary adjustments (consisting only of normal recurring adjustments) have been included in the amounts stated below to present fairly the unaudited quarterly results when read in conjunction with the Consolidated Financial Statements and Notes thereto appearing elsewhere in this Report. The operating results for any quarter are not necessarily indicative of the results to be expected for any future period.

F-24

Table of Contents

Fiscal Year Ended December 31,

	1997	1998	1999
Net sales Cost of goods sold	100.0% 86.2	100.0% 88.7	100.0 % 89.1
Gross profit	13.8	11.3	10.9
Selling, general and administrative expenses	9.8	6.7	6.7
Restructuring charge	_	_	1.5
Depreciation and amortization expense	1.1	1.1	1.2
Total operating expenses	10.9	7.8	9.4
Operating income	2.9	3.5	1.5
Interest expense	1.4	0.6	0.4
Income before income taxes and extraordinary item	1.5	2.9	1.1
Income tax expense	0.1	1.4	0.9
Income before extraordinary item	1.4	1.5	0.2
Extraordinary item, net of tax effect	-	(0.4)	-
Not income	1 4		
Net income Dividends on convertible preferred stock subject to redemption	1.4 0.2	1.1 0.9	0.2 0.2
Net income attributable to common stockholders	1.2%		⊎.∠ —%
MET THEOMIE STELLINGESTE TO COMMINION SERVENCED SELECTION OF SELECTION	1.2%	0.2%	—% ———————————————————————————————————

1998

		19	998			19	999	
	First Quarter	Second Quarter	Third Quarter	Fourth Quarter	First Quarter	Second Quarter	Third Quarter	Fourth Quarter
_				thousands, exc				
Net sales Cost of goods sold	\$100,504 87,443	\$122,920 108,031	\$149,809 132,561	\$220,108 198,078	\$168,569 149,476	\$201,687 179,282	\$181,577 162,026	\$215,410 192,508
Gross profit	13,061	14,889	17,248	22,030	19,093	22,405	19,551	22,902
Selling, general and administrative expenses Depreciation and	8,025	9,067	10,551	12,028	11,606	12,988	13,330	14,273
amortization Restructuring charge	1,201 -	1,560 —	1,626 -	1,944 —	2,021 _	2,458 -	2,296 12,800	2,111 (1,513)
Total operating expenses	9,226	10,627	12,177	13,972	13,627	15,446	28, 426	14,871
Operating income (loss) Interest expense	3,835 1,709	4,262 963	5,071 605	8,058 1,041	5,466 738	6,959 774	(8,875) 586	8,031 744
Income (loss) before income taxes and extraordinary item Income tax expense	2,126	3,299	4,466	7,017	4,728	6,185	(9,461)	7,287
(benefit)	1,033	1,618	2,179	3,438	2,317	3,013	(4,063)	5,681
Income (loss) before extraordinary item Extraordinary item, net	1,093	1,681	2,287	3,579	2,411	3,172	(5,398)	1,606
of tax effect	_	(2,338)	_	-	_	_	-	_
Net income (loss)	\$ 1,093	\$ (657)	\$ 2,287	\$ 3,579	\$ 2,411	\$ 3,172	\$ (5,398)	\$ 1,606
Per share attributable to common stockholders: Income before								
extraordinary item — Basic Extraordinary item, Net	\$ 0.12	\$ 0.16	\$ (0.20)	\$ 0.30	\$ 0.18	\$ 0.23	\$ (0.44)	\$ 0.10
of tax effect Net income (loss) —	\$ -	\$ (0.30)	\$ -	\$ -	\$ -	\$ -	\$ -	\$ -
Basic (1033)	\$ 0.12	\$ (0.14)	\$ (0.20)	\$ 0.30	\$ 0.18	\$ 0.23	\$ (0.44)	\$ 0.10
Income before extraordinary Item — diluted Extraordinary item, net	\$ 0.11	\$ 0.16	\$ (0.20)	\$ 0.27	\$ 0.17	\$ 0.21	\$ (0.44)	\$ 0.10
of tax effect	\$ -	\$ (0.22)	\$ -	\$ -	\$ -	\$ -	\$ -	\$ -
Net income (loss) — Diluted	\$ 0.11	\$ (0.06)	\$ (0.20)	\$ 0.27	\$ 0.17	\$ 0.21	\$ (0.44)	\$ 0.10

F-25

Table of Contents

Our quarterly net sales and operating results may vary significantly as a result of a variety of factors, including, but not limited to, changes in the supply and demand for commercial mid-range servers, peripheral equipment, software and related services, the cost, timing and integration of acquisitions, the addition or loss of a key vendor or customer, the introduction of new technologies, changes in manufacturers' prices, price protection policies or stock rotation privileges, changes in market development funds, changes in the level of operating expenses, product supply shortages, disruption of warehousing or shipping channels, inventory adjustments, increases in the amount of accounts receivable written off, price competition, changes in the mix of products sold through distribution channels and in the mix of products purchased by OEMs. Operating results could also be adversely affected by general economic and other conditions affecting the timing of customer orders and capital spending, a downturn in the market for commercial mid-range servers, and order cancellations or rescheduling. In addition, the computer distribution industry experiences both seasonal trends and, within each quarter, tends to sell a substantial amount of its products at the end of the quarter. For example, our largest vendor, IBM, sells up to 35% of its products in the last calendar quarter. Historically, a substantial portion of our net sales has been made in the last few days of a quarter.

Accordingly, our quarterly results of operations are difficult to predict and delays in the completion of sales near the end of a quarter could cause quarterly net sales to fall substantially short of anticipated levels and, to a greater degree, adversely affect profitability. Because of the many factors that can affect our operating results, we believe that period-to-period comparisons of our operating results are not necessarily meaningful and should not be relied upon as an indication of future performance. Our future operating results are expected to continue to fluctuate as a result of these and other factors, which could have a material adverse effect on our business, financial condition and results of operations.

Liquidity and Capital Resources

We have required substantial capital to finance accounts receivable, inventories, capital expenditures and acquisitions. In the past, we have financed these requirements primarily through borrowings under credit facilities, cash generated from operations, the issuance and sale of Common and Series A Preferred Stock and Subordinated Notes.

In September 1998, we agreed with IBMCC to increase the IBMCC Credit Facility line from \$75.0 million to \$125.0 million. For the period December 27, 1999 through March 31, 2000 the line was temporarily increased to \$145 million. Product purchases from IBM and cash advances from IBMCC are directly charged to the credit line and are paid by us based on payment terms outlined in the IBMCC Credit Facility. Borrowings under the IBMCC Credit Facility are based on eligible accounts receivable and inventory, as defined. The IBMCC Credit Facility expires on August 31, 2000 and contains restrictive covenants which include the maintenance of minimum current, tangible net worth and net profit after tax to revenue ratios, as defined. As of December 31, 1998 and 1999, we had outstanding obligations under the IBMCC Credit Facility of \$113.7 million and \$117.7 million, respectively. Of the total outstanding borrowings, \$6.2 million and \$6.0 million represented cash advances at December 31, 1998 and 1999, respectively. Cash advances bear interest at the prime rate plus 1.875% (10.375% at December 31, 1999). Based on eligible assets, as of December 31, 1999, we had additional borrowings available under the IBMCC Credit Facility of approximately \$24.4 million.

On April 23, 1999, we executed an amendment to the IBMCC Credit Facility, pursuant to which we obtained a loan of \$5,000,000 to consummate the Enlaces transaction. On September 8, 1998, we executed another amendment to the IBMCC Credit Facility, under which we obtained an additional loan of \$15,000,000 to consummate the REAL transaction. These loans bear interest at prime plus 2.0% (10.5% at December 31, 1999) and are due in monthly installments through September 2000.

Operating activities for 1999 provided cash in the amount of \$21.3 million. For this period, cash was provided primarily as a result of a decrease in accounts receivable of \$12.3 million. Favorable accounts payable terms with IBMCC and increased leasing of product purchases by customers through IBMCC and third party leasing vendors resulted in the generation of cash in 1999. For 1998, net cash of \$15.6 million was provided. This was primarily attributable to growth in sales and the resulting increases in accounts payable and accrued expenses, offset by an increase in accounts receivable.

Investing activities for 1999 used cash in the amount of \$16.8 million. For this period, cash was used for the purchase of Enlaces and Infinite and continuing leasehold and computer hardware and software investments made at the headquarters, sales office and warehouse and integration center sites. For 1998, \$23.7 million was used in investing activities, principally for the acquisitions of REAL and UDC as well as infrastructure additions.

In 1999, cash used by financing activities was \$510,000, which resulted primarily from borrowings and payments under the IBMCC Credit Facility. Cash provided by financing activities for 1998 was \$11.0 million, consisting of a \$15.0 million loan from IBMCC and proceeds from a secondary offering of our common stock of \$28.6 million offset by payments on the Subordinated Notes, SMS Seller Notes and the IBMCC Credit Advance totaling \$32.6 million.

We believe we have sufficient funds, or alternate sources of funds, to carry on our business as presently conducted through 2000.

Backlog

Although we receive purchase orders for products to be delivered to customers over a specified time period, there can be no assurance that such orders will result in sales, as most orders are subject to revision or cancellation without penalty. Consequently, we do not believe that backlog is a meaningful indicator of sales for future periods.

Inflation

We do not believe that inflation in the United States, Canada or Mexico in recent years has had a significant effect on our results of operations.

Recent Pronouncements

In June 1998, the Financial Accounting Standards Board ("FASB") issued Statement of Financial Accounting Standards ("SFAS") No. 133, "Accounting for Derivative Instruments and Hedging Activities." SFAS No. 133, as amended by SFAS No. 137, is effective for all fiscal quarters of all fiscal years beginning after June 15, 2000. SFAS No. 133 requires that all derivative instruments be recorded on the balance sheet at their fair value. Changes in the fair value of derivatives which meet the definition of a hedge transaction are recorded each period in either earnings or other comprehensive income. We believe that the impact of the pronouncement will be minimal, as historically, we have not entered into any derivative or hedging transactions.

Item 7A. Quantitative and Qualitative Disclosures About Market Risk

Market Risks

Certain of the Company's operations, primarily its international subsidiaries, occasionally purchase and sell products in currencies other than their functional currencies. This subjects the Company to the risks associated with

fluctuations of foreign currency exchange rates. The Company reduces this risk by utilizing natural hedging (offsetting receivables and payables). The Company does not hedge either its investments in its foreign operations or its floating interest rate exposures.

The Company's primary interest rate exposure results from the revolving credit facility's floating rate pricing mechanism. If interest rates were to increase 100 basis points from March 20, 2000 levels, the additional annual expense would be approximately \$200,000 on a pre-tax basis.

F-27

Table of Contents

Item 8. Financial Statements and Supplementary Data

REPORT OF INDEPENDENT ACCOUNTANTS

To the Board of Directors and Stockholders

Savoir Technology Group, Inc.

In our opinion, the consolidated financial statements listed in the index appearing under Item 14(a)(1) on page F-59 present fairly, in all material respects, the consolidated financial position of Savoir Technology Group, Inc. and its subsidiaries at December 31, 1999 and 1998, and the consolidated results of their operations and their cash flows for each of the three years in the period ended December 31, 1999, in conformity with accounting principles generally accepted in the United States. In addition, in our opinion, the financial statement schedule listed in the index appearing under Item 14(a)(2) on page F-59 presents fairly, in all material respects, the information set forth therein when read in conjunction with the related consolidated financial statements. These financial statements and financial statement schedule are the responsibility of the Company's management; our responsibility is to express an opinion on these financial statements and financial statement schedule based on our audits. We conducted our audits of these statements in accordance with auditing standards generally accepted in the United States, which require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, and evaluating the $% \left(1\right) =\left(1\right) \left(1\right) \left($ overall financial statement presentation. We believe that our audits provide a reasonable basis for the opinion expressed above.

As discussed in Note 18, the Company has revised the consolidated financial statements as of December 31, 1999 and 1998 to present \$14,580,000 and \$15,729,000, respectively, of convertible preferred stock subject to redemption as convertible preferred stock outside of stockholders' equity.

As a Percentage of Net Sales

Net sales Cost of goods sold	100.0% 87.0	100.0% 87.9	100.0% 88.5	100.0% 90.0	100.0% 88.7	100.0% 88.9	100.0% 89.2	100.0% 89.4
Gross profit	13.0	12.1	11.5	10.0	11.3	11.1	10.8	10.6
Selling, general and administrative expenses Depreciation and amortization Restructuring charge	8.0 1.2 —	7.4 1.2 -	7.0 1.1 —	5.5 0.8 —	6.9 1.2 —	6.4 1.3 —	7.3 1.4 7.0	6.6 1.0 (0.7)
Total operating expenses	9.2	8.6	8.1	6.3	8.1	7.7	15.7	6.9
Operating income (loss) Interest expense	3.8 1.7	3.5 0.8	3.4 0.5	3.7 0.5	3.2 0.4	3.4 0.3	(4.9) 0.3	3.7 0.4
Income (loss) before income taxes and extraordinary item Income tax expense	2.1	2.7	2.9	3.2	2.8	3.1	(5.2)	3.3
(benefit)	1.0	1.3	1.4	1.6	1.4	1.5	(2.2)	2.6
Income (loss) before extraordinary item Extraordinary item, net of	1.1	1.4	1.5	1.6	1.4	1.6	(3.0)	0.7
tax effect	_	(1.9)	_	_	_	_	_	_
Net income (loss)	1.1%	(0.5)%	1.5%	1.6%	1.4%	1.6%	(3.0)%	0.7%

Austin, Texas

January 27, 2000, except for Note 17, as to which the date is March 2, 2000

F-28

Table of Contents

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

CONSOLIDATED BALANCE SHEETS

PricewaterhouseCoopers LLP

The accompanying notes are an integral part of these consolidated financial statements.

F-29

Table of Contents

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF OPERATIONS

	1998 (as revised)	1999 (as revised)
	(In thousands, e	xcept share amounts)
SSETS		
urrent assets: Cash	\$ 5,820	\$ 9,863
Trade accounts receivable, net of allowance for doubtful	ψ 3,620	φ 9,003
accounts of \$2,036 in 1999 and \$1,100 in 1998	156,953	145,328
Other receivables	11,443	14,331
Inventories	38,913	36,986
Other current assets	4,574	6,588
Total current assets	217,703	213,096
operty and equipment, net	5,526	5,998
cess of cost over acquired net assets and other intangibles,		
t	83,810	100,969
her assets	1,863	4,630
Total assets	\$308,902	\$324,693
LIABILITIES, CONVERTIBLE PREFERRED AND STOCKHOLDER:		
rrent liabilities:		
Notes payable	\$ 21,240	\$ 21,018
Current portion of long-term debt	1,020	2,969
Accounts payable	184,915	177,161
Accrued expenses and other current liabilities	7,073	6,379
Total current liabilities	214,248	207,527
ng-term debt, less current portion	1,087	4,376
Total liabilities	215,335	211,903
ommitments and contingencies (Notes 4, 10 and 12)		
nvertible preferred stock subject to redemption, \$0.01 par		
nvertible preferred stock subject to redemption, \$0.01 par lue; 10,000,000 shares authorized; issued and outstanding:		
nvertible preferred stock subject to redemption, \$0.01 par lue; 10,000,000 shares authorized; issued and outstanding: ries A: 1,850,012 shares in 1999 and 1,986,500 shares in 1998;		
nvertible preferred stock subject to redemption, \$0.01 par lue; 10,000,000 shares authorized; issued and outstanding: ries A: 1,850,012 shares in 1999 and 1,986,500 shares in 1998; ries B: 10 shares in 1999 and 1998; liquidation preference:	15 720	14 590
nvertible preferred stock subject to redemption, \$0.01 par lue; 10,000,000 shares authorized; issued and outstanding: ries A: 1,850,012 shares in 1999 and 1,986,500 shares in 1998; ries B: 10 shares in 1999 and 1998; liquidation preference:	15,729	14,580
nvertible preferred stock subject to redemption, \$0.01 par lue; 10,000,000 shares authorized; issued and outstanding: ries A: 1,850,012 shares in 1999 and 1,986,500 shares in 1998; ries B: 10 shares in 1999 and 1998; liquidation preference: 7,691 in 1999 and \$18,996 in 1998	15,729	14,580
nvertible preferred stock subject to redemption, \$0.01 par lue; 10,000,000 shares authorized; issued and outstanding: ries A: 1,850,012 shares in 1999 and 1,986,500 shares in 1998; ries B: 10 shares in 1999 and 1998; liquidation preference: 7,691 in 1999 and \$18,996 in 1998 ockholders' equity Common stock, \$0.01 par value; 25,000,000 shares	15,729	14,580
nvertible preferred stock subject to redemption, \$0.01 par lue; 10,000,000 shares authorized; issued and outstanding: ries A: 1,850,012 shares in 1999 and 1,986,500 shares in 1998; ries B: 10 shares in 1999 and 1998; liquidation preference: 7,691 in 1999 and \$18,996 in 1998 ockholders' equity Common stock, \$0.01 par value; 25,000,000 shares authorized; issued and outstanding: 13,165,311 shares in	,	
nvertible preferred stock subject to redemption, \$0.01 par lue; 10,000,000 shares authorized; issued and outstanding: ries A: 1,850,012 shares in 1999 and 1,986,500 shares in 1998; ries B: 10 shares in 1999 and 1998; liquidation preference: 7,691 in 1999 and \$18,996 in 1998 Dockholders' equity Common stock, \$0.01 par value; 25,000,000 shares authorized; issued and outstanding: 13,165,311 shares in 1999 and 10,698,010 shares in 1998	15,729 107	132
nvertible preferred stock subject to redemption, \$0.01 par lue; 10,000,000 shares authorized; issued and outstanding: ries A: 1,850,012 shares in 1999 and 1,986,500 shares in 1998; ries B: 10 shares in 1999 and 1998; liquidation preference: 7,691 in 1999 and \$18,996 in 1998 ockholders' equity Common stock, \$0.01 par value; 25,000,000 shares authorized; issued and outstanding: 13,165,311 shares in 1999 and 10,698,010 shares in 1998 Shareholder note receivable	107	132 (2,513)
nvertible preferred stock subject to redemption, \$0.01 par lue; 10,000,000 shares authorized; issued and outstanding: ries A: 1,850,012 shares in 1999 and 1,986,500 shares in 1998; ries B: 10 shares in 1999 and 1998; liquidation preference: 7,691 in 1999 and \$18,996 in 1998 ockholders' equity Common stock, \$0.01 par value; 25,000,000 shares authorized; issued and outstanding: 13,165,311 shares in 1999 and 10,698,010 shares in 1998 Shareholder note receivable Additional paid-in capital	107 _ 76,101	132 (2,513) 98,866
nvertible preferred stock subject to redemption, \$0.01 par lue; 10,000,000 shares authorized; issued and outstanding: ries A: 1,850,012 shares in 1999 and 1,986,500 shares in 1998; ries B: 10 shares in 1999 and 1998; liquidation preference: 7,691 in 1999 and \$18,996 in 1998 ockholders' equity Common stock, \$0.01 par value; 25,000,000 shares authorized; issued and outstanding: 13,165,311 shares in 1999 and 10,698,010 shares in 1998 Shareholder note receivable	107	132 (2,513)
convertible preferred stock subject to redemption, \$0.01 par clue; 10,000,000 shares authorized; issued and outstanding: pries A: 1,850,012 shares in 1999 and 1,986,500 shares in 1998; cries B: 10 shares in 1999 and 1998; liquidation preference: 17,691 in 1999 and \$18,996 in 1998 Cockholders' equity Common stock, \$0.01 par value; 25,000,000 shares authorized; issued and outstanding: 13,165,311 shares in 1999 and 10,698,010 shares in 1998 Shareholder note receivable Additional paid-in capital	107 _ 76,101	132 (2,513) 98,866
authorized; issued and outstanding: 13,165,311 shares in 1999 and 10,698,010 shares in 1998 Shareholder note receivable Additional paid-in capital Retained earnings	107 - 76,101 1,630	132 (2,513) 98,866 1,725

The accompanying notes are an integral part of these consolidated financial statements.

F-30

Table of Contents

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY

Year Ended December 31,

	1997	1998	1999
Net sales	(In thous	ands, except per share \$593,341	amounts) \$767,243
Cost of goods sold	205,089	526,113	683,292
Gross profit	32,795	67,228	83,951
Selling, general and administrative expenses Restructuring charge	25,969 —	46,002 —	61,083 11,287
Operating income	6,826	21,226	11,581
Interest expense	3,181	4,318	2,842
Income before income taxes and extraordinary item	3,645	16,908	8,739

Income tax expense	335	8,268	6,948
Income before extraordinary item Extraordinary item, net of tax effect of \$2,246	3,310	8,640 (2,338)	1,791
Net income Dividends on convertible preferred stock subject to redemption	3,310 (484)	6,302 (5,439)	1,791 (1,432)
Net income attributable to common stockholders	\$ 2,826	\$ 863	\$ 359
Net income per share attributable to common stockholders: Income before extraordinary item — basic Extraordinary item, net of tax effect	\$ 0.57	\$ 0.37 (0.27)	\$ 0.03
Net income per share — basic	\$ 0.57	\$ 0.10	\$ 0.03
Income before extraordinary item — diluted Extraordinary item, net of tax effect	\$ 0.55	\$ 0.34 (0.25)	\$ 0.03
Net income per share — diluted	\$ 0.55	\$ 0.09	\$ 0.03
Number of shares used in per share calculations: Basic	4,902	8,714	12,387
Diluted	5,976	9,343	12,951

The accompanying notes are an integral part of these consolidated financial statements.

F-31

Table of Contents

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

CONSOLIDATED STATEMENTS OF CASH FLOWS

	Common Stock		Additional Paid-In	Additional Retained Paid-In Earnings		
	Shares	Amount	Capital	(Deficit)	Note Receivable	Total
Balances, January 1, 1997 Exercise of stock options	4,488,131 24,375	\$ 45 _	(In thousands, \$17,914 74	except share amounts) \$(2,245)	\$ -	\$15,714 74
Issuance of common stock in business combinations	809,898	8	6,817	_		6,825
Issuance of common stock under employee stock purchase plan Dividend on convertible preferred stock	31,044	1	255	_	_	256
subject to redemption Common stock warrants issued in	4,230	_	50	(100)		(50)
connection with debt offerings Issuance of common stock warrants Tax benefit from exercise of stock		_ _	1,330 1,000	_ _		1,330 1,000
options Net income	- -	_ _	489 —	3,310	_ _	489 3,310
Balances, December 31, 1997 (as revised) Exercise of stock options	5,357,678 54,611	54 1	27,929 308	965		28,948 309
Issuance of common stock in business combinations Issuance of common stock under employee	1,193,510	12	9,727	=	-	9,739
stock purchase plan Issuance of common stock under secondary	33,773	_	297	_	_	297
offering Dividend on convertible preferred stock subject to redemption	3,000,000 171,252	30 2	28,559 1,801	(1,803)	_	28,589 _
Conversion of convertible preferred stock subject to redemption to common	•		,	(, ,		
stock Dividend on convertible preferred stock subject to redemption	262,853 612,533	2 6	2,182 3,793	(3,799)	_	2,184
Cash dividend on convertible preferred stock subject to redemption	-	_	_	(35)	_	(35)
Exercise of warrants Revalue of warrants in connection with debt offering	11,800	_	114 1,391	-	_	114 1,391
Net income				6,302		6,302
Balances, December 31, 1998 (as revised) Exercise of stock options	10,698,010 583,614	107 6	76,101 2,927	1,630 _	<u>-</u> -	77,838 2,933

		_				
Balances, December 31, 1999 (as revised)	13,165,311	\$132	\$98,866	\$ 1,725	\$(2,513)	\$98,210
Net income	_	_	_	1,791	_	1,791
Shareholder note receivable	_	_	_	_	(2,513)	(2,513)
options	_	_	553	_	_	553
Tax benefit from exercise of stock				` '		, ,
Cash dividend on convertible preferred stock subject to redemption	_	_	_	(120)	_	(120)
stock	163,147	2	1,147	_	_	1,149
Conversion of convertible preferred stock subject to redemption to common						
subject to redemption	156,790	1	1,575	(1,576)	_	_
Dividend on convertible preferred stock	34,420	-	022			020
combinations Issuance of common stock under employee stock purchase plan	84,428	1	622	_	_	623
Issuance of common stock in business	1,479,322	15	15,941	_	_	15,956

The accompanying notes are an integral part of these consolidated financial statements.

F-32

Table of Contents

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

1. Summary of Significant Accounting Policies:

Nature of Operations:

Savoir Technology Group, Inc. (the "Company"), formerly Western Micro Technology, Inc., is a value-added distributor of commercial mid-range servers, peripheral equipment (including wireless networking equipment, storage products, printers and terminals) and software. The Company believes that it is one of the leading distributors of IBM's AS/400 and RS/6000 commercial mid-range servers. The Company also distributes commercial mid-range servers, peripheral equipment and software manufactured by NCR, Aironet and Telxon. The Company primarily distributes commercial mid-range servers and related products to VARs who generally incorporate commercial applications software and sell integrated computer systems to end user customers. The Company also integrates and configures personal computers, workstations and departmental servers for OEMs and VARs, and provides and remarkets installation and technical support services. The Company's primary sales offices and distribution centers, from which it ships products to customers throughout the United States, are located in Northern California and San Antonio, Texas. In addition to these locations, the Company has distribution centers in Massachusetts, Southern California, Georgia and Illinois, sales offices throughout the United States, Mexico and Canada.

Consolidated Financial Statement Presentation:

The consolidated financial statements include the accounts of the Company and its wholly owned subsidiaries. All significant intercompany accounts and transactions have been eliminated.

Estimates:

In preparing financial statements in conformity with generally accepted accounting principles, management is required to make estimates and assumptions that affect the reported amounts of assets and liabilities and the disclosure of contingent assets and liabilities at the date of the financial statements and the reported amount of revenues and expenses during the reporting period. Actual results could differ from those estimates.

Certain Risks and Concentrations:

The Company maintains cash balances with four major financial institutions. The Company sells its products to a broad geographic and demographic base of customers, extends trade credit, and generally does not require supporting collateral. To reduce credit risk, the Company performs ongoing credit evaluations of its customers, maintains an allowance for doubtful accounts and has credit insurance. One customer accounted for 16% and 22% of the outstanding accounts receivable balance at December 31, 1999 and 1998, respectively. No other customer accounted for more than 10% of the outstanding accounts receivable balance at December 31, 1999 and 1998.

Revenues are concentrated with a relatively limited number of customers and the providers of certain systems are concentrated among a few manufacturers. The loss of a major customer or the interruption of certain supplier relationships could adversely affect operating results. During the years ended December 31, 1997, 1998 and 1999, approximately 65%, 80% and 82%, respectively, of the Company's revenue was generated from the sale of products purchased from one of the Company's vendors, International Business Machines Corporation ("IBM").

Fair Value of Financial Instruments:

The carrying amounts of the Company's financial instruments including cash, accounts receivable, notes payable and accounts payable approximate fair value due to their short maturity. Based on borrowing rates currently available to the Company for similar debt, the carrying value of long-term debt approximates fair value.

F-33

Table of Contents

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS - (Continued)

Revenue Recognition:

The Company records revenue, net of allowance for estimated returns, at the time of product shipment.

Inventories:

Inventories, consisting primarily of purchased product held for resale, are stated at the lower of cost or net realizable value. Cost is determined using average and specific cost methods. The Company's inventories include high technology computer systems that may be specialized in nature and subject to rapid technological obsolescence. The Company does, however, have certain return privileges with many of its vendors. While the Company attempts to minimize the required inventories on hand and considers technological obsolescence when estimating required reserves to reduce recorded amounts to market values, it is reasonably possible that such estimates could change in the near term.

Property and Equipment:

Property and equipment are recorded at cost. Depreciation is recorded on a straight-line basis over the estimated useful lives, typically two to ten years. Leasehold improvements are amortized over the useful lives of the improvements or lease term, whichever is shorter.

When assets are sold or retired, the cost and related accumulated depreciation are removed from the accounts and the resulting gains or losses are included in income.

Excess of Cost over Acquired Net Assets and Other Intangibles:

The excess of cost over acquired net assets is being amortized on a straight-line basis over 15 and 20 year periods. Other intangibles are being amortized on a straight-line basis over their estimated useful lives which is typically 3 to 5 years. Amortization expense was \$1,400,000, \$4,636,000 and \$6,296,000 in 1997, 1998 and 1999, respectively. The Company reviews the carrying value of excess costs over acquired net assets and other intangibles for impairment whenever events or changes in circumstances indicate the carrying amount may not be recoverable. At December 31, 1999, the net unamortized balance of goodwill is not considered to be impaired.

Income Taxes:

The Company accounts for its income taxes using the liability method under which deferred tax assets and liabilities are determined based on differences between the financial reporting and tax bases of assets and liabilities and are measured using enacted tax rates and laws that will be in effect when the differences are expected to reverse. Valuation allowances are established when necessary to reduce deferred tax assets to amounts expected to be realized.

Market Development Funds:

Primary vendors provide the Company with market development funds in an amount that is generally based on purchases of the vendors' products and services. These funds typically range from 1% to 3% of such purchases and are required to be used to market and promote the vendors' products and services. The Company records these funds when earned as a reduction to offset direct costs of marketing, selling, general, and administrative expenses.

Stock-Based Compensation:

The Company accounts for stock-based compensation using the

F-34

Table of Contents

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

options is measured as the excess, if any, of the quoted market price of the Company's stock at the date of the grant over the amount an employee must pay to acquire the stock. The Company has adopted the disclosure-only provisions of Statement of Financial Accounting Standards No. 123, "Accounting for Stock-Based Compensation."

Net Income (Loss) Per Share Attributable to Common Stockholders:

Basic earnings per share ("EPS") is computed by dividing income attributable to common stockholders by the weighted average number of common shares outstanding for the period. Diluted EPS is computed giving effect to all dilutive potential common shares that were outstanding during the period. Dilutive potential common shares consist of incremental shares issuable upon exercise of stock options and warrants and conversion of preferred stock outstanding.

Recent Pronouncements:

In June 1998, the Financial Accounting Standards Board ("FASB") issued Statement of Financial Accounting Standards ("SFAS") No. 133, "Accounting for Derivative Instruments and Hedging Activities." SFAS No. 133, as amended by SFAS No. 137, is effective for all fiscal quarters of all fiscal years beginning after June 15, 2000. SFAS No. 133 requires that all derivative instruments be recorded on the balance sheet at their fair value. Changes in the fair value of derivatives which meet the definition of a hedge transaction are recorded each period in either earnings or other comprehensive income. The Company believes that the impact of the pronouncement will be minimal, as historically, the Company has not entered into any derivative or hedging transactions.

Reclassifications:

Certain amounts in the financial statements have been reclassified to conform with the current year's presentation. These classifications did not change previously reported total assets, liabilities, stockholders' equity or net income.

2. Long Lived Assets:

Voor	Endod	Docombor	24
Year	Fnded	December	31.

	1997	1998	1999
		(In thousands)	
sh flows from operating activities:	ф 2 21O	\$ 6.302	\$ 1.791
Net income Adjustments to reconcile net income to net cash provided by operating activities:	\$ 3,310	\$ 6,302	\$ 1,791
Write down of fixed assets and intangibles			7,492
Depreciation and amortization	2,670	6,331	8,886
Extraordinary loss on retirement of debt	2,070	2,773	0,000
Loss on sale of equipment	_	117	1
Provision for doubtful accounts receivable	472	842	1,702
Deferred taxes	(1,551)	(59)	(171
Accretion on long-term debt obligations	344	732	(
Change in assets and liabilities:	0	. 02	
Accounts receivable	(23,282)	(77, 229)	12,273
Inventories	(7,505)	(2,089)	1,848
Other current assets	(1,564)	(8,570)	(4,451
Accounts payable	32,312	71,080	(9,318
Accrued expenses and other liabilities	655	15,331	1,271
The trade of the trade t			
Net cash provided by operating activities	5,861	15,561	21,324
h flows from investing activities:			
Acquisition of businesses, net of cash acquired	(35,166)	(18,573)	(9,977
Proceeds from sale of fixed assets	- .	6	11
Acquisition of other assets	(988)	(2,665)	(2,660
Acquisition of property and equipment	(1,592)	(2,423)	(4,395
Proceeds from sale of investment	=	=	250
Net cash used in investing activities	(37,746)	(23,655)	(16,771
h flows from financing activities:			
Proceeds from short-term borrowings	72,761	301,023	487,532
Payments on short-term borrowings	(81,175)	(301,440)	(487,654
Payments on long-term debt obligations	(236)	(32,643)	`(1,311
Proceeds from issuance of common stock		28,589	· · · -
Proceeds from exercise of stock options and warrants	74	423	420
Proceeds from short term loan	_	15,000	5,000
Payments on short-term loan	_	· –	(5,000
Proceeds from employee stock purchase plan	256	297	623
Proceeds from issuance of long-term			
debt net of issuance costs	23,099	_	_
Proceeds from issuance of convertible preferred stock subject	•		
to redemption and warrants, net	19,082	(219)	_
Proceeds from equipment loans	559		_
Payment of cash dividend on convertible preferred stock			
subject to redemption	_	(35)	(120
Net cash provided by (used in) financing activities	34,420	10,995	(510
increase in cash	2,535	2,901	4,043
h — beginning of period	384	2,919	5,820
sh — end of period	\$ 2,919	\$ 5,820	\$ 9,863

F-35

Table of Contents

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

Excess of cost over acquired net assets and other intangibles:

December 31,			
1998	1999		
(In thousands)			

Computer and office equipment Leasehold improvements	\$10,119 1,482	\$10,995 1,937
Accumulated depreciation and amortization	11,601 (6,075)	12,932 (6,934)
	\$ 5,526	\$ 5,998

3. Borrowing Arrangements:

Notes Payable:

Excess of cost over net assets acquired Other intangibles

Accumulated amortization

December	31,	
		10

1998	1999
(In	thousands)
\$85,544	\$108,291
4,633	3,636
90,177	111,927
(6,367)	(10, 958)
\$83,810	\$100,969

The Company has an inventory and working capital financing agreement (the "IBMCC Credit Facility") with IBM Credit Corporation ("IBMCC"), an affiliate of IBM, whereby purchases from IBM and cash advances from IBMCC are directly charged to the IBMCC Credit Facility and are paid by the Company based on payment terms outlined in the agreement. Total borrowings under the IBMCC Credit Facility are based on eligible accounts receivable and inventory, as defined, and are limited to \$125,000,000. For the period December 27, 1999 through March 31, 2000 the line was temporarily increased to \$145,000,000. The IBMCC Credit Facility expires on August 31, 2000 and contains restrictive covenants which include the maintenance of minimum current ratio, tangible net worth and net profit after tax to revenue ratios, as defined, and is collateralized by substantially all assets of the Company. As of December 31, 1998 and 1999, the Company had outstanding borrowings under this agreement of \$113,697,000 and \$117,671,000, respectively. Of the total outstanding borrowings, \$6,228,000 and \$6,018,000 represented cash advances at December 31, 1998 and 1999, respectively, with the remainder included in accounts payable which amounted to \$107,469,000 and \$111,652,000, respectively. Cash advances bear interest at prime (8.5% and 7.75% as of December 31, 1999 and 1998, respectively) plus 1.875%. Based on eligible assets, as of December 31, 1999, the Company had borrowings available of approximately \$24,356,000. The weighted average interest rates for the Company's cash advances during 1998 and 1999 were 10.3% and 9.8%, respectively.

On September 8, 1998, the Company executed an amendment to its credit facility with IBMCC, pursuant to which the Company obtained a loan of \$15,000,000 to consummate the REAL transaction (See Note 12). On April 23, 1999, the Company executed an additional amendment to the IBMCC Credit Facility, pursuant to which the Company obtained an additional loan of \$5,000,000 to consummate the Enlaces transaction (See Note 12). These loans, included in other notes payable, bear interest at prime (8.5% and 7.75% at December 31, 1999 and 1998, respectively) plus 2.0% and are due in monthly installments through September 2000. The balance remaining at December 31, 1999 was \$15,000,200. In January 2000, the maturity date of each of the \$15,000,000 and \$5,000,000 loans was amended and extended through June 2001.

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

Long-Term Debt:

Working capital line

December 31,			
1998	1999		
(In	thousands)		
\$ 6,228	\$ 6,018		
15,012	15,000		
\$21,240	\$21,018		
Ψ==/=:0	421, 616		

Principal payments for long-term debt at December 31, 1999 are as follows:

Enlaces note payable Various notes payable
Less amounts due within one year
Long-term debt due after one year

December 31,			
1998	1999		
(In thou	sands)		
\$ -	\$ 5,000		
2,107	2,345		
2,107	7,345		
(1,020)	(2,969)		
\$ 1,087	\$ 4,376		

On April 23, 1999, in connection with the purchase of Enlaces (see Note 12), the Company recorded \$5,000,000 in long-term debt due in annual payments through April 2001.

On May 15, 1998, in connection with the acquisition of assets from UniDirect Corporation (see Note 12) the Company issued a \$1,700,000 promissory note to UniDirect. The promissory note bears interest at 8.25% and is payable over a two-year period.

4. Operating Lease Commitments:

The Company leases its warehouse and office space under operating leases. These leases expire on various dates from 2000 through 2004 and provide for payment of insurance, maintenance and property taxes. In addition, the Company leases certain equipment under operating leases and rental arrangements extending for periods of up to seven years.

The total rent expense, net of sublease income, was \$1,600,000, \$2,169,000 and \$2,275,000 for 1997, 1998 and 1999, respectively.

Future minimum rental commitments for all noncancelable operating leases are as follows:

2000 2001	\$2,969
	3,692
2002	174
2003	127
2004	383
	\$7,345

F-37

Table of Contents

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

5. Income Taxes:

Years Ending Decembe	r 31,
	(In thousands)
2000	\$ 3,116
2001	2,659
2002	2,515
2003	1,745
2004	678
Thereafter	423
	\$11,136

The Company's effective tax rate differs from the U.S. federal statutory tax rate as follows:

	Federal	State	Foreign	Total
1000		(In tho	usands)	
1999: Current Deferred	\$ 5,013 (171)	\$ 882 —	\$1,224 _	\$ 7,119 (171)
	\$ 4,842	\$ 882	\$1,224	\$ 6,948
1998: Current Deferred	\$ 6,055 (44)	\$1,585 (15)	\$ 687 —	\$ 8,327 (59)
	\$ 6,011	\$1,570	\$ 687	\$ 8,268
1997: Current Deferred	\$ 1,578 (1,301)	\$ 308 (250)	\$ - -	\$ 1,886 (1,551)
	\$ 277	\$ 58	\$	\$ 335

The components of the net deferred tax asset (included in other current assets) are as follows (In thousands):

Benefit resulting from utilization of federal NOL State taxes, net of federal benefit
Change in valuation reserve Foreign rates in excess of federal statutory rate Other

34%	35%	34%
15	10	28
(17)	_	_
8	6	10
(35)	_	-
_	_	6
4	(2)	1
9%	49%	79%

F-38

Table of Contents

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

The components of income before income tax expense and extraordinary item are as follows (in thousands):

	December 31,	
	1998	1999
Deferred tax assets: Accounts receivable reserve Accumulated depreciation and amortization Uniform inventory capitalization Inventory reserve Other nondeductible reserves Other	\$ 443 (13) 216 399 319 246	\$ 469 407 113 432 162 197
	\$1,610	\$1,780

6. Convertible Preferred Stock Subject To Redemption:

On September 19, 1997, the Company completed the private placement of 1,121,250 units (the "Units"). Each Unit consists of two shares of the Company's Series A Preferred Stock, par value \$0.01 per share, for an aggregate of 2,242,500 shares, at a purchase price of \$9.5625 per share, and one Common Stock purchase warrant (which expires in five years), par value \$0.01 per share, for an aggregate of 1,121,250 shares, at a purchase price of \$.125 per warrant and exercisable at a price of \$9.6875 per share for a total purchase price of \$19.25 per Unit. The Series A Preferred Stock has an eight percent (8%) cumulative dividend, payable in cash or Company Common Stock at the election of the Company and a potential special dividend should the price of the Company's Common Stock fall below \$9.5625 on each anniversary of the private placement. The special dividend may not exceed \$1.9125 per share each year. The Series A Preferred Stock is convertible at the option of the Holders, at any time, into Common Stock of the Company. The conversion price is \$9.3125 and is subject to adjustment if the Company issues any stock or securities at less than the conversion price. During the year ended December 31, 1998 and 1999, 256,000 and 136,488, respectively, of these shares were converted into 262,853 and 163,147, respectively, shares of the Company's common stock. Subsequent to September 19, 1998, the Company may redeem the Preferred Stock provided that the Company's Common Stock is trading at one hundred fifty percent (150%) of the conversion price (as adjusted) and the daily trading volume of the Company's stock is in excess of 125,000 shares, as defined. Subsequent to September 19, 2001, the Company may redeem the Preferred Stock at the conversion price (as adjusted). The holders of the preferred

stock are entitled to effectively redeem their shares of the preferred stock in the event of a change in control of the Company, as defined, at a value of \$9.6581 per share, payable in cash or property with similar value. In connection with the transaction, the Company issued warrants for the purchase of 112,125 shares of Common Stock to placement agents. The warrants are exercisable at \$9.6875 per share and expire in five years. Net proceeds totaled approximately \$18,900,000. The Company used the proceeds to pay down its line of credit and for general working capital purposes.

On September 19, 1998, the Company declared a special dividend, payable in common stock, due to the holders of the Company's Series A Preferred Stock. The Company issued 612,533 shares of its common stock with a fair market value of approximately \$3,800,000. The dividend was paid as a result of the average common stock price falling below \$9.5625 in the five trading days prior to September 19, 1998, as stipulated in the Series A Preferred Stock Agreement. At a special shareholders' meeting held on April 6, 1999, the Company's stockholders approved the amendment of certain provisions of its Series A Preferred Stock. Included in the adopted changes was the elimination of the special dividend provision, which potentially occurred on an annual basis dependent on the average stock price of the Company's Common Stock falling below a predetermined amount. Also included in the adopted changes was the lowering of the conversion price of the Preferred Stock from \$9.31 to \$8.00 per share.

F-39

Table of Contents

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

Activity of the convertible preferred stock subject to redemption is as follows:

	Domestic	Foreign	Total
1999	\$ 7,006	\$1,733	\$ 8,739
1998	15,346	1,562	16,908
1997	3,614	31	3,645

7. Stockholders' Equity:

Common Stock:

On April 29, 1998, the Company completed the public offering of 3,000,000 shares of its Common Stock at a price of \$10.50 per share. After deducting the underwriting discount and offering expenses, the net proceeds to the Company were \$28,589,000. The Company used the proceeds of the offering to reduce its outstanding debt obligations, exclusive of the outstanding cash advances on the IBMCC Credit Facility.

Warrants:

At December 31, 1998 and 1999, warrants were outstanding to purchase a total of 1,721,575 shares of Common Stock at exercise prices ranging from \$4.77 to \$9.69 per share. The warrants, which were issued in connection with various debt and equity financings, expire from 2002 to 2004. At December 31, 1998 and 1999, the Company had reserved 1,721,575 shares of Common Stock for issuance upon exercise of these warrants. During the year ended December 31, 1998, 11,800 warrants were exercised. No warrants were exercised in 1999.

Stock Option Plan:

Under the terms of the 1987 and 1994 Stock Option Plans, the Company may grant nonqualified or incentive stock options at prices not less than 85% and 100% of the market value at the grant date, respectively. To date, most options have been granted at 100% of the market value as of the date of grant. Generally, options vest and become exercisable in equal annual increments over four years beginning one year after the date of grant and expire five years after they become exercisable.

	Number of Shares	Amount (In thousands)
Balances, January 1, 1997 Issuance of convertible preferred stock subject to redemption, net	2,242,500	
Balances, December 31, 1997 Conversion of convertible preferred stock subject to	2,242,500	18,132
redemption to common stock Additional issuance costs	(256,000) 	(2,184) (219)
Balances, December 31, 1998 Conversion of convertible preferred stock subject to redemption to common stock	1,986,500 (136,488)	15,729 (1,149)
Balances, December 31, 1999	1,850,012	\$14,580

F-40

Table of Contents

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

	Shares		Options Outstanding		
	Available for Grant	Number of Shares	Price per Share	Total (in thousands)	
Balances, January 1, 1997. Additional shares reserved	64 700,000	929,581	\$2.00 - \$10.34	\$5,430	
Options granted	(610,679)	610,679	\$8.00 - \$12.75	6,506	
Options exercised	· · · · - ·	(24, 375)	\$2.25 - \$ 7.00	(74)	
Options terminated	79,875	(79,875)	\$2.13 - \$12.25	(629)	

At December 31, 1999, there were 2,580,045 shares of Common Stock reserved for issuance under the Company's stock option plans. Outstanding options for 991,507, 598,330 and 360,581 shares of Common Stock were exercisable, at December 31, 1999, 1998 and 1997, respectively, at weighted average exercise prices per share of \$9.02, \$6.43 and \$5.44.

The Company implemented an Employee Stock Purchase Plan (the "Plan") in November 1995, under which 675,000 shares of common stock have been reserved for issuance. The Plan is qualified under Section 423 of the Internal Revenue Code. The Plan allows for the purchase of stock at 85% of the lower of the closing stock price at the beginning or the end of each six-month purchase period. As of December 31, 1999, 191,080 shares have been issued under this Plan.

The following information concerning the Company's stock option and employee stock purchase plans is provided in accordance with SFAS No. 123, "Accounting for Stock-Based Compensation." The Company accounts for such plans in accordance with APB No. 25 and related interpretations.

The following table summarizes information with respect to stock options outstanding at December 31, 1999:

Outstanding and Exercisable By

Price Range as of 12/31/99

	Options Outstanding			
	Shares Available for Grant	Number of Shares	Price per Share	Total (in thousands)
Balances, December 31, 1997 Additional shares reserved	169,260 925,000	1,436,010	\$2.00 - \$12.75	\$11,233
Options granted	(1,178,500)	1,178,500	\$5.63 - \$13.00	9,846
Options exercised	(1,170,300)	(54,611)	\$2.00 - \$ 9.00	(309)
Options terminated	123,500	(123, 500)	\$2.25 - \$13.00	(1,047)
Balances, December 31, 1998	39,260	2,436,399	\$2.00 - \$13.00	\$19,723
Additional shares reserved	700,000		* *	
Options granted	(738, 500)	738,500	\$4.50 - \$11.44	4,866
Options exercised	=	(583,614)	\$2.13 - \$10.25	(2,933)
Options terminated	332,250	(332, 250)	\$2.00 - \$13.00	(3,118)
Plan shares expired	(6,875)	(5,125)	N/A	(57)
Balances, December 31, 1999	326,135	2,253,910	\$2.00 - \$13.00	\$18,481

[Additional columns below]

[Continued from above table, first column(s) repeated]

Options	Outstanding
----------------	--------------------

Number Outstanding As of 12/31/99	Weighted Average Remaining Contractual Life	Weighted Average Exercise Price
81,221	7.34	\$3.89
375,000	9.80	4.63
56,250	6.00	5.08
227,375	8.26	5.63
233,875	7.31	6.51
41,000	8.69	8.66
	0utstanding As of 12/31/99 81,221 375,000 56,250 227,375 233,875	Outstanding As of 12/31/99 Remaining Contractual Life 81,221 7.34 375,000 9.80 56,250 6.00 227,375 8.26 233,875 7.31

F-41

Table of Contents

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

Options Exercisable

Number Exercisable As of 12/31/99	Weighted Average Exercise Price
46,221	\$3.38
0	0.00
56,250	5.08
73,314	5.63
118,875	6.50
9, 250	8.74

[Additional columns below]

[Continued from above table, first column(s) repeated]

Options Outstanding

Range of Exercise Prices	Number Outstanding As of 12/31/99	Weighted Average Remaining Contractual Life	Weighted Average Exercise Price
\$ 9.00 - \$ 9.00	333,000	7.81	9.00
\$ 9.06 - \$10.00	110,500	8.08	9.61
\$10.25 - \$10.25	237, 250	8.09	10.25
\$10.34 - \$13.00	558, 439	7.72	11.71
\$ 3.38 - \$13.00	2,253,910	8.12	\$ 8.22

The fair value of each option grant has been estimated on the date of grant using the Black-Scholes option pricing model with the following weighted average assumptions used for grants in 1997, 1998 and 1999:

Options Exercisable

Number Exercisable As of 12/31/99	Weighted Averag Exercise Price	
159,000	9.00	
42,750	9.65	
179,000	10.25	
306,847	11.47	
991,507	\$ 9.02	

The weighted average expected life was calculated based on the exercise behavior of each group. Group A represents officers and directors who are a smaller group holding a greater average number of options than other option holders and who tend to exercise later in the vesting period. Group B are all other option holders, virtually all of whom are employees. This group tends to exercise earlier in the vesting period.

The weighted average fair value of those options granted in 1997, 1998 and 1999 was \$6.90, \$4.27 and \$6.26, respectively.

The Company has also estimated the fair value for the purchase rights issued under the Company's Employee Stock Purchase Plan under the Black-Scholes valuation model using the following assumptions:

Group A					
1997	1998	1999	1997	1998	1999
6.30	4.96	5.88	6.39	5.14	6.00
5 years 75.00%	5 years 56.00%	5 years 186.55%	4 years 75.00%	4 years 56.00%	4 years 186.55%

The weighted average fair value of those purchase rights granted in 1997, 1998 and 1999 was \$5.10, \$3.56 and \$5.80, respectively.

The following pro forma net income information has been prepared following the provisions of SFAS No. 123 (In thousands except per share data):

	1997	1998	1999
Risk-free interest rate	5.46	5.41	6.00
Expected life	0.5 years	0.5 years	0.5 years
Volatility	75.00%	56.00%	186.55%

The above pro forma effects on net income may not be representative of the effects on net income for future years as option grants typically vest over several years and additional options are generally granted each year.

F-42

Table of Contents

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

8. Earnings Per Share:

In accordance with the disclosure requirements of SFAS 128, a reconciliation of the numerator and denominator of basic and diluted EPS is provided as follows:

	Year Ended December 31,		
	1997	1998	1999
Net income (loss) — pro forma	\$1,624	\$ (252)	\$1,227
Basic net income (loss) per share — pro forma	\$ 0.23	\$(0.03)	\$ 0.10
Diluted net income (loss) per share — pro forma	\$ 0.21	\$ (0.03)	\$ 0.09

Options to purchase 1,419,144, 906,582 and 211,720 shares of Common Stock were outstanding at December 31, 1999, 1998 and 1997, respectively, but were not included in the calculation of net income per share because the options exercise price was greater than the average market price of the Common Shares. Additionally, the common stock issuable upon the assumed conversion of the convertible preferred stock subject to redemption was not included in the diluted earnings per share calculations for 1998 and 1999 as it would have been anti-dilutive.

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

9. Supplemental Cash Flow Information:

Supplemental Disclosures of Cash Flow Information:

Cash paid for interest and income taxes was:

Year Ended December 31,

	1997	1998	1999
	(In thou	sands, except per share	amounts)
umerator-basic and diluted EPS attributable to common	•		-
tockholders Income attributable to common stockholders, before			
extraordinary item	\$2,826	\$ 3,201	\$ 359
Extraordinary item, net of tax effect	_	(2,338)	-
Net income basic	2,826	863	359
Plus: impact of assumed preferred conversion	484	_	_
Net income diluted	\$3,310	\$ 863	\$ 359
enominator-basic EPS			
Weighted average shares outstanding	4,902	8,714	12,387
Basic EPS			
Income before extraordinary item	\$ 0.57	\$ 0.37	\$ 0.03
Extraordinary item, net of tax effect	-	(0.27)	_
Basic earnings per share	\$ 0.57	\$ 0.10	\$ 0.03
		_	
enominator-diluted EPS Denominator-basic EPS	4,902	8,714	12,387
Effect of dilutive securities:	4,902	0,714	12,307
Common stock options and warrants	453	629	564
Preferred stock subject to redemption	621	_	_
	5,976	9,343	12,951
Diluted EPS			
Income before extraordinary item	\$ 0.55	\$ 0.34	\$ 0.03
Extraordinary item, net of tax effect	=	(0.25)	=
Diluted earnings per share	\$ 0.55	\$ 0.09	\$ 0.03

Supplemental Disclosures of Noncash Investing and Financing Activities:

Year Ended December 31,

1997	1998	1999
\$2,700 \$ 706	(In thousands) \$3,495 \$4,495	\$ 2,769 \$11,225

Interest Income taxes

10. Contingencies:

The Company is from time to time subject to routine legal proceedings arising in the ordinary course of the Company's business and incidental to its business, none of which the Company expects to have a material adverse impact upon the Company's business, financial position or results of

On June 18, 1999 a complaint was filed in the Superior Court of Orange County, California by Lee Adams against Western Micro Technology, Inc., WMT Acquisition Corp. and the Company. Mr. Adams' complaint purports to allege causes of action for breach of contract, specific performance, accounting, common count and false promise, arising out of an Agreement and Plan of Reorganization, dated March 17, 1997, by and among Western Micro Technology, Inc., WMT Acquisition Corp., Target Solutions, Inc. and Lee Adams. In addition to equitable relief, the complaint seeks to recover compensatory damages in the amount of \$10 million and punitive damages according to proof. Mr. Adams has subsequently dismissed without prejudice his cause of action for accounting. The Company believes the complaint is without merit and intends to defend itself vigorously against this complaint. In addition, the Company has filed a cross-complaint for breach of contract in which it is seeking compensatory damages in the amount of \$1.5 million. The case has been set for trial on July 17, 2000.

The Company has employment agreements with several of its officers. Under the employment agreements, the officers are entitled to receive 6 to 24 months of their base salary in the event they are terminated without cause or for certain specified reasons. During the year ended December 31, 1999, the Company paid \$138,019 in connection with terminations under the employment agreements. The Company made no payments in 1997 and 1998.

F-44

Table of Contents

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS – (Continued)

11. Savings and Retirement Plan:

The Company maintains the "Western Micro Technology Savings and Retirement Plan," qualified under section 401(a) of the Internal Revenue Code. The Plan provides for tax deferred automatic salary deductions and alternative investment options. Employees are eligible to participate after completion of six months of employment. Participants may apply for loans from their accounts.

The Plan permits Company contributions determined quarterly by the Board of Directors. No contributions were made in the years ended December 31, 1997, 1998 or 1999.

12. Acquisitions and Investments:

On April 27, 1999, the Company completed the acquisition of Enlaces, an IBM distributor headquartered in Monterrey, Mexico, for approximately \$5,200,000 in cash and 235,638 shares of the Company's Common Stock (valued at \$8.49 per share). The acquisition has been accounted for as a purchase with the result

that Enlaces' operations are included in the Company's financial statements from the date of purchase. The agreement between the Company and Enlaces contains an earnout provision which allows the former shareholders of Enlaces to earn up to an additional \$8,000,000, of which \$5,000,000 is guaranteed. The guaranteed earnout has been recorded as a liability by the Company. In connection with the acquisition, the Company recorded approximately \$11,600,000 of goodwill and other intangible assets.

Enlaces consists of a group of companies that sell IBM AS/400 and RS/6000 mid-range products and services, and includes Enlaces Computacionales and Enlaces y Asociados. It also includes Instituto de Educacion Avanzada, which is an IBM training center and provides sales, technical support and administrative services. For the year ended December 31, 1998, Enlaces had unaudited revenues of approximately \$17,000,000.

On January 4, 1999, the Company acquired certain assets of Infinite Solutions, Inc. ("Infinite") for \$2,750,000 in cash and 88,560 shares of the Company's Common Stock (valued at \$8.47 per share). The acquisition has been accounted for as a purchase with the result that Infinite's operations are included in the Company's financial statements from the date of purchase. In connection with the acquisition, the Company recorded approximately \$3,600,000 of goodwill and other intangible assets.

Infinite is an Atlanta, Georgia based specialty distributor and integrator of computer systems and software. For the year ended December 31, 1998, Infinite had unaudited revenues of approximately \$19,000,000.

On September 8, 1998, the Company acquired the distribution segment of REAL Applications, Ltd. ("REAL") from its parent company, El Camino Resources, Ltd. ("El Camino") for \$12,875,000. In connection with the acquisition, the Company recorded approximately \$13,141,000 of goodwill and other intangible assets. The acquisition has been accounted for as a purchase with the result that REAL operations are included in the Company's financial statements from the date of purchase.

REAL is a reseller of IBM commercial mid-range servers, including the AS/400 and RS/6000, as well as high end data storage and IBM Netfinity servers. REAL had audited revenues of approximately \$80,000,000 for the year ended April 30, 1998.

The following presents unaudited pro forma combined net sales, net income and earnings per share of the Company and REAL as if the combination had occurred at the beginning of the earliest period presented. The pro forma information is presented for informational purposes only, and is not necessarily indicative of the operating results that would have occurred if the REAL acquisition had been consummated at the earliest period presented, nor is it indicative of future operating results.

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

Year Ended December 31,

	1997	1998	1999
		(In thousands)	
Tax benefit from exercise of stock options	\$ 489	\$ -	\$ 553
Common stock issued in connection with business combinations	6,825	9,739	15,956
Dividend on preferred stock	50	1,803	1,576
Special dividend on preferred stock	_	3,799	· –
Common stock warrants issued in connection with notes payable			
issuance	1,330	_	_
Common stock warrants issued in connection with preferred stock	1,000	_	_
Common stock warrants revaluation	· –	1,391	_
Shareholder note receivable (see Note 16)	_	· _	2,513

On June 5, 1998, the Company acquired all of the outstanding shares of MCBA Systems, Inc.'s ("MCBA") common stock for \$487,000 of canceled indebtedness and 852,854 shares of the Company's common stock. The acquisition has been accounted for as a purchase with the result that MCBA operations are included in the Company's financial statements from the date of purchase. In connection with the acquisition, the Company recorded approximately \$9,800,000 of goodwill and other intangible assets. The fair values of assets acquired from MCBA were approximately \$4,045,000 and liabilities assumed were approximately \$3,945,000. In addition, the selling stockholders of MCBA earned an additional 1,500,000 shares of the Company's common stock based upon attainment of certain performance goals in calendar 1998 and 1999. These shares were accounted for as an increase in the purchase price which resulted in the Company recording an additional \$11,000,000 of goodwill and other intangible assets.

MCBA is a value-added wholesale distributor and reseller of IBM commercial mid-range servers, including the AS/400 and RS/6000, and software. MCBA is also one of four authorized distributors of IBM's S/390 mainframe systems. In addition to the products it offers, MCBA provides network configuration and technical support services to its customers. MCBA's audited revenues for the year ended December 31, 1997 were approximately \$26,900,000 with net income of approximately \$26,000.

On May 15, 1998, the Company purchased substantially all of the UniDirect catalog and VarCity distribution business and related electronic software distribution organization of UniDirect Corporation ("UDC") for \$4,600,000, comprised of \$2,900,000 in cash and a \$1,700,000 promissory note. The promissory note bears interest at 8.25% and is payable over a two-year period. This transaction has been accounted for as a purchase with the result that UDC operations are included in the Company's financial statements from the date of purchase. In connection with the acquisition, the Company recorded approximately \$4,900,000 of goodwill and other intangible assets. During the year ended December 31, 1997, the acquired businesses had unaudited revenues of approximately \$18,000,000. The Company has included the unamortized balance of goodwill and other intangible assets of the VarCity distribution business in the restructuring charge for the year ended December 31, 1999 (see Note 15).

UDC specializes in the sales and support of technically advanced software products. Software solutions include products for areas such as web and intranet development and related management tools, UNIX and Windows NT applications and utilities, client/server applications, advanced operating systems and databases, and PC connectivity applications.

On September 30, 1997, the Company acquired all of the capital stock of SMS for an aggregate of \$42,150,000 in cash at closing, 460,000 shares of the Company's Common Stock, valued at \$3,887,000, and additional cash payments of approximately \$6,870,000, made in May of 1998. In addition, the selling stockholders have earned \$5,000,000 in cash payments based upon attainment of certain performance goals. The acquisition has been accounted for as a purchase with the result that SMS operations are included in the Company's financial statements from the date of purchase. In connection with the acquisition, the Company recorded approximately \$47,600,000 of goodwill and other intangible assets. The fair value of assets acquired from SMS was approximately \$41,400,000 and liabilities assumed were approximately \$36,100,000. SMS was a holding company for a family of companies including Star Data Systems, Inc., dba Sirius Computer Solutions ("Sirius"), a value-added distributor for high technology mid-range solutions in the IBM AS/400 and RS/6000 systems market. Sirius also sells systems directly to end user customers as an industry remarketer. Prior to the closing of the SMS acquisition, SMS completed a spin-off of the Sirius end user

F-46

Table of Contents

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

business as a separate unaffiliated company. Upon acquiring SMS, the distribution arm was renamed Business Partner Solutions, Inc. ("BPS"). Upon completion of the SMS acquisition, BPS became a wholly owned subsidiary of the Company. For the eleven months ended September 30, 1997, the distribution business of Sirius had revenues of approximately \$86,500,000 and income from operations of approximately \$1,100,000. The following presents unaudited pro forma combined net sales, net income and earnings per share of the Company and SMS (excluding the Sirius end user business) for the fiscal year ended December 31, 1997. The pro forma information is presented for informational purposes only, and is not necessarily indicative of the operating results that would have occurred if the SMS acquisition had been consummated at the beginning of the earliest period presented, nor is it indicative of future operating results.

	Year Ended December 31, 1997	Year Ended December 31, 1998
Net Sales	\$299,927,000	\$646,575,000
Net Income	\$ 3,360,000	\$ 8,300,000
Net Income per share — basic	\$ 0.58	\$ 0.33
Net Income per share — diluted	\$ 0.56	\$ 0.31

The above amounts do not include pro forma adjustments for sales that would have occurred between the distribution business of SMS and the end user business if the spinoff of the end user business had occurred at the beginning of such period presented. Including these sales amounts, pro forma combined net sales would have been approximately \$336,000,000 for the year ended December 31, 1997.

On March 17, 1997, the Company acquired all of the common stock of Target Solutions, Inc. ("TSI"), a privately

held company, for approximately \$2,200,000, paid in common stock (220,273 shares) of the Company. Additional consideration, up to \$10,000,000 in cash and stock, can be earned by TSI by meeting certain defined gross profit targets through fiscal year 1998. An additional 62,578 shares were issued by the Company on March 17, 1997, and placed in escrow for the earn-out provision. No shares were earned in 1997 or 1998. The unearned shares will be returned to the Company. The acquisition has been accounted for as a purchase with the result that TSI operations are included in the Company's financial statements from the date of purchase. Additional amounts payable under contingent earn-out provisions will be accounted for as an increase in the purchase price when such amounts become determinable. In connection with the acquisition, the Company recorded approximately \$2,600,000 of goodwill and other intangible assets. The fair value of assets acquired from TSI was approximately \$1,141,000 and liabilities assumed were approximately \$1,484,000. The Company has included the unamortized balance of goodwill and other intangible assets of the TSI operations in the restructuring charge for the year ended December 31, 1999 (See Note 15).

13. Segment Reporting:

The Company has adopted the Financial Accounting Standards Board's Statement of Financial Accounting Standards No. 131, "Disclosures about Segments of an Enterprise and Related Information" ("SFAS 131").

The Company has two operating segments: the Mid-Range Systems Division ("MRS") and the Computer and Peripherals Group ("CPG"). Products distributed by MRS include mid-range servers that run on UNIX, OS/400 and NT operating systems, peripheral equipment (including wireless networking equipment, storage products, printers and terminals) and software. Through CPG, the Company offers its customers value-added systems integration services up to, and including, installation (e.g., "turnkey" systems assembly of departmental servers, workstations, hardware and software "bundling" and light manufacturing). The accounting policies of the segments are the same as those described in the summary of significant accounting policies. Although management measures the profitability of its business through the results of

F-47

Table of Contents

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

these two segments, the Company's segments have similar economic characteristics and, as such, the results of operations have been aggregated and separate disclosure is not presented.

Foreign sales for the years ended December 31, 1997, 1998 and 1999 were approximately \$6,700,000, \$31,000,000 and \$47,200,000, respectively.

During the years ended December 31, 1997, 1998 and 1999, approximately 65%, 80% and 82%, respectively, of the Company's net sales were generated from the sale of IBM products.

One customer accounted for approximately 11%, 18% and 16% of the Company's net sales in 1997, 1998 and 1999, respectively. No other single customer accounted for more than 10% of the Company's net sales.

Long-lived assets in foreign countries as of December 31, 1999 and 1998 were approximately \$11,066,642 and \$4,440, respectively. There were no long-lived assets in foreign countries as of December 31, 1997.

14. Extraordinary Charge:

During the year ended December 31, 1998, in connection with the repayment of its outstanding debt obligations, the Company recorded an extraordinary charge of \$2,338,000, net of tax, resulting from a prepayment penalty, the write off of unamortized discounts relating to certain warrants issued to debt holders and other related expenses.

15. Restructuring Charge:

The consolidated statements of operations for 1999 include pre-tax charges totaling \$11.3 million related to the restructuring of corporate functions and the recognition of impaired assets. Of this amount, \$2.8 million represents costs associated with the consolidation of facilities and elimination of redundant back-office functions. The costs associated with the restructuring principally include lease termination costs (\$500,000), leasehold and other asset disposals (\$1.4 million) and personnel (a total of 38 employees, primarily in administration, finance and information technology) severance costs (\$900,000). The remaining \$8.5 million represents asset impairment charges related to recorded goodwill and fixed assets. These impairment charges principally include the write off of goodwill (\$7.0 million) related to unprofitable business activities acquired in previous purchases (Target Solutions, Inc., International Data Products, LLC and UniDirect, Inc.), the operations of which the Company has terminated as part of its restructuring. The balance of the impairment charge relates to the write off of unproductive electronic commerce assets (\$1.5 million).

As of December 31, 1999, the Company had approximately \$800,000 remaining in its restructuring and asset impairment reserve, primarily consisting of severance. The Company expects that all activities and expenditures related to this restructuring will be completed by June 30, 2000.

Activity in the accrued restructuring liabilities recorded in the third and fourth quarter of 1999 is as follows (in thousands):

Year Ended December 31, 1997

Net sales Net income Net income per share-basic Net income per share-diluted \$306,000,000 \$ 899,000 \$ 0.17 \$ 0.14

SAVOIR TECHNOLOGY GROUP, INC. AND SUBSIDIARIES

NOTES TO CONSOLIDATED FINANCIAL STATEMENTS — (Continued)

16. Executive Retention Agreement:

During the year ended December 31, 1999 the Company executed an Executive Retention Agreement (the "Agreement") with its Chief Executive Officer, P. Scott Munro ("Mr. Munro"). Included in the agreement is a provision for a recourse loan ("loan") to Mr. Munro for approximately \$2,500,000 with a stated interest rate of 4.9%. Mr. Munro executed the loan provision on May 10, 1999 and used the proceeds to exercise approximately 480,000 options under the revised terms of the Company's Employee Stock Option Plan. The loan is due and payable, including principal and interest, in May 2002. If Mr. Munro remains employed with the Company, the loan, and accrued interest, will be forgiven as follows: \$800,000 after the first quarter of 2000 and the balance on May 9, 2002. The \$2,500,000 note receivable from Mr. Munro is shown as a reduction to stockholders' equity. The loan provision of the Agreement also allows for Mr. Munro to borrow up to an additional \$1,100,000 under the same terms and conditions noted above. On September 10, 1999, Mr. Munro executed the remaining portion of the loan provision and borrowed the additional \$1,100,000 available to him. The \$1,100,000 note receivable from Mr. Munro is included in other assets. Upon the occurrence of a Change of Control (as defined in the agreement), Mr. Munro's obligation to pay the then outstanding balance of the loan (including unpaid principal and accrued interest thereon) shall be forgiven and the promissory notes shall be canceled. As noted in Note 17 the Company has agreed to merge with a subsidiary of Avnet, Inc. ("Avnet"). If the acquisition of the Company by Avnet is completed, the loan will be forgiven.

17. Subsequent Event:

On March 2, 2000 the Company agreed to merge with a subsidiary of Avnet, so that it would become a wholly owned subsidiary of Avnet. The acquisition has been approved by the boards of directors of both companies and is subject to approval by the stockholders of the Company and regulatory review. If the merger is completed, depending upon the average of the closing trading price of Avnet's common stock during the 15 days ending on the fifth day prior to the vote of the Company's stockholders, the Company's common stockholders will receive between 0.15494 and 0.11452 of a share of Avnet common stock for each share of the Company's stock they own. The transaction is expected to close in June 2000. If the merger is not completed, the Company may be required to pay Avnet a termination fee of \$750,000, and the Company may be required to pay Avnet an additional fee of \$3.75 million if it enters into an acquisition agreement with another company within a year of termination of the Avnet merger agreement.

Avnet is a Phoenix, Arizona based distributor of semiconductors, interconnect, passive and electromechanical components and computer products. At its last fiscal year end, Avnet reported

sales of approximately \$6.35 billion.

18. Revision:

Previously, the Company classified its outstanding convertible preferred stock as a component of stockholders' equity. As a result of the change in control provision discussed in Note 6, the Company has revised the consolidated financial statements to present the convertible preferred stock subject to redemption as convertible preferred stock outside of stockholders' equity. The impact of this revision on stockholders' equity is as follows:

	Restructuring Charge	Paid/ Written Off	Accrued at December 31, 1999
Goodwill and electronic commerce asset	\$ 8,500	\$ 8,435	\$ 65
Leasehold and other asset disposals	1,446	1,446	_
Severance	897	531	366
Lease termination	444	111	333
Total restructuring costs	\$11,287	\$10,523	\$764

This revision did not affect previously reported net income.

F-49

Table of Contents

Item 9. Changes in and Disagreements With Accountants on Accounting and Financial Disclosure

Not applicable.

PART III

Item 10. Directors and Executive Officers of the Registrant

Executive Officers And Directors

The following table sets forth certain information regarding the executive officers and directors of the Company as of February 24, 2000.

	December 31,		
	1997	1998	1999
Stockholders' equity, as previously reported Revision	\$ 47,080 (18,132)	(in thousands) \$ 93,567 (15,729)	\$112,790 (14,580)
Stockholders' equity, as revised	\$ 28,948	\$ 77,838	\$ 98,210

Dogombor 21

Name	Age	Position
P. Scott Munro(1)	43	Chairman of the Board, Chief Executive Officer and Secretary
Carlton Joseph Mertens II	34	Chief Operating Officer, President and Director
Dennis J. Polk	33	Senior Vice President of Corporate Finance
Robert O'Reilly	47	Senior Vice President of Human Resources
Angelo Guadagno(2)	57	Director
Mike Gunnells(3)	39	Director
Guy M. Lammle(1)	47	Director
K. William Sickler(3)	50	Director
J. Larry Smart(1)(2)	51	Director

P. Scott Munro has served as Chief Executive Officer and Secretary of the Company since July 1995. Mr. Munro has also served as Chairman of the Board since January 1998 and served as a Director of the Company from July 1995. From January 1993 to July 1995, Mr. Munro was President, Computer Systems Division of the Company, and from July 1990 to January 1993, he served as Senior Vice President, Computer Systems Division of the Company. Prior to 1990, Mr. Munro served as a General Manager for both Future Electronics, Inc., a distributor of electronic components, and Arrow Electronics, Inc., a distributor of computer products.

Carlton Joseph Mertens II has served as a Director of the Company and as Chief Operating Officer since September 1997. From 1984 to September 1997, Mr. Mertens was an active member of the management team of SMS, a computer systems integration company, and served as Executive Vice President of SMS from 1991 to September 1997 prior to its acquisition by the Company.

Dennis J. Polk currently serves as Senior Vice President of Corporate Finance, but he has resigned effective March 31, 2000. From 1988 to 1995, Mr. Polk held staff and management positions at Grant Thornton LLP. In 1995, Mr. Polk joined Savoir as corporate controller and has held several positions in finance and operations. Mr. Polk is a Certified Public Accountant.

Robert O'Reilly has served as Senior Vice President of the Company since September 1997 and prior to that time served as Vice President of Human Resources from February 1996 and Director of Human Resources from September 1995. Prior to joining the Company, Mr. O'Reilly was a Director of Human Resources, with special emphasis on recruitment, training and development, during his 10-year career with Future Electronics, Inc., a distributor of electronic components.

Angelo Guadagno has served as a Director of the Company since August 1997. Currently Mr. Guadagno is an adjunct professor at Boston University and Chief Executive Officer of Skycady, Inc., a Boston,

F-50

Table of Contents

During his tenure at Data General he also served as: Vice President of U.S. Sales from 1994 to 1996; Vice President, American Sales and Services from 1990 to 1994; and Vice President of North American Sales Division from 1989 to 1990.

Mike Gunnells has served as a Director of the Company since October 1999. Since June 1998, he has served as President and CEO of MCBA Inc., an e-commerce consulting company. From 1992 to 1998, Mr. Gunnells was President and CEO of MCBA Systems, a computer systems integration company.

- Guy M. Lammle has served as a Director of the Company since August 1998. He has served as Chairman of the Board, Chief Executive Officer and Director of NxTrend Technology, Inc., a software and sales company, since February 1980 and assumed the additional title of President in October 1996. From 1974 to 1980, Mr. Lammle was employed with the Burroughs Corporation, a computer hardware company, progressing from Sales Representative to Zone Manager and finally District Product Manager.
- K. William Sickler has served as a Director of the Company since July 1993. Mr. Sickler has served as Chief Executive Officer and President of Gadzoox Networks, Inc., a provider of gigabit fibre channel networking products, since April 1996. From July 1995 to April 1996, he was Executive Director of Software Business Development for Seagate Technology, a software developer and manufacturer of disk drives, with responsibility for analysis of potential software company acquisitions. From December 1992 to July 1995, Mr. Sickler was President and Chief Executive Officer of Network Computing, Inc., a provider of network management software for local area networks.
- J. Larry Smart has served as a Director of the Company since October 1995. From October 1995 to January 1998, he also served as Chairman of the Board of the Company. Since July 1999, Mr. Smart has served as Chairman of the Board of Visioneer, Inc., a developer of personal desktop management products. From March 1997 to March 1999, Mr. Smart served as President and Chief Executive Officer of Visioneer, Inc. (prior to Visioneer, Inc. merging with ScanSoft, Inc.), and served as Chairman of the Board of Directors of that company from February 1997 until assuming the position of President and Chief Executive Officer. From July 1995 until March 1997, he was Chairman of the Board, President and Chief Executive Officer of StreamLogic Corporation, a data storage company. From March 1994 to February 1995, Mr. Smart was President and Chief Executive Officer of Maxtor Corporation, a data storage company. From July 1991 to February 1995, Mr. Smart was President and Chief Executive Officer of Southwall Technologies, Inc., a materials sciences company.

The Company currently has authorized seven (7) members of the Board of Directors. All directors are elected to hold office until the next annual meeting of stockholders of the Company and until their successors have been duly elected and qualified. Officers are elected at the first meeting of the Board of Directors following the stockholders' meeting at which the directors are elected and serve at the discretion of the Board of Directors. There are no family relationships among any of the directors or executive officers of the Company.

Section 16(a) Beneficial Ownership Reporting Compliance.

Section 16(a) of the Securities Exchange Act of 1934,
as amended (the "Exchange Act"), and the rules of the
Securities and Exchange Commission (the "Commission")
thereunder require the Company's directors, executive
officers and the beneficial owners of more than ten percent of
the Company's common stock to file reports of their

ownership and change in ownership of the Company's common stock with the Commission. Based solely upon a review of such reports filed electronically with the Commission, the Company believes that all reports required by Section 16(a) of the Exchange Act to be filed by its directors, executive officers and ten percent beneficial stockholders during the last fiscal year were filed on time, except that Mike Gunnells, appointed to the Board in October 1999, will be filing a late Form 3.

F-51

Table of Contents

Item 11. Executive Compensation

The following table provides certain summary information for the years ended December 31, 1997, 1998 and 1999 concerning compensation paid to the Company's Chief Executive Officer and to the Company's three other named executive officers whose compensation exceeded \$100,000 in 1999 and one individual who would have been a named executive officer but for the fact that he was not serving as an executive officer at December 31, 1999 (the "Named Executive Officers").

Summary Compensation Table

- (1) Member of the Nominating Committee.
- (2) Member of the Compensation Committee.
- (3) Member of the Audit Committee.

	Annual Compensation					
					Long-Term Compensation Awards	
Name and Principal Position	Fiscal Year	Salary(\$)	Bonus(\$)	Other Annual Compensation(\$)	Securities Underlying Options(#)	All Other Compensation(\$)
P. Scott Munro	1999	495,000	229,500			
Chairman of the Board, Chief	1998	400,406	261,049	_	450,000	_
Executive Officer and	1997	302,485	119,398	_	150,000	_
Secretary	1999	350,000	128,587	_	100,000	_
Carlton J. Mertens(1)	1998	270,000	148,717	_	_	_
President and Chief	1997	67,579	32,500	_	45,716	_
Operating Officer	1999	235,917	37,500	_	-, -	_
James W. Dorst(2)	1998	227,750	74,610	_	50,000	_
Chief Financial Officer	1997	192,333	51,872	_	30,000	_
Dennis J. Polk(3) Senior Vice President of Corporate Finance	1999	166, 675	39, 375	-	75,000	-
Robert O'Reilly(4)	1999	200,000	42,500	_	25,000	_
Senior Vice President of	1998	176,000	62,175	_	50,000	_
Human Resources	1997	142,517	36,578	_	30,000	_

F-52

The following table sets forth certain information regarding options granted during the fiscal year ended December 31, 1999 to the Named Executive Officers.

Option Grants in Last Fiscal Year

- (1) Mr. Mertens became an executive officer of the Company during 1997.
- (2) Mr. Dorst resigned from the Company during 1999.
- (3) Mr. Polk became an executive officer of the Company in 1999 and has resigned effective March 31, 2000.
- (4) Mr. O'Reilly became an executive officer of the Company during 1997.

	Securities Total Opti Underlying Granted	Percent of Total Options Granted to Employees in	ns Exercise or	Expiration	Potential Realizable Value at Assumed Annual Rates of Stock Price Appreciation for Option Term(\$)(3)	
Name	Granted(#)(1)	Fiscal Year(2)	(\$/Share)	Date	5%	10%
P. Scott Munro						
Carlton J. Mertens	100,000	13.54%	9.00	1/4/09	566,005	1,434,368
James W. Dorst	_	=	_	_	_	_
Dennis J. Polk	25,000	3.39%	9.00	1/4/09	141,501	358,592
	50,000	6.77%	4.63	10/19/09	145,182	368,303
Robert O'Reilly	25,000	3.39%	4.63	10/19/09	72,591	184,151

The following table shows the stock options exercised during the fiscal year ended December 31, 1999 and the number of shares of Common Stock represented by outstanding stock options held as of December 31, 1999 by each of the Named Executive Officers.

Aggregate Option Exercises in Last Fiscal Year and Fiscal Year-End Option Values(1)

- (1) These options have a 10-year term and vest at the rate of twenty-five (25%) per year over a four-year period. These options have accelerated vesting upon a change of control of the Company.
- (2) Based on options to purchase an aggregate of 738,500 shares of Common Stock granted during fiscal 1999.
- (3) Amounts represent hypothetical gains that could be achieved for the respective options if exercised at the end of the option term. These gains are based on assumed rates of stock appreciation of five percent (5%) and ten percent (10%) compounded annually from the date the respective options were granted to their expiration date and are not presented to forecast possible future appreciation, if any, in the price of the Common Stock. The gains shown are net of the option exercise price, but do not include deductions for taxes or other expenses associated with the exercise of the options or the sale of the underlying shares of Common Stock. The actual gains, if any, on the stock option exercises will depend on the future performance of the Common Stock, the optionee's continued employment through applicable vesting periods and the date on which the options are exercised.

	Acquired Shares Ex	of Shares on Exercise ercised Value lized(\$)	Number of Securities Underlying Unexercised Options at Fiscal Year-End Exercisable/Unexercisable	Value of Unexercised In- the-Money Options at Fiscal Year-End(\$) Exercisable/Unexercisable	
P. Scott Munro(2)	476,718	\$1,777,749	25,000/ -	\$28,125/ —	
Carlton J. Mertens	_	_	22,858/122,858	-/ -	
James W. Dorst	50,000	179,371	6,250/ -	7,031/ -	
Dennis J. Polk	· –	· —	30,000/110,000	19,688/\$150,063	
Robert O'Reilly	_	_	53,750/ 96,250	32,344/ 99,406	

- (1) Based on a per share price of \$7.125, the closing price of the Common Stock as reported by The Nasdaq National Market on December 31, 1999, the last trading day of the fiscal year.
- (2) Pursuant to an Amendment of Stock Option Agreements, dated as of May 10, 1999, between the Company and Mr. Munro, the Company has accelerated the vesting of 385,000 shares underlying stock options so that they are fully vested. In addition, the Company has permitted Mr. Munro to exercise stock

Compensation of Directors

The Company's outside directors (i.e., those who are not employees of the Company) receive an annual retainer of \$20,000, plus \$750 for each board meeting attended. The Company pays for directors' liability insurance and has entered into indemnification agreements with each of its directors. At the 1994 Annual Meeting of Stockholders, the stockholders approved the adoption of the 1994 Stock Option Plan, which has been amended several times since its adoption. The 1994 Stock Option Plan, as amended, provides for the grant of an option of 15,000 shares of the Common Stock to certain directors who are not employees following their initial election or appointment and an option for 4,000 shares of Common Stock at every regular annual meeting at which they are re-elected. The exercise price of the options is the fair market value of the Common Stock on the date of each respective grant and the options vest over a four-year period. In September 1998 and June 1999, the Board of Directors made discretionary grants of nonstatutory stock options to each of the outside directors to purchase 10,000 and 4,000 shares, respectively, of Common Stock at an exercise price of \$6.00 and \$9.50 per share, respectively.

Change in Control

Stock options held by the Company's directors and Named Executive Officers under the Company's Stock Option Plans will become fully vested and exercisable following a change in control of the Company. A "change in control" means the occurrence of any of the following events: (i) stockholder approval of a merger or consolidation of the Company with any other corporation resulting in a change in fifty percent (50%) or more of the total voting power of the Company; (ii) stockholder approval of a plan of complete liquidation of the Company or an agreement for the sale or disposition of all or substantially all of the Company's assets; or (iii) any person becomes the beneficial owner of more than fifty percent (50%) of the Company's total outstanding securities. Further arrangements concerning change in control are described below under "Employment Agreements."

Employment Agreements

The Company has entered into an employment agreement with P. Scott Munro dated January 22,1998. Pursuant to the terms of the agreement, as amended, Mr. Munro receives a base salary of \$495,000 per year and is eligible to receive a bonus of up to \$270,000 per year, subject to achievement of certain performance goals. If Mr. Munro is terminated without cause, he will be entitled to receive his base salary for twelve (12) months following his termination. The Company has entered into an Executive Retention Agreement with P. Scott Munro dated as of May 10, 1999. Included in the Executive Retention Agreement is a provision which provides that if Mr. Munro is terminated without cause or his responsibilities are reduced within twelve (12) months following a change in control and such reduction in responsibilities is not for cause, any resignation of employment by Mr. Munro as a consequence of such reduction in responsibilities will be treated as a termination of employment without cause and he will receive a severance payment equal to 200% of his annual base salary plus his annualized target incentive. Also included in the Executive

Retention Agreement is a provision for a recourse loan to Mr. Munro for approximately \$2,500,000 with a stated interest rate of 4.9%. Mr. Munro executed the loan provision on May 10, 1999 and used the proceeds to exercise approximately 480,000 options under the revised terms of the Company's Employee Stock Option Plan. The loan is due and payable, including principal and interest, in May 2002. If Mr. Munro remains employed with the Company, the loan, and accrued interest, will be forgiven as follows: \$800,000 after the first quarter of 2000 and the

F-54

Table of Contents

balance on May 9, 2002. The loan provision of the Executive Retention Agreement also allows for Mr. Munro to borrow up to an additional \$1,100,000 under the same terms and conditions noted above. On September 10, 1999, Mr. Munro executed the remaining portion of the loan provision and borrowed the additional \$1,100,000 available to him. Upon the occurrence of a change of control (as defined in the Executive Retention Agreement), Mr. Munro's obligation to pay the then outstanding balance of the loan (including unpaid principal and accrued interest thereon) shall be forgiven and the promissory notes shall be canceled. If the acquisition of the Company by Avnet is completed, Mr. Munro's loan will be forgiven.

The Company has also entered into employment agreements with other management personnel as follows: (i) an employment agreement with Dennis J. Polk dated November 1, 1998, amended on January 11, 1999 and November 1, 1999, pursuant to which Mr. Polk receives a base salary of \$225,000 per year and is eligible to receive a bonus of six months base salary in the event of a change in control; and (ii) an employment agreement with Robert O'Reilly dated January 22, 1998, pursuant to which Mr. O'Reilly receives a base salary of \$200,000 per year and is eligible to receive a bonus of up to \$50,000 per year. In addition to the foregoing, pursuant to each of their agreements with the Company, if Mr. Polk or Mr. O'Reilly is terminated for cause, such person will be entitled to receive his base salary and bonus through the date of his termination. If Mr. Polk or Mr. O'Reilly is terminated without cause, such person will be entitled to receive his base salary for six (6) months following his termination. If Mr. Polk's or Mr. O'Reilly's responsibilities are reduced twelve (12) months following a change in control and such reduction in responsibilities is not for cause, any resignation of employment as a consequence of such reduction in responsibilities will be treated as a termination of employment without cause.

The Company has entered into an employment agreement with Carlton Joseph Mertens II dated September 30, 1997, pursuant to which Mr. Mertens receives a base salary of \$350,000 per year and is eligible to receive a bonus of up to \$150,000 per year, subject to achievement of certain performance goals. If Mr. Mertens is terminated for cause, he will be entitled to receive his base salary and bonus due through the date of his termination. If Mr. Mertens is terminated without cause or if Mr. Mertens terminates his employment with the Company for certain specified reasons, he will be entitled to receive his base salary for nine (9) months following his termination; such reasons include assignment or alteration by the Company of Mr. Mertens' duties, responsibilities or obligations materially inconsistent with his position with the Company after notice of Mr. Mertens' objections thereto, failure of the Company to provide to Mr. Mertens the salary or bonuses

described above, relocation of the Company's IBM Operational Headquarters outside of San Antonio, Texas, any requirement by the Company for Mr. Mertens to relocate anywhere other than San Antonio, Texas and instructions by the Company given to Mr. Mertens to violate any applicable law after notice of Mr. Mertens' objections.

F-55

Table of Contents

Compensation Committee Interlocks and Insider Participation

Angelo Guadagno and J. Larry Smart are the members of the Compensation Committee.

Item 12. Security Ownership of Certain Beneficial Owners and Management

The following table sets forth information as to the beneficial ownership of the Capital Stock of the Company as of February 24, 2000, by: (i) each person known to the Company to beneficially own more than five percent (5%) of the Capital Stock of the Company; (ii) each of the Company's directors; (iii) certain of the Company's executive officers; and (iv) all executive officers and directors of the Company as a group.

options to purchase 400,000 shares prior to vesting in the underlying shares. Exercises of the 400,000 stock options are subject, if not otherwise accelerated, to the Company's right to repurchase at cost upon Mr. Munro's termination from service with the Company or attempted disposition of unvested shares. The 400,000 shares vest ratably on a monthly basis over the vesting periods set forth in the stock option agreements for each of the option grants. If the acquisition of the Company by Avnet is completed, all of Mr. Munro's shares will be fully vested and the Company's right of repurchase will lapse. For a discussion of vesting of stock options following a change in control, see the subsection below entitled "Change in Control."

Shares **Beneficially** Name of Beneficial Owner(1) Owned(1) Percent(2) Astoria Capital Partners, L.P.(3) 6600 SW 92nd Avenue, Suite 370 Portland, OR 97223
Robert Fleming, Inc.(4)
320 Park Avenue, 11th Floor
New York, NY 10022
Strome Susskind Investment Management, L.P.(5) 1,877,430 13.56% 1,487,112 10.26% 100 Wilshire Blvd., 15th Floor Santa Monica, CA 90401 John M. Harkins(6) 1,259,697 8.91% 4955 Corporate Drive Huntsville, AL 35806 Michael N. Gunnells — Director(7) 1,200,000 8.90% 4955 Corporate Drive Huntsville, AL 35806 ROI Capital Management, Inc.(8) 8.78% 1,185,000 17 E. Sir Francis Drake Blvd., Suite 225 Larkspur, CA 94939 P. Scott Munro — Director and Executive Officer(9) 6.98% 960,826 254 Hacienda Avenue Campbell, California 95008
Carlton Joseph Mertens II - Director & Executive Officer(10) 516,458 3.82% 6550 North Loop 1604 East San Antonio, Texas 778247 Dennis Polk — Executive Officer(11) 460,000 3.41% 22,083 Robert O'Reilly - Executive Officer(12)
Guy M. Lammle - Director
J. Larry Smart - Director(13)
K. William Sickler - Director(14) 30.855 33,537 26,000 7,500

2,500 2,283,933

16.81%

* Less than one percent (1%).

F-56

Table of Contents

Unless otherwise indicated, the beneficial owner has sole voting and dispositive power over the shares reported in the table. Includes shares of Common Stock and shares of Series A Preferred Stock on an as-converted basis. Information with respect to beneficial ownership is based upon information obtained from the stockholders and from the Company's transfer agent. To the Company's knowledge, unless otherwise indicated, the persons and entities named in the table have sole voting and sole investment power with respect to all shares beneficially owned, subject to community property laws where applicable. Beneficial ownership is determined in accordance with the rules of the Securities and Exchange Commission and includes voting and investment power with respect to securities. Shares of Common Stock issuable upon conversion of shares of Series A Preferred Stock, upon exercise of stock

options exercisable within 60 days of February 24, 2000 or upon exercise of warrants that are currently exercisable or exercisable within 60 days of February 24, 2000 are deemed to be outstanding and to be beneficially owned by the person presently entitled to exercise for the purpose of computing the percentage ownership of such person but are not treated as outstanding for the purpose of computing the percentage ownership of any other person. Each share of Series A Preferred Stock is currently convertible at any time into shares of the Common Stock at a conversion ratio of 1.1953125 shares of Common Stock for each share of Series A Preferred Stock and is entitled to vote, without conversion, together with the Common Stock as a single class on an as-converted basis.

- Based on 13,489,569 shares of Common Stock, 1,850,012 shares of Series A Preferred Stock (equivalent to shares of Common Stock on an as-converted basis) outstanding as of February 24, 2000.
- Includes 1,526,500 shares of Common Stock, 207,000 shares of Series A Preferred Stock and warrants to purchase 103,500 shares of Common Stock. Common Stock ownership is based on a Schedule 13G/ A dated February 15, 2000.
- Includes 487,510 shares of Common Stock, 533,000 shares of Series A Preferred Stock and warrants to purchase 362,500 shares of Common Stock. Common Stock ownership based on a Schedule 13G dated February 9, 2000.
- Based on a Schedule 13G/ A filed March 18, 1999 by Strome Susskind Investment Management, L.P., (5) includes 603,591 shares of Common Stock, 387,012 shares of Series A Preferred Stock and warrants to purchase 193,506 shares of Common Stock.
- Ownership totals are based on a Schedule 13D dated June 5, 1998 filed by Mr. Harkins, as well as (6) shares issued to Mr. Harkins thereafter.
- Ownership totals are based on a Schedule 13D dated June 5, 1998 filed by Mr. Gunnells, as well as (7) shares issued to Mr. Gunnells thereafter.
- Includes 689,576 shares of Common Stock, 160,000 shares of Series A Preferred Stock and warrants to purchase 80,000 shares of Common Stock. Common Stock ownership is based on a Schedule 13G/ A dated February 15, 2000.
- Includes 25,000 shares subject to stock options that are presently exercisable. Also includes shares subject to a right of repurchase held by the Company pursuant to an Amendment of Stock Option Agreements, dated as of May 10, 1999, which vest ratably on a monthly basis over the vesting periods set forth in the stock option agreements for each of the option grants. If the acquisition of the Company by Avnet is completed, all of Mr. Munro's shares will be fully vested and the Company's right of repurchase will lapse. For a discussion of vesting of stock options following a change in control, see the subsection entitled "Change in Control" under Item 11 above.

Changes in Control

On March 2, 2000 the Company agreed to merge with a subsidiary of Avnet, Inc., so that it would become a wholly owned subsidiary of Avnet. The acquisition has been approved by the boards of directors of both entities and is subject to approval by the stockholders of Savoir and regulatory review. If the merger is completed, depending upon the average of the closing trading price of Avnet's common stock during the 15 days ending on the fifth day prior to the vote of the Company's stockholders, the Company's common stockholders will receive between 0.15494 and 0.11452 of a share of Avnet common stock for each share of Savoir stock they own. The transaction is expected to close in June. If the merger is not completed, the Company may be required to pay Avnet a termination fee of \$750,000, and the Company may be required to pay Avnet an additional fee of \$3.75 million if it enters into an acquisition agreement with another company within a year of termination of the Avnet merger agreement.

Avnet, Inc. is a Phoenix, Arizona based distributor of semiconductors, interconnect, passive and electromechanical components and computer products. At its last fiscal year end, Avnet reported sales of approximately \$6.35 billion.

Item 13. Certain Relationships and Related Transactions

(a) Transactions with Management and Others

The Company has entered into an Executive Retention Agreement with P. Scott Munro dated as of May 10, 1999. Included in the Executive Retention Agreement is a provision for a recourse loan to Mr. Munro for approximately \$2,500,000 with a stated interest rate of 4.9%. Mr. Munro executed the loan provision on May 10, 1999 and used the proceeds to exercise approximately 480,000 options under the revised terms of the Company's Employee Stock Option Plan. The loan is due and payable, including principal and interest, in May 2002. If Mr. Munro remains employed with the Company, the loan, and accrued interest, will be forgiven as follows: \$800,000 after the first quarter of 2000 and the balance on May 9, 2002. The loan provision of the Executive Retention Agreement also allows for Mr. Munro to borrow up to an additional \$1,100,000 under the same terms and conditions noted above. On September 10, 1999, Mr. Munro executed the remaining portion of the loan provision and borrowed the additional \$1,100,000 available to him. Upon the occurrence of a change of control (as defined in the Executive Retention Agreement), Mr. Munro's obligation to pay the then outstanding balance of the loan (including unpaid principal and accrued interest thereon) shall be forgiven and the promissory notes shall be canceled. If the acquisition of us by Avnet is completed, Mr. Munro's loan will be forgiven.

The Company has also entered into an Amendment of Stock Option Agreements with Mr. Munro, dated as of May 10, 1999. Pursuant to this amendment, the Company has accelerated the vesting of 385,000 shares underlying stock options so that they are fully vested. In addition, the Company has permitted Mr. Munro to exercise stock options to purchase 400,000 shares prior to vesting in the underlying shares. Exercises of the 400,000 stock options are subject, if not otherwise accelerated, to the Company's right to repurchase at cost upon Mr. Munro's termination from service with the Company or attempted disposition of unvested shares. The 400,000 shares vest ratably on a monthly basis over the vesting periods set forth in the stock option agreements for each of the option

grants. If the acquisition of the Company by Avnet is completed, all of Mr. Munro's shares will be fully vested and the Company's right of repurchase will lapse. For a discussion of vesting of stock options following a change in control, see the subsection entitled "Change in Control" under Item 11 above.

(b) Certain Business Relationships

Not applicable.

(c) Indebtedness of Management

For a discussion of the loan to and promissory notes of P. Scott Munro, see Item 13(a) above.

F-58

Table of Contents

PART IV

Item 14. Exhibits, Financial Statement Schedules, and Reports on Form 8-K

(a) 1. Financial Statements

The financial statements listed below appear on the pages indicated:

- (10) Ownership totals are based on a Schedule 13D dated June 5, 1998 filed by Mr. Mertens.
- (11) Includes 21,250 shares subject to stock options that are presently exercisable or will become exercisable within 60 days of February 24, 2000.
- (12) Includes 28,750 shares subject to stock options that are presently exercisable or will become exercisable within 60 days of February 24, 2000.
- (13) Includes 12,500 shares subject to stock options that are presently exercisable or will become exercisable within 60 days of February 24, 2000.
- (14) Includes 7,500 shares subject to stock options that are presently exercisable or will become exercisable within 60 days of February 24, 2000.
- (15) Includes 2,500 shares subject to stock options that are presently exercisable or will become exercisable within 60 days of February 24, 2000.
- (16) Includes 97,500 shares subject to stock options that are presently exercisable or will become exercisable within 60 days of February 24, 2000.
 - 2. Financial Statement Schedules

The financial statement schedules listed below appear on the pages indicated:

	Number
Consolidated Balance Sheets, December 31, 1998 and 1999	F-29
Consolidated Statements of Operations for the years ended December 31, 1997, 1998 and 1999	F-30
Consolidated Statements of Stockholders' Equity for the years ended becember 31, 1997, 1998 and 1999	F-31
Consolidated Statements of Cash Flows for the years ended December 31,	F-32
Notes to Consolidated Financial Statements	F-33

3. Exhibits

See the exhibits listed under Item 14(c) or filed or incorporated by reference herein. Each management contract or compensation plan or arrangement required to be filed has been identified.

(b) Reports on Forms 8-K

None.

(c) Exhibits

The exhibits listed below are filed or incorporated by reference herein.

Schedule II $-\mbox{\sc Valuation}$ and Qualifying Accounts and Reserves

F-63

Page

F-59

Table of Contents

Exhibit	
2.1	Agreement and Plan of Reorganization dated March 17, 1997 by and among the Company, WMT Acquisition Corp., Target Solutions, Inc. and Lee Adams, filed as Exhibit 2.1 to the Company's Current Report on Form 8-K filed with the
2.2†	Commission on April 1, 1997, and incorporated herein by this reference. Stock Purchase Agreement dated June 4, 1997 by and among the Company, Star Management Services, Inc., Harvey E. Najim and Carlton Joseph Mertens II, filed as Exhibit 2.1 to the Company's Current Report on Form 8-K filed with
2.3†	the Commission on July 17, 1997, and incorporated herein by this reference. Third Amendment to Stock Purchase Agreement dated September 30, 1997 by and among the Company, Star Management Services, Inc., Harvey E. Najim and Carlton Joseph Mertens II, filed as Exhibit 2.1 to the Company's Current
2.4	Report on Form 8-K filed with the Commission on October 10, 1997, and incorporated herein by this reference. Purchase Agreement Assignment between the Company and the Agent dated September 30, 1997, filed as Exhibit 10.5 to the Company's Current Report on Form 8-K filed with the Commission on October 10, 1997, and incorporated
2.5	herein by this reference. Asset Purchase Agreement dated November 29, 1996 by and among the Company, International Data Products, LLC, Oliver-Allen Corporation, Inc., International Data Products and Financial, Ltd., Alan M. Bynder and Michael R. Duhaime, filed as Exhibit 10.25 to the Company's Annual Report on Form 10-K for the year ended December 31, 1996 and incorporated herein by reference.

F-60

Exhibit	
2.6	Certificate of Ownership and Merger dated as of November 21, 1997, filed as Exhibit 2.1 to the Company's Current Report on Form 8-K filed with the Commission on November 24, 1997, and incorporated herein by this reference.
2.7†	Agreement and Plan of Reorganization dated November 22, 1997, by and among Savoir Technology Group, Inc., MCBA Systems, Inc., Michael N. Gunnells and John Harkins, filed as Exhibit 2.1 to the Company's Current Report on Form 8-K filed with the Commission on December 22, 1997, and incorporated herein by this reference.
2.8	Second Amendment to Agreement and Plan of Reorganization dated April 23, 1998, by and among Savoir Technology Group, Inc., STG Acquisition Corp., MCBA Systems, Inc., Michael N. Gunnells and John Harkins, filed as Exhibit 2.1 to the Company's quarterly report on Form 10-Q for the quarterly period ended June 30, 1998, and incorporated herein by this reference.
2.9	First Amendment to Agreement and Plan of Reorganization dated March 27, 1998, by and among Savoir Technology Group, Inc., STG Acquisition Corp., MCBA Systems, Inc., Michael N. Gunnells and John Harkins, filed as Exhibit 2.12 to the Company's Amendment on Form S-2/ A to its Registration Statement on Form S-2 filed with the Commission on April 3, 1998 and incorporated herein by this reference.
2.10*	Asset Purchase Agreement by and between Business Partner Solutions, Inc., a subsidiary of Savoir Technology Group, Inc. and REAL Applications Ltd. dated as of September 8, 1998, filed as Exhibit 2.1 to the Company's Current Report on Form 8-K filed with the Commission on September 23, 1998 and incorporated herein by this reference.
3(i)(a)	Restated Certificate of Incorporation of Savoir Technology Group, Inc., a Delaware corporation, filed as Exhibit 3(ii) to the Company's Amendment on Form 8-K/ A to its Current Report on Form 8-K dated July 23, 1997, filed on August 14, 1997 and incorporated herein by this reference.
3(i)(b)	Certificate of Amendment of the Certificate of Designation, Preferences and Rights of the Company's Series A Preferred Stock, filed as Exhibit 3.1(b) to the Company's Registration Statement on Form S-3 dated April 29, 1999, and incorporated herein by this reference.
3(i)(c)	Certification of Designation, Preferences and Rights of the Company's Series B Preferred Stock, filed as Exhibit 3.1 to the Company's Current Report on Form 8-K dated October 10, 1997, and incorporated herein by this reference.
3(ii)	Amended and Restated Bylaws of Savoir Technology Group, Inc., a Delaware corporation, filed as Exhibit 3.2 to the Company's Annual Report on Form 10-K for the year ended December 31, 1997 and incorporated herein by this reference.
4.1	Registration and Put Rights Agreement among the Company and the Purchasers dated September 30, 1997, filed as Exhibit 4.2 to the Company's Current Report on Form 8-K dated October 10, 1997, and incorporated herein by this reference.
4.2	Warrant Agreement of Western Micro Technology, Inc. between the Company and the Purchasers dated September 30, 1997 filed as Exhibit 4.3 to the Company's Current Report on Form 8-K dated October 10, 1997, and incorporated herein by this reference.
4.3	Common Stock Purchase Warrant in favor of Robert Fleming Inc., filed as Exhibit 4.4 to the Company's Current Report on Form 8-K dated October 10, 1997, and incorporated herein by this reference.
4.4	Common Stock Purchase Warrant in favor of CanPartners Investments IV, LLC filed as Exhibit 4.5 to the Company's Current Report on Form 8-K dated October 10, 1997, and incorporated herein by this reference.
4.5	Warrant Agreement of Western Micro Technology, Inc. between the Company and ICC dated September 30, 1997, filed as Exhibit 4.9 to the Company's Current Report on Form 8-K dated October 10, 1997, and incorporated herein by this reference.
4.6	Registration and Put Rights Amendment between the Company and ICC, filed as Exhibit 4.10 to the Company's Current Report on Form 8-K dated October 10, 1997, and incorporated herein by this reference.
4.7	Common Stock Purchase Warrant in favor of ICC, filed as Exhibit 4.11 to the Company's Current Report on Form 8-K dated October 10, 1997, and incorporated herein by this reference.
	- 4

F-61

Table of Contents

10.11

10.12

10.13

Exhibit Voting Agreement among the Company and certain holders of the Series A Preferred Stock of the Company dated as of December 31, 1998, filed as Annex B to the Company's Definitive Proxy Statement on Schedule 14A dated 9.1 April 6, 1999, and incorporated herein by this reference. Lease Agreement dated February 2, 1998, by and between Green Mountain Ventures I, Ltd., a Texas limited partnership and the Company's subsidiary Business Partner Solutions, Inc., a Texas corporation, filed as Exhibit 10.1 to the Company's Current Annual Report on Form 10-K for the year ended December 31, 1997 and incorporated herein by this reference. 10.1 Employment Letter between the Company and P. Scott Munro dated January 22, 1998, filed as Exhibit 10.17 to the Company's Annual Report on Form 10-K for the year ended December 31, 1997 and incorporated herein by this reference. 10.2** Employment Letter between the Company and James W. Dorst dated January 22, 1998, filed as Exhibit 10.18 to the Company's Annual Report on Form 10-K for the year ended December 31, 1997 and incorporated herein by 10.3** this reference. 10.4** Employment Letter between the Company and Robert O'Reilly dated January 22, 1998, filed as Exhibit 10.19 to the Company's Annual Report on Form 10-K for the year ended December 31, 1997 and incorporated herein by this reference. 10.5** Employment Letter between the Company and Carlton Joseph Mertens II dated September 30, 1997, filed as part of Exhibit 2.1 to the Company's Current Report on Form 8-K filed with the Commission on July 16, 1997 and incorporated herein by this reference. 10.6** Amendment of Stock Option Agreements by and between the Company and P. Scott Munro, filed as Exhibit 10.1 to the Company's Quarterly Report on Form 10-Q for the quarterly period ended June 30, 1999, and incorporated herein by this reference. Executive Retention Agreement, dated as of May 10, 1999, by and between the Company and P. Scott Munro, filed as Exhibit 10.2 to the Company's Quarterly Report on Form 10-Q for the quarterly period ended June 30, 1999, and incorporated herein by this reference. 10.7** 10.8**/†† Email Agreement, dated March 13, 2000, between P. Scott Munro and the Company. Promissory Note, dated May 10, 1999, of P. Scott Munro in favor of the Company, filed as Exhibit 10.3 to the Company's Quarterly Report on Form 10-Q for the quarterly period ended June 30, 1999, and incorporated 10.9* herein by this reference. herein by this reference. Promissory Note, dated September 13, 1999, of P. Scott Munro in favor of the Company. Solution Provider Agreement by and between Business Partner Solutions, Inc., a subsidiary of Savoir Technology Group, Inc. and REAL Applications Ltd. dated as of September 8, 1998, filed as Exhibit 10.1 to the Company's Current Report on Form 8-K dated September 8, 1998 and incorporated herein by this reference. Inventory and Working Capital Financing Agreement dated September 4, 1998 by and among the Company, Business Partners Solutions, Inc. and MCBA Systems, Inc., filed as Exhibit 10.2 to the Company's Quarterly Report on Form 10-Q for the quarterly period ended September 30, 1998, and incorporated herein by this reference. 10.10**/††

Amendment to Inventory and Working Capital Financing Agreement, dated as of May 21, 1999, by and among the Company, Business Partner Solutions, Inc., MCBA Systems, Inc. and IBM Credit Corporation, filed as

Exhibit 10.4 to the Company's Quarterly Report on Form 10-Q for the quarterly period ended June 30, 1999, and incorporated herein by this reference.
Industry Remarketer Affiliate Agreement between the Company and Sirius Computer Solutions, Ltd. (included in Exhibit 2.3).
Amendment Number One to Industry Remarketer Affiliate Agreement between the Company and Sirius Computer Solutions, Ltd., dated as of December 31, 1998, filed as Exhibit 10.7 to the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 1998, and incorporated herein by this reference.
Amendment Number Two to Industry Remarketer Affiliate Agreement between the Company and Sirius Computer Solutions, Inc., dated as of November 30, 1999.
The 1994 Stock Option Plan of Western Micro Technology, Inc. as amended and restated on May 18, 1997, filed as Exhibit A to the Company's Definitive Proxy Statement as filed with the Commission on June 27, 1997, is hereby incorporated by reference.

Exhibit Amendment to the 1994 Stock Option Plan of Western Micro Technology, Inc., dated as of June 29, 1999. The 1995 Employee Stock Purchase Plan of Western Micro Technology, Inc., filed with the Company's Revised Definitive Proxy Statement as filed with the Commission on September 13, 1995, is hereby incorporated by 10.18†† 10.19 reference. 10.20†† Amendment to the 1995 Employee Stock Purchase Plan of Western Micro Technology, Inc., dated as of June 29, 1999. Employment Letter between the Company and Dennis J. Polk, dated November 1, 1998, and amendments thereto 10.21**/†† dated January 11, 1999 and November 1, 1999. List of Subsidiaries. 21.1†† Consent of Independent Accountants Power of Attorney 23.1 27.1 Financial Data Schedule.

- † Confidential Treatment was granted by the Commission with respect to certain portions of this exhibit. Schedules omitted from this exhibit will be furnished to the Commission upon request.
- †† Filed previously.
- Confidential Treatment requested pursuant to a request for confidential treatment filed with the Commission on September 23, 1998. The portions of the exhibit for which confidential treatment has been requested have been omitted from the exhibit. The omitted information has been filed separately with the Commission as part of the confidential treatment request.
- (d) Financial Statement Schedules

The financial statement schedules listed below appear on the pages indicated:

** Indicates management contract or compensation plan or arrangement.

All other schedules have been omitted since the required information is not present or not present in amounts sufficient to require submission of the schedule or because the information required is included in the Consolidated Financial Statements or Notes thereto.

F-62

Table of Contents

Schedule II

SAVOIR TECHNOLOGY GROUP, INC.

VALUATION AND QUALIFYING ACCOUNTS AND RESERVES

(in thousands)

	Balance at Beginning of Period	Additions(1)	Other(2)	Deductions(3)	Balance at End of Period
Year ended December 31, 1997: Allowance for doubtful accounts Inventory valuation reserve	\$ 411 \$ 785	\$ 472 \$ 180	\$ 80 \$150	\$644 \$893	\$ 319 \$ 222
Year ended December 31, 1998: Allowance for doubtful accounts Inventory valuation reserve	\$ 319 \$ 222	\$ 842 \$1,370	\$ 60 \$ -	\$121 \$601	\$1,100 \$ 991
Year ended December 31, 1999: Allowance for doubtful accounts Inventory valuation reserve	\$1,100 \$ 991	\$1,702 \$1,155	\$ 60 \$ -	\$826 \$910	\$2,036 \$1,236
		F-63			

Table of Contents

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized on May 11, 2000.

- (1) Charged to costs and expenses.
- (2) Reserves recorded for acquisitions.
- (3) Accounts written off against the reserve.

SAVOIR TECHNOLOGY GROUP, INC. By /s/ P. SCOTT MUNRO

 $\begin{array}{c} \text{P. Scott Munro} \\ \text{Chief Executive Officer} \\ \text{By /s/ TERRY V. JOHNSON} \end{array}$

Pursuant to the requirements of the Securities Exchange Act of 1934, this report has been signed below by the following persons on behalf of the Registrant and in the capacities and on the dates indicated.

Terry V. Johnson Chief Financial Officer

F-64

Information Not Required in Prospectus

Item 20. Indemnification of Directors and Officers.

Section 54 of the Registrant's By-laws provides as follows:

Signature	Title	Date
/s/ P. SCOTT MUNRO	Chairman of the Board, Chief Executive Officer and	May 11, 2000
P. Scott Munro	Secretary (Principal Executive Officer)	
/s/ TERRY V. JOHNSON	Chief Financial Officer	May 11, 2000
Terry V. Johnson	(Principal Financial and Accounting Officer)	
*	Chief Operating Officer, President and Director	May 11, 2000
Carlton Joseph Mertens II	-	
*	Director	May 11, 2000
Angelo Guadagno	-	
*	Director	May 11, 2000
Mike Gunnells	-	
*	Director	May 11, 2000
Guy M. Lammle	-	
*	Director	May 11, 2000
K. William Sickler	-	
*	Director	May 11, 2000
J. Larry Smart	-	

*By: /s/ P. SCOTT MUNRO

P. Scott Munro Attorney-in-Fact

Section 721 of the New York Business Corporation Law (the "BCL") provides that no indemnification may be made to or on behalf of any director or officer of the Registrant if "a judgment or other final adjudication adverse to the director or officer establishes that his acts were committed in bad faith or were the result of active and deliberate dishonesty and were material to the cause of action so adjudicated, or that he personally gained in fact a financial profit or other advantage to which he was not legally entitled."

Section 54B of the Registrant's By-laws includes the foregoing statutory language.

The rights granted under Section 54 of the By-laws are in addition to, and are not exclusive of, any other rights to indemnification and expenses to which any director or officer may otherwise be entitled. Under the BCL, a New York corporation may indemnify any director or officer who is made or threatened to be made a party to an action by or in the right of such corporation against "amounts paid in settlement and reasonable expenses, including attorneys' fees," actually and necessarily incurred by him in connection with the defense or settlement of such action, or in connection with an appeal therein, if such director or officer acted, in good faith, for a purpose which he reasonably believed to be in the best interests of the corporation, except that no indemnification shall be made in respect of (1) a threatened action which is

settled or otherwise disposed of, or (2) any claim, issue or matter as to which such director or officer shall have been adjudged liable to the corporation, unless and only to the extent that a court determines that the director or officer is fairly and reasonably entitled to indemnity (BCL Section 722(c)). A corporation may also indemnify directors and officers who are parties to other actions or proceedings (including actions or proceedings by or in the right of any other corporation or other enterprise which the director or officer served at the request of the corporation) against "judgments, fines, amounts paid in settlement and reasonable expenses, including attorneys' fees," actually or necessarily incurred as a result of such actions or proceedings, or any appeal therein, provided the director or officer acted in good faith, for a purpose which he reasonably believed to be in the best interests of the corporation (or in the case of service to another corporation or other enterprise at the request of such corporation, not opposed to the best interests of such corporation) and, in criminal cases, that he also had no reasonable cause to believe that his conduct with unlawful (BCL Section 722(a)). Any indemnification under Section 722 may be made only if authorized in the specific case by disinterested directors, or by the board of directors upon the opinion in writing of independent legal counsel that indemnification is proper, or by shareholders (BCL Section 723(b)), but even without such authorization, a court may order indemnification in certain circumstances (BCL Section 724). Further, any director or officer who is "successful, on the merits

II-1

Table of Contents

or otherwise," in the defense of an action or proceeding is entitled to indemnification as a matter of right (BCL Section 723(a)).

A New York corporation may generally purchase insurance, consistent with the limitation of New York insurance law and regulatory supervision, to indemnify the corporation for any obligation which it incurs as a result of the indemnification of directors and officers under the provisions of the BCL, so long as no final adjudication has established that the directors' or officers' acts of active and deliberate dishonesty were material to the cause of action so adjudicated or that the directors or officers personally gained in fact a financial profit or other advance (BCL Section 726).

The Registrant's directors and officers are currently covered as insureds under directors' and officers' liability insurance. Such insurance, subject to annual renewal and certain rights of the insurer to terminate, provides an aggregate maximum of \$50,000,000 of coverage for directors and officers of the Registrant and its subsidiaries against claims made during the policy period relating to certain civil liabilities, including liabilities under the Securities Act of 1933.

Item 21. Exhibits and Financial Statement Schedules.

(a) Exhibits

A. The Corporation shall indemnify, and advance the expenses of, any director, officer or

employee to the full extent permitted by the New York Business Corporation Law as the same now exists or may hereafter be amended.

- B. The indemnification and advancement of expenses granted pursuant to this Section 54 shall not be exclusive or limiting of any other rights to which any person seeking indemnification or advancement of expenses may be entitled when authorized by (i) a resolution of shareholders, (ii) a resolution of directors or (iii) an agreement providing for such indemnification; provided that no indemnification may be made to or on behalf of any such person if a judgment or other final adjudication adverse to such person establishes that his acts were committed in bad faith or were the result of active and deliberate dishonesty and were material to the cause of action so adjudicated, or that he personally gained in fact a financial profit or other advantage to which he was not legally entitled.
- C. No amendment, modification or rescission of these By-laws shall be effective to limit any person's right to indemnification with respect to any alleged cause of action that accrues or other incident or matter that occurs prior to the date on which such modification, amendment or rescission is adopted.
- (b) Financial Statement Schedules

Not applicable with respect to the registrant, since Item 14(e) of Form S-4 is not applicable to it. None with respect to Savoir, since the schedules enumerated in Items 17(a) and 17(b) of Form S-4 are inapplicable to Savoir.

II-2

Table of Contents

(c) Report, Opinion or Appraisal

The opinion of Alliant Partners is set forth as Appendix D to the proxy statement/prospectus constituting Part I of this registration statement.

Item 22. Undertakings.

- (a) The undersigned registrant hereby undertakes that, for purposes of determining any liability under the Securities Act of 1933, each filing of the registrant's annual report pursuant to Section 13(a) or 15(d) of the Securities Exchange Act of 1934 (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the Securities Exchange Act of 1934) that is incorporated by reference in the registration statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (b) The undersigned registrant hereby undertakes that prior to any public reoffering of the securities registered hereunder through use of a prospectus which is a part of this registration statement, by any person or party who is deemed to be an underwriter within the meaning of Rule 145(c), the issuer undertakes that such reoffering prospectus will contain the information called for by the applicable registration form with respect to reofferings by persons who may be deemed underwriters, in addition to the information called for by the other items of the applicable form.

- (c) The registrant undertakes that every prospectus:
 (i) that is filed pursuant to paragraph (b) immediately preceding, or (ii) that purports to meet the requirements of Section 10(a)(3) of the Act and is used in connection with an offering of securities subject to Rule 415, will be filed as a part of an amendment to the registration statement and will not be used until such amendment is effective, and that, for purposes of determining any liability under the Securities Act of 1933, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (d) Insofar as indemnification for liabilities arising under the Securities Act of 1933 may be permitted to directors, officers and controlling persons of the registrant pursuant to the provisions referred to in Item 20 above, or otherwise, the registrant has been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the registrant of expenses incurred or paid by a director, officer or controlling person of the registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Act and will be governed by the final adjudication of such issue.
- (e) The undersigned registrant hereby undertakes to respond to requests for information that is incorporated by reference into the prospectus pursuant to Item 4, 10(b), 11 or 13 of this form, within one business day of receipt of such request, and to send the incorporated documents by first class mail or other equally prompt means. This includes information contained in documents filed subsequent to the effective date of the registration statement through the date of responding to the request.
- (f) The undersigned registrant hereby undertakes to supply by means of a post-effective amendment all information concerning a transaction, and the company being acquired involved therein, that was not the subject of and included in the registration statement when it became effective.

II-3

Table of Contents

SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, the registrant has duly caused this Registration Statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the city of Phoenix, State of Arizona, on May 12, 2000.

Exhibit No.	Description
2.1	Amended and Restated Agreement and Plan of Merger, dated as of March 2, 2000, among Avnet, Inc., Tactful Acquisition Corp. and Savoir Technology Group, Inc. included as Appendix A to the proxy statement/prospectus constituting Part I of this registration statement. The copy so included does not include the schedules to the Agreement and Plan of Merger as listed in the table of contents thereto. The registrant undertakes to furnish any such schedules to the Commission upon its request.
2.2	Inducement Agreement dated as of March 2, 2000, between Savoir Technology Group, Inc. and Avnet, Inc. included as Appendix B to the proxy statement/prospectus constituting Part I of this registration statement.
2.3	Option Agreement dated as of March 2, 2000 included as Appendix C to the proxy statement/prospectus constituting Part I of this registration statement.
4	Note: The registrant hereby agrees to provide the Commission, upon request, copies of such instruments defining the rights of holders of long-term debt of the registrant and its subsidiaries as are specified in Item 601(b)(4)(iii) (A) of Regulation S-K.
5	Òpínion of David R. Birk, Senior Vice President and General Counsel of Avnet
8	Tax opinion of Carter, Ledyard & Milburn
23.1	Consent of Arthur Andersen LLP
23.2	Consent of PricewaterhouseCoopers LLP
23.3	Consent of Carter, Ledyard & Milburn (included in Exhibit 8)
23.4	Consent of David R. Birk (included in Exhibit 5)
24	Powers of Attorney
99.1	Form of proxy card of Savoir
99.2	Consent of Alliant Partners

AVNET, INC. By: /s/ RAYMOND SADOWSKI

Pursuant to the requirements of the Securities Act of 1933, this Registration Statement has been signed on May 12, 2000, by the following persons in the capacities indicated.

> Raymond Sadowski Senior Vice President and Chief Financial Officer

> > II-4

Table of Contents

Signature	Capacity
/s/ ROY VALLEE	Chairman of the Board, Chief Executive Officer and Director
Roy Vallee	(principal executive officer)
/s/ RAYMOND SADOWSKI	Senior Vice President and Chief Financial Officer and (principal financial officer)
Raymond Sadowski	(principal financial officer)
/s/ JOHN F. COLE	Controller (principal accounting officer)
John F. Cole	
*	Director
Eleanor Baum	
*	Director
J. Veronica Biggins	
*	Director
Joseph F. Caligiuri	
*	Director
Lawrence W. Clarkson	
*	Director
Ehud Houminer	
*	Director
James A. Lawrence	
*	Director

Table of Contents

EXHIBIT INDEX

Signature	Capacity
*	Director
Frederic Salerno	
*	Director
Frederick S. Wood	
*By: /s/ RAYMOND SADOWSKI	
Raymond Sadowski	
Attorney-in-Fact	

AVNET Corporate Services Group

May 5, 2000

Board of Directors Avnet, Inc. 2211 South 47th Street Phoenix, Arizona 85034

Re: Registration Statement on Form S-4

Ladies and Gentlemen:

I refer to the Registration Statement on Form S-4 (the "Registration Statement") to be filed by Avnet, Inc. (the "Company") with the Securities and Exchange Commission in connection with the registration under the Securities Act of 1933, as amended (the "Act"), of up to 3,151,667 shares of common stock, par value \$1.00 per share (the "Shares"), of the Company to be issued in connection with the merger (the "Merger") of Tactful Acquisition Corp., a wholly-owned subsidiary of the Company ("Tactful"), with and into Savoir Technology Group, Inc. ("Savoir").

I have examined and am familiar with the Amended and Restated Agreement and Plan of Merger, dated as of March 2, 2000, by and among the Company, Tactful and Savoir (the "Merger Agreement"). I also have examined and am familiar with originals, or copies the authenticity of which has been established to my satisfaction, of such documents and instruments as I have deemed necessary to express the opinions hereinafter set forth.

Based upon the foregoing, it is my opinion that when the Registration Statement relating to the Shares has become effective under the Act, and the Merger has become effective under the Merger Agreement, the Shares will be validly issued, fully paid and nonassessable.

I consent to the use of this opinion as Exhibit 5 to the Registration Statement and to the reference to me under the caption "Legal Matters" in the prospectus constituting Part I thereof.

Very truly yours,

/s/ David R. Birk

David R. Birk Senior Vice President and General Counsel

[LETTERHEAD OF CARTER, LEDYARD AND MILBURN]

May 5, 2000

Avnet, Inc. 2211 South 47th Street Phoenix, AZ 85034

Gentlemen:

We have acted as counsel for Avnet, Inc. ("Parent") in connection with the proposed merger (the "Merger") of Tactful Acquisition Corp., a Delaware corporation and a wholly-owned subsidiary of Avnet ("Buyer"), with and into Savoir Technology Group, Inc. (the "Company"). Under the Amended and Restated Agreement and Plan of Merger dated March 2, 2000 (the "Agreement") by and among Parent, the Buyer and the Company, at the effective time of the Merger, (i) the Company will become a wholly-owned subsidiary of the Parent and (ii) the common stock and series A preferred stock of the Company will be converted into the right to receive shares of common stock of Parent (and cash instead of fractional shares). You have requested our opinion as to certain federal income tax consequences of the Merger. All capitalized terms used herein which are not otherwise defined have the same meaning as in the Agreement.

In connection with this opinion, we have examined and are familiar with originals or copies, certified or otherwise identified to our satisfaction, of the Agreement, and such other documents as we have deemed necessary or appropriate in order to enable us to render the opinion below. In our examination, we have assumed the genuineness of all signatures, the legal capacity of all natural persons, the authenticity of all documents submitted to us as originals, the conformity to original documents of all documents submitted to us as certified, conformed or photostatic copies and the authenticity of the originals of such copies. In addition, we have assumed that (i) the Merger will be consummated in the manner contemplated and in accordance with the provisions of the Agreement, and (ii) the representations made to us by Parent and the Company in their officer's certificates (the "Certificates") dated May 1, 2000 and May 3, 2000, respectively, and delivered to us for purposes of this opinion, are accurate, and will be accurate as of the effective time of the Merger. Where a representation in a Certificate is limited to the knowledge of the officer, we nevertheless have assumed the subject statement to be true.

In rendering our opinion, we have considered the applicable provisions of the Internal Revenue Code of 1986, as amended (the "Code"), Treasury Regulations, pertinent judicial authorities, interpretive rulings of the Internal Revenue Service and such other authorities as we have considered relevant.

Based upon and subject to the foregoing, we are of the opinion that (i) the Merger will, under current law, constitute a reorganization under Section 368(a) of the Code, (ii) each of Parent, Buyer and Company will be a party to such reorganization within the meaning of Section 368(b) of the Code, and (iii) Parent, Buyer and Company will not recognize any gain as a result of the Merger.

Except as set forth above, we express no opinion as to the tax consequences to any party, whether Federal, state, local or foreign, of the Merger or of any transactions related to the Merger or contemplated by the Agreement. This opinion is being furnished only to you in connection with the Merger and solely for your benefit in connection therewith and may not be used or relied upon for any other purpose and may not be circulated, quoted or otherwise referred to for any other purpose without our express written consent.

We hereby consent to the filing of this opinion as an exhibit to the Registration Statement filed in connection with the Merger and to the references to this opinion in the Registration Statement. In giving such consent, we do not thereby admit that we are in the category of persons whose consent is required under Section 7 of the Securities Act of 1933.

Sincerely,

/s/ Carter, Ledyard and Milburn

CONSENT OF INDEPENDENT PUBLIC ACCOUNTANTS

As independent public accountants, we hereby consent to (1) the incorporation by reference in this Registration Statement on Form S-4 of our report dated August 4, 1999, included in Avnet, Inc.'s Annual Report on Form 10-K for the year ended July 2, 1999, (2) the incorporation by reference in this Registration Statement on Form S-4 of our report dated August 25, 1999, on the consolidated financial statements of Marshall Industries, which report is included in Avnet, Inc.'s Current Report on Form 8-K bearing cover date of October 20, 1999 and (3) all references to our firm included in this Registration Statement.

Phoenix, Arizona

ARTHUR ANDERSEN LLP

May 11, 2000

CONSENT OF INDEPENDENT ACCOUNTANTS

We hereby consent to the use in this Registration Statement on Form S-4 of Avnet, Inc. of our report dated January 27, 2000, except for Note 17, as to which the date is March 2, 2000, relating to the consolidated financial statements and financial statement schedule of Savoir Technology Group, Inc., which are included in such Registration Statement in Appendix F. We also consent to the reference to us under the heading "Experts" in such Registration Statement.

PricewaterhouseCoopers LLP

Austin, Texas May 11, 2000

KNOW ALL PERSONS BY THESE PRESENTS:

The undersigned does hereby make, constitute and appoint David R. Birk and Raymond Sadowski, and either of them, his attorneys-in-fact and agents with full power of substitution and resubstitution, to execute for and on behalf of the undersigned in any and all capacities a Registration Statement under the Securities Act of 1933, any amendments to such Registration Statement (including post-effective amendments), and any other documents incidental thereto, relating to the offer and sale of securities of Avnet, Inc. in connection with its acquisition of Savoir Technology Group, Inc., and to file the same, with all exhibits thereto and all other required documents, with the Securities and Exchange Commission. The undersigned further grants unto said attorneys-in-fact and agents, and either of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done in connection with the said filings, as fully to all intents and purposes as the undersigned might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents and/or any of them or their or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has executed this power of attorney this 10th day of April, 2000.

/s/ Roy Vallee
-----Roy Vallee

KNOW ALL PERSONS BY THESE PRESENTS:

The undersigned does hereby make, constitute and appoint David R. Birk and Raymond Sadowski, and either of them, her attorneys-in-fact and agents with full power of substitution and resubstitution, to execute for and on behalf of the undersigned in any and all capacities a Registration Statement under the Securities Act of 1933, any amendments to such Registration Statement (including post-effective amendments), and any other documents incidental thereto, relating to the offer and sale of securities of Avnet, Inc. in connection with its acquisition of Savoir Technology Group, Inc., and to file the same, with all exhibits thereto and all other required documents, with the Securities and Exchange Commission. The undersigned further grants unto said attorneys-in-fact and agents, and either of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done in connection with the said filings, as fully to all intents and purposes as the undersigned might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents and/or any of them or their or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has executed this power of attorney this 10th day of April, 2000.

/s/ Eleanor Baum -----Eleanor Baum

KNOW ALL PERSONS BY THESE PRESENTS:

The undersigned does hereby make, constitute and appoint David R. Birk and Raymond Sadowski, and either of them, her attorneys-in-fact and agents with full power of substitution and resubstitution, to execute for and on behalf of the undersigned in any and all capacities a Registration Statement under the Securities Act of 1933, any amendments to such Registration Statement (including post-effective amendments), and any other documents incidental thereto, relating to the offer and sale of securities of Avnet, Inc. in connection with its acquisition of Savoir Technology Group, Inc., and to file the same, with all exhibits thereto and all other required documents, with the Securities and Exchange Commission. The undersigned further grants unto said attorneys-in-fact and agents, and either of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done in connection with the said filings, as fully to all intents and purposes as the undersigned might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents and/or any of them or their or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has executed this power of attorney this 10th day of April, 2000.

/s /J. Veronica Biggins

J. Veronica Biggins

KNOW ALL PERSONS BY THESE PRESENTS:

The undersigned does hereby make, constitute and appoint David R. Birk and Raymond Sadowski, and either of them, his attorneys-in-fact and agents with full power of substitution and resubstitution, to execute for and on behalf of the undersigned in any and all capacities a Registration Statement under the Securities Act of 1933, any amendments to such Registration Statement (including post-effective amendments), and any other documents incidental thereto, relating to the offer and sale of securities of Avnet, Inc. in connection with its acquisition of Savoir Technology Group, Inc., and to file the same, with all exhibits thereto and all other required documents, with the Securities and Exchange Commission. The undersigned further grants unto said attorneys-in-fact and agents, and either of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done in connection with the said filings, as fully to all intents and purposes as the undersigned might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents and/or any of them or their or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has executed this power of attorney this 10th day of April, 2000.

/s/ Joseph F. Caliguiri ------Joseph F. Caligiuri

KNOW ALL PERSONS BY THESE PRESENTS:

The undersigned does hereby make, constitute and appoint David R. Birk and Raymond Sadowski, and either of them, his attorneys-in-fact and agents with full power of substitution and resubstitution, to execute for and on behalf of the undersigned in any and all capacities a Registration Statement under the Securities Act of 1933, any amendments to such Registration Statement (including post-effective amendments), and any other documents incidental thereto, relating to the offer and sale of securities of Avnet, Inc. in connection with its acquisition of Savoir Technology Group, Inc., and to file the same, with all exhibits thereto and all other required documents, with the Securities and Exchange Commission. The undersigned further grants unto said attorneys-in-fact and agents, and either of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done in connection with the said filings, as fully to all intents and purposes as the undersigned might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents and/or any of them or their or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has executed this power of attorney this 10th day of April, 2000.

/s/ Lawrence W. Clarkson
Lawrence W. Clarkson

KNOW ALL PERSONS BY THESE PRESENTS:

The undersigned does hereby make, constitute and appoint David R. Birk and Raymond Sadowski, and either of them, his attorneys-in-fact and agents with full power of substitution and resubstitution, to execute for and on behalf of the undersigned in any and all capacities a Registration Statement under the Securities Act of 1933, any amendments to such Registration Statement (including post-effective amendments), and any other documents incidental thereto, relating to the offer and sale of securities of Avnet, Inc. in connection with its acquisition of Savoir Technology Group, Inc., and to file the same, with all exhibits thereto and all other required documents, with the Securities and Exchange Commission. The undersigned further grants unto said attorneys-in-fact and agents, and either of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done in connection with the said filings, as fully to all intents and purposes as the undersigned might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents and/or any of them or their or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has executed this power of attorney this 10th day of April, 2000.

/s/ Ehud Houminer -----Ehud Houminer

KNOW ALL PERSONS BY THESE PRESENTS:

The undersigned does hereby make, constitute and appoint David R. Birk and Raymond Sadowski, and either of them, his attorneys-in-fact and agents with full power of substitution and resubstitution, to execute for and on behalf of the undersigned in any and all capacities a Registration Statement under the Securities Act of 1933, any amendments to such Registration Statement (including post-effective amendments), and any other documents incidental thereto, relating to the offer and sale of securities of Avnet, Inc. in connection with its acquisition of Savoir Technology Group, Inc., and to file the same, with all exhibits thereto and all other required documents, with the Securities and Exchange Commission. The undersigned further grants unto said attorneys-in-fact and agents, and either of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done in connection with the said filings, as fully to all intents and purposes as the undersigned might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents and/or any of them or their or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has executed this power of attorney this 10th day of April, 2000.

/s/ James A. Lawrence
James A. Lawrence

KNOW ALL PERSONS BY THESE PRESENTS:

The undersigned does hereby make, constitute and appoint David R. Birk and Raymond Sadowski, and either of them, his attorneys-in-fact and agents with full power of substitution and resubstitution, to execute for and on behalf of the undersigned in any and all capacities a Registration Statement under the Securities Act of 1933, any amendments to such Registration Statement (including post-effective amendments), and any other documents incidental thereto, relating to the offer and sale of securities of Avnet, Inc. in connection with its acquisition of Savoir Technology Group, Inc., and to file the same, with all exhibits thereto and all other required documents, with the Securities and Exchange Commission. The undersigned further grants unto said attorneys-in-fact and agents, and either of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done in connection with the said filings, as fully to all intents and purposes as the undersigned might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents and/or any of them or their or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has executed this power of attorney this 10th day of April, 2000.

/s/ Salvatore J. Nuzzo
Salvatore J. Nuzzo

KNOW ALL PERSONS BY THESE PRESENTS:

The undersigned does hereby make, constitute and appoint David R. Birk and Raymond Sadowski, and either of them, his attorneys-in-fact and agents with full power of substitution and resubstitution, to execute for and on behalf of the undersigned in any and all capacities a Registration Statement under the Securities Act of 1933, any amendments to such Registration Statement (including post-effective amendments), and any other documents incidental thereto, relating to the offer and sale of securities of Avnet, Inc. in connection with its acquisition of Savoir Technology Group, Inc., and to file the same, with all exhibits thereto and all other required documents, with the Securities and Exchange Commission. The undersigned further grants unto said attorneys-in-fact and agents, and either of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done in connection with the said filings, as fully to all intents and purposes as the undersigned might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents and/or any of them or their or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has executed this power of attorney this 10th day of April, 2000.

/s/ Frederic Salerno
-----Frederic Salerno

KNOW ALL PERSONS BY THESE PRESENTS:

The undersigned does hereby make, constitute and appoint David R. Birk and Raymond Sadowski, and either of them, his attorneys-in-fact and agents with full power of substitution and resubstitution, to execute for and on behalf of the undersigned in any and all capacities a Registration Statement under the Securities Act of 1933, any amendments to such Registration Statement (including post-effective amendments), and any other documents incidental thereto, relating to the offer and sale of securities of Avnet, Inc. in connection with its acquisition of Savoir Technology Group, Inc., and to file the same, with all exhibits thereto and all other required documents, with the Securities and Exchange Commission. The undersigned further grants unto said attorneys-in-fact and agents, and either of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done in connection with the said filings, as fully to all intents and purposes as the undersigned might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents and/or any of them or their or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has executed this power of attorney this 10th day of April, 2000.

/s/ Frederick Wood
-----Frederick Wood

KNOW ALL PERSONS BY THESE PRESENTS:

The undersigned does hereby make, constitute and appoint David R. Birk and Raymond Sadowski, and either of them, his attorneys-in-fact and agents with full power of substitution and resubstitution, to execute for and on behalf of the undersigned in any and all capacities a Registration Statement under the Securities Act of 1933, any amendments to such Registration Statement (including post-effective amendments), and any other documents incidental thereto, relating to the offer and sale of securities of Avnet, Inc. in connection with its acquisition of Savoir Technology Group, Inc., and to file the same, with all exhibits thereto and all other required documents, with the Securities and Exchange Commission. The undersigned further grants unto said attorneys-in-fact and agents, and either of them, full power and authority to do and perform each and every act and thing requisite and necessary to be done in connection with the said filings, as fully to all intents and purposes as the undersigned might or could do in person, hereby ratifying and confirming all that said attorneys-in-fact and agents and/or any of them or their or his substitute or substitutes, may lawfully do or cause to be done by virtue hereof.

IN WITNESS WHEREOF, the undersigned has executed this power of attorney this 10th day of April, 2000.

/s/ John F. Cole ------John F. Cole

SAVOIR TECHNOLOGY GROUP, INC. PROXY FOR SPECIAL MEETING OF STOCKHOLDERS TO BE HELD ON , 2000

The undersigned stockholder of Savoir Technology Group, Inc. revoking all prior proxies, hereby appoints P. Scott Munro, Carlton Joseph Mertens II, or Terry Johnson, or each of them acting singly, proxies, with full power of substitution, to vote all shares of Savoir common stock which the undersigned is entitled to vote at the special meeting of stockholders to be held at 44931 Industrial Boulevard, Fremont, California 94538 on , 2000, beginning at 10 a.m., local time, and at any adjournment or postponement of the meeting, upon matters set forth in the Notice of Special Meeting dated , 2000, and the related proxy statement/prospectus, copies of which have been received by the undersigned, and in their discretion upon any adjournment of the meeting or upon any other business that may properly be brought before the meeting by the Savoir board of directors. Attendance of the undersigned at the meeting or any adjourned session of the meeting will not be deemed to revoke this proxy unless the undersigned will affirmatively indicate the intention of the undersigned to vote the shares represented hereby in person prior to the exercise of this proxy.

THIS PROXY IS SOLICITED ON BEHALF OF THE SAVOIR BOARD OF DIRECTORS. A STOCKHOLDER WISHING TO VOTE IN ACCORDANCE WITH THE RECOMMENDATIONS OF THE BOARD OF DIRECTORS NEED ONLY SIGN AND DATE THIS PROXY AND RETURN IT IN THE ENCLOSED ENVELOPE

To Vote by Telephone:

--- Call our toll-free number from any Touch-Tone telephone in the United States or Canada: 1-800-______.

--- When prompted, enter your "control number," followed by the # sign. Your control number is printed below.

--- Follow the recorded instructions.

--- When you finish, you will hear a recorded recap. If it is correct, press "2" to register your vote.

To Vote by Mail:

--- Mark, sign and date the voting card which is attached below.

--- Return it in the postage-paid envelope we have provided. Make sure the pre-printed address shows through the envelope window.

[Reverse]

|X| PLEASE MARK YOUR VOTES AS IN THIS EXAMPLE.

To adopt the Amended and Restated Agreement and Plan of Merger dated as of March 2, 2000, among Avnet, Inc., Tactful Acquisition Corp., a wholly owned subsidiary of Avnet, and Savoir Technology Group, Inc.

FOR / /
AGAINST / /
ABSTAIN / /

THE SHARES REPRESENTED BY THIS PROXY WILL BE VOTED AS DIRECTED OR, IF NO DIRECTION IS GIVEN WITH RESPECT TO THE PROPOSAL SET FORTH ABOVE, WILL BE VOTED FOR SUCH PROPOSAL.

DATED: , 2000
------Signature of Stockholder(s)

Please promptly complete, date and sign this proxy promptly and mail it in the enclosed envelope to assure representation of your shares. No postage need be affixed if mailed in the United States. PLEASE SIGN EXACTLY AS NAME(S)

APPEARS ON THE STOCK CERTIFICATE.

If stockholder is a corporation, please sign full corporate name by president or other authorized officer and, if a partnership, please sign full partnership name by an authorized partner or other persons. If shares are held by joint tenants, both should sign. Attorneys-in-fact, executors, administrators, trustees, guardians or others signing in a representative capacity should indicate the capacity in which they are signing.

Mark here if you plan to attend the meeting / /

[LETTERHEAD OF ALLIANT PARTNERS]

We hereby consent to the inclusion of our opinion letter dated April 19, 2000 to the Board of Directors of Savoir Technology Group, Inc. ("Savoir") as Appendix D to the Proxy Statement/Prospectus relating to the proposed merger of Tactful Acquisition Corp., a wholly-owned subsidiary of Avnet, Inc., with and into Savoir, and to the references to our opinion letter in the Proxy Statement/ Prospectus. In giving this consent, we do not admit that we come within the category of persons whose consent is required under, and we do not admit that we are "experts" for purposes of, the Securities Act of 1933 and the rules and regulations promulgated thereunder.

Palo Alto, California May 4, 2000

By: /s/ James L. Kochman

James L. Kochman Managing Partner Alliant Partners

Exhibit No.	Description
2.1	Amended and Restated Agreement and Plan of Merger, dated as of March 2, 2000, among Avnet, Inc., Tactful Acquisition Corp. and Savoir Technology Group, Inc. included as Appendix A to the proxy statement/prospectus constituting Part I of this registration statement. The copy so included does not include the schedules to the Agreement and Plan of Merger as listed in the table of contents thereto. The registrant undertakes to furnish any such schedules to the Commission upon its request.
2.2	Inducement Agreement dated as of March 2, 2000 included as Appendix B to the proxy statement/prospectus constituting Part I of this registration statement.
2.3	Option Agreement dated as of March 2, 2000, between Savoir Technology Group, Inc. and Avnet Inc. included as Appendix C to the proxy statement/prospectus constituting Part I of this registration statement.
4	Note: The registrant hereby agrees to provide the Commission, upon request, copies of such instruments defining the rights of holders of long-term debt of the Registrant and its subsidiaries as are specified in Item 601(b)(4)(iii) (A) of Regulation S-K.
5	Opinion of David R. Birk, Senior Vice President and General Counsel of Avnet
8	Tax opinion of Carter, Ledyard & Milburn
23.1	Consent of Arthur Andersen LLP
23.2	Consent of PricewaterhouseCoopers LLP
23.3	Consent of Carter, Ledyard & Milburn (included in Exhibit 8)
23.4	Consent of David R. Birk (included in Exhibit 5)
24	Powers of Attorney
99.1	Form of proxy card of Savoir
99.2	Consent of Alliant Partners